Associations between the dynamics of socioeconomic status and children's externalizing behaviour, internalizing behaviour, and language development

Scott Orr

School of Education, Humanities and Social Sciences

University of Strathclyde

A thesis manuscript submitted to the University of Strathclyde for the degree of Doctor of Philosophy

February 2023

Authors Declaration

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed: Sealt DA

Date: 26/02/2023

Acknowledgements

I would first of all like to thank my supervisors, Dr Edward Sosu and Dr Markus Klein for their time, support and patience. Throughout the process, their guidance, encouragement and support had been invaluable, and this thesis would never have materialised without them. It has been a great experience working with you both and I have learned a lot.

I would also like to thank my family for their support and encouragement over the past few years. It has been a long slog and you have always encouraged me to press on. Thanks to my friends for their patience and understanding and for providing some much needed distractions. Thanks to Dr Mark Elliott and Dr Allan McGroarty for the unhealthy breakfasts and Abigail for listening to my list of complaints, the encouragement, and the reassurance. It is very much appreciated. My thanks also go to my fellow postgrads and the PsyPAGers. Meeting you all has been one of the highlights of this process and I am so grateful to be friends with you all. Our "writing retreats" and conferences will always be memorable.

Table of contents

Authors Declaration	3
Acknowledgements	4
Table of contents	5
Glossary of acronyms and abbreviations	.10
List of figures	.13
List of tables	.15
Abstract	.18
Chapter 1: Introduction	.20
1.1 Overview 1.2 Focus of this study 1.3 The rationale for the study 1.4 Aims of this study 1.5 Research questions 1.6 Structure of the study	. 20 . 21 . 22 . 25 . 26 . 27
Chapter 2 – Literature review	.29
2.1 AIM OF THIS CHAPTER 2.2 Context 2.3 Policies to address developmental inequalities 2.4 Child Development	. 29 . 29 . 31 . 38
2.4.1 THEORY OF CHILD DEVELOPMENT	.41 .43 46 47 49 49 51
2.5.4.3 INCOME 2.5.5 ALTERNATIVE MEASURES 2.5.6 CONCEPTUALISATION OF FAMILY SES IN THIS STUDY 2.6 SES AND CHILDREN'S DEVELOPMENTAL OUTCOMES 2.6.1 THEORETICAL CONSIDERATIONS	.51 53 56 .57 57

	го
2.6.1.2 FAMILY STRESS MODEL	
2.6.1.3. EVIDENCE SUPPORTING THE FAMILY STRESS	IVIODEL60
2.6.1.4 THE FAMILY INVESTMENT MODEL	
2.6.1.5. EVIDENCE SUPPORTING THE FAMILY INVESTIN	MENT INIODEL
2.6.2 EVIDENCE ON SES AND DEVELOPMENTAL OUTCO	DMES
2.6.2.1 EVIDENCE ON SES AND LANGUAGE DEVELOPM	/ENT
2.6.2.2 EVIDENCE ON SES AND SOCIO-EMOTIONAL DE	57 EVELOPMENT
2.6.2.3 SUMMARY	
2.7 EVIDENCE OF DIFFERENT SES DIMENSIONS AND CHILI	
2.8. A DYNAMIC VIEW OF SES AND DEVELOPMENTAL OU	TCOMES
2.8.1 THEORETICAL CONSIDERATIONS	
2.8.1.1 SENSITIVE PERIOD MODELS	
2.8.1.2 ACCUMULATION OF RISK MODELS	
2.8.1.3 TRAJECTORY MODELS	
2.8.2 EVIDENCE OF SES DIFFERENCES IN DEVELOPMEN	ITAL OUTCOMES BY STAGE OF CHILDHOOD
2.8.3.1 EVIDENCE OF SES TRAJECTORIES AND DEVELO	PMENTAL OUTCOMES85
2.8.3.2 EVIDENCE OF DEVELOPMENTAL OUTCOME TRA	AJECTORIES93
2.9 RESEARCH AIMS	
2.10 SUMMARY	
Chapter 3 - Methodology	
Chapter 3 - Methodology	
Chapter 3 - Methodology 3.1 AIM OF THIS CHAPTER 3.2 DESIGN	
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample	
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample 3.4 Data collection procedure	
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample 3.4 Data collection procedure 3.5 Sample achieved by the MCS at each wave	
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample 3.4 Data collection procedure 3.5 Sample achieved by the MCS at each wave 3.5.1 Analytic sample	103 103 103 103 104 104 107 108 109
 Chapter 3 - Methodology	
 Chapter 3 - Methodology	103 103 103 103 103 104 104 107 108 109 110 110 110
 Chapter 3 - Methodology	103 103 103 103 103 103 104 104 107 107 108 109 110 110 111 111
 Chapter 3 - Methodology	103 103 103 103 103 104 104 107 107 108 109 110 110 111 111 112
 Chapter 3 - Methodology	103 103 103 103 103 103 104 104 107 108 109 110 110 111 111 112 112 112 112 112 112
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample 3.4 Data collection procedure 3.5 Sample achieved by the MCS at each wave 3.5.1 Analytic sample 3.6 Measures 3.6.1 Language Ability 3.6.2 Age 3 (MCS2) Language Ability 3.6.3 Age 5 (MCS3) Language Ability 3.6.4 Age 7 (MCS4) Language Ability 3.6.5 Age 11 (MCS5) Language Ability	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 112 112 112 113
 Chapter 3 - Methodology	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 111 112 112 113 113
 Chapter 3 - Methodology	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 112 112 112 113 113 113 113 115
Chapter 3 - Methodology	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 111 112 112 113 113 113 113 113 115 117
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample 3.4 Data collection procedure 3.5 Sample achieved by the MCS at each wave 3.5.1 Analytic sample 3.6 Measures 3.6.1 Language Ability 3.6.2 Age 3 (MCS2) Language Ability 3.6.3 Age 5 (MCS3) Language Ability 3.6.4 Age 7 (MCS4) Language Ability 3.6.5 Age 11 (MCS5) Language Ability 3.6.6 Age 14 (MCS6) Language Ability 3.6.7 Internalizing and Externalizing Behaviour 3.6.8 Measures of socio-economic status 3.6.9 Family Income	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 111 112 112 113 113 113 113 113 113
 Chapter 3 - Methodology	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 111 112 112 113 113 113 113 113 113
 Chapter 3 - Methodology	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 111 112 112 113 113 113 113 113 113
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample 3.4 Data collection procedure 3.5 Sample achieved by the MCS at each wave 3.5.1 Analytic sample 3.6 Measures 3.6.1 Language Ability 3.6.2 Age 3 (MCS2) Language Ability 3.6.3 Age 5 (MCS3) Language Ability 3.6.4 Age 7 (MCS4) Language Ability 3.6.5 Age 11 (MCS5) Language Ability 3.6.6 Age 14 (MCS6) Language Ability 3.6.7 Internalizing and Externalizing Behaviour 3.6.8 Measures of socio-economic status 3.6.10 Education 3.6.11 Occupation (parental class) 3.6.12 Covariates	103 103 103 103 103 104 104 107 107 108 109 109 110 110 111 111 112 112 113 113 113 113 113 113
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 Data and sample 3.4 Data collection procedure 3.5 Sample achieved by the MCS at each wave 3.5.1 Analytic sample 3.6 Measures 3.6.1 Language Ability 3.6.2 Age 3 (MCS2) Language Ability 3.6.3 Age 5 (MCS3) Language Ability 3.6.4 Age 7 (MCS4) Language Ability 3.6.5 Age 11 (MCS5) Language Ability 3.6.6 Age 14 (MCS6) Language Ability 3.6.7 Internalizing and Externalizing Behaviour 3.6.8 Measures of socio-economic status 3.6.10 Education 3.6.11 Occupation (parental class) 3.6.12 Covariates	103 103 103 103 103 103 104 104 107 107 108 109 109 110 110 110 111 111 112 112 113 113 113 113 113 113
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 DATA AND SAMPLE 3.4 DATA COLLECTION PROCEDURE 3.5 SAMPLE ACHIEVED BY THE MCS AT EACH WAVE 3.5.1 ANALYTIC SAMPLE 3.6 MEASURES 3.6.1 LANGUAGE ABILITY 3.6.2 AGE 3 (MCS2) LANGUAGE ABILITY 3.6.3 AGE 5 (MCS3) LANGUAGE ABILITY 3.6.4 AGE 7 (MCS4) LANGUAGE ABILITY 3.6.5 AGE 11 (MCS5) LANGUAGE ABILITY 3.6.6 AGE 14 (MCS6) LANGUAGE ABILITY 3.6.7 INTERNALIZING AND EXTERNALIZING BEHAVIOUR 3.6.8 MEASURES OF SOCIO-ECONOMIC STATUS 3.6.10 EDUCATION 3.6.11 OCCUPATION (PARENTAL CLASS) 3.6.12 COVARIATES 3.8 MISSING DATA	103 103 103 103 103 104 104 107 107 108 109 109 110 110 110 111 111 112 112 113 113 113 113 113 114 115 115 117 117 118 118 122 126 128 131
Chapter 3 - Methodology 3.1 Aim of this chapter 3.2 Design 3.3 DATA AND SAMPLE 3.4 DATA COLLECTION PROCEDURE 3.5 SAMPLE ACHIEVED BY THE MCS AT EACH WAVE 3.5.1 ANALYTIC SAMPLE 3.6 MEASURES 3.6.1 LANGUAGE ABILITY 3.6.2 AGE 3 (MCS2) LANGUAGE ABILITY 3.6.3 AGE 5 (MCS3) LANGUAGE ABILITY 3.6.4 AGE 7 (MCS4) LANGUAGE ABILITY 3.6.5 AGE 11 (MCS5) LANGUAGE ABILITY 3.6.6 AGE 14 (MCS6) LANGUAGE ABILITY 3.6.7 INTERNALIZING AND EXTERNALIZING BEHAVIOUR 3.6.8 MEASURES OF SOCIO-ECONOMIC STATUS 3.6.10 EDUCATION (PARENTAL CLASS) 3.6.12 COVARIATES 3.7 DATA SCREENING 3.8 MISSING DATA 3.9 DATA ANALYSIS PROCEDURES	103 103 103 103 103 104 107 107 108 109 110 110 110 111 111 111 112 112 112 113 113 113 113

4.1 AIM OF THIS CHAPTER	12
4.2 INTRODUCTION	12
4.2.1 THEORETICAL FRAMEWORK FOR UNDERSTANDING THE ROLE OF SES IN CHILDREN'S DEVELOPMENT14	14
4.2.2 EVIDENCE OF SES DIMENSIONS AND CHILDREN'S DEVELOPMENT	ł7
4.2.3 FAMILY SES AND CHILD DEVELOPMENT AT DIFFERENT STAGES OF CHILDHOOD	19
4.3 METHODOLOGY	;2
4.3.1 SAMPLE	52
4.3.2 Measures	52
4.3.3 LANGUAGE ABILITY15	53
4.3.4 Age 3 (MCS2) language ability15	53
4.3.5 Age 7 (MCS4) language ability15	54
4.3.6 Age 14 (MCS6) LANGUAGE ABILITY15	55
4.3.7 INTERNALIZING AND EXTERNALIZING BEHAVIOUR15	56
4.3.8 Measures of socio-economic status15	58
4.3.9 COVARIATES	50
4.3.10 ANALYTIC STRATEGY16	51
4.4 RESULTS	52
4.4.1 DESCRIPTIVE STATISTICS	52
4.4.2 Associations between family SES and children's language ability during early, middle, and	
LATE CHILDHOOD	55
4.4.3 Associations between family SES and children's internalizing behaviour during early,	
MIDDLE, AND LATE CHILDHOOD	57
4.4.4 Associations between family SES and children's externalizing behaviour during early,	
MIDDLE, AND LATE CHILDHOOD	59
4.5 DISCUSSION	1/
4.5.1 Strengths and weaknesses of the study17	<i>'</i> 6
4.5.2 Implications of the study17	77
4.6 SUMMARY	/8

5.1 AIMS OF THIS CHAPTER	179
5.2 1 THEORETICAL FRAMEWORK FOR LINDERSTANDING THE ROLE OF SES TRAJECTORIES IN CHILDREN'S	179
DEVELOPMENT.	182
5.2.2 EVIDENCE OF SES TRAJECTORIES AND CHILDREN'S DEVELOPMENT	184
5.2.3 THE CURRENT STUDY	187
5.3 METHODOLOGY	188
5.3.1 SAMPLE	188
5.3.2 Measures	189
5.3.3 Language Ability	189
5.3.4 INTERNALIZING AND EXTERNALIZING BEHAVIOUR	190
5.3.5 Measures of socio-economic status	191
5.3.6 Covariates	192
5.3.7 Analytic strategy	192
5.4 Results	194
5.4.1 DESCRIPTIVE STATISTICS	194
5.4.2 Trajectories of SES dimensions	200
5.4.3 Trajectories of family income	201
5.4.4 Associations between income trajectories and children's outcomes	203

5.4.5 TRAJECTORIES OF PARENT'S LEVEL OF EDUCATION	206
5.4.6 Associations between parent education trajectories and children's outcomes	208
5.4.7 TRAJECTORIES OF PARENT'S LEVEL OF OCCUPATION	211
5.4.8 Occupation trajectories on children's outcomes	212
5.5 DISCUSSION	215
5.5.1 LIMITATIONS OF THE STUDY	222
5.5.2 IMPLICATIONS OF THE STUDY	223
5.6 SUMMARY	224

Chapter 6 – Associations between family SES trajectories and children's language,	
internalizing, and externalizing trajectories	225

6.1 AIM OF THIS CHAPTER	225
6.2 11 THEORETICAL ERAMEWORK FOR LINDERSTANDING THE ROLE OF FAMILY SES TRAJECTORIES IN CH	225
DEVELOPMENTAL TRAJECTORIES	110KLN 3
6.2.2 EVIDENCE OF SES ASSOCIATIONS WITH CHILDREN'S DEVELOPMENTAL OUTCOME TRAJECTORIES	220
6.2.2 EVIDENCE OF SES ASSOCIATIONS WITH CHIEDREN'S DEVELOPMENTAL OUTCOME TRAJECTORIES	230
6 3 METHODOLOGY	
6 3 1 SANDIE	233
6 3 2 MEASURES	232
6 3 3 LANGUAGE ABILITY	23 234
6.3.4 Age 3 (MCS2) LANGUAGE ABULTY	20 735
6 3 5 AGE 5 (MCS2) LANGUAGE ABILITY	233 236
6.3.6 AGE 7 (MCSA) LANGUAGE ABILITY	23C 23F
6.3.7 Age 11 (MCS5) Language ability	230
6.3.8 Age 14 (MCS6) Language ability	237
6.3.9 INTERNALIZING AND EXTERNALIZING BEHAVIOLIR	,237 230
6 3 10 MEASURES OF SOCIO-ECONOMIC STATUS	202
6 3 11 COVARIATES	241 741
6 3 12 ANALYTIC STRATEGY	241 242
6.4 RESULTS	
6.4.1 DESCRIPTIVE STATISTICS	
6.4.2 TRAJECTORIES OF CHILDREN'S DEVELOPMENTAL OUTCOMES.	
6.4.3 TRAJECTORIES OF LANGUAGE ABILITY	
6.4.4 TRAJECTORIES OF LITERNALIZING BEHAVIOUR	
6.4.5 TRAJECTORIES OF EXTERNALIZING BEHAVIOUR	255
6.4.6 PROBABILITIES OF CHILDREN'S MEMBERSHIP TO OUTCOME TRAJECTORIES	
6.4.7 PROBABILITY OF BELONGING TO OUTCOME TRAJECTORIES BASED ON FAMILY INCOME	
6.4.8 PROBABILITY OF BELONGING OUTCOME TRAJECTORIES BASED ON PARENT EDUCATION	
6.4.9 PROBABILITY OF BELONGING OUTCOME TRAJECTORIES BASED ON PARENT OCCUPATION	
6.5 DISCUSSION	272
6.5.1 LIMITATIONS OF THE STUDY.	
6.5.2 IMPLICATIONS OF THE STUDY	278
6.6 SUMMARY	280
hapter 7 – General Discussion	281
7.1 AIM OF THIS CHAPTER 7.2 AIM OF THIS THESIS	281 281

7.4 Key findings and their Implications	284
7.4.1 SES DIMENSIONS	
7.4.2 Stage of childhood	
7.4.3 Trajectories	
7.5 Overall Implications	297
7.5.1 MULTIDIMENSIONAL SUPPORT	
7.5.2 Support over the childhood timeline	
7.5.3 Addressing stress as a mechanism between SES and child development	
7.6 LIMITATIONS	310
7.7 FUTURE RESEARCH	313
7.7.1 EXAMINING PATHWAYS BETWEEN SES AND CHILDREN'S DEVELOPMENTAL OUTCOMES	
7.7.2 Considering sex differences	
7.7.3 Considering neurodiversity	
7.7.4 Considering different contexts	
7.7.5 INTERNATIONAL RESEARCH	
7.8 CONCLUSION	321
References	377
	JZZ
Appendix	428

Glossary of acronyms and abbreviations

ADHD	Attention deficit hyperactivity disorder
AIC	Akaike's information criterion
АРА	American Psychiatric Association
APU	Applied Psychology Unit
ASD	Autism spectrum disorders
BAS	British Ability Scale
BAS-NV	British Ability Scale - Naming Vocabulary subscale
BAS-WR	British Ability Scale - Word Reading subscale
BAS-WS	British Ability Scale - Word Similarities subscale
BIC	Bayesian information criterion
BLRT	Bootstrapped likelihood ratio test
BST	Bioecological Systems Theory
BTEC	Business and Technology Education Council
CFI	Comparative fit index
CLT	Central Limit Theorem
DWP	Department for Work and Pensions
FIM	Family Investment Model
FIML	Full information maximum likelihood
FSM	Family Stress Model
FSS	Fair Start Scotland
GCM	Growth Curve Modelling
GCSE	General Certificate of Secondary Education

GMM	Growth Mixture Modelling
GNVQ	General National Vocational Qualification
GSVQ	General Scottish Vocational Qualification
HNC	Higher National Certificate
HND	Higher National Diploma
ISD	Information Services Division
LCGA	Latent Class Growth Analyses
LCT	Life Course Theory
LD	Learning difficulties
LMR	Vong-Lo-Mendell-Rubin test
MAR	Missing at random
MCAR	Missing completely at random
MCS	Millennium Cohort Study
MI	Multiple imputation
MNAR	Missing not at random
NOLB	No One Left Behind
NS-SEC	National Statistics Socio-economic Classification
NVQ	National Vocational Qualification
OECD	Organization for Economic Cooperation and Development
ONS	Office for National Statistics
RMSEA	Root mean square error of approximation
RSA	Royal Society of Arts
SAAS	Student Awards Agency Scotland

SCOTVEC	Scottish Vocational Educational Council
SD	Standard Deviation
SDQ	Strength and Difficulties Questionnaire
SEG	Socioeconomic Groups
SES	Socioeconomic status
SIMD	Scottish Index of Multiple Deprivation
SSS	Subjective Social Status
SVQ	Scottish Vocational Qualification
TLI	Tucker–Lewis index
VIF	Variance inflation factor

List of figures

Figure 1. Ecological Systems Model42
Figure 2. Family Stress Model60
Figure 3. Family Investment Model61
Figure 5.1. Six-class model of income trajectories203
Figure 5.2. Parent level of education trajectories208
Figure 5.3. Parent level of occupation trajectories212
Figure 6.1. Five-class model of language ability trajectories250
Figure 6.2. Five-class model of internalizing behaviour trajectories
Figure 6.3. Five-class model of externalizing behaviour trajectories
Figure 6.4. Children's likelihood of belonging to language trajectories based on family
income trajectories
Figure 6.5. Children's likelihood of belonging to internalizing behaviour trajectories based on
family income trajectories261
Figure 6.6. Children's likelihood of belonging to externalizing behaviour trajectories based
on family income trajectories263

Figure 6.7. Children's likelihood of belonging to language trajectories based on parent's
education trajectories264
Figure 6.8. Children's likelihood of belonging to internalizing behaviour trajectories based on
parent's education trajectories266
Figure 1.9. Children's likelihood of belonging to externalizing behaviour trajectories based
on parent's education trajectories267
Figure 6.10. Children's likelihood of belonging to language trajectories based on parent's
occupation trajectories
Figure 6.11. Children's likelihood of belonging to internalizing behaviour trajectories based
on parent's occupation trajectories270
Figure 6.12. Children's likelihood of belonging to externalizing behaviour trajectories based
on parent's occupation trajectories271

List of tables

Table 3.1. Strata, electoral wards, and expected sample size from the MCS106
Table 3.2. MCS productive sample and reason for attrition at each wave
Table 3.3. Descriptive statistics and bivariate correlations for language ability at all data
collection points114
Table 3.4. Average household weekly income (£) at each wave
Table 3.5. Parent's level of education at each wave 121
Table 3.6. Parent's level of household occupation at each wave 124
Table 3.7. Bivariate correlations between SES dimensions at each wave 130
Table 2.8. Missing values from highest to lowest132
Table 3.9. Summary of the goodness of fit indices 140
Table 4.1. Descriptive statistics and Bivariate Correlations for language ability at all data
collection points156
Table 4.2. Parent's level of education and occupation at each time point
Table 4.3. Descriptive statistics of all variables in the analyses

Table 4.4 Regression models examining the associations between SES and children's
language ability in early, middle, and late childhood166
Table 4.5 Regression models examining the associations between SES and children's
internalizing behaviour in early, middle, and late childhood168
Table 4.6 Regression models examining the associations between SES and children's
externalizing behaviour in early, middle, and late childhood170
Table 5.1. Summary of the goodness of fit indices
Table 5.2 Descriptive statistics of all variables in the analyses
Table 5.3 Parent's level of education at each wave 197
Table 5.4 Parent's level of household occupation at each wave
Table 5.5 Indicators of model fit for family income 202
Table 5.6. Associations between family income trajectories and children's language ability,
internalizing and externalizing behaviour205
Table 5.7. Indicators of model fit for parent's level of education 207
Table 5.8. Associations between parent education trajectories and children's language
ability, internalizing and externalizing behaviour210
Table 5.9 Indicators of model fit for parent's level of occupation 211

Table 5.10. Associations between parent occupation trajectories and children's language
ability, internalizing and externalizing behaviour214
Table 6.1. Correlations between language ability measures at all data collection points239
Table 6.2. Descriptive statistics of all variables in the analyses
Table 6.3. Parent's level of education and occupation at each time point 248
Table 6.4. Indicators of model fit for children's language ability
Table 6.5 Indicators of model fit for children's internalizing behaviour 254
Table 6.6 Indicators of model fit for children's externalizing behaviour 257

Abstract

This thesis examined the associations between the dynamics of family socioeconomic status (SES) and children's language ability, internalizing behaviour, and externalizing behaviour. Longitudinal national cohort data from the Millennium Cohort Study was used which provided data from six time points when children were 9 months, 3, 5, 7, 11, and 14 years old. SES was conceptualised as family income, education, and occupation (social class) and was recorded at each time point. Children's internalizing and externalizing behaviour were recorded using the parental report Strengths and Difficulties Questionnaire (SDQ) when children were 3, 5, 7, 11, and 14 years old. Children's language ability was measured by British Ability Scale (BAS) Naming Vocabulary Scale when children were 3 and 5 years old, the BAS Word Reading scale when they were 7 years old, the BAS Word Similarities scale when they were 11 years old, and by the Applied Psychology Unit Standardised Vocabulary Test when they were 14 years old. This first study examined the extent to which family SES was associated with children's language ability, internalizing behaviour, and externalizing behaviour at different stages of childhood. The childhood stages examined were early, middle, and late childhood. Results identified that family income was associated with all child outcomes at all stages of childhood. Education was associated with all child outcomes at all stages of childhood apart from internalizing behaviour during late childhood. Occupation was associated with all child outcomes at all stages of childhood apart from internalizing behaviour during middle and late childhood. The second study examined whether different family SES trajectories could be found in the sample and whether the different trajectories were associated with children's language ability, internalizing behaviour, and externalizing behaviour. Results found several family income, education, and occupation trajectories and indicated that persistently low SES trajectories were associated with the most

language and behavioural problems. It also found that increasing SES trajectories over the life course was associated with improved child outcomes. The third study examined whether family SES trajectories were associated with children's likelihood of belonging to different trajectories of language ability, internalizing behaviour, and externalizing behaviour. Results found several outcome trajectories and showed that children were more likely to be in problematic outcome trajectories if their family SES trajectories were persistently low. Children were also more likely to be in high language ability and low behaviour problem trajectories if their family SES trajectories were also improving over the life course. The findings have implications for policymakers interested in improving children's outcomes by increasing family SES. A multidimensional approach to improving family SES was discussed as the results suggest that targeting parents' education and occupation might be beneficial in addition to current policies that target family income. A prolonged approach to supporting family SES at different stages of childhood was also discussed as the result indicated that each SES dimension was associated with children's outcomes at all stages of childhood and that different SES trajectories were associated with children's developmental trajectories over time. The possibility of targeting stress as a mechanism through which SES is associated with children's development was also discussed as an alternative to targeting SES dimensions directly.

Chapter 1: Introduction

1.1 Overview

Widening economic disparities continue to be a problem in the UK and the past decade has seen a steady increase in the number of children living in poverty with approximately one in three children in the UK living in poverty (Social Mobility Commission, 2020). Socioeconomic inequality is prevalent in the United Kingdom which is ranked 32nd out of 38 countries (OECD, 2022). This is problematic as research consistently shows that family socioeconomic status (SES) is associated with a myriad of child developmental outcomes such as language ability (Barnes et al., 2010; Buckingham et al., 2013), numeracy and problem solving ability (Khanam & Nghiem, 2016; Fomby & Musick, 2018), internalizing and externalizing behaviour (Zachrisson & Dearing, 2015), educational attainment (Reardon et al., 2013) and lower future employment and earnings (Chetty et al., 2011; Erola et al., 2016). Children with problematic developmental outcomes disproportionately belong to families with low SES which has led researchers and policymakers to focus on better understanding the inequality gaps in children's development (Clarke & Thévenon, 2022; ISD, 2019; Youth Wellbeing NI, 2020; NHS Digital, 2021; Scottish Government, 2022). Although the associations between SES and children's development are well established in the literature, there remain complexities such as the dynamics of SES that can influence the associations that need further examination. The dynamics of SES refer to factors such as the timing, change, movement, or stability of SES over time and they have been found to be important factors to consider in the associations between SES and children's development (Zachrisson and Dearing, 2015; Björkenstam et al., 2017; Miller et al., 2021). For instance, the associations between SES and children's development outcomes can vary depending on the stage of childhood and the trajectories of family SES over the life course (Mazza et al.,

2017; Burchinal et al, 2018; Miller et al., 2021). Examining whether different dynamics of SES influence the associations between SES and children's development will provide researchers and policymakers with better insight into how inequality gaps might be addressed.

1.2 Focus of this study

This study examined the associations between SES dynamics and children's language ability and socioemotional behaviour. Children's language ability and socioemotional behavioural development were examined due to the significance they have for other developmental domains and their later life outcomes (Zdebik et al., 2019; Straatmann et al., 2018; Hukkelberg et al., 2019; Hulme et al., 2020). For instance, children's language abilities are critical for gaining access to areas of the education curriculum (Hulme et al, 2015; Chow & Jacobs, 2016; Hulme et al., 2020), maintaining social relationships and interacting with their environments (Gillespie-Lynch et al., 2012; Duff et al., 2015) and their future educational and economic outcomes (Duncan et al., 2010, Johnson et al, 2010; Wade et al., 2018). Children's socioemotional behavioural development is important for creating and maintaining healthy social relationships (Smith & Smith 2010; Barthel et al., 2018), maintaining good mental and physical health (Liu, 2004; Sagatun, 2014; Peverill et al., 2021), learning to behave appropriately in society (Barthel et al., 2018), and achieving positive economic outcomes (Liu, 2004; Goodman et al., 2011). Across the UK, between 18% and 33.1% of children failed to meet their communication and language targets and between 11% and 21.9% reported borderline or abnormal socioemotional behavioural problems with children showing language and behaviour problems disproportionally belonging to disadvantaged families (StatsWales, 2019; ISD, 2019; Department for Education, 2019; Public Health England, 2020; Youth Wellbeing NI, 2020; NHS Digital, 2021; Public Health

Wales, 2021). Due to their significance, policymakers are keen to improve children's language ability and socioemotional behaviour and reduce the economic gap in their development (Youth Wellbeing NI, 2020; NHS Digital, 2021; Public Health Wales, 2021; Scottish Government, 2022).

This study defined SES as a multidimensional construct comprising of family income, parent's level of education, and parent's level of occupation which is the most widely used definition of SES (Bradley & Corwyn, 2002; APA, 2007) and is consistent with theory (Weber, 1968; Bourdieu, 1986; Coleman, 1988). This study examined two types of SES dynamics, the stage of childhood (timing) and trajectories of SES, because evidence indicates that children are sensitive to the negative effects of low SES during different stages of childhood (Blakemore, 2014) and because SES is not a static construct meaning people can transition between different social circumstances over their life to create different trajectories (Elder, 2003).

1.3 The rationale for the study

Although a lot of attention has been paid to the associations between SES and children's outcomes, there are some gaps in the literature that need to be addressed. First, SES is a multidimensional construct and includes measures such as income, education, occupation, poverty, and neighbourhood deprivation (McLoyd, 1998). The majority of the literature has examined the associations that family income or composite measures of SES have on children's language and behavioural outcomes which do not provide information about which SES dimensions have the largest associations with children's outcomes (Kiernan & Huerta, 2008; Violato et al., 2011; Letourneau et al., 2011; Khanam & Nghiem, 2016; Cooper & Stewart, 2017; Fitzsimons et al., 2017). Each SES dimension is theoretically distinct (Weber, 1968), and is associated with unique pathways known to be important for children's development (Coddington et al., 2014). For instance, parents' education is associated with cognitive and linguistic stimulation while family income is associated with standard of living each of which is associated with children's development (Coddington et al., 2014). It is therefore likely that the extent to which different SES dimensions are associated with children's outcomes varies. Examining the associations between different SES dimensions and children's language and behavioural outcomes would be useful as UK policy is interested in improving family income, level of education, and occupation (Welfare Reform Act, 2016). However, current policies largely focus on increasing family income (Northern Ireland Executive, 2016; Welsh Government, 2015; Scottish Government, 2018), so evidence that different SES dimensions are also important for children's language and behavioural outcomes might be useful for policymakers interested in targeting other SES dimensions.

Second, the extent to which SES is associated with children's developmental outcomes varies depending on the stage of childhood. It has generally been accepted that early childhood is the most important stage as it is when brain development is of critical importance for establishing neural structures and functions that are responsible for future cognitive and behavioural processes (Brummelte et al., 2017) and therefore would be most susceptible to the adverse effects of low SES. Consequently, the majority of studies examine the SES associations with children's language and behavioural outcomes during early childhood (e.g., Schoon et al., 2011; 2013; Dickerson & Popli, 2012; 2016; Burchinal et al., 2018). However, a growing body of literature has found that SES during middle and late childhood is also important (e.g., Sullivan & Brown, 2013; Mazza et al., 2017; Miller, et al., 2021). However, it is difficult to determine which stage of childhood (Schoon et al., 2011; 2013; Burchinal et al.

al., 2018) or middle and late childhood (Sullivan & Brown, 2013; Mazza et al., 2017; Miller, et al., 2021). Examining at which childhood stage family income, education, and occupation are associated with children's language and behavioural development would help policymakers understand when children would benefit from intervention or if prolonged intervention would be best especially since several policies already focus on early childhood (e.g., Child Payment is available when children are under 6 years old, Best Start Grant ends when the child enters schooling, early learning and childcare end when children are 4 (5 in Scotland)).

Third, SES is not a static construct and can change over the life course to form trajectories (Elder, 2003) and different SES trajectories have different associations with children's language and behavioural outcomes (e.g., Jackson et al., 2017; Björkenstam et al., 2017; Hoyt et al., 2019). However, the current SES literature only focuses on family income (Dearing et al., 2001; Schoon et al., 2011; Duncan et al., 2014; Zachrisson & Dearing, 2015; Jackson et al., 2017; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Comeau et al., 2018; Hoyt et al., 2019) which means the extent to which other SES dimensions have various trajectories and whether they are associated with children's outcomes remains unclear. Further, the studies that examine associations between income trajectories and children's language and behavioural outcomes tend to focus on short timeframes such as early childhood (Dearing et al., 2001; Zachrisson & Dearing, 2015; Jackson et al., 2017) or from mid-childhood to latechildhood (e.g., Miller & Votruba-Drzal 2017; Comeau & Boyle, 2018). Since trajectories are formed based on experiences and life events of the life course, when more data points over a longer period of time are used more trajectories are generally found (Cavanagh et al., 2016; Björkenstam et al., 2017; Oshio et al., 2018). Therefore, examining income, education, and occupation trajectories over the entire childhood timeline would give policymakers more

detailed information about how different SES trajectories over the life course are likely to impact children's later language and behavioural outcomes. This would be particularly useful given current policy interventions are aimed at improving family income, education, and occupation (e.g., No One Left Behind and Restart programmes (Scottish Government, 2020; Department for Work and Pensions, 2020).

Fourth, similar to SES, children's language and behavioural outcomes are not static and can also form trajectories over the life course (Mangin et al., 2017; Miller & Votruba-Drzal, 2017; Comeau, 2018) and static measures of SES have been associated with children's membership to different developmental trajectories (Ukoumunne et al., 2012; McKean et al., 2017; Hentges et al., 2019). Since SES is associated with children's developmental trajectories it is likely that SES trajectories are also associated with children's developmental trajectories, however, this has not been examined in previous studies. Examining the likelihood of children belonging to a particular language and behavioural trajectory based on their family SES trajectories would provide useful information for policymakers. For example, it would provide evidence on the extent to which children are likely to belong to improving language and behavioural trajectories if policy supported family SES dimensions.

1.4 Aims of this study

The overall aim of this study was to address the gaps in the literature mentioned above by examining the associations between different dynamics of family socioeconomic status and children's language ability, internalizing behaviour, and externalizing behaviour. First, this study aims to examine during which stages of childhood family income, parents' education, and parents' occupation are associated with children's language ability, internalizing behaviour and externalizing behaviour. By using secondary longitudinal cohort data from the millennium cohort study this study can examine SES associations during early, middle and late childhood as the data were collected from 9 months until 14 years old. Second, this study aims to explore trajectories of family income, parent education, and parent and occupation trajectories across the entire childhood timeline and examine the associations that each trajectory has with children's language and behavioural outcomes. Since using more data improves the likelihood of finding new trajectories compared to studies that used shorter timeframes (Cavanagh et al., 2016; Oshio et al., 2018), this study examined trajectories by using data collected from 6 time points (when children were 9 months, 3, 5, 7, 11, and 14 years old). It is also likely that new trajectories of parents' education and occupation will be found as this is not examined by current SES trajectory literature. Third, this study seeks to explore whether trajectories of family income, parents' education, and parents' occupation are associated with trajectories of children's language ability, internalizing behaviour, and externalizing behaviour. Predicted probabilities of belonging to particular outcome trajectories will be produced to examine the likelihood that children will belong to outcome trajectories based on their family income trajectories.

1.5 Research questions

This study attempts to address the following research questions:

1a) To what extent are dimensions of family SES (income, education, and occupation) uniquely associated with children's developmental outcomes (language development, internalizing, and externalizing behaviours)?

1b) Are some SES indicators more strongly associated with children's language and behavioural outcomes and is this consistent through to adolescence?

2) To what extent does the association between family SES and children's outcomes vary by stage of development?

3. To what extent does family income, parents' education, and parents' occupation change or remain stable from childbirth through late childhood?

4. To what extent are family income, parent's education, and parent's occupation trajectories associated with children's language and behavioural outcomes during late childhood?5. To what extent do children belong to different developmental (language, internalizing and externalizing behaviour) trajectories?

6. To what extent does the likelihood of belonging to different developmental (language ability, internalizing and externalizing behaviour) trajectories differ by family SES trajectory (stable, increasing, or decreasing SES)?

1.6 Structure of the study

Chapter 1 gives an overview of the study including a brief introduction to the context and rationale for this study. Chapter 2 gives a more detailed overview of the context and policy that is relevant to this study before presenting an overview of child development and a discussion on why language ability and socioemotional behaviour are examined in this study. The chapter then introduces SES and discusses its different conceptualisations before explaining how it is defined in this study. Theory that helps explain the associations between SES and children's development is then introduced followed by a literature review on the associations between SES and children's language ability and socioemotional behaviour. Following this, SES dynamics are introduced and a literature review of the importance of the stage of childhood for the association between SES and children's language and behavioural development. Further literature reviews on the associations between SES trajectories and children's developmental outcomes are presented. Research aims and questions are then outlined. Chapter 3 outlines the methodology of this study. It presents the background of the dataset that was used, how the data were collected, and the sample used. The outcome

measures were then described followed by independent variables and covariates. Data screening and missing data procedures are then discussed followed by an overview of the analytical procedures used for each of the studies.

Chapter 4 presents the first study examining the associations that family income parents' education and parents' occupation have with children's language ability, internalizing behaviour, and externalizing behaviour at different stages of childhood. It presents a literature review and research questions, the methodology, and a discussion on the regression models used. The results and their implications are then discussed. Chapter 5 presents the second study examining the different family income, parent education, and parent occupation trajectories and their associations with children's language ability, internalizing behaviour, and externalizing behaviour. It presents a literature review and research questions, the methodology and an overview of latent class growth analysis used to generate the trajectories and the relevant model fit criteria. The results and their implications were then discussed. Chapter 6 presents the third study examining the different trajectories of children's language ability, internalizing behaviour, and externalizing behaviour. It presents a literature review, research questions, and the methodology. The results were then presented in a series of graphs for improved interpretability and their implications were discussed. Chapter 7 provides a general discussion of this study. It gives an overview of the aims of the study before discussing the key findings and their implications. Overall implications were presented followed by limitations and future research recommendations. A final conclusion to end the study was presented.

Chapter 2 – Literature review

2.1 Aim of this chapter

This chapter has five sections. The first section sets out the context of the socioeconomic inequalities in children's language development and behavioural outcomes and the legislation in place to improve family socioeconomic status (SES) to reduce developmental inequalities. The second section gives a brief overview of child development and the key theories that explain how children develop. The third section presents different conceptualizations and measures of SES from the literature and presents how SES is conceptualized in the current study. In the fourth section, an overview of theory explaining why SES is associated with children's development is provided before moving on to present the literature on the associations between SES and children's language development, internalizing behaviour, and externalizing behaviour. This section will also discuss that the associations between SES and children's development differ depending on the outcomes being examined and the SES dimensions being measured. In the fifth section theory about why different SES dynamics (stage of childhood, duration, and trajectories) are associated with children's outcomes. An overview of the current literature is presented on the association between SES and child development at different childhood stages, between different durations of low SES and children's development, between different trajectories of SES and children's development, and between SES trajectories and children's developmental outcome trajectories.

2.2 Context

Socioeconomic inequalities in children's language proficiency and socioemotional behavioural development are widespread across Organization for Economic Cooperation and Development countries (Clarke & Thévenon, 2022; OECD, 2022). In all 38 OECD countries, children with the best language ability and socioemotional outcomes disproportionately belong to families with a higher socioeconomic status indicating that inequality remains a pervasive issue for children's development (Clarke & Thévenon, 2022). Socioeconomic inequality is particularly problematic in the United Kingdom as according to the most recent OECD data, the UK has one of the highest levels of economic inequality among OECD countries ranking 32nd out of 38 countries (OECD, 2022). Across the four UK nations, between 18% and 33.1% of children failed to meet their communication and language targets and between 11% and 21.9% of children were reported to show borderline or abnormal socioemotional behavioural problems (StatsWales, 2019; ISD, 2019; Department for Education, 2019; Public Health England, 2020; Youth Wellbeing NI, 2020; NHS Digital, 2021; Public Health Wales, 2021; Scottish Government, 2022; Moore et al., 2022). In each UK nation, children with language and socioemotional problems disproportionally belonged to disadvantaged families (ISD, 2019; Youth Wellbeing NI, 2020; NHS Digital, 2021; Scottish Government, 2022). For instance, at the time of writing the most recent data in the UK comes from Scotland which shows that 80.7% of the least deprived children but only 56% of the most deprived children achieve their expected literacy score (Scottish Government, 2022). Similarly, 25% of the most deprived children but only 9% of the least deprived children show borderline or abnormal socioemotional behavioural problems (Scottish Government, 2022). This is more concerning since child poverty has been forecast to rise to approximately 38% even prior to the COVID-19 pandemic (e.g., Scottish Government, 2018a). Addressing inequality related developmental gaps is therefore a priority for the UK.

2.3 Policies to address developmental inequalities

Legislation in effect in the UK and the devolved nations mandate that local authorities improve children's language and socio-emotional development and reduce the developmental inequality gap (The Childcare Act (2006); Well-Being of Future Generations (Wales) Act (2015); Children's Services Co-operation (Northern Ireland) Act (2015); The Education (Scotland) Act (2016)). Based on this legislation, The Early Years Foundation Stage (England), the National Improvement Framework (Scotland), Children and Young People's Strategy (Northern Ireland), and the Early Years Outcomes Framework (Wales) are presented to outline the objectives of each UK nation and strategies to achieve them. These frameworks have made reducing the socioeconomic inequality gap in children's developmental outcomes an explicit priority with children's language skills and socioemotional development being key indicators. For example, The National Improvement Framework (2021, pp. 6) in Scotland outlines the following four key objectives:

- "Improvement in attainment, particularly in literacy and numeracy.
- Closing the attainment gap between the most and least disadvantaged children and young people.
- Improving children and young people's health and wellbeing
- Improving employability skills and sustained, positive school-leaver destinations for all young people".

Since low-income children are most likely to have problematic development, policymakers have put in place legislation to support low-income families with children. For instance, the Child Poverty (2010) Act pledged to end child poverty by 2020. This act required each UK nation to publish their strategies for tackling poverty and set the following income targets:

- "Less than 10 per cent of children live in households that are in relative poverty.
- Less than 5 per cent of children live in households that are in absolute poverty.
- Less than 5 per cent of children live in households that are in combined low income and material deprivation.
- Less than 5 per cent of children live in households that are in persistent poverty."

In 2010 the UK government contended that narrowly focusing on income-based poverty targets encouraged policymakers to focus on mitigating the effects of low income rather than addressing the causes of low income in the first place (Department for Education, 2011). The government argued that a multidimensional approach was required to reduce current child poverty and prevent future child poverty (Department for Education, 2011). This resulted in The Welfare Reform Act (2012) which aimed to target the underlying social causes of poverty including education and employment in addition to low earnings. The Welfare Reform and Work Act (2016) repealed the statutory requirement the report on income-based poverty targets and publish child poverty strategies. Instead, the Welfare Reform and Work Act (2016) required authorities to measure, report, and improve levels of household worklessness and educational attainment at age 16 as they were seen as 'root causes' of poverty (Department of Work and Pensions, 2017). This framework allowed policy to target broader indicators of poverty rather than income alone.

The devolved nations of the UK have maintained the income-based poverty targets set by the Child Poverty (2010) Act and continue to measure and publish child poverty plans (e.g.,

Northern Ireland Executive, 2016; Welsh Government, 2015; Scottish Government, 2018b). The Scottish Government has more devolved powers than the other devolved nations in the UK and has more control over their policies. The Scottish government states that child poverty is primarily due to income and passed The Child Poverty (Scotland) Act 2017 (Scottish Government, 2017) which stipulated that the targets set out in the Child Poverty Act (2010) should be met by 2030. Therefore, Scotland's child poverty plan is largely focused on income but also includes some focus on improving employment and education. Despite differences in reporting requirements between the four UK nations, policies tackling child poverty are similar and take the form of three broad categories: *Increasing income from employment, social security, and reducing the cost of living*.

Increasing income from employment is the UK government's primary focus for reducing child poverty and they aim to achieve this by improving families working status and income from work. Policies like The Work Programme and The Restart Scheme (Department for Work & Pensions, 2011; 2022) are aimed at getting people into work who have been unemployed for over a year and the Help to Work scheme designed to help them beyond that if they are still unemployed by giving training or work experience. Similar policies are in place in Wales (e.g., the ReAct+ programme, Communities for Work programme, and Parent's Childcare, and Employment programme; Welsh Government, 2022), Scotland (e.g., No One Left Behind and Fair Start Scotland; Chief Economist Directorate, 2022), and Northern Ireland (e.g. the Step Up to Sustainable Employment programme, Northern Ireland Executive, 2021).

Policies also aim to encourage working by making it more beneficial to be in work than not. For example, the Benefit Cap assumes that benefits would not total more than the average

weekly wage and the rollout of Universal Credit incentivises work as it allows families to work more hours without revoking their benefits which leaves them better off (Department for Work & Pensions, 2017; Low Pay Commission, 2022). The National Living wage of £7.20 was also introduced in 2016 which at the time reflected an additional £910 per year for full-time employees who were on minimum wage (Low Pay Commission, 2022). The National living wage has been increased year on year and is currently at £9.50 (Low Pay Commission, 2022). The income tax allowance has also allowed families to keep more of their earnings (Low Pay Commission, 2022).

Policy is also in place to help workers progress in their jobs through training and raising their level of qualifications with the goal of increasing their income (Welfare Reform and Work Act, 2016; Scottish Government, 2018b; 2023; Welsh Government, 2022; Northern Ireland Executive, 2021). Across England and Wales, the National Careers Service currently helps parents choose what type of education would benefit them and provides free English and maths education for low-educated parents and funding is available for adult apprenticeships and traineeships (Department for Business Innovation & Skills, 2012). Parents can also enrol in college or university courses with some support from grants such as the Lone Parent Grant and the Lone Parents' Child Grant and they are exempted from council tax if studying full time. Wales also offers Community Employability Programmes (Communities for Work+ and Parents Childcare and Employment) designed to support disadvantaged people get into work by providing training and work experience (Welsh Government, 2022). Northern Ireland also does not deviate far from the UK government policies however it also offers the Step Up to Sustainable Employment programme which is designed to reduce the barriers to employment by helping people with appropriate training and education (Northern Ireland Executive, 2021). In Scotland, if parents are under 26 years old their tuition fees are funded

by Student Award Agency Scotland. In Scotland, Fair Start Scotland and No One Left Behind programmes help parents from the most disadvantaged families overcome their barriers to employment so that they can gain sustained employment (Chief Economist Directorate, 2022). Fair Start Scotland also helps currently employed low-income parents progress in their careers with the aim of increasing their income (Chief Economist Directorate, 2019). There is an acknowledgement that the skills required to participate in the labour market are changing which risks job loss. Policy aims to help working families by providing education, training and reskilling for higher-paid positions and job retention. For example, the Flexible Workforce Development Fund is in place for employers to fund academic or non-academic training to up-skill and re-skill their current employees (Scottish Government, 2018b). The National Transition Fund also aims to support new jobs, educate, and reskill people for jobs of the future such as high-tech and high-skilled jobs and green jobs (Scottish Government, 2018b).

Social security entitlements aim to increase family income. The primary social security policy in the UK is Universal Credit which combines Employment and Support Allowance, Jobseeker's Allowance, Housing Benefit, Working Tax Credit, Child Tax Credit, and Income Support into one payment (Welfare Reform Act, 2012). Families can also claim child benefits for children under 16 or under 20 if they are in full-time education. Additionally, in Scotland, low-income families also receive the Scottish Child Payment for children under 6 years old which was initially £20 per week (Scottish Government, 2018b) and was increased to £25 per week in 2022 (Scottish Government, 2022b). The Best Start Grant also provides families with a payment at key stages of children's lives; The Pregnancy and Baby Payment from pregnancy until the child is 6 months old; the Early Learning Payment when the child is between 2 and 3.5 years old; and the School Age Payment when the child is old enough to go

to school (Scottish Government, 2018b). Similarly, the Best Start Food Grant provides monthly payments on to a prepaid food card to families until the child is 3 years old (Scottish Government, 2018b). Due to fewer devolved powers, Wales and Northern Ireland have limited power to implement social security benefits beyond what is available at the UK level (e.g., Universal Credit) and therefore do not have specific payments for poor families with children (e.g., like the Scottish Child Payment). However, it has been increasingly recommended that they both implement a similar policy to help lift children out of poverty (Birt et al., 2022). Although they have limited powers, the Welsh Government provide support to low-income families with children under 5 from the Child Investment Fund (Welsh Government, 2022). Although Northern Ireland does not have specific benefits for child poverty, the Fresh Start Agreement (Northern Ireland Executive, 2015) provides financial top-ups to low-income families who have been negatively affected by the benefit cap. Welfare Supplementary Payments are also available for low-income families that have continually been in receipt of benefits since 2016 in an attempt to continue to support these families (House of Commons, 2022).

Reducing the cost of living helps families retain as much of their income as possible. Several policies such as the Warm Home Discount, Water Sure scheme, and Healthy Start are all designed to mitigate the cost of essential consumables in each of the four UK nations (Department for Environment Food and Rural Affairs 2012; Health and Social Care Act, 2012; The Healthy Start Scheme and Welfare Food (Miscellaneous Amendments) Regulations, 2020; Department for Business, Energy, & Industrial Strategy, 2022). Free School Meals are also a crucial benefit designed to support families and have been extended to all primary school pupils in Scotland and they are available all year round including holidays (Scottish Government, 2018b). Wales are also a iming to provide free school meals
to all primary school children by 2024 and offer a Pupil Development Grant to ensure children have a school uniform and equipment for school (Welsh Government, 2023). In Northern Ireland, free school lunches are only available for children if their parents receive Universal Credit, have a parent studying full time, and have an income of less than £14,000 (Birt et al., 2022).

Early learning and childcare provisions are available for low-income families in Scotland, England, and Wales allowing families to receive free childcare, but at present, there are no free childcare provisions in Northern Ireland (Department for Education, 2018; Scottish Government, 2018b). In Northern Ireland, parents may apply for 12.5 hours of free preschool education during term time but have no specific free childcare (Employers for Childcare, 2022). Eligible families in England are allowed up to 30 hours of free childcare per week during term time for 3-4-year-old children totalling 1,140 hours of free childcare per year (Department for Education, 2018). Scotland offers the same 1,140 hours of free childcare per year but has increased flexibility by allowing families to distribute their 1,140 hours across the entire year while England only allows families to use their 1,140 hours during term time. Scotland also provides 1,140 hours of early learning and childcare for 2year-old children for low-income families while all families are entitled to the same service for their 3- and 4-year-old children (Scottish Government, 2018b). In Wales, families can get up to 30 hours of free childcare per week for 48 weeks for 3-4-year-olds totalling 1,440 hours per year (Welsh Government, 2022). This simultaneously improves the parents' availability to work more flexible hours whilst reducing the childcare costs for doing so. Similarly, low-income families are entitled to free after-school care for children so that they can be more available for work. Allocation is now available via The Pupil Premium in the UK and Pupil Equity Fund (PEF) in Scotland for schools to absorb the costs of childcare for

older children by providing after-school care or study allowing parents of older children to be more available for work (Scottish Government, 2018b). The Pupil Premium and PEF are discretionary funds that can be used by schools to improve children's chances and reduce the financial costs of schooling for parents (e.g., for school uniforms, activities, after school tuition).

Overall, apart from a spike during the 2008 recession, income inequality in the UK has remained consistent for the past 20 years despite the above-mentioned policies that have been implemented (Cribb et al., 2022). However, over the same 20-year period the child poverty rates have fluctuated. In 1999, after the government pledged and took action to end child poverty by 2020 (Piachaud & Sutherland, 2001), the child poverty rates began to decline from approximately 33% in 2000 to 28% in 2005 (Joseph Rowntree Foundation, 2022). However, the child poverty rates began to increase leading up to the 2008 recession peaking at 31%. After the 2008 recession and the implementation of the Child Poverty Act (2010), child poverty declined from 31% to 27% and remained at 27% for several years (Joseph Rowntree Foundation, 2022). However, from 2014 onward child poverty steadily rose and was back to 31% in 2020 (Joseph Rowntree Foundation, 2022). The current forecast for child poverty was anticipated to rise in 2022/2023 and this was before the energy crisis, cost of living crisis, and rising inflation (Francis-Devine, 2022). It is therefore likely that child poverty will be worse than the forecast if no interventions are implemented.

2.4 Child Development

Child development refers to the ways in which children change over time from when they are born until they reach adulthood (Kail, 2015). There are many domains of child development including physical, cognitive, language, social, and emotional development (Kail, 2015). Since the domains of child development cover a wide spectrum of abilities, skills and dispositions, I will focus on children's language ability and socioemotional behaviour (internalizing and externalizing behaviour) because of their significance for other developmental domains (Liu, 2004; Schoon et al., 2010; Park et al., 2011; Broeren et al., 2013; Hukkelberg et al., 2019; Hulme et al., 2020) and the impact that they can have on life course outcomes (Johnson et al, 2010; Duncan et al., 2010; Goodman et al., 2011; Wade et al., 2018; Pedersen et al., 2019).

Children's language abilities are critical for their future development and learning because it determines children's ability to understand, interact, and communicate competently with people in their environments (Gillespie-Lynch et al., 2012; Duff et al., 2015) and gives them access to other areas of the education curriculum such as reading and numeracy (Hulme et al, 2015; Chow & Jacobs, 2016; Hulme et al., 2020). Children's language ability is one of the most reliable predictors of their school readiness (Hoff, 2013), educational attainment (Wade et al., 2018), occupational outcomes (Johnson et al, 2010), and future income (Duncan et al., 2010). It is also important for children's social relationships as language development is essential for communication. Children with lower vocabularies are less able to express themselves and are more likely to have health issues (Schoon et al., 2010; Park et al., 2011), and relationship problems (Britto et al., 2017). Deficits in language ability are therefore detrimental to children's overall development and their ability to thrive in their societies.

Socioemotional behavioural development (SEB) enables children to create and maintain social relationships, behave appropriately, and navigate their social worlds (Barthel et al., 2018). SEB normally comprises externalizing and internalizing behaviour (Löytömäki et al., 2022). Internalizing behaviours are directed toward the self rather than towards others and include anxiety, depression, poor self-regulation, and other mental health problem (Broeren et al., 2013). Childhood internalizing behavioural problems are associated with poorer developmental outcomes such as education attainment (Gustafsson et al., 2010; Smith & Smith, 2010; Pedersen et al., 2019), mental and physical health (Sagatun, 2014; Peverill et al., 2021), social relationships (Smith & Smith 2010), and future SES (Goodman et al., 2011). Internalizing problems experienced during childhood can persist into adolescence (Zdebik et al., 2019; Straatmann et al., 2018). Externalizing behaviour on the other hand refers to behaviours that are projected outward towards the environment (Eisenberg et al., 2001). During early life externalizing behaviour can be seen as benign such as lying or arguing (APA, 2013) but can escalate as the child gets older to include behaviours such as physical and verbal aggression, delinquency and hyperactivity (Liu, 2004). Children with childhood externalizing behaviour problems are more likely to be involved in crime later in life that leads to criminal convictions (Liu, 2004), have poorer educational outcomes and occupational attainment, lower future income, have poor mental health (Liu, 2004), struggle to maintain social relationships and have an overall lower level of social competence (Hukkelberg et al., 2019). Overall, socioemotional behavioural problems hinder children's ability to successfully develop in a healthy way that helps them thrive in their societies.

Due to the significance that children's language ability and socioemotional behavioural development have for other developmental domains and their later life outcomes, they have both attracted significant policy attention in each of the four UK nations with policymakers keen to improve children's language and behavioural outcomes and reduce the economic gap in their development (StatsWales, 2019; ISD, 2019; Department for Education, 2019; Public Health England, 2020; Youth Wellbeing NI, 2020; NHS Digital, 2021; Public Health Wales, 2021; Scottish Government, 2022). This study will therefore examine the associations between different family socioeconomic dynamics and children's language and socioemotional behavioural outcomes.

2.4.1 Theory of child development

The Bioecological Systems Theory (BST) is an evolving theory which presents an explanation as to how children's environments are associated with their development (Bronfenbrenner, 1986; Bronfenbrenner & Morris, 2006). It presents the environments as four concentric layers ranging in proximity from the developing child called the microsystem, mesosystem, exosystem, and macrosystem (see figure 1; Bronfenbrenner, 1976; 1979). The microsystem is the most immediate system that the child interacts with and includes the contexts that the child has direct contact with (e.g., family, home, and school; Bronfenbrenner, 1974; 1979). These are the environments that children can engage with directly and how they engage will influence their development (Bronfenbrenner, 1976; 1979). The mesosystem is where two different microsystems interact to influence children's development (e.g., do the home and school environments interact to influence development; Bronfenbrenner, 1979). The exosystem are environments that can influence children's development indirectly (e.g., parents' place of work; Bronfenbrenner, 1974; 1978; 1979). The macrosystem represents the wider societal context that the child lives in (e.g., cultural, institutional, and political contexts; Bronfenbrenner, 1976; 1978;1979). A fifth system, called the chronosystem, refers to 'time' and focuses on how the timing of events, consistency or changes in the environment, and life transitions can influence children's development (Bronfenbrenner, 1986; 1989).



Figure 1. Ecological Systems Model (Bronfenbrenner, 1979)

The BST proposes that complex bidirectional interactions between the child and their environment are the primary drivers of child development (Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Morris, 1998; Bronfenbrenner 1999; Bronfenbrenner & Evans, 2000). These bidirectional interactions are termed proximal processes and include interactions such as parents reading and playing with children, and children interacting with objects in their homes (Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Morris, 1998). The BST makes two main proposals about what is needed for proximal processes to facilitate child development. First, interactions between the child and their environment must be frequent, reciprocal, happen over extended periods of time, and be increasingly complex Bronfenbrenner (Bronfenbrenner, 1986; Bronfenbrenner & Ceci, 1993). Second, the extent to which interactions between children and their environments have positive or negative associations with children's development is dependent on the characteristics of the child, their immediate and remote environments, and the outcomes being examined (Bronfenbrenner & Morris, 1998; 2006). Bronfenbrenner states that better development occurs when positive relationships are formed with people in the environment and if the environment provides the opportunity to observe and engage in activities with assistance from more skilled people to facilitate learning (Bronfenbrenner, 1973; 1979b).

Family socioeconomic status (SES) has consistently been highlighted as a key environmental factor in children's development (Sirin, 2005; Letourneau et al., 2011; Cooper & Stewart, 2013; 2017). It is proposed that family SES is important for children's development because it can shape the home environment and influence the interactions that occur within it (Conger & Donnellan, 2007; Conger & Conger, 2010) and these interactions are ultimately what shape children's development (Bronfenbrenner, 1986; Bronfenbrenner & Morris, 2006). The next section will provide an overview of SES by discussing how SES has been conceptualized in the literature and how it is defined in this study.

2.5 Conceptualisation of SES

The following section will discuss how SES is conceptualised in this study. It does this by discussing the theoretical considerations that helped define SES and the different measures that can be used to represent SES. There will be a discussion on key factors that led to this study conceptualising SES as a multidimensional construct and why family income, education, and occupation were chosen to represent the different SES dimensions.

2.5.1 Theoretical considerations

SES is a multidimensional construct but has no universally accepted definition, and the appropriate measurement of SES has been subject to debate for decades (e.g., Mueller & Parcel, 1981; White, 1982; Entwisle, & Astone, 1994; Gottfried, 1985; McLoyd, 1998; Sirin,

2005). The issue with defining SES is that agreement on how best to measure the different economic and social resources has not been reached. Researchers conceptualize SES based on the measures that they choose to use, which are determined by their specific research questions. This leads to considerable debate about how best to measure and define SES. With no current single definition or standard measurement of SES (Oakes & Rossi, 2003), researchers often draw on the broadly accepted general understanding that SES is a combination of economic and social resources and refers to an individual's social position based on their ability to access and deploy financial, human, and cultural resources (APA, 2007). In the literature, SES is most commonly operationalised by measuring income, education, and occupation.

The theoretical rationale for conceptualising SES as a combination of economic and social resources and measuring it with income, education, and occupation is largely credited to theories proposed by Marx (1967) and Weber (1968). They each present how social stratification occurs. Marx (1967) proposed that an individual's SES was determined at birth and could be explained entirely by their social class, which was measured by an individual's relationship to the labour market. His theory proposed a two-class stratification system whereby individuals' social classes are differentiated based on whether they owned the means of production (e.g., land, factories, business etc.) or were an employee.

Weber's (1968) theory also indicates that social class is an important factor in determining SES. However, his work differs from Marx in several ways. First, Weber believes that there are multiple social class categories that people are stratified into (e.g., upper class, upper middle, lower middle, and lower class) rather than the two categories proposed by Marx (owner and employee). Second, Weber takes a multidimensional approach to understanding

SES, while Marx proposes a unidimensional approach. Weber suggests that people are stratified based on their positions across several different indices in addition to social class. He proposed three dimensions that could be used to stratify people: classes, status groups, and parties. Classes can be represented by economic resources such as income or occupation, status groups can be represented by the level of prestige in the community such as an individual's level of occupation or education and skills (Liberatos et al., 1988), and parties refers to one's access to power which can be represented as income and occupation. Weber argues that an individual's SES is determined by their ability to utilise their education, skills, networks, and other qualities to their advantage to navigate their social environment. People are therefore stratified into SES groups which are comprised of individuals who score similarly across several different dimensions. Third, Weber notes that SES is not a static construct and that individuals are active participants in generating their SES as they can improve their standing across different SES indices over time (e.g., increased income, education, or occupation) ultimately improving their overall SES.

Like Weber, other theorists also suggest a multidimensional approach to conceptualising SES such as Bourdieu (1986) and Coleman (1988). They both propose that stratification is based on an individual's access to, and ability to deploy, different forms of capital (e.g., economic, human, and social). *Economic capital* refers to the economic resources that people can access immediately to purchase goods or services such as income. *Social capital* refers to the formal (institutions such as occupations) and informal (such as peer groups) social networks that individuals accrue over time and have available to them. *Human capital* refers to the education, skills, and knowledge that people have gained in their environments. Each form of capital is measured differently and represents a distinct component of SES.

2.5.2 Individual vs composite measures

Researchers have used the different measures of SES individually or combined to create a composite measure of SES. Using a composite measure of SES is advantageous in some research scenarios as it is more likely to capture the greater overall association that SES has with the outcome of interest as it will encompass the additive effects of the multiple dimensions (Cowan et al., 2012). In this scenario it would be a useful way for researchers to determine an overall association between SES and outcomes of interest where the specific mechanisms between SES and the outcome were not the focus of the study. A composite measure of SES may also be useful for research that wish to use SES as a covariate in the relationship between another independent variable and an outcome.

Researchers have utilised several SES indicators to generate their own composite measures of SES from data available to them. For example, in the UK Britton et al. (2021) used local deprivation measures and free school meal eligibility as an SES scale. Hanscombe et al. (2012) generated a composite SES index using parental education, occupation, and income. Schoon et al. (2021) constructed an SES composite using parental education, social class, parental worklessness, family income, and home ownership. Duckworth & Schoon (2012) use a socioeconomic index of parental education, parent social class, lone parent, teen parent, housing tenure, and worklessness. Dearden et al. (2011) generated a measure of SES using equivalised income, mother and father's class, housing tenure, the experience of financial difficulties, and material deprivation. Recently the Civil Service recommended that employers interested in the SES of their employees create a composite score from their occupation, the highest qualification of their parent, the type of secondary school the individual attended, and whether they were eligible for free school meals. They note that any number of other measures could be used in addition to their recommendations including subjective measures of SES (Civil Service, 2018).

The downside of using a composite measure is that it does not allow for a proper examination of the individual associations between the different SES indicators and the outcome of interest (Eriksson et al., 2021). This is because the associations of each SES indicator could be masked by the associations of other indicators if they are combined as a composite. The unique mechanisms through which different SES indicators operate are also neglected when they are combined as a composite (Cowan et al., 2012; Eriksson et al., 2021). Composite measures of SES are limited in their ability to help inform policy beyond understanding that SES is important for children's developmental outcomes as they limit the amount of information that we can see about the associations between SES and children's development (Cowan et al., 2012). To properly benefit policy, we need to know which SES indicators are being measured and understand the extent to which each SES indicator is associated with children's outcomes to ensure that attention and resources are targeted at the appropriate dimension. This is evident in the literature which indicates that different SES indicators have different associations with different outcomes of interest but also that they have different associations when regressed on to the same outcome (Darin-Mattsson, 2017). This suggests that more useful information is likely to be gathered when using individual SES indicators. Therefore, this study will examine SES indicators independently.

2.5.3 The importance of SES dimensions

Researchers that opt to use individual SES measures must select which indicators to use. When selecting which SES indicators to use, the literature indicates that SES is most often conceptualised as family income, parental education, and parental occupation (McLoyd, 1998; Bornstein & Bradley, 2003; Darin-Mattsson et al., 2017). It has been argued in the literature that SES indicators can be used interchangeably such that the associations between education and child outcomes are used to support the associations between income or occupation and child outcomes (Geyer et al., 2006). This practice largely stems from Lazarsfeld's (1939) research that found different measures of economic status had similar associations with outcomes. From this, it was argued that it was likely that the different indicators of economic status measured the same underlying construct and therefore the measures could be used interchangeably.

However, this is problematic. Although income, education, and occupation are correlated with each other (Darin-Mattsson et al., 2017; Festin et al., 2017) they represent theoretically distinct constructs that represent the different economic and social resources a family can access (Weber, 1968; Bourdieu, 1986; Colemann, 1988; Jæger, 2007; Bukodi & Goldthorpe, 2013; Harding et al., 2015; Festin et al., 2017). It is also noted in the literature that the correlations between different SES dimensions are generally small to moderate and therefore ideally one should not be used to represent another (Geyer, 2006; Hoffmann et al., 2019). Each indicator is proposed to operate via unique pathways (Conger et al., 2010; Harding et al., 2015; Erola et al., 2016; Davis-Kean et al., 2021; Thaning, 2021) so even if the overall association between different indicators and outcomes were similar, it may not be due to the same reasons (Gayer et al., 2006). Using different SES indicators interchangeably is also problematic for the interpretation of results as claims made about low SES measured by different SES indicators may be less beneficial for policymakers because they can differ depending on the SES dimension measured or the outcome being examined (Avlund et al 2003; Davis-Kean, 2005; Letourneau et al., 2011; Sullivan & Brown, 2013; Boe et al., 2014; Coddington et al., 2014; Jackson et al., 2017; Cooper & Stewart, 2017; Fitzsimons et al., 2017; Gonzalez et al., 2020). For example, Boe et al (2014) indicate that family income had the strongest association with children's internalizing problems, but parents' education had the strongest association with children's externalizing problems. Therefore, policymakers

would benefit from a more accurate understanding of which SES dimensions are more important for children's language and behavioural outcomes.

2.5.4 SES Measurement

Drawing on the above theories gives researchers a wide scope of possible measures to define SES in their studies. At the individual and family level, objective SES measures often include family income, parental education, and parental occupational class, which are referred to as 'The big 3' (Willms & Tramonte 2019). Income, education, and occupation are generally accepted to be the main indicators of SES (Duncan et al., 1972; Bradley & Corwyn, 2002) and are the most commonly measured indicators (Smith & Graham, 1995; Bradley & Corwyn, 2002; Bornstein et al., 2003; Duncan & Magnuson, 2003). They can be used individually or in different combinations as a composite latent construct to measure SES.

2.5.4.1 Occupation

Occupation is considered a primary measure of SES and has been used to determine the social class in the UK for some time. For example, between 1921 and 2001 the Registrar General's Social Class (SC or Social Class based on Occupation) and Socioeconomic Groups (SEG) were used to measure social class via occupation. In 2001 the National Statistics Socio-economic Classification (NS-SEC) replaced both of these measures by drawing on their advantages. The NS-SEC is based on Goldthorpe's schema (1997) which allows occupations to be collapsed into a small number of meaningful categories by considering employment relations (Rose & Pevalin, 2003). Determining social class via occupation employment relations reflects people's working conditions, skill requirements, and their social networks (Fujishiro et al., 2010; Muntaner et al., 2010). Employment relations refer to the nature of the relationship between the employee and the employer and are usually

determined by service relationships or labour contracts. Service relationships and labour contracts generally stem from the ease at which the employer can define and monitor the job requirements. If the employer can define the role requirements, then they can more easily assign earnings by monitoring productivity (e.g., hourly rate) and dictating terms of the labour contract (Goldthorpe, 2007).

Labour contracts are more likely to be assigned to working-class roles because they are normally easier to define and monitor because less specialist knowledge is required. This means that the employer can assign an hourly rate based on the amount of work and productivity (Chandola, 2001) and tie a worker's earnings to their productivity. These roles generally have poorer working conditions, less autonomy as the role requirements are defined, less working flexibility, work under supervision, and employees are more easily replicable as specialist skills are not required (Goldthorpe, 2007; Williams, 2016). Service relationships are more likely for upper-class roles where skills and specialist knowledge are required. Due to the required knowledge, it is more difficult for the employer to define the precise job requirement and monitor the employee's progress, so it is difficult for the employer to assign earnings based on productivity. Since skilled employees are more difficult to replace it is in the employer's best interest to incentivise the employee to continue working in the interest of the business. Employees with a service relationship, are therefore likely to have more working autonomy and flexibility, less supervision, greater job security, and promotion opportunities (Goldthorpe 2007; Williams, 2016). Data on occupational employment relations are normally gathered by asking parents to state their current or most recent occupation through open-ended questions where probes can be used to better ascertain the requirements of their job and more accurately determine their relationship to the labour

market. Alternatively, parents are asked to report their occupation from a list of occupational categories.

2.5.4.2 Education

Education level is another common measure of family SES. Education reflects parents' knowledge and skills that can be utilised to improve their children's life chances (e.g., increased communication, positive family values towards education, higher expectations in terms of developmental outcomes and behaviour etc.). Since parents' education is usually completed during their youth it is considered a more stable indicator of SES compared to income and occupations which can decrease over time (Duncan et al., 2002). This means that parental education is a particularly good indicator of children's early life SES as parents' education is likely to be established before pregnancy, remain the same through birth, and cannot decrease. It is normally measured as either a continuous or categorical variable. If measured as a continuous variable, then parents would be asked to indicate the number of years spent in education and as a categorical variable, they would be asked to indicate the highest level of qualification achieved (Galobardes, 2006).

2.5.4.3 Income

Income reflects a family's financial resources and the purchasing power that individuals have access to which enables them to buy the materials and services that they need to thrive in their society. Income information has been gathered in different ways across the literature. It has been gathered by asking families to report their weekly, monthly, or annual income or asking families to indicate their income band from a selection (Office for National Statistics, 2020). Income data can also be equivalized to account for families' needs based on their household composition so that families with different compositions can more accurately be

compared (Office for National Statistics, 2020). Poverty status is also a common measure of income in the literature (Galobardes, 2006) and can be defined as absolute or relative (Townsend, 1979). Absolute poverty measures one's ability to meet a certain standard of living against a static threshold, meaning that whether the society is better or worse off, the threshold does not change to reflect it. Relative poverty measures one's ability to participate in their society, meaning the poverty threshold changes to reflect society's current state. In the UK relative poverty is defined as those who have an income that is more than 60% below the median household income. Income measures, therefore, relate to the financial risks of poverty (e.g., Short, 2016). Where information on family income is not available, proxy measures such as family wealth and free school meals can be used (Galobardes, 2006; Gorad, 2012).

Although family wealth can be used as a proxy for family income it is distinct in that it is a combination of household wealth (e.g., property asset value minus debt) and financial wealth (e.g., family savings and investments minus debt) while income indicates liquid assets able to pay for goods and services (Miller et al., 2021). However, wealth does not necessarily represent purchasing power or the liquid assets that people can immediately deploy for goods and services and is a more difficult SES dimension to measure than family income as it requires wealth indices to adequately represent the different facets of wealth (Poirier et al., 2019; Miller et al., 2021). Wealth is also less dynamic than family income (Duncan et al., 2002) with studies examining the intergenerational transfer of wealth showing that family wealth is relatively stable compared to family income (Hällsten & Thaning, 2022). Therefore, family income is more suitable for the purpose of the current study which focuses on the dynamics of SES over time. Children's eligibility for free school meals is also a common proxy measure of family income however it is less dynamic than family income due to it being a binary measure (Gorad, 2012). Children are eligible for free school meals if their

parents are unemployed or on an income support benefit such as Universal Credit (Taylor, 2017). While eligibility for free school meals is a common and convenient measure in research, some evidence indicates that it does not accurately represent the lowest income children once all benefits are accounted for (Gorard, 2012). For example, Hobbs and Vignoles (2010) found that only 25-50% of children eligible for free school meals belonged to the lowest income households meaning that some of the lowest income children were missed by the free school meal indicator. Similarly, Taylor (2017) found that approximately 8% of children living in chronic poverty did not meet the free school meal eligibility criteria.

2.5.5 Alternative measures

There are also alternative economic and social indicators of SES other than the 'big 3'. Housing circumstances are often used as an SES indicator which measures a family's material circumstances and accumulated wealth rather than a reflection of their purchasing power (Dalstra et al., 2006). Housing circumstances can be measured by tenure, conditions, and access to amenities. Housing tenure assesses whether the property is owned (either outright or through a mortgage) or rented. Housing conditions refer to whether the house is safe and clean to live in, for example, is it free from dampness and overcrowding. Access to amenities is measured by asking whether the house has clean running water, heating, or utilities such as washing machines. In the UK, the most frequently used measure of housing circumstances is housing tenure (Nightingale et al., 2018).

Housing circumstances are not examined in this study as they are closely linked to family wealth (Office of National Statistics, 2022, Advani et al., 2020) and therefore share similar limitations as family wealth as discussed above. For example, housing tenure is less dynamic than other SES dimensions as once a person becomes a homeowner, they tend not to become

a renter and recent data shows that homeowners tend to stay in the same home for over 10 years and over 20 years for those who have paid off their mortgage (Scottish Government 2019c, Ministry of Housing, Communities, and Local Government, 2020). People renting social housing on average stay in the same home for 12 years (Scottish Government 2019c, Ministry of Housing, Communities, and Local Government, 2020), while private renters live in the same home for between 2 and 8 years (Scottish Government 2019c, Ministry of Housing, Communities, and Local Government, 2020). Similarly, housing tenure is normally measured as a dichotomous variable assessing whether an individual is a homeowner or renter (Grundy & Holt, 2001; Khalatbari-Soltani et al., 2021) which is also less dynamic than other SES dimensions that offer more possible categories means it is less useful for this study focussing on SES dynamics.

Neighbourhood SES is another alternative measure of SES used in the literature and is measured using composite scales that reflect the level of deprivation of a neighbourhood based on the quality of the social and physical environment and the characteristics of the people that live there. In the UK, for example, the Townsend Deprivation Index (Townsend et al., 1988) measures unemployment, overcrowding, the proportion of households with no car ownership, and homeownership. The Carstairs and Morris index (1991) is similar and measures area SES as a composite of car ownership, occupation, household overcrowding, and male unemployment. More recently, the UK nations have used a multiple deprivation index that is slightly different based on their specific needs. They all measure income, education, occupation, health, disability deprivation, and area crime. England includes additional measures of barriers to housing and services and the family's physical environment (McLennan et al., 2019). In Scotland, the Scottish Index of Multiple Deprivation (SIMD) also measures access to services, and housing. Northern Ireland measures access to services and the physical environment, and Wales measures access to services, housing, and the physical environment. Neighbourhood SES was not used in this study because it does not always reflect an individual's SES since it is an aggregate measure of SES in a given geographical area (Galobardes et al., 2006) which is especially problematic in rural areas due to the greater heterogeneity rural populations (Pardo-Crespo et al., 2013; Xie et al., 2020). Additionally, as discussed in section 2.3.2 above, since neighbourhood SES is a composite measure, it does not allow for examination of individual dimensions of SES (Eriksson et al., 2021).

Subjective Social Status (SSS) measures are used less frequently in the literature as an alternative to objective SES measures. SSS reflects an individual's perception of their own SES compared to others in society. For example, the MacArthur scale of Subjective Social Status (Adler & Stewart, 2007) presents respondents with a hierarchical ladder of 10 points and an explanation that the top of the ladder is the most well-off people with high level jobs, education, high income, and the most respect, while people at the bottom of the ladder are worst off. Participants are then asked to select where they believe they belong on the ladder. Although SSS measures explicitly ask participants to think about dimensions of SES when determining their subjective SES, there is evidence that SSS is a better measure of social indicators (e.g., individuals' control over their life and social influence) than economic dimensions (e.g., income, education, and occupation) (Galvan et al., 2022). SSS is also proposed to operate via different pathways compared to objective measures of SES such as via emotions (Gallo & Matthews, 2003; Destin et al 2012), sleep quality (Moore et al., 2002), and health issues (Zahodne et al., 2018).

SSS was not measured in this study because it does not differentiate between SES dimensions which is what this study aimed to do. Instead, it is hypothesised in the literature that cognitive

averaging takes place (Singh-Manoux et al., 2003; Andersson, 2015; Präg, 2020) which means that participants consider their income, occupation, education, and standard of living as a whole in relation to others in society and indicate where they believe they belong on a social ladder (e.g., Langlois et al., 2020). Furthermore, people can report their perceived SES based on different factors as subjective measures are open to participants interpretations. For example, some people base their subjective SES on their past, present, and future potential SES as opposed to only their current SES (Singh-Manoux et al., 2005). People may also have different understandings of the levels of income, education, and occupation in society and therefore ascribe different subjective assessments on SES dimensions (Adler et al., 2000; Kraft et al, 2022). Therefore, SSS is arguably more useful as an overall indicator of SES, similar to composite measures, than a tool for examining individual dimensions of SES (Singh-Manoux et al., 2003; Zahodne et al., 2018; Amir et al., 2019; Präg, 2020).

2.5.6 Conceptualisation of family SES in this study

In this study, I conceptualise SES as a multidimensional construct comprised of a combination of economic and social resources and will measure family income, parents' level of education, and occupation. I chose these measures since the theories discussed in section 2.3.1 outline that each dimension represents a different type of resource that families have access to and can draw upon which should mean they each have unique associations with children's outcomes (Weber, 1968; Bourdieu, 1986; Coleman, 1988). Family income was chosen to represent the financial resources that families can access to purchase goods and services. It was chosen over proxy measures of income such as family wealth because wealth does not necessarily reflect a family's purchasing power and is less dynamic than income (Miller et al., 2021; Hällsten & Thaning, 2022). Parent education is measured because this most closely represents the human resources that people have (Alan et al., 2008). Occupation

was measured to represent social capital because it is likely that parents with different employment relations have different social networks through their work (Meanley et al., 2020).

The conceptualisation of SES in this study is also in line with current policy as the UK aims to reduce the socioeconomic inequalities in children's outcomes (The Childcare Act (2006); Well-Being of Future Generations (Wales) Act, 2015; Children's Services Co-operation (Northern Ireland) Act, 2015; The Education (Scotland) Act, 2016) by improving income, education, and occupation outcomes (Department for Work and Pensions, 2011;2022; Department for Education, 2021). To monitor progress toward these goals the UK government measures income, education, and occupation (Welfare and Work Reform act, 2016; Northern Ireland Executive, 2016; Welsh Government, 2015; Scottish Government, 2018b). Overall, the UK nations acknowledged that income, education, and occupation are all important SES dimensions for improving family SES and reducing the developmental inequality gap. However, they place different levels of importance on different SES measures. Researching the association between the different SES dimensions and children's outcomes will therefore be useful in determining which measures are more important in a UK context.

2.6 SES and children's developmental outcomes

2.6.1 Theoretical considerations

This section will outline the theoretical considerations that help explain why and how family SES is associated with children's language and behavioural outcomes to outline the framework that informed this study. First, the bioecological systems theory is outlined as it provides the overarching framework explaining why SES is important for children's

developmental outcomes by emphasising the role of the environment. Second, encompassed within the broader bioecological framework, the family stress and investment models outline how family SES is associated with children's developmental outcomes by emphasising pathways between family SES and children's outcomes. Evidence supporting the family stress and investment models is also provided.

2.6.1.1 Bioecological framework

As established in section 2.2.1, the Bioecological Systems Theory (BST: Bronfenbrenner, 1986) proposes that child development takes place due to complex bidirectional interactions (*proximal processes*) between the child and their environment (Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Morris, 1998; Bronfenbrenner 1999; Bronfenbrenner & Evans, 2000). *Proximal processes* have the strongest influence on children's development when they occur in the most proximal environments because these are the environments that children are exposed to most frequently (Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Morris, 1998). The *microsystem* is the innermost layer of children's environments and helps explain why SES is associated with children's development (Bronfenbrenner, 1979). In the microsystem, children have frequent direct contact with their environments (e.g., family and home) and the people in them and is therefore where *proximal processes* have the strongest influence on children's development and the strongest influence on children's development for the strongest influence on children's development (Bronfenbrenner, 1979; Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Morris, 1998).

Family SES is associated with children's development because it can determine the nature of microsystems that children directly interact with (Bradley & Corwyn, 2002; Sarsour et al, 2011). This means the type and quality of bidirectional interactions that occur between the child and their microsystems are likely to vary based on family SES (e.g., Conger et al.,

2010). This process is presented in the family stress and investment models, which propose several pathways in children's microsystems through which SES is associated with child outcomes (Conger & Donnellan, 2007; Conger et al., 2010). They each indicate that low SES can impair these pathways and therefore impair children's bidirectional interactions with them. These impaired interactions are proposed to lead to poorer developmental outcomes.

2.6.1.2 Family Stress Model

The Family Stress Model (FSM: Conger et al., 2010) outlines that SES is associated with children's development because it influences their microsystems which impairs the quality of bidirectional interpersonal interactions between children and their parents (a microsystem). The FSM proposes that SES is associated with children's development through parental distress and child-parent interactions. It states that economic hardship (e.g., low income) leads to economic pressures such as being unable to pay bills and buy nutritious food, clothing or other material needs. These economic pressures lead to parental distress (e.g., depression, anxiety, and stress) and increased conflict between parents (e.g., arguments and separations), which lead to more harsh or inconsistent parenting, neglect, and poorer parent-child interactions. These impaired bidirectional interactions lead to poorer developmental outcomes (see figure 2).



Figure 2. Family Stress Model (figure in Masarik & Conger, 2017)

2.6.1.3. Evidence Supporting The Family Stress Model

Evidence supports the FSM as low income (Katz, 2007; Shaw & Shelleby, 2014; Shelleby et al., 2014) and low occupation (Brooks-Gunn et al., 2010; Hadzic et al., 2013; Dockery et al., 2016; Cho, 2018; Moreira et al., 2019) are associated with increased parental distress which leads to poorer cognitive and behavioural outcomes (Kiernan & Manesh, 2009; Marryat & Martin, 2010; Avan et al., 2010; Neppl et al., 2016; Sosu & Schmidt 2017; Ahun et al., 2017). Increased parental distress can also lead to harsh, neglectful, and inconsistent parenting styles (Kohl et al., 2011; Micklewright et al., 2012; Rijlaarsdam et al., 2013; Shaw & Shelleby, 2014; Sutton et al., 2017) which are associated with worse child outcomes (Kiernan & Manesh, 2011; Micklewright et al., 2012; Shelleby et al., 2014; Sun et al., 2015; Sosu & Schmidt, 2017). Increased parental distress also increases family conflict (Benner & Kim, 2010; Ponnet, 2014; Landers-Potts et al, 2015; Fisher et al., 2015; Simons et al., 2016) and increases work-family conflict (Johnson et al., 2013; Dockery et al., 2016; Moreira et al., 2016; Moreira et al., 2013; Dockery et al., 2016; Moreira et al., 2016; Moreira et al., 2013; Dockery et al., 2016; Moreira et al., 2016; Moreira et al., 2013; Dockery et al., 2016; Moreira et al., 2016; Moreira et al., 2013; Dockery et al., 2016; Moreira et al., 2016; Moreira et al., 2013; Dockery et al., 2016; Moreira et al., 2016; Mo

2019) which are associated with children's behavioural outcomes (Johnson et al., 2013; Dockery et al., 2016).

2.6.1.4 The Family Investment Model

The Family Investment Model (FIM: see figure 3) states that family SES is associated with children's development because it determines parent's ability to invest in the resources required for good child development such as nutritious food, educational materials, activities, and a good environment (Conger & Donnellan, 2007). Low SES families have fewer resources to invest and must concentrate on providing the basic requirements to children (e.g., food and shelter) while high SES families can provide developmental support (e.g., extra tuition and educational experiences). Family SES therefore determines the microsystems that children have access to and interact with which ultimately shapes their development (Bronfenbrenner, 1986).



Figure 3. Family Investment Model (figure in Xiaofeng et al., 2018)

2.6.1.5. Evidence Supporting The Family Investment Model

Research indicates that family investments in children's education improve their language ability (Sohr-Preston et al., 2013) and externalizing behaviour (Sosu & Schmidt, 2017). When parents invest in a cognitively stimulating home environment (e.g., purchasing books and reading with children) then children's cognitive and behavioural outcomes improve (Prevoo et al., 2014; Longo et al., 2017). Investing in educational activities and support improves children's cognitive and behavioural outcomes (Fomby & Musick, 2018). Investing in a safe home environment is also associated with better cognitive and behavioural outcomes (Longo et al., 2017). Higher educated parents are more likely to invest more time, money, and effort into creating cognitively stimulating environments which lead to better cognitive and behavioural outcomes (Kalil, et al., 2012; Sohr-Preston et al., 2013; Jeong et al., 2017).

2.6.1.6 Summary

Overall, following a bioecological framework helps explain why children's environments are important factors in children's language and behavioural development by emphasizing the role of their most proximal environments and the interactions within them. Ultimately, since children's interactions are largely limited to their parents and their home for the first few years of life and remain their primary source of interactions throughout childhood it is these interactions that have a significant influence on their development (Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Morris, 1998). The family stress and investment models within the bioecological framework help explain how SES is important for children's language and behavioural outcomes by showing that family SES can determine the quality of children's environments and the interactions that occur within them such as influencing family conflicts, parent-child interactions, stress, quality of cognitive stimulation, and investments each of which are associated with children's language ability and behavioural development (Rowe, 2012; Landers-Potts et al, 2015; Longo et al., 2017; Jeong et al., 2017; Sosu & Schmidt 2017).

2.6.2 Evidence on SES and developmental outcomes

This section provides a brief overview of the evidence on the associations between SES and children's language ability and socio-emotional outcomes. The literature overall consistently finds that family SES is associated with children's language and behavioural outcomes in cross-sectional and longitudinal studies, and meta-analyses (e.g., Sirin, 2005; Berger et al., 2009; Kiernan & Mensah, 2009, 2011; Violato et al., 2011; Letourneau et al., 2011; Cooper & Stewart, 2013; 2017; Fitzsimons et al., 2017). However, there are differences between language ability and behavioural outcomes. For instance, in Letourneau et als (2011) metaanalysis SES was associated with children's cognitive and language ability in all seven studies (g = 0.35, CI = [0.24, 0.46]), with externalizing behaviour in six of seven studies (g =0.06, CI = [0.02, 0.11]), and with internalizing behaviour in only two out of five studies (g = 0.08, CI = [0.01, 0.16]). Studies in the meta-analysis examining cognitive and language outcomes focussed on early childhood while studies examining behavioural outcomes focussed on later childhood. These findings indicate that there are likely several factors such as different mechanisms and dynamics (e.g., stage of childhood) that can influence the associations between SES and children's language and behavioural outcomes. This section will discuss this further by presenting evidence for language ability and socio-emotional behaviour.

2.6.2.1 Evidence on SES and language development

Theory indicates that children develop language through frequent exposure to language and bidirectional interactions with their parents (e.g., Vygotsky, 1978; Bronfenbrenner, 1986). Without exposure, it is likely that language development would be hindered. Low SES children likely get less exposure due to impaired parenting practices and child-parent interactions, and less investment in additional materials and educational services, as outlined in the family stress and investment models (Conger & Donnellan, 2007; Conger et al., 2010). These factors limit the quantity and quality of interactions between children and their caregivers which hinders language development.

The quantity of interactions is important for development as hearing parents speak improves children's language processing and ability to improve their vocabulary (Hoff, 2003; Marchman, & Fernald, 2008; Rowe, 2012). Low-SES parents speak to their children less frequently than high-SES parents (Hart & Risley, 1995; Rowe, 2008) and up to 18 times fewer words per day (Weisleder & Fernald, 2013). The family investment model suggests this is because low SES families have fewer resources to invest in educational materials and activities for children's development leading to fewer interactions and few words being spoken. This affects vocabulary growth and the speed of processing words (Weisleder & Fernald, 2013) as evidence indicates that the amount of talking that is heard is associated with children's vocabulary growth (Rowe, 2012; Weisleder & Fernald, 2013). Quality of childparent interactions (e.g., better speech, responsiveness, and warmth) is also associated with children's language development (Hirsh-Pasek et al., 2015; Justice 2019). Higher SES parents are more likely to have higher quality speech (more complex words and grammar) due to higher levels of education (Huttenlocher et al., 2010; Zauche et al., 2016) and this

higher quality speech is associated with their children having better language development (Rowe, 2012; Cartmill et al., 2013).

Responsiveness (engaging warmly in a timely manner, Wilson & Durbun, 2013) is an important component of language development (Madigan, 2019). When parents are responsive to children during interactions then there are periods of joint sustained attention which are important for language development (Farrant & Zubrick, 2012). A recent metaanalysis found that children have 2.8 times better language ability if their parents show higher responsiveness and warmth during their interactions (Madigan, 2019). The family stress model indicates low SES parents are less likely to be responsive to their children and more likely to display poorer parenting practices (e.g., neglect) due to distress which will negatively impact their language development.

The Intergenerational-Transfer-of-Socioeconomic-Resources Model (Davis-Kean et al., 2021) states that higher SES families have higher expectations for their children, and they are more able to invest resources into their children's learning as outlined in the Family Investment Model (Conger & Donnellan, 2007). This is seen in the literature as higher SES families are more likely to place emphasis on education, be involved in schooling, invest more in activities to boost language development such as invest in reading materials, and read to their children. Each of these are important to children's language development (Bhattacharya, 2010; Hindman & Morrison, 2012; Yeo et al., 2014; Tichnor-Wagner et al., 2015; Gay et al., 2021). Lower SES parents are less likely to be involved in children's educational activities (Sonnenschein et al., 2014) and less able to provide the resources required (Conger & Donnellan, 2007; Hornby & Lafaele, 2011; Hornby & Blackwell, 2018). Lack of these resources can hinder language development (Tichnor-Wagner et al., 2015).

SES-related language differences can be seen at various stages of childhood from as early as infancy and continue to be seen across children's life course into adolescence (Betancourt et al., 2015; Brownell et al., 2016; Sharkins et al., 2017, Miller et al., 2021). Even during pregnancy parents' SES can have an impact on their children's language development when they are born (Brownell et al., 2016; Ribeiro et al., 2016; Berthelon et al., 2021). Family stress models explain this association because low-SES mothers are more likely than high-SES mothers to experience stress while pregnant (Conger et al., 2010). Research indicates that if mothers are stressed during pregnancy their children can have poorer language development (Laplante et al., 2004; King & Laplante, 2005; Berthelon et al., 2021; Ribeiro et al., 2016). Low SES continues to be associated with children's language ability throughout their life course as SES differences in vocabulary can be seen at 7 months (Betancourt et al., 2015) and by 18 months there are significant SES-related differences in vocabulary size and processing. By 24 months low SES is associated with poorer language outcomes (Spann et al., 2020; Markant et al., 2016) and there is up to a 6-month gap in language proficiency between low and high SES children (Fernald et al., 2013) which is an indicator of future language ability. SES is associated with vocabulary and sentence comprehension at 48 months (Schmerse et al., 2018), receptive language at 36-48 months (Morales et al., 2021), and persists through to 60-72 months (Miller et al., 2013; Morales et al., 2021), and language comprehension from 5 years to 14 years (Burneo-Garcés et al., 2019; Miller et al., 2021).

Overall, cross-sectional and longitudinal studies consistently indicate that low family SES is associated with children's current and future language ability (Feinstein 2003; Hansen & Joshi 2007; Berger et al., 2009; Blanden & Machin 2010; Goodman & Gregg, 2010; Washbrook & Waldfogel 2010; Violato et al., 2011; Buckingham et al., 2013; Waldfogel 2013; Platt et al. 2014; Kiernan & Mensah, 2009, 2011; Dickerson & Popli, 2014, 2016; Hernandez-Alava & Popli, 2017). Several meta-analyses have also examined the associations between SES and children's language development (e.g., Sirin, 2005; Letourneau et al., 2011; Cooper & Stewart, 2013; 2017). A recent meta-analysis by Cooper and Stewart (2017) found that family income was associated with children's language outcomes in 25 out of 30 studies with effect sizes ranging from .01 -.37. The authors cite methodological limitations as the cause of the five studies with non-significant results (e.g., controlling for variables that should be used as mediators weakened the income effects).

2.6.2.2 Evidence on SES and socio-emotional development

Social learning theory explains that children learn behaviour through observing the behaviours of people in their environments (primarily the parents), encoding the behaviours, and then imitating them (Bronfenbrenner, 1986; Vygotsky, 1978; Bandura, 1986). When children perform an observed behaviour, their parents can reinforce this behaviour positively or negatively to encourage or discourage the behaviour. Within the bioecological framework, these interactions would take place in the microsystem and are termed proximal processes. The reason SES is associated with children learning behaviour is that low SES can generate family conflict and impaired parenting practices (Conger et al., 2010). If children observe conflict in the household, then they are more likely to encode and imitate it at a later date. Similarly, if children experience harsh parenting practices and impaired parent-child interactions then children are less likely to receive positive reinforcement for desirable behaviours.

These theoretical assumptions are supported in the literature as low SES is associated with higher levels of parental stress (Zhu et al., 2019) which is associated with children's internalizing problems (Cote et al., 2018). Similarly, parental psychological distress (anxiety

and depression) has a direct association with children's externalizing problems (Brook et al., 2012). SES is also associated with children's internalizing problems via parental anxiety alone (Zhu et al., 2019). Low SES has been linked to disrupted and harsh parenting (Neppl et al., 2016), lack of support (Newland et al., 2013), and lower quantity and quality interactions between children and their parents (Iruka et al., 2012), each of which is associated with children's internalizing problems (Wu & Lee, 2020).

Ma et al. (2012) found that when parents exhibited punitive parenting practices 14-year-old children showed greater externalizing behavioural problems while children with parents who showed more warmth had fewer externalizing behavioural problems. A recent study by Kaiser et al. (2017) examined the pathways between poverty status and children's internalizing and externalizing behaviours. They found that mothers life satisfaction and parenting style significantly mediated the association between SES and children's behaviours with parenting style having a larger effect. Research also indicates that family conflicts such as abuse, sexual assault, and neglect also explain the relationship between low SES and children's internalizing problems (Liu & Merritt, 2018). Investment models also argue that low SES families are less likely to be able to invest in basic needs (e.g., nutrition requirements for children) and food insecurity has been associated with internalizing and externalizing and externalizing and externalizing and externalizing and several problems (Slopen et al., 2010).

The literature examining the relationship between SES and internalizing, and externalizing behaviour is smaller than the literature for language development, but it does indicate that SES is associated with both internalizing and externalizing behaviour (Letourneau et al., 2011). Socioeconomic inequalities in internalizing and externalizing problems are evident in early childhood (Marryat et al., 2018). At age 3 SES inequalities are already associated with children's internalizing (Bøe et al., 2014; Ng-Knight & Schoon, 2017) and externalizing

problems (Bøe et al., 2014). Lower SES children have significantly worse internalizing (Evans & Cassles 2014; Bøe et al., 2014; Zachrisson & Dearin 2015; Ryan et al., 2015) and externalizing (Dearing et al., 2006; Bøe et al., 2014) outcomes than their non-poor peers, and this trend has been found to persists at all ages from 3-19 years of age (Peverill et al., 2021). Longitudinal studies indicate that SES during early childhood is associated with internalizing and externalizing outcomes later in life (Bøe et al., 2014; Hernandez-Alava & Popli, 2017; Hosokawa & Katsura, 2017; 2018; Straatmann et al., 2019).

Overall, cross-sectional and longitudinal studies consistently indicate that low family SES is associated with children's current and future socioemotional behavioural development (Berger et al., 2009; Violato et al., 2011; Wilson et al., 2012; Bøe et al., 2014; Bitsko et al., 2016; Flouri & Midouhas, 2017; Hosokawa & Katsura, 2017; Hernandez-Alava & Popli, 2017; Fitzsimons et al., 2017; Marryat et al., 2018; Tamura et al., 2020; Fuller et al., 2022). Meta-analyses have also examined the associations between SES and children's socioemotional behavioural development (e.g., Sirin, 2005; Letourneau et al., 2011; Cooper & Stewart, 2013; 2017). Cooper and Stewart's (2017) meta-analysis found that family income was associated with children's behavioural outcomes in 12 out of 16 studies. Effect sizes for internalizing behaviour ranged between .02 - .11 and for externalizing behaviour between .03 - .11. Of the four studies that found no effect of SES, two were experimental involving monthly cash transfers (Morris & Gennetian, 2003; Manley et al., 2015), one examined lottery winners (Cesarini et al., 2016), and one was observational (Violato et al., 2011). Although they had no association with behavioural outcomes, two found income associations with cognitive and language outcomes (Violato et al., 2011; Manley et al., 2015) which seems to indicate that SES associations could differ by outcome being measured. However, neither Violato et al. (2011) or Manley et al. (2015) distinguish between the type of behaviour (internalizing and externalizing) so it cannot be determined if this finding would be

consistent for all behaviour. Violato et al. (2011) only examined early childhood behaviour so it could be that behaviour outcomes are more likely to be associated with SES during later childhood. Letourneau et al. (2011) also conducted a meta-analysis distinguishing between dimensions of behaviour to examine associations between family SES and children's internalizing behaviour and externalizing behaviour. Six of seven studies found that low SES was associated with increased externalizing behaviour problems (g = 0.06, CI = [0.02, 0.11]) but only two out of five studies found an association between family SES and children's internalizing behaviour (g = 0.08, CI = [0.01, 0.16]) indicating that SES may have different associations with different dimensions of behaviour.

2.6.2.3 Summary

Taken together, evidence has consistently shown that family SES is associated with children's current and future language ability and socioemotional behavioural outcomes (Feinstein 2003; Hansen & Joshi 2007; Berger et al., 2009; Kiernan & Mensah, 2009, 2011; Schoon et al., 2010; Blanden & Machin 2010; Goodman & Gregg, 2010; Washbrook & Waldfogel 2010; Hansen et al. 2010; Violato et al., 2011; Buckingham et al., 2013; Waldfogel 2013; Platt et al. 2014; Bøe et al., 2014; Dickerson & Popli, 2014, 2016; Bitsko et al., 2016; Flouri & Midouhas, 2016; Brownell et al., 2016; Sharkins et al., 2017; Hosokawa & Katsura, 2017; Hernandez-Alava & Popli, 2017; Fitzsimons et al., 2017; Tamura et al., 2020; Fuller et al., 2022). Studies also indicate that different dimensions of SES (e.g., family income, parent occupation, and parent education) are associated with children's language and behavioural outcomes (Huttenlocher et al., 2010; Bøe et al., 2017; Zauche et al., 2016; Hernandez-Alava & Popli, 2017; Hosokawa & Katsura, 2017; 2018; Miller et al., 2021) and that the size of the associations between SES and children's developmental outcomes differ by outcome (Letourneau et al., 2011; Cooper & Stewart, 2013; 2017; Kiernan & Huerta, 2013; Votruba-Drzal, 2021).

The benefit of longitudinal studies is that they help identify complexities in the associations between SES and children's developmental outcomes by examining the SES effects over extended periods of time at different stages of childhood (Park & Rainsberry, 2020). The stage of childhood is an important consideration because evidence indicates that the strength of associations between SES and outcomes differs between studies but also between stages of childhood (Burnett & Farkas, 2009; Dickerson & Popli, 2014; 2016; Miller et al., 2021; Whitfield et al., 2021). Further evidence indicates that the duration of being in a low SES family has an impact on the associations between SES and children's language and socioemotional behavioural development (Kiernan & Mensah, 2009; Fitzsimons et al., 2017). Studies indicate that SES is not static, it changes over time generating different SES trajectories, and that the different SES trajectories have different associations with children's language and socioemotional behavioural outcomes (Fitzsimons et al., 2017). The literature examining the SES association with children's language and socioemotional behavioural is broad and inconsistent which makes it difficult to draw conclusions about what outcomes SES is most strongly associated with, at which time during childhood SES associations are strongest, and whether different exposures to SES are important. Meta-analyses have also been useful for drawing findings together to generate key points to consider. Several metaanalyses have examined the overall trends in the associations between SES and children's language and behavioural outcomes and have found that SES is consistently associated with children's language ability but there has been some variation in the SES association with behavioural outcomes (Sirin, 2005; Letourneau et al., 2011; Cooper & Stewart, 2013; 2017), particularly internalizing behaviour (Letourneau et al., 2011). Studies in the meta-analyses examining language outcomes focussed on early childhood while studies examining behavioural outcomes focussed on later childhood. This further indicates that the stage of

childhood is an important factor to consider in this study and will be discussed in later sections.

Overall, the current evidence highlights several important factors regarding the associations between SES and children's development. *First*, the extent to which family SES is associated with children's developmental outcomes differs depending on the outcomes being examined. *Second*, the associations between SES and children's developmental outcomes may differ depending on the SES dimensions being examined. *Third*, the dynamics (e.g., stage of childhood, duration, trajectories) of SES may influence the associations between SES and children's language and behavioural outcomes. The next section will discuss the association between the different dimensions of SES and children's development.

2.7 Evidence of different SES dimensions and children's development

As established, SES is associated with children's language ability, internalizing behaviour, and externalizing behaviour but it is unclear which dimensions of SES are most important for each outcome. The majority of the literature has examined the associations that income measures or composite measures of SES have on children's language and behavioural outcomes which do not provide information about which SES dimensions have the largest associations with children's outcomes (Kiernan & Huerta, 2008; Violato et al., 2011; Letourneau et al., 2011; Khanam & Nghiem, 2016; Cooper & Stewart, 2017; Fitzsimons et al., 2017). However, theoretically, each SES dimension represents a different resource that families have access to (Weber, 1968; Bourdieu, 1986; Coleman, 1988). These SES dimensions and the resources they represent are associated with children's development differently via both unique and overlapping pathways as outlined in the family stress model and investment models. Researchers have tried to identify whether there are unique
associations between different SES dimensions and children's language ability and behavioural outcomes. Compared to the broader SES literature, fewer studies have examined the unique associations between income, education, and occupation as individual dimensions and children's outcomes. Studies that have done this could reveal which SES dimensions are most important and help inform policy.

Education appears to be the most important SES indicator for language ability when only family income and parents' education are measured. Examining children's language ability, Davis-Kean (2005) found a direct association between parents' education and children's language ability, but not with income. Both parents' education and income were indirectly associated with children's language ability through parents' educational expectations for their children and parents' education had the largest association with parents' expectations. Similar results were found by Coddington et al. (2014) when examining the association between maternal education and family income and children's language ability through several different pathways. Maternal education, but not family income, was directly associated with children's language ability. Mother's education was also associated with all mediating pathways (standard of living, home ownership, cognitive stimulation, linguistic stimulation, publicly and privately funded early childcare and education) while family income was only associated with standard of living and private early childcare and education. This suggests that mother's education is the most important indicator for children's language ability especially for children with no access to private care and education. However, standard of living was the most important mediator which was significantly associated with both income and education indicating that they both are important SES indicators for children's language, but education is arguably more important as it had the strongest association with standard of living.

Including parents' occupation into their analysis, Gonzalez et al., (2020) examined the independent associations between parents' occupational class and level of education and children's language at age 4-5. They found that mother's education had a larger association with children's language scores than occupational social class or employment status. This indicates that education and occupation operate differently but it does not examine the unique effects of all three SES dimensions as it neglects family income.

Although these studies indicate that parents' education has the largest associations with children's language ability it seems to be dependent on the mediating variables being measured. The studies focus on education related mediators (e.g., parent's educational expectations and cognitive and linguistic stimulation) and it is likely that parent's education would be strongly associated with these educational factors as the Intergenerational-Transferof-Socioeconomic-Resources Model indicates that parents with higher levels of education expect better outcomes from their children (Davis-Kean et al., 2021). This can be seen in Coddington et al. (2014) study where family income was associated with mediators that can be purchased (e.g., standards of living and private childcare and education) but was not associated with mediators that cannot be purchased (e.g., parents' linguistic stimulation). This shows that different SES dimensions can operate via different pathways, but it does not allow conclusions to be drawn in terms of which SES dimension is most important. Instead, it raises questions about whether parents' education is also the most important SES dimension for non-cognitive outcomes (e.g., internalizing, and externalizing behaviour).

When other child outcomes are considered, there are inconsistencies about which SES indicators are most important. Jackson et al. (2017) examined the associations between

mothers' education, family income, and children's age 5 language ability and externalizing behaviour in a UK and US sample. In the UK, mothers' education and family income were both associated with children's language and externalizing behaviour. However, in the US, only family income was associated with children's language and externalizing behaviour. The differences between the UK and US samples could be attributed to differences in policy (e.g., healthcare and childcare) but it does indicate that parents' education could be important for children's behavioural outcomes in the UK indicating a need to understand this association with UK data. Similarly, Boe et al (2014) examined associations between family income, parents' education and children's internalizing and externalizing behaviour. They found family income was directly associated with children's internalizing problems and had the biggest overall association with internalizing problems. Parents' education had the only direct association with children's externalizing problems, but negative parenting practices had the largest overall association with externalizing problems. Both income and maternal education were associated with negative parenting, but income had a slightly larger association with it. These results suggest income is the most SES dimension for children's internalizing problems, but it is unclear which is most important for externalizing problems. Hosokawa and Katsura (2018) found similar inconsistencies as family income was the only SES dimension associated with children's internalizing behaviours at 6 years old. However, both family income and maternal education were significantly associated with children's externalizing behaviours with education having a slightly stronger association. These studies indicate that family income is more important than parents' education for children's internalizing behaviours during early childhood, but it cannot tell us whether it is more important than occupation as it was not measured. The findings are less clear about which SES dimension is most important for externalizing behaviour as the differences between income and education associations are small.

Measuring all three SES indicators, Sullivan and Brown (2013) examined unique associations between income, education, and occupational class on children's age 16 arithmetic, spelling, and vocabulary scores. They found that income and education were associated with all outcomes, while occupation was only significantly associated with children's arithmetic. They also found that education had the strongest association with all outcomes, followed by income, and occupation. When parents reading behaviour at home was controlled for education still had the strongest association with all outcomes while income was only significantly associated with arithmetic. Occupation had no significant associations with any outcomes. This would suggest that higher educated parents were more likely to read with their children which improves their language ability. Different conclusions are drawn by Miller et al. (2021) who also examined the associations between family income, mothers' education, mothers' employment, and children's externalizing and internalizing behaviour, maths ability, and language ability. When children were 5-6 years old, mothers' employment had the largest association with children's externalizing outcomes, mothers' education had the largest association with internalizing problems and language ability, and income had the largest association with mathematics ability. At age 9-10, the findings were the same apart from income, rather than education, had the largest association with children's language ability. At age 13-14, mothers' occupation had the largest association with internalizing and externalizing problems, mothers' education had the largest association with language ability, and income had the largest association with mathematics ability. These inconsistent findings represent the problem with the current literature as it indicates that SES dimensions, outcomes, and childhood stage are all important factors to consider in research.

Overall, the literature examining the independent unique associations between different SES dimensions and children's language and behavioural outcomes is inconsistent and it is difficult to determine which SES dimension is most important. With studies measuring different dimensions, it makes it difficult to compare findings. Across the studies, each dimension has been found to be the most important which suggests that the relationship between SES dimensions and children's outcomes is complex. The evidence does indicate that each SES dimension has a unique association with children's outcomes which supports theory (e.g., Weber, 1968), but it indicates that the associations are dependent on the outcomes being assessed and the SES dynamics (e.g., stage of childhood). Few studies examine the associations between income, education, and occupation and children's language ability, internalizing and externalizing behaviour across different stages of childhood. With the UK currently focused on addressing language and behavioural inequalities (Childcare Act, 2006; Well-Being of Future Generations Act, 2015; Children's Services Co-operation Act, 2015; The Education Act, 2016) it raises the question of which SES dimensions are most important for different child developmental outcomes in the UK context and where policy should be focussed. This study will add to the literature by examining this whilst considering the *dynamics* of SES which will be discussed in the following section.

2.8. A dynamic view of SES and developmental outcomes

This section will discuss the dynamics of SES (e.g., the stage of childhood, duration of exposure to low SES, and trajectories) and their associations with children's language ability and socioemotional behavioural outcomes. The above literature highlighted that the stage of childhood is an important factor to consider in the associations between SES and children's language and behavioural outcomes (Dickerson & Popli, 2014; 2016; Miller et al., 2021; Whitfield et al., 2021). Theory and longitudinal studies also indicated that the duration of

exposure to SES is also an important factor in the association between SES and children's language and behavioural outcomes (Kiernan & Mensah, 2009; Fitzsimons et al., 2017). Unique exposure to different SES circumstances over time form different SES trajectories that are associated with children's outcomes (Fitzsimons et al., 2017). The following subsections will provide an overview of the theoretical considerations that help researchers understand why SES dynamics are important for children's language and behavioural outcomes and then present the current empirical evidence.

2.8.1 Theoretical considerations

As indicated in the literature, SES dynamics (e.g., stage of childhood, duration of exposure, and trajectories) can influence the association between SES and children's language ability and behavioural outcomes (Fitzsimmons et al., 2017; Miller et al., 2021). As established, family SES is associated with children's developmental outcomes because it can determine the nature of the microsystem environments that children interact with (Bronfenbrenner, 1979; 1986; Conger et al., 2010; Conger & Donnellan, 2007). The family stress model indicates that low SES is associated with poorer outcomes through children's bidirectional interactions with impaired pathways such as poor parenting and conflict (Conger et al., 2010) However, within Bronfenbrenner's bioecological systems theory, the chronosystem acknowledges that children's environments can change over time (Bronfenbrenner, 1979; 1986). When environments change, the bidirectional interactions that children have with them are also likely to change which will impact children's outcomes. Therefore, the extent to which SES is associated with children's language and behavioural outcomes is likely to vary based on SES dynamics. Life course epidemiology provides different models to explain why different SES dynamics are associated with children's developmental outcomes (Kuh et al., 2003). Life course models study how exposure to low SES over the life course is

associated with children's developmental outcomes. The different models include the *sensitive period, accumulation of risk, and trajectory models* (Kuh et al., 2003).

2.8.1.1 Sensitive period models

Sensitive period models explain why the association between SES and children's outcomes differ depending on the stage of childhood (Hertzman et al., 2001; Kuh et al., 2003). The sensitive period hypothesis states that there are certain times during the life course when people are more sensitive to risk factors such as low SES (Kuh et al., 2003). We know from family stress and investment models that low SES is associated with poorer developmental outcomes because low SES leads to impaired bidirectional child-parent interactions and less investment in their development (Conger et al., 2010; Conger & Donnellan, 2007). However, the *sensitive period model* argues that children may be more susceptible to the adverse effects of impaired interactions and low investment during certain times during childhood, particularly during early childhood (Kuh et al., 2003).

2.8.1.2 Accumulation of risk models

Accumulation of risk models (Power et al., 1991; Kuh et al., 2003) explain why the association between SES and children's outcomes differ depending on the length of time families are in low SES. These models posit that children's exposure to low SES at several time points each have an association with children's outcomes, but also accumulate to produce a cumulative adverse effect on children's outcomes (Power et al., 1991; Kuh et al., 2003). Therefore, children in low SES households for longer durations will have worse developmental outcomes than their peers because they will have more cumulative impaired bidirectional child-parent interactions and go longer periods with less investment in their

development. This build-up of exposure to stressful and impaired interactions are associated with worse developmental outcomes (Conger et al., 2010; Conger & Donnellan, 2007).

2.8.1.3 Trajectory models

Trajectory models explain that SES is not a static construct that people can have different SES trajectories and that each trajectory can have a different association with children's outcomes. Trajectories highlight that SES and development do not stop after critical or sensitive periods but instead continue from birth to death (Elder et al., 2003). People can have several different trajectories simultaneously that can be independent or associated with each other (e.g., income, educational, occupational, and developmental trajectories) and each trajectory encompasses transitions into and out of states and major life events that occur across the life span. Life course research has generated the social mobility hypothesis which states that individuals can continuously move between different social circumstances over the life course and that these upward and downward trajectories can have an association with children's developmental outcomes (Duncan, 1988; Lynch et al., 1994). Family stress model explains that this is because as family SES changes, then family conflict and parents' interactions with their children are likely to change in the same direction which can increase or decrease the SES association with children's outcomes. Similarly, the investment model states that changes in family SES will be reflected in their ability to invest in their children's development and therefore impact the association between family SES and children's outcomes.

2.8.2 Evidence of SES differences in developmental outcomes by stage of childhood When following sensitive period models, researchers examine the associations between exposure to risk factors at particular time points and children's developmental outcomes (Green & Popham, 2017; Green et al., 2018). Exposure to particular factors during these critical periods can have positive or negative effects on the developmental outcomes of interest (Kuh et al., 2003). In the case of SES, it would be expected that exposure to low SES during a critical period would have a negative effect on children's current and future developmental outcomes. These 'sensitive' or 'critical' period can vary depending on the outcome being measured and the external factor that the children are exposed to. This is particularly important as childhood and adolescents are times of significant change for children (Gariepy et al., 2017).

The developmental-contextual model presented by Schoon et al. (2002) proposes that current SES risk is associated with future SES risk, current developmental outcomes, and future developmental outcomes. Testing their model, they found that SES risk during early childhood was more strongly associated with early (current) academic adjustment than late academic adjustment. They also found that late SES risk was more strongly associated with adolescent (future) academic adjustment than late (current) academic adjustment. This indicates that SES risk can have more or less importance on children's outcomes depending on when they experience SES risk. The model indicates that early childhood is a particularly sensitive period to SES risk for academic outcomes. Sensitive periods are not exclusive to early childhood.

It is important to consider the stage of childhood when examining the associations between SES and children's development because evidence indicates that associations differ depending on the stage of childhood (Miller et al., 2021). Few studies have examined the association between family SES at different stages of childhood and children's language ability, internalizing behaviour, and externalizing behaviour. One study by Burchinal et al

(2018) examined how experiencing poverty at 0-2 years and 3-5 years was associated with a variety of children's language and behavioural outcomes. They found that experiencing poverty between 0-2 years was worse for children's problem solving, phonological awareness, and internalizing behaviour problems. However, experiencing poverty for the first time between 3 and 5 years old was worse for children's receptive vocabulary, word identification, and executive functioning. There was no difference regarding externalizing behaviour problems. However, behaviour problems. These findings indicate that family SES at different childhood stages is important, but it does not provide any understanding about whether mid or late childhood are more or less important as the time frame only included early childhood age ranges. It is also difficult to determine at which stage in childhood SES is most important from the broader literature because the studies often focus on a short timeframe or findings are inconsistent.

The broader literature largely focuses on early childhood as being the most important critical period in a child's development (particularly cognitive outcomes) and is the most sensitive to the adverse effects of low SES. This is because the early years of childhood are considered to be a sensitive period for children's development (Sylva 1997; Kuhl et al., 2005). It is likely that the early years are particularly sensitive to the adverse effects of low SES because it is when brain development is of critical importance for establishing neural structures and functions that are responsible for future cognitive and behavioural processes (Brummelte et al., 2017). Evidence supports that early SES is most important for later life outcomes (e.g., Brooks-Gunn et al., 1997; Joinson et al., 2017). This assumption tends to hold true for research examining cognitive outcomes where the earlier a child experiences low SES the worse their cognitive outcomes tend to be. For example, in the UK, Schoon et al. (2011; 2013) found that experiencing income poverty during the first year of life had a stronger negative effect on children's cognitive development at 5 years of age than experiencing

poverty at age 3 or 5. Dickerson and Popli (2012; 2016) also compared the effects of experiencing income poverty at birth and at age 7 on children's cognitive outcomes at 7 years old. They found that early income poverty had a larger association with children's cognitive outcomes than current poverty at age 7. Like Burchinal et al. (2018), these studies are limited to early-mid childhood which means they do not help draw conclusions about whether SES during the later years of childhood is also important for children's outcomes.

Emerging evidence suggests that sensitive periods can differ depending on the outcome being considered (e.g., Campbell et al., 2000; Costello et al., 2005). Low SES during earlier years appears to be particularly problematic for cognitive and language outcomes but low SES during later childhood and adolescence is important for socioemotional problems (National Institute of Child Health and Human Development Early Child Care Research Network, 2005; Costello et al., 2011; Gibb et al., 2012; Lai et al., 2019). Evidence indicates that adolescence is a sensitive period for behavioural problems due to new experiences such as being exposed to stressors such as bullying (Blakemore, 2014; Fuhrmann et al., 2015; Efstathopoulos et al., 2018). Support for this assumption can be seen in the literature. Duncan et al. (2010) examined whether there were differences in children's internalizing and externalizing behaviour based on whether they experienced low income during early childhood, mid-late childhood, and adolescence. Both internalizing and externalizing behaviour were only associated with low income during adolescence (Duncan et al., 2010). Lai et al. (2019) found the same when comparing the associations between different patterns of poverty on children's socioemotional behavioural problems. They found that income poverty during late childhood had a stronger association with children's socioemotional behavioural problems than income poverty during early childhood. Tsal et al., (2005) also showed that low income during mid-late childhood (6-12 years) was associated with ADHD (externalizing) problems beyond the effect of low income during early childhood (0-5 years).

This evidence indicates that low SES during later childhood is also important for children's outcomes and that children may be sensitive to the adverse effects of low SES across the entire childhood timeline. However, the behaviour measures used in each of the studies measure different indicators of behaviour making it is difficult to draw comparisons between studies. The findings are also inconsistent with Mazza et al (2017) who examined low SES during early (0-3 years), middle (5-7 year), and late (8-12 years) childhood and found that being exposed to poverty during early childhood had the strongest association with externalizing behaviour problems (hyperactivity, physical aggression, and oppositional behaviour). This was closely followed by experiencing poverty between 8-12 years indicating that proximal or most recent experiences of poverty were also associated with behavioural outcomes. This contradicts Burchinal et al. (2018) findings that low SES during early childhood was not associated with externalizing problems and differs from research stating that there was no association at all between SES and externalizing problems. These inconsistencies are problematic for policymakers and research is required to make these associations clearer.

Overall, the literature examining the associations between family SES at different childhood stages and children's language ability, internalizing problems, and externalizing problems is limited. At present, the evidence indicates that SES at different times has different associations with children's development, but it is unclear which childhood stage is most important for language ability, internalizing problems, and externalizing problems. Since these outcomes are a focus of current UK policy (The Childcare Act, 2006; Well-Being of Future Generations Act, 2015; Children's Services Co-operation Act, 2015; The Education Act, 2016), this study will address this gap in the literature by expanding on Burchinal et al's (2018) study by examining the SES associations with these outcomes at all stages of childhood. The literature also largely focuses on the associations between income at sensitive

periods and children's developmental outcomes and neglects to consider that children may be more sensitive to their parents' education or occupation at different stages of childhood. For example, children's language development is sensitive to mothers' education during early childhood as this provides scaffolding to build upon as they age (Vygotsky, 1978; Carr & Pike, 2012), however, mothers' education is also associated with children's behaviour problems during later childhood and adolescence (Miller et al., 2021). This indicates that different SES dimensions can be important at different stages of childhood.

Study 1 will address this gap by examining the associations that each SES dimension (income, education, and occupation) has on children's language, internalizing, and externalizing outcomes at different stages of childhood. Current policy is in place to improve family income, education, and occupation at different times (e.g., Child Payment is available when children are under 6 years old and the Best Start Grant ends when the child enters schooling). Similarly, free early learning and childcare ends when the child is 4 years old which places stress on parents' availability to work. This would reduce their income from employment which is counterintuitive to the policy goal of maximizing family income through employment. Evidence (e.g., Blakemore, 2014) indicates that there are likely to be more sensitive periods at different stages of childhood for different outcomes which could mean that when policy support ends it could have a negative effect on children's sensitive periods. Therefore, examining when children are most sensitive to different SES dimensions will help policymakers understand the effect that current policies have and indicate at what time policy could be most beneficial.

2.8.3.1 Evidence of SES trajectories and developmental outcomes

Life course theories explain that trajectories are formed by the dynamic sequence of events and experiences that individuals go through over the course of their lives and represent the longitudinal patterns of change and stability in any given sphere of a person's life and can follow them from birth to death (Elder 2003). In terms of SES, trajectories could be formed by encompassing an individual's educational milestones, occupational job loss or gain, and fluctuations in income over time. Depending on the individual's experiences of SES over the life course they may have stable SES trajectories meaning nothing has changed over time, increasing SES trajectories meaning their SES has improved over time or decreasing SES trajectories meaning their SES has worsened over time. These different trajectories have different associations with children's developmental outcomes (Duncan, 1988; Lynch et al., 1994). Bronfenbrenner (1986) tells us that the different trajectories are associated with children's outcomes due to the long-term bidirectional interactions between children and their parents. The accumulation of risk models and family stress and investment models also tell us that cumulative low SES over time is worse for children's development because families with low SES for longer will experience more familial conflicts, more impaired parent-child interactions, and longer durations of low investment, than families who are in low SES for less time (Conger & Donnellan, 2007: Conger et al., 2010).

In general, the literature supports the accumulation of risk models as persistent poverty (e.g., accumulated poverty over several time points) usually has larger negative associations with children's cognitive outcomes than episodic or transitory poverty (Kiernan & Mensah, 2009; 2011; Schoon et al., 2010; 2012; Holmes & Kiernan, 2013; Dickerson & Popli, 2012; 2014; 2016). For example, Schoon et al. (2012) found persistent poverty (poor from birth through to age 5) had the largest negative association with children's age 5 language ability (B = -10.314) followed by being poor twice (B = -7.064), being poor once at birth (B = -6.195) and being poor once at age 5 (B = -2.929). In addition to supporting the accumulation of risk model, the study also indicates that poverty trajectories could be worse if the onset of poverty

begins earlier in life, which is a finding echoed in a series of studies by Dickerson and Popli (2012; 2014; 2016).

Not all studies find that persistent poverty is the worst for children's developmental outcomes as some studies suggest that transitioning into poverty is more important for children's cognitive (e.g., Guo, 1998) and behavioural (Fitzsimons et al., 2017) outcomes. Fitzsimons et al. (2017) found that only persistent poverty between 9 months and 5 years was associated with children's age 5 internalizing and externalizing behaviours. However, transitioning into poverty between 5-11 years old had a stronger association than persistent poverty with children's age 11 peer problems (B = .044 versus .0386), conduct problems (B = .0721 versus .0474), and hyperactivity (B = .0274 versus .0117). Guo (1998) drew similar conclusions as longer exposure to poverty did not result in greater adverse associations with cognitive outcomes. Instead, later onset poverty was associated with adverse cognitive outcomes. The National Institute of Child Health and Human Development Early Child Care Research Network (2005) also found that persistent poverty had the largest negative association with children's age 5 language outcomes, while later onset poverty had the largest negative association with children's externalizing and internalizing behaviour. These studies indicate that while persistent poverty is consistently associated with children's outcomes, particularly during early childhood, it does not always have the largest association. Therefore, it is important to consider other trajectories as findings may vary by the outcome and the stages of childhood being examined.

While the above evidence supports the accumulation of risk models indicating that accumulated and transitory exposure to poverty is both associated with children's outcomes and that accumulated poverty is generally worse, the studies are limited in the number of trajectories that can be examined due to poverty status being dichotomous and can only measure whether a family had entered, left, or remained in poverty which is less dynamic than other SES dimensions. For example, family income can increase or decrease without changing one's poverty status which provides the opportunity to identify more trajectories and examine their associations with children's outcomes. There is a small literature examining different family income trajectories and their associations with children's developmental outcomes which will be presented next.

The literature examining SES trajectories is relatively small but has identified between threeand five-income trajectories (e.g., Jackson et al., 2017; Björkenstam et al., 2017; Kendzor et al., 2012) with three trajectory studies finding a variation of always high, always medium, and always low income. Four trajectory studies found a variation of high, low, increasing and decreasing income trajectories, and five trajectory studies found a variation of highest, second highest, lowest, increasing, and decreasing income trajectories. The literature examining the associations between family SES trajectories and children's developmental outcomes is relatively small and is mostly associated with children's health outcomes such as children's weight (Demment et al., 2014), obesity (Kendzor et al., 2012), asthma (Kozyrsky et al., 2010), and oral health (Peres et al., 2011). Some researchers also examined the associations between family income trajectories and children's behaviours such as smoking and drinking (Poonawalla et al., 2014), violence (Mok et al., 2018) and mental health problems (Björkenstam et al., 2017). Some studies examined the association between family income trajectories and children's language ability (Dohoon, 2014; Jackson et al., 2017).

Findings indicate that children belonging to persistently low-income trajectories almost always fare worse than those from high-income, increasing-income, or decreasing-income trajectories (Chen et al., 2007; Peres et al., 2007; Björkenstam et al., 2017; Boe et al., 2016; Mok et al., 2018) indicating that the longer a child experiences low income, the worse their outcomes will be. *Persistently low* income leads to the poorest language outcomes (Dearing et al., 2001; Schoon et al., 2011; Duncan et al., 2014; Jackson et al., 2017). Children belonging to an *increasing* income trajectory show improved receptive language and expressive language (Dearing et al., 2001), and improved vocabulary (Jackson et al., 2017), and are more likely to graduate compared to their persistently poor peers (Dohoon, 2014). Belonging to a *decreasing* income trajectory leads to poorer language outcomes and are less likely to graduate compared to increasing or high-income trajectories but still better than persistently low income (Dohoon, 2014; Jackson et al., 2017). *Persistently low* income leads to more internalizing and externalizing behavioural problems (Zachrisson and Dearing, 2015; Comeau et al., 2018). Belonging to an *increasing* income trajectory leads to improvements in children's internalizing and externalizing problems (Hoyt et al., 2019) while belonging to a *decreasing* income trajectory leads to an increase in internalizing and externalizing problems (Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Hoyt et al., 2019) compared to their stable high- and increasing-income peers, but still not as problematic as stable low-income children (Hoyt et al., 2019).

The current SES trajectory literature is limited in three ways. *First*, the literature examining the associations between SES trajectories and children's language ability, internalizing behaviour, and externalizing behaviour is small which makes drawing conclusions difficult. The studies that do focus on these outcomes use different samples, methodologies, and do not include all of the outcomes in a single study (Dearing et al., 2001; Zachrisson and Dearing, 2015; Jackson et al., 2017; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Comeau et al., 2018; Hoyt et al., 2019). By examining all these outcomes in the same study, it will provide a better understanding of which SES trajectories are better for each outcome. *Second*, current literature examining associations between SES trajectories and children's language

ability and behavioural development tends to focus on early childhood from age 1-3 years (Dearing et al., 2001; Zachrisson & Dearing, 2015) or 1-5 years (e.g., Jackson et al., 2017) or from mid-childhood to early-adolescence (e.g., Miller & Votruba-Drzal 2017; Comeau & Boyle, 2018). This means families are classified into income trajectories based on short periods of time. Given that trajectories are created over the life course and influenced by life events and life changes (Elder, 2003), it is likely that more family SES variation will be captured if data are used over a longer duration which could identify additional trajectories. This would be beneficial as it would give insight into the different trajectories that exist over the entire childhood timeline rather than specific stages of childhood only. This would also give long-term insight into the efficacy of policy long-term. *Third*, the SES trajectory literature almost exclusively focuses on income trajectories. Evidence in the next paragraph indicates that parents' occupations and education can change over time and are likely to have several unique trajectories that will have associations with children's language development, externalizing behaviour.

The social mobility theories indicate that occupational change is likely which is supported by recent research that found over a 12-year period 51% of people experienced occupational mobility with 31% increasing and 20% decreasing their occupational level (Jonsson et al., 2017). The family stress model tells us that decreases in occupational level would result in financial pressures, parental distress, impaired family interactions and parenting and impact child development. This would suggest that decreasing occupation trajectories would be associated with poorer developmental outcomes. However, theory also suggests that occupational movements in either direction generate stress in parents (Sorokin, 1959; Newman, 1999) and it is well documented that parental stress has a negative effect on

children's cognitive outcomes (Coddington et al, 2014; Lohndorf et al., 2017), internalizing behaviour (Rijlaarsdam et al., 2013), and externalizing behaviour (Sun et al., 2015). Theoretically then it is unclear which occupation trajectory would have the strongest association with children's outcomes.

Parents can also increase their level of education over time (Harding, 2015). Since higher levels of education have better quality interactions with their children (Huttenlocher et al., 2010; Zauche et al., 2016) it is possible that if parents increase their level of education, then their quality of interactions will continue to improve and benefit children's language development (Conger et al., 2010). Some evidence shows that when mothers gain additional qualifications over the first 3 years of their life their children's vocabulary comprehension and expressive language improved (Magnuson et al., 2009), and children's age 3 and 4 cognitive scores continued to improve as their mother's education level increased (Harding, 2015). Children are more likely to do well in education if their mothers graduated high school, enrolled in college, and earned a bachelor's degree indicating that as mothers' education increases their children's developmental outcomes improve simultaneously (Monaghan, 2017). However, other evidence indicates that improvements in mothers' education do not improve children's age 14 cognitive skills or reduce behavioural problems (Augustine & Negraia, 2018). This finding could be due to the different outcome measures or the children's ages. Harding (2015) and Magnuson (2009) show the positive effects of improving education for early childhood outcomes while Augustine and Negraia, (2018) show that there is no association with late childhood outcomes suggesting long-term mother educational improvements are not significant for child development. However, these studies use cross-sectional data, so to further understand the long-term associations between mothers' education and children's outcomes study 2 of this thesis will examine the educational trajectories.

Theoretically, because each SES dimension represents a family's access to different resources and each dimension can operate via different pathways, the trajectories of each dimension will represent the longitudinal access to these unique resources (Weber, 1968; Conger et al., 2010; Davis-Kean et al., 2021). Therefore, by examining the associations that income, education, and occupation trajectories have with children's language and behavioural outcomes, findings will provide insight into whether longitudinal access to each resource is associated with children's different trajectories of language development, internalizing behaviour, and externalizing behaviour. This will allow researchers and policymakers to be more confident about the long-term impact of their policies addressing the SES gap in children's outcomes.

Overall, trajectories of income, education, and occupation are likely associated with children's developmental outcomes. Although there is evidence that income, education, and occupation can change over time the trajectory literature almost exclusively focuses on how income trajectories are associated with children's developmental outcomes. This is likely due to income being the most prone to change and showing short-term trajectories while education may take years to improve. Income was also the focal point of policy whereas UK policy is now increasingly focussing on the importance of improving occupation and education so it will be beneficial to examine how their different trajectories are associated with children's language development and socio-emotional behavioural outcomes are key components of UK policy but since few studies examine the associations between SES

trajectories and children's language and behavioural outcomes it is currently difficult to draw conclusions confidently.

Study 2 will address these limitations by examining family income, education, and occupation trajectories over the first 14 years of their child's life and their associations with children's language ability and internalizing and externalizing behaviours. Current policies such as No One Left Behind and Restart (Scottish Government, 2020; Department for Work and Pensions, 2020) aim to improve parents' occupations and education and although there is large uptake in these programmes (Scottish Government, 2023), it is unclear whether improvements are likely to benefit children as the data are not collected. This study would allow policymakers to be more confident in understanding how their decisions to improve family SES dimensions will affect children's later language and behaviour outcomes. This will also benefit life course research by examining trajectories of different SES indicators in line with the theory that states several different trajectories exist simultaneously.

2.8.3.2 Evidence of developmental outcome trajectories

As discussed, life course theories explain that individuals can have several different trajectories in any given sphere of life simultaneously (Elder 2003). Like SES dimensions, children's developmental outcomes are not static constructs, and they can change over time to form developmental trajectories. The different developmental trajectories are likely to have different associations with family SES (Duncan, 1988; Lynch et al., 1994). Accumulation of risk, family stress, and family investment models suggest that children with longer exposure to low SES will belong to poorer developmental trajectories due to more impaired parent-child interactions and lower parental investment (Kuh et al., 2003; Conger & Donnellan, 2007; Conger et al., 2010).

The child development literature supports the life course theory as several language trajectories (Mangin et al., 2017; Yaari et al., 2018) and behavioural trajectories (Miller & Votruba-Drzal, 2017; Comeau, 2018; Gutman, 2019; Tamura et al., 2020) have been identified in the literature. Evidence has identified between three and five language development trajectories in the literature (Landa et al., 2012; Ukoumunne et al., 2012; Brian et al., 2014; Longard et al., 2017; McKean et al., 2017; Nguyen et al., 2018; Hentges et al., 2019; Jin et al., 2020; Riva et al., 2021). Studies normally find a variant of low-decreasing, stable low, stable mid, high-decreasing, and stable high language trajectories (Nguyen et al., 2018). Researchers have also identified between three and six internalizing behaviour trajectories, but most find 4 trajectories labelled as a variant of stable low, stable high, increasing, and decreasing (Melchior et al., 2014; Korhonen et al., 2014; Hauser-Cram & Woodman; 2016; Miller and Votruba-Drzal, 2017; Gutman & McMaster, 2020; Kuang & Flouri, 2020; Tamura et al., 2020). Studies that have examined externalizing behaviour trajectories have identified between three and five trajectories (Thompson et al., 2011; Hauser-Cram & Woodman, 2016; Kjeldsen et al., 2016; Lee et al., 2016; Miller and Votruba-Drzal 2017; Gerstein et al., 2017; Olsen et al., 2017; Figge, 2018; Min et al., 2018; Perry et al., 2018; Gutman et al, 2019; Tamura et al., 2020). They typically find a variant of stable low, stable high, stable mid, increasing, and decreasing trajectories in children between 6-11 (Miller & Votruba-Drzal, 2017).

Since SES is associated with children's developmental outcomes in later life it is likely that different levels of family SES are associated with children's membership to particular developmental trajectories. Looking specifically at language development outcomes, Hentges et al., (2019) found that low income was associated with children's membership to lower language communication trajectories and was not associated with high or increasing language communication trajectories. Ukoumunne et al., (2012) found that children with mothers with higher levels of education were more likely to belong to higher language development trajectories while those with low-educated parents were more likely to belong to an impaired language development trajectory. Armstrong et al (2017) found that children were twice as likely to belong to a decreasing language trajectory if their parents did not complete secondary education compared to children whose parents did complete secondary education. They also found that children were more than twice as likely to belong to a deteriorating language trajectory if their family income was below the poverty line compared to those that had income above the poverty line. McKean et al (2017) found that children are more likely to belong to poorer language trajectories if they have social disadvantage and if their mothers have less time spent in education.

Research has also focussed on the associations between SES and children's membership of internalizing and externalizing trajectories. Flouri and Sarmadi (2015) found children living in low-income areas were most likely to belong to the highest behavioural problem trajectory. Flouri and Midouhas (2017) also found that those with the highest levels of socioeconomic disadvantage belonged to the highest externalizing problem trajectory and found that children of families that experienced more adverse life events (parental job loss or decreased income) belonged to the highest internalizing and externalizing problem trajectories but were unable to determine which life events had the largest association with outcome trajectories. Comeau (2018) found that children who experienced poverty between 5 and 14 years old were most likely to belong to the highest internalizing and externalizing problem trajectories and that never being poor meant that children were more likely to belong to lower internalizing and externalizing problem trajectories internalizing and externalizing problem trajectories and that never being poor meant that children were more likely to belong to lower internalizing and externalizing problem trajectories internalizing and externalizing problem trajectories and that never being poor meant that children were more likely to belong to lower internalizing and externalizing and externalizing internalizing and externalizing problem trajectories and that never being poor meant that children were more likely to belong to lower internalizing and externalizing problem trajectories.

poverty were most likely to belong to the middle internalizing and externalizing problem trajectory. Similarly, Miller and Votruba-Drzal (2017) found that children from stable income families were most likely to belong to the lowest internalizing and externalizing problem trajectories, while those who experienced income loss were more likely to be in the increasing problems or high behavioural problems groups compared to those with income gains or stable income.

Overall, these studies provide an overview of the associations between SES and children's membership to different trajectories of language ability, internalizing behaviour, and externalizing behaviour. However, there are several limitations with the current literature examining the association between SES and children's language and behavioural trajectories. *First*, the studies all use static measures of SES from single time points to examine their associations with children's developmental trajectories. Since SES is not static over time and has its own trajectories, it raises the question of whether different SES trajectories are associated with children's different outcome trajectories. If children's SES trajectories, then it provides useful information in terms of policy. For example, do policies that target short-term income support benefit children's language and behavioural trajectories? Or is long-term income support required for sustained membership to beneficial language and behavioural trajectories? If membership to the best outcome trajectories is most likely for stable high-income families, then it is unlikely that short-term income support policies will help reduce the developmental inequality gap over the life course.

Second, current literature also mainly focuses on the associations between income and children's developmental trajectories. This does not provide information about whether

education and occupation trajectories are associated with children's language and behavioural trajectories. Since parents' level of education and occupation are both associated with their children's later life outcomes (Magnuson, 2009; Harding, 2015; Flouri & Midouhas, 2017; Miller & Votruba-Drzal, 2017; Perry-Jenkins et al., 2020), it is worthwhile to examine if different education and occupation trajectories are associated with children's membership to language and behavioural trajectories. Examining this will provide evidence about whether long-term support for improving education or occupation would be beneficial for children's membership to language and behavioural trajectories, especially since policy is increasingly focussing on improving occupation and education. Third, current studies also examine children's outcome trajectories from early childhood to middle childhood (Ukoumunne et al., 2012; Hentges et al., 2019) or middle childhood to late childhood and adolescence (e.g., Miller & Votruba-Drzal, 2016; Comeau, 2018). This is problematic as outcome trajectories beginning from later childhood may miss key events from earlier sensitive periods. For example, children with poor language ability during early childhood but good language ability during late childhood would be assigned to a stable high language ability trajectory when they should really be categorised into increasing language development if the full childhood timeline was examined. By using data over a longer timeframe research is more likely to find more outcome trajectories that incorporate more life events and transitions (Cavanagh et al., 2016; Oshio et al., 2018). This will help identify which outcome trajectories children are likely to be in based on their SES trajectories.

Study 3 will address these limitations by examining whether income, education, and occupation trajectories are associated with children's language ability, internalizing behaviour, and externalizing behaviour trajectories from birth until 14 years old. Current policy is interested in reducing the developmental inequality gaps in children's language

ability and socio-emotional behavioural development and one of the primary method of doing so is by reducing socioeconomic inequalities (Well-Being of Future Generations Act, 2015; Children's Services Co-operation Act, 2015; The Education Act, 2016). This study will provide useful insights into whether improving family SES dimensions is likely to improve the trajectory of children's language and behavioural outcomes and whether policy focused on improving SES dimensions is likely to help disadvantaged children 'catch up' and reduce inequality gaps. This would be useful information for policies such as Job Entry Targeted Support (National Audit Office, 2022), Restart Scheme (Department for Work and Pensions, 2020), and No One Left Behind (Scottish Government, 2020) which aim to improve family income, education and training, and occupation.

2.9 Research Aims

This study aims to examine the associations between family SES dynamics and children's language and socioemotional behavioural outcomes. These outcomes were selected as they are critical in unlocking access to areas of the education curriculum (Hulme et al, 2015; Chow & Jacobs, 2016; Hulme et al., 2020), improving well-being, generating human capital and maintaining social relationships (Britto et al., 2017; Hukkelberg et al., 2019), and are predictors of future socioeconomic status (Johnson et al, 2010; Wade et al., 2018; Pedersen et al., 2019). For these reasons improving children's language and socioemotional outcomes has gained attention from policymakers who aim to reduce the socioeconomic gap in these outcomes (e.g., The Childcare Act, 2006; The Education (Scotland) Act, 2016). As discussed in section 2.3 this study conceptualised SES as a multidimensional construct and will measure family income, parents' level of education and occupation consistent will theory (Weber, 1968; Bourdieu, 1986; Coleman, 1988) and in line with current UK policy interests (Department for Work and Pensions, 2011;2022; Department for Education, 2021). Family

income was chosen to represent the financial resources that families can access to purchase goods and services (Galobardes, 2006; Miller et al., 2021; Hällsten & Thaning, 2022). Parent education is measured because this represents the human resources, skills, and knowledge that people have (Galobardes, 2006; Alan et al., 2008). Occupation was measured as it represents social class (Goldthorpe, 1997; Muntaner et al., 2010; Meanley et al., 2020). To achieve these aims, three studies were conducted.

First, study 1 will examine whether the stage of childhood influences the associations between SES dimensions and children's language and behavioural outcomes during early, middle, and late childhood. At present, the evidence is unclear about at which stage of childhood SES is most important for language ability, internalizing problems, and externalizing problems as studies focus on short time frames (e.g., Schoon et al., 2011; 2013; Dickerson and Popli; 2012; 2016; Burchinal et al., 2018). The literature is also unclear about whether different dimensions of SES are important for children's language ability, internalizing problems, and externalizing problems at different stages of childhood as studies tend to examine family income (Schoon et al., 2011; 2013; Burchinal et al., 2018; Lai et al., 2019). Examining when children are most sensitive to different SES dimensions will help policymakers understand the effect that current policies have and indicate at what time policy could be most beneficial. The Specific research questions being addressed in study 1 are:

1a. To what extent are dimensions of family SES (income, education, and occupation) uniquely associated with children's developmental outcomes (language development, internalizing, and externalizing behaviours)?

1b. Are some SES indicators more strongly associated with children's language and behavioural outcomes and is this consistent through to adolescence?

2. To what extent does the association between family SES and children's outcomes vary by stage of development?

Second, study 2 will examine family income, education, and occupation trajectories over the first 14 years of their child's life and their associations with children's language ability and internalizing and externalizing behaviours. At present, research tends to focus on income trajectories meaning that it is unclear whether different education and occupation trajectories are associated with children's language and behavioural outcomes (Dearing et al., 2001; Schoon et al., 2011; Duncan et al., 2014; Zachrisson and Dearing, 2015; Jackson et al., 2017). Research also tends to focus on short timeframes which limit the potential to identify more SES trajectories as more data from longer timeframes can encompass more key life stages and events (Cavanagh et al., 2016; Oshio et al., 2018). Examining the association between different SES trajectories and children's language and behavioural outcomes will allow policymakers to understand how their decisions to improve family SES dimensions might affect children's language and behaviour outcomes. The Specific research questions being addressed in study 2 are:

1. To what extent does family income, parents' education, and parents' occupation change or remain stable from childbirth through late childhood?

2. To what extent are family income, parent's education, and parent's occupation trajectories associated with children's language and behavioural outcomes during late childhood?

Third, study 3 will examine whether income, education, and occupation trajectories are associated with children's language ability, internalizing behaviour, and externalizing behaviour trajectories from birth until 14 years old. Prior research focuses on the association between static measures of SES and children's developmental trajectories (Flouri & Sarmadi, 2015; Flouri & Midouhas, 2017; McKean et al., 2017; Hentges et al., 2019) or associations between SES trajectories and children's developmental trajectories (Zachrisson and Dearing, 2015; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Hoyt et al., 2019) which does not provide information on the associations between trajectories of SES and trajectories of language and behaviour. Examining this will provide information about the likelihood of children belonging to language ability, internalizing behaviour, and externalizing behaviour trajectories based on their family SES trajectories. The Specific research questions being addressed in study 3 are:

1. To what extent do children belong to different developmental (language, internalizing and externalizing behaviour) trajectories?

2. To what extent does the likelihood of belonging to different developmental (language ability, internalizing and externalizing behaviour) trajectories differ by family SES trajectory (stable, increasing, or decreasing SES)?

2.10 Summary

This chapter provided a literature review explaining why the associations between SES dynamics and children's language ability and socioemotional behaviour were examined in this study. The context of the study and the reason why children's language ability and socioemotional behaviour were outcomes of interest were presented. Definitions of SES were

discussed and the conceptualisation of SES that was used in this study was presented. Theoretical considerations that help explain why and how SES is associated with children's language ability and socio-emotional behaviour were outlined followed by empirical research showing that different SES dimensions (income, education, and occupation) and SES dynamics (stage of childhood, SES trajectories, and developmental trajectories) are important factors to consider in the relationship between SES and children's language ability and socioemotional behaviour. Finally, the research aims section provided a brief recap of the chapter and presented the research questions that were examined in each study. The next chapter will discuss the methodology.

Chapter 3 - Methodology

3.1 Aim of this chapter

This chapter provides an overview of the methodology underpinning the three studies in this thesis. It details the design, data, and sample used in this thesis followed by a discussion of the key variables, and covariates. I then describe the recoding of variables and how I chose to handle the missing data for the thesis. Last, an overview is provided of the statistical techniques that were used to address the thesis research questions.

3.2 Design

A longitudinal secondary data analysis design was followed using the Millennium Cohort Study (MCS). This design benefits from being able to follow individuals across multiple time points meaning that more comprehensive and detailed information is available for each individual (Caruana et al., 2015) than would be available in other designs (e.g., crosssectional designs). At the time of writing the MCS provides data on families across seven waves collected at ages 9 months, 3, 5, 7, 11, 14, and 17 years (Fitzsimons, 2020). Data on this scale would be unobtainable without longitudinal secondary data. This design allows the examination of the association between early circumstances and later outcomes. It also allows for examining dynamic concepts (e.g., SES dynamics) rather than static concepts (Caruana et al., 2015) and places information and events in chronological order so that timing, stability, and change can be examined (Gustafsson, 2010). Longitudinal prospective designs also benefit from reduced recall bias as respondents are asked about their current circumstances rather than being asked to remember past information (Caruana et al., 2015).

Longitudinal secondary data analysis design also comes with inherent problems and the primary issue is sample attrition. In the MCS attrition occurs when respondents are

uncontactable at a known address, cannot be located, or refuse to continue (Plewis, 2007). Attrition leads to reduced sample size and power as well as potential attrition bias. Attrition adds to the complexity of longitudinal secondary data analysis as additional steps must be taken to account for attrition (e.g., applying non-response weights; Mostafa, 2014). Other potential issues are that the data may be collected and stored differently to what the researcher may have preferred which could add additional steps to preparing the data such as recoding (Boslaugh, 2007). Since cohort studies follow a selected sample rather than representative of the population, conclusions drawn can only be applied to the cohort (Gustafsson, 2010).

3.3 Data and sample

The Millennium Cohort Study (MCS) is the most recent UK longitudinal national cohort study that follows the lives of 18851 families who have children that were born in the UK between 2000 and 2002 (Fitzsimons, 2020). At the time, national cohort studies did not include UK wide samples, so the MCS was designed to represent the population of the UK by providing data about children and their families in each of the four UK countries (Plewis, 2007). The MCS was also designed with the goal to provide a sufficient amount of usable data about advantaged and disadvantaged subgroups, ethnic subgroups, and children living in Scotland, Northern Ireland, and Wales (Plewis, 2007).

To ensure the design goals were met, children living in disadvantaged areas, children with ethnic minority backgrounds, and children living in Scotland, Northern Ireland, and Wales were oversampled. To achieve this, within each country (Scotland, England, Ireland, Wales) electoral wards were stratified into *ethnic minority strata, disadvantaged strata*, and *advantaged strata* (Plewis, 2007; Ipsos MORI, 2017). Electoral wards were used because at the time there was no individual level ethnic and family disadvantage information available

to the researchers, but area level information was readily available (Plewis, 2007; Hansen, 2014). Ethnic minority strata were wards that contained an ethnic minority population greater than 30% (Plewis, 2007; Hansen, 2014). Disadvantaged strata contained children who lived in the poorest 25% of wards that were not already included in the ethnic minority strata (Plewis, 2007; Hansen, 2014). The advantaged strata contained children living in any ward not part of the ethnic minority or disadvantaged strata (Plewis, 2007; Hansen, 2014). Scotland, Ireland, and Wales only had two strata (disadvantaged and advantaged) because the number of ethnic minority groups were too small (approximately 1% of the population; Plewis, 2007; Hansen, 2014).

A total of 398 electoral wards were selected for the MCS sample (Plewis, 2007; Ipsos MORI, 2017). Disadvantaged and ethnic minority electoral wards were oversampled, as were wards in Scotland, Wales, and Northern Ireland to ensure that an adequate sample size for each stratum could be achieved while accounting for attrition (Plewis, 2007; Ipsos MORI, 2017). Table 3.1 below presents the number of wards and expected sample sizes by country and strata. As can be seen, the expected sample size was generally oversampled for disadvantage and ethnic minority.

Country - Ward Strata	Number of Electoral Wards	Expected Sample Size
England – Disadvantaged	71	5258
England – Advantaged	110	5511
England – Ethnic Minority	19	2606
Scotland – Disadvantaged	30	1285
Scotland – Advantaged	32	1243
Wales – Disadvantaged	50	2219
Wales – Advantaged	23	897
Northern Ireland – Disadvantaged	40	1322
Northern Ireland – Advantaged	23	762
Total UK	398	21103

Table 3.1. Strata, electoral wards, and expected sample size from the MCS

Note. Source: Plewis (2007).

At the time of writing, the MCS had collected seven waves of data when children were aged 9 month, 3, 5, 7, 11, 14, and 17 years (Connelly & Platt, 2014) with data from age 17 children (MCS 7) recently made available (Fitzsimons et al, 2020). This thesis did not use age 17 data as they were not available in time to be included. To be eligible for inclusion, babies from England and Wales had to be born between September 1st 2000 and August 31st 2001 (52 weeks of births). Babies from Scotland and Northern Ireland had to be born between November 23rd 2000 and November 22nd 2001 (52 weeks of births) to avoid overlap with families taking part in an infant feeding survey that was ongoing during September and October 2000 (Plewis, 2007; Connelly & Platt, 2014). Where there was a shortfall in eligible children, the eligible weeks of birth for taking part were extended from 52 weeks to 59 weeks making the latest possible eligible date of birth January 11th 2002 (Plewis, 2007). Eligible children were identified based on birthdate from the Child Benefit register, and if they were living in one of the selected electoral wards across the UK (Plewis, 2007).

3.4 Data collection procedure

Data from families and children were collected from parents during face-to-face interviews and self-report questionnaires. A total of 324 interviewers were recruited across the UK to collect data from eligible families. When children were approximately 7 months old, the Department for Work and Pensions send out information sheets and opt out forms to eligible families prior to interviews being conducted. Families that did not opt out were then contacted and interviewed when the child was approximately 9.5 months old (Hansen, 2014). At all data collection waves the interviews included a household questionnaire, main respondent interview, and a partner interview. (NatCen, 2003). When children were 3, 5, 7, 11, and 14 years old, a cognitive assessment was administered, and a physical health assessment was completed (NatCen, 2003; Fitsimmons, 2020).

Before the main and partner interviews began the household survey was administered. The household survey included questions such as who lives in the home, the relationship between the adults and the child member of the cohort, and the working status of the adults in the home etc. (Fitzsimons, 2020). The main respondent and partner interviews contained several modules about the cohort child (e.g., modules on child behaviour and child health) and personal questions about themselves (e.g., modules on education and occupation, the family context, parents' health) (Fitzsimons, 2020). Children's cognitive assessments (detailed below) were performed on the interviewer's tablet (Fitzsimons, 2020). Children's physical health (e.g., height, weight, and body fat were taken by the interviewers if consent was given (Fitzsimons, 2020).

3.5 Sample achieved by the MCS at each wave

As seen in table 2 below the initial issued sample was 19,243 families. However, during wave 1 692 families were not identified soon enough to participate meaning the sample was 18,551 families (Plewis, 2007). The reason the 692 families were not identified soon enough was due to delays in the Department for Work and Pensions identifying them as receiving child benefits (Plewis, 2007). Since the child benefit register was used to identify eligible families this means that the 692 families could not take part (Plewis, 2007). Attrition occurred at each wave of the MCS due to a family being categorised as either *ineligible, uncertain eligibility*, unproductive (Hansen, 2014). Ineligible refers to families who have moved to an ineligible address, those who had an error in the Child Benefit record, or if the baby had died. Uncertain eligibility includes families that were not issued to an interviewer or were untraceable (e.g., could not establish if the resident lived at an address within a sample ward; unable to confirm that the child lived at an eligible address; moved home and address could not be ascertained). Unproductive refers to; non-contact with families; refusal to take part; other non-response reasons (e.g., sensitive family circumstances). Families that were categorised as ineligible, uncertain eligibility, or unproductive in two consecutive waves were not issued in the proceeding waves (Hansen, 2014; Ipsos MORI, 2017; Fitzsimons, 2020). To account for panel attrition the MCS provides non-response weights (Plewis, 2007) which are described further in the missing data section 3.8.
	MCS1	MCS2	MCS3	MCS4	MCS5	MCS6
	Age 9 months	Age 3 years	Age 5 years	Age 7 years	Age 11 years	Age 14 years
	N (%)	N (%)	N (%)	N (%)	N (%)	N (%)
Productive	18,551 (96.4)	15,590 (81)	15,246 (79.2)	13,857 (72)	13,287 (69)	11,726 (60.9)
Refusal		1,739 (9)	2,315 (12)	1,811 (9.4)	2,195 (11.4)	3,029 (15.7)
Ineligible		167 (0.9)	300 (1.6)	126 (0.7)	78 (0.4)	45 (0.2)
Untraceable		686 (3.6)	546 (2.8)	706 (3.7)	388 (2)	428 (2.2)
Non-contact		930 (4.8)	546 (2.8)	123 (0.6)	438 (2.3)	75 (0.4)
Not issued	692 (3.6)			2,212 (11.5)	2,851 (14.8)	3,828 (19.9)
Unproductive		131 (0.7)	290 (1.5)	408 (2.1)	6 (0.1)	112 (0.6)
Total	19,243 (100)	19,243 (100)	19,243 (100)	19,243 (100)	19,243 (100)	19,243 (100)

 Table 3.2. MCS productive sample and reason for attrition at each wave

Note. Source: Millennium Cohort Study: User Guide (Fitzsimons, 2020).

3.5.1 Analytic sample

Data from all 6 waves of the MCS were used in this thesis. The productive samples at each wave were 18,551, 15,590, 15,246, 13,857, 13,287, and 11,726 families (Ketende 2010; Mostafa, 2014; Fitzsimons, 2020). The sample used in this thesis was the productive sample from wave 6 (N = 11,726) to ensure that the sample only included participants with longitudinal weights which were used to account for attrition. From the wave 6 productive sample 947 cases were excluded as they did not have any data points from any wave as missing data was handled by Full Information Maximum Likelihood (discussed further in missing data section 3.8) which requires at least one valid value to include a case into the estimation (Muthén et al., 2017). As discussed by Enders (2010) Full Information Maximum Likelihood can only produce estimates by "borrowing" information from observed values. A

further 96 cases were identified as having extreme outlier values on the dependent variables (discussed further in data cleaning section 3.7). This process left a final analytic sample of 10,683 families.

3.6 Measures

The dependent variables in this thesis are children's language ability, internalizing behaviour, and externalizing behaviour. Data on the dependent variables were obtained when children were three (MCS 2), five (MCS 3), seven (MCS 4), eleven (MCS 5), and fourteen (MCS 6) years old. The independent variables were family SES (i.e., family income, maternal education, and maternal occupation). The covariates used were age, gender, ethnicity, birth weight, single or dual parent household. Data on the independent variables and covariates apart from birth weight were available at all six data collection waves when children were 9 months old (MCS1), three (MCS 2), five (MCS 3), seven (MCS 4), eleven (MCS 5), and fourteen (MCS 6) years old. Birth weight was recorded at wave 1.

3.6.1 Language Ability

Children's language ability was measured by testing their verbal vocabulary knowledge when they were 3, 5, 7, 11, and 14 years old (Moulton et al., 2020; McElroy et al 2021). Different measures were used to measure language ability as the children got older. At ages 3, 5, 7, and 11, British Ability Scale (BAS) subscales are used to measure language ability and at age 14 Applied Psychology Unit (APU) Vocabulary Test (Closs, 1976) was used. To allow meaningful comparison across all of the different language measures children's scores were standardised (e.g., Pereyra-Elias et al., 2022). For each British Ability Scale subscale (at 3, 5, 7, and 11 years old) age adjusted standardised T-scores are used to account for children's different ages when the tests were administered. T-scores were used at age 14 but were not age standardised as the Applied Psychology Unit Vocabulary test was a self-completion test conducted online when children were the same age. The following section provides a description of each language ability measure at age 3, 5, 7, 11, and 14.

3.6.2 Age 3 (MCS2) language ability

Children's language ability was tested when they were 3 years old using the BAS Naming Vocabulary subscale. The Naming Vocabulary subscale was designed to assesses the verbal vocabulary knowledge (both spoken and expressive) of children aged between 2 years and 6 months to 7 years and 11 months. The BAS-NV comprised 36 pictures of objects that children were asked to identify and name when shown. The pictures were numbered 1 through 36 with 1 being the easiest and 36 being the most difficult. Children were presented with a first tranche of 16 pictures and only asked to go further if they made fewer than 3 mistakes. If the children progressed to the second tranche, then they saw items 17 through 30. If they made fewer than 3 mistakes, they were then presented with the third tranche containing items 31-36. The precise starting point for each child was dependent on their age and ability which meant that they were not presented with items too easy or difficult for them in order to protect their self-esteem and avoid frustration. This means that children's raw scores (summing the total correct responses) were not comparable. As a result, the raw scores were adjusted for item difficulty and age. Children's raw scores were a minimum of 0 and maximum of 30 correct answers to the naming vocabulary test with a mean of 16.62 (SD=4.811). Previous analyses have found the measure to be reliable (alpha = .81; Elliott, 1997).

111

3.6.3 Age 5 (MCS3) language ability

The BAS Naming Vocabulary subscale was used again to measure language ability children were 5 years old, so the procedure was the same as above when children were age 3. Children scored a minimum of 0 and maximum of 25 correct answers to the naming vocabulary test with a mean of 14.14 (SD=3.826).

3.6.4 Age 7 (MCS4) language ability

The BAS Word Reading subscale was used to measure children's language ability when they were 7 years old. The BAS word reading subscale assesses children's reading vocabulary knowledge between the ages of 5 and 17 years and 11 months, so was suitable for the current study when children were 7 years old. The child is presented with a series of words on a card. There is a total of 90 words divided into 9 blocks of 10 words. Each block of words increased in difficulty as the child progressed through block 1 to 10. Children were asked to read each word aloud with correct pronunciation. The number of blocks the child was asked to attempt was dependent on their performance in the previous block (e.g., if the child struggled with their current block, they would not be asked to attempt the next more difficult block. Example words from the easiest block of words (block 1) are "box", "fish", "jump" and from the most difficult block (block 10) "catastrophe", "regurgitate", and "archaic". Children's raw scores on the word reading task were calculated by summing the total number of correct responses ranging from 0-90 with a mean score of 44.4 (SD=19.278). This scale has been shown to have good reliability in previous studies (Stuart, 1999; Hatcher et al, 1994).

3.6.5 Age 11 (MCS5) language ability

The BAS Word Similarities subscale assesses children's verbal vocabulary knowledge at age 11. There are a total of 21 items with each item containing 3 words that shared a common theme. The task was that upon hearing the 3 words, the child would be asked to provide a 4^{th} word that matched the theme and then state what the theme was. For example, the interviewer would say "apple, orange, banana" and the child could say "strawberry" as the 4^{th} item and "fruit" as the theme. Children only score one point if they provide both the correct word and the correct theme. Therefore, children could get a maximum of 21 points. The data show that the raw scores range from 0- 20 with a mean of 8.69 (SD=3.637).

3.6.6 Age 14 (MCS6) language ability

The Applied Psychology Unit Standardised Vocabulary Test was used to assess children's verbal vocabulary knowledge at age 14. The vocabulary test is designed to measure respondents understanding and meaning of words and is suitable for those aged 14 and over. The word activity involved presenting respondents with a list of 20 target words. For each target word, 5 additional words were presented. Children were asked to identify which of the 5 additional words had the same meaning as the target word. The words used were subsets of those used in the vocabulary test from the 1970 British Cohort Study which were originally from the standardised vocabulary test designed by the Applied Psychology Unit at the University of Edinburgh in 1976. Children were given 1 point for each correct answer and total scores were a sum of all correct answers meaning 20 was the maximum score. In the current sample, the raw scores ranged from 0-19 with a mean of 7.09 (SD= 2.625). Previous analyses have found the measure to be reliable (alpha = .83 - .95; Levy & Goldstein, 1984).

Previous studies using the abovementioned measures of language ability have found small to moderate correlations (r = .2 - .56) indicating a small to moderate degree of association

113

between the measures which provides confidence that the different measures are recording the same underlying construct (Girard et al., 2017; Hernández-Alava & Popli 2017; Forrest et al 2018; Toseeb & St Clair 2020; Kromydas et al.,2022; Pereyra-Elias et al., 2022). The correlation between the APU vocabulary test and the BAS naming vocabulary test is similar to the correlations observed between the BAS subscales at r = .36 (Hill, 2005; Girard et al., 2017; Sullivan et al., 2017). Correlations were run between the measures to ensure that the data used in this thesis had similar correlations with the literature. This was found to be the case as correlations ranged between .247 and .496 which is within the expected range (see table 1).

 Table 3.3. Descriptive statistics and bivariate correlations for language ability at all data

 collection points

	Age 3 BAS	Age 5 BAS	Age 7 Word	Age 11 Word	Age 14 Vocabulary
	Naming	Naming	Reading Task	Similarities Task	Test
	Vocabulary	Vocabulary			
Age 3 BAS Naming	-				
Vocabulary					
Age 5 BAS Naming	.496***	-			
Vocabulary					
Age 7 Word Reading	.278***	.386***	-		
Task					
Age 11 Word	.271***	.378***	.358***	-	
Similarities Task					
Age 14 Vocabulary Test	.247***	.317***	.410***	.354***	-
N	10683	10683	10683	10683	10683
М	49.971	49.894	49.767	49.431	49.248
SD	9.766	9.840	9.841	9.536	9.831

Note. Standardised T-scores used to generate these statistics. *** = p < .001.

3.6.7 Internalizing and Externalizing behaviour

The Strength and Difficulties Questionnaire (SDQ; Goodman, 1997) was used to measure children's internalising and externalizing behaviours at ages 3, 5, 7, 11, and 14 years old and was completed by the parents. The SDQ contains 25 items measuring five dimensions of behavioural development in children aged between 3 and 16. The five dimensions are emotion symptoms, conduct problems, hyperactivity, peer problems, and pro-social behaviour. Each item in the subscales is answered on a 3-point scale ("not true" = 0, "somewhat true" = 1, "certainly true" = 2). Total score on each subscale can range from 0-10. Goodman provides information and syntax for scoring the SDQ and deriving internalizing and externalizing behaviour scales on his website (Goodman: SDQ Info, 2023).

The internalizing behaviour scale is derived by summing the items from the emotional subscale ("often complains of headaches, stomach aches or sickness", "many worries, often seems worried", "often unhappy, down-hearted or tearful", "nervous or clingy in new situations, easily loses confidence", and "many fears, easily scared") and the peer problems subscale ("rather solitary, tends to play alone", "has at least one good friend", "generally liked by other children", "picked on or bullied by other children", and "gets on better with adults than with other children"). Normal internalizing scores range from 0-4, borderline problems scores range from 5-7, and abnormal scores range from 8-20 (Maurice-Stam et al., 2018).

The externalizing behaviour scale is derived by summing the items from the hyperactivity subscale ("*restless, overactive, cannot stay still for long*", "*constantly fidgeting or squirming*", "*thinks things out before acting*", "*sees tasks through to the end, good attention span*", and "*easily distracted, concentration wanders*") and the conduct problems subscale

("often has temper tantrums or hot tempers", "generally obedient, does what adults request", "often fights with other children or bullies them", "can be spiteful to others", and "often argumentative with adults"). Normal externalizing scores range from 0-5, borderline scores range from 6-8, and abnormal scores range from 9-20 (Maurice-Stam et al 2018).

Confirmatory factor analysis indicates that three-factor model (internalizing behaviour, externalizing behaviour, and prosocial behaviour) and a five-factor model (all five subscales) have similar model fit but the 5-factor model was slightly better (CFI = .901; TLI = .954; RMSEA = .049) than the three-factor model (CFI = .900; TLI = .953; RMSEA = .049) (Goodman et al., 2010). Sosu & Schmidt (2017) found a similar result when comparing the three-factor model and five-factor model. They found the five-factor model to be a better fit (CFI = .95: TLI = .94: RMSEA = .02) than the three-factor model (CFI = .921; TLI = .92:RMSEA = .02) but the three-factor model was an adequate fit in both Goodman et al. (2010) and Sosu and Schmidt (2017). However, Goodman et al., (2010) also found substantial overlap between the emotion and peer problems subscales (r between .66 - .71) and between the hyperactivity and conduct problems subscales (r between .71 - .81). Consequently, the convergent and discriminant validity of the internalizing and externalizing model was better than that of the 5-subscale model (e.g., Goodman et al., 2010; Croft et al., 2015) meaning that differentiating between internalizing and externalizing behaviour is clearer than differentiating between each of the 5 individual subscales. As a result, Goodman et al. (2010) suggest that the internalizing and externalizing behaviour scales may produce greater explanatory power than in low-risk non-clinical populations while the 5 subscales are better at identifying specific problem behaviour in high-risk populations (e.g., with mental health problems) (Goodman et al., 2000; Fitzsimmons et al., 2017). Supporting the above suggestion, when Sosu and Schmidt (2017) tested a two-factor model (internalizing and

116

externalizing behaviour without the prosocial subscale) they found good model fit (CFI = .94; TLI = .94) and suggested that research focusing on problem behaviour use a two factor model.

Research indicates that there is good reliability for both the internalizing behaviour scale with α ranging from .73 - .82, and the externalizing behaviour scale with α ranging from .78 - .83 (Palmieri & Smith 2007; Goodman et al., 2010; Hessel, et al., 2017). Sosu and Schmidt (2017) also indicate that the SDQ is an appropriate measure for longitudinal research as they tested a metric invariance model to determine whether the relationship between items in the SDQ remained the same over time. Their metric invariance model showed a good fit (CFI = .95: TLI = .94: RMSEA = .01) meaning that SDQ items measure the same intended constructs across time. This suggests that any changes in children's SDQ outcomes over time is genuine change and not due to changes in the respondents interpretation of the measure or the measure itself.

3.6.8 Measures of socio-economic status

3.6.9 Family income

Information on family household income was gathered at every data collection wave during the face-to-face interviews by asking respondents to state their take home income after tax and other deductions (net income). Responses were then equivalised using the Organisation for Economic Cooperation and Development (OECD) equivalisation scales to generate net equivalised weekly income variables. Equivalising is a method of adjusting household income based on the financial requirements of the household (e.g., adjusts for the size and composition of the household). For equivalization to be done, each family's needs are assigned a value relative to a couple with no children. In the MCS, a couple with no children is given a value of 1. The assigned value of each type of person is as follows: Main respondent = .67, partner = .33, each child between 14 and 18 = .33, each child under 14 =.20. For example, a family of one parent and one child under 14 would be assigned a value of .87 (.67 + .20). Table 3.4 below presents the average household income at each data collection wave. As seen the average household income increased from £317.51 when children were 9 months old to £417.44 when children were 14 years old.

Child age	N	М	SD
9 months	10,683	317.51	204.608
3 years	10,683	343.89	223.367
5 years	10,683	364.98	222.884
7 years	10,683	398.86	232.388
11 years	10,683	418.07	178.661
14 years	10,683	417.44	178.067

Table 3.4. Average household weekly income (£) at each wave

3.6.10 Education

Parents' highest level of education was measured by asking parents to indicate their highest level of academic or vocational education at every data collection wave. Parents who had qualifications could select from thirteen categories of qualifications which are mapped onto National Vocational Qualification (NVQ) system ranging from NVQ 1 (lowest level of qualification) – NVQ 5 (highest level of qualification). The following outlines the guidance on NVQ1-NVQ5:

- NVQ 1 comprises: NVQ level 1, Scottish Vocational Qualification (SVQ) Level 1, General National Vocational Qualification (GNVQ) Foundation Level, General Scottish Vocational Qualification (GSVQ) Level 1; Business and Technology Education Council (BTEC) qualification, Scottish Vocational Educational Council (SCOTVEC) first or general certificate/SCOTVEC modules; City & Guilds part 1/ Royal Society of Arts (RSA) Stage 1, 2, 3/Junior certificate.
- NVQ 2 comprises: NVQ or SVQ Level 2/GNVQ Intermediate or GSVQ Level 2; BTEC, SCOTVEC first or general diploma; City & Guilds Craft or Part II/RSA Diploma.
- NVQ 3 comprises: NVQ or SVQ Level 3/GNVQ Advanced or GSVQ Level 3; Ordinary National Diploma, Ordinary National Certificate, BTEC National, SCOTVEC National Certificate; City & Guilds advanced craft, Part III/RSA Advanced Diploma.
- *NVQ 4* comprises: Nursing or other medical qualifications (below degree level); NVQ or SVQ level 4 or 5; Higher National Diploma (HND), Higher National Certificate (HNC), BTEC Higher Level, RSA Higher Diploma.
- 5. *NVQ 5* comprises professional qualifications at degree level (e.g., graduate member of professional institute, chartered accountant, or surveyor).

Since each category of the NVQ contains several different qualifications, they are further clustered. Studies often summarise the content of the NVQ levels as follows: NVQ I = GCSE/ O Level below grade C; NVQ 2 = GCSE/ O Level grade A-C; NVQ 3 = A Levels/

Highers; *NVQ 4* = Teaching qualification, higher education, or first degree; and *NVQ 5* = Higher degree/ postgraduate qualification (e.g., Quigley et al., 2012). Similar to previous studies (e.g., Quigley et al., 2012; Cheng & Furnham 2012; Furnham & Cheng, 2019) when responses were mapped onto the NVQ, the results were coded such that larger scores indicated higher level of qualification where 0= no qualification 1= GCSE/ O Level below grade C, 2 = GCSE/ O Level grade A-C, 3 = A Levels/ Highers, 4 = Teaching qualification, higher education, or first degree, 5 = Higher degree/ postgraduate qualification (e.g., Quigley et al., 2012). On average, parent's level of parent's education increased from 2.87 at 9 months to 3.252 at age 14 indicating that education increased a classification from GCSE/ O Levels to A Levels/ Highers. Table 3.5 below shows the percentage of parents in each NVQ level at each time point.

Child age	Parent's level of education	%
9 months		
	NVQ1	6.1
	NVQ2	25.5
	NVQ3	17.9
	NVQ4	41.9
	NVQ5	8.6
3 Years		
	NVQ1	5.9
	NVQ2	25.2
	NVQ3	17.5
	NVQ4	42
	NVQ5	9.4
5 Years		
	NVQ1	5.8
	NVQ2	24.9
	NVQ3	17.4
	NVQ4	42.4
	NVQ5	9.4
7 Years		
	NVQ1	4.8
	NVQ2	23
	NVQ3	17.3
	NVQ4	41.7

Table 3.5. Parent's level of education at each wave (N=10,683).

	NVQ5	13.2
11 Years		
	NVQ1	4.8
	NVQ2	20.6
	NVQ3	16.2
	NVQ4	40.9
	NVQ5	17.6
14 Years		
	NVQ1	4.8
	NVQ2	19.6
	NVQ3	15.7
	NVQ4	40.3
	NVQ5	19.7

3.6.11 Occupation (parental class)

Information about parental class was gained by measuring the highest level of household occupation at every data collection wave and was measured on a categorical scale by asking parents to indicate their occupation. Occupations are recorded by the National Statistics Socio-economic Classification (NS-SEC) which has a total of 41 possible responses which is used to derive occupational classes (Office of national statistics, 2016). The MCS collapses the 41 possible responses of the NS-SEC into the 7-class, 5-class, and 3-class versions. The NS-SEC 7-class analytical version was used in this thesis over the 5- and 3-class versions as it may provide more detailed information due to the extra categories and has been found to

categorise people's employment circumstances accurately during face-to-face interviews (Birch & Beerten, 2002).

The 7-class NS-SEC contains the following categories: (1) Large employers and higher managerial and professional occupations, (2) Lower managerial and professional occupations, (3) Intermediate occupations, (4) Small employers and own account workers, (5) Lower supervisory and technical occupations, (6) Semi-routine occupations, (7) Routine occupations. The data from the NS-SEC 7 were recoded such that the larger scores indicated increasingly professional categories of employment where 1= routine occupation, 2= Semi-routine occupations, 3= Lower supervisory and technical occupations, 6= Lower managerial and professional occupations, and 7 = Large employers and higher managerial and professional occupations. On average parents' level of occupation decreased from 4.639 at 9 months to 4.378 at age 14. Although the average level of occupation decreased the average remains in the same category (small employers and own account workers). Table 3.6 below shows the percentage of parents in each NS-SEC class at each time point.

Child age	Parent's level of occupation	%
9 months		
	Routine	4.3
	Semi routine	19.2
	Lower supervisory	12.6
	Small employers	9.2
	Intermediate occupations	10.6
	Lower managerial and professional	22.7
	higher managerial and professional	21.3
3 Years		
	Routine	3.6
	Semi routine	18.7
	Lower supervisory	13.7
	Small employers	11.6
	Intermediate occupations	9.2
	Lower managerial and professional	22.9
	higher managerial and professional	20.4
5 Years		
	Routine	3.8
	Semi routine	19.4
	Lower supervisory	14.4
	Small employers	11.5
	Intermediate occupations	9
	Lower managerial and professional	22.5

Table 3.6. Parent's level of household occupation at each wave (N=10,683)	
---	--

	higher managerial and professional	19.3
7 Years		
	Routine	4.1
	Semi routine	20.3
	Lower supervisory	14
	Small employers	12.2
	Intermediate occupations	8.8
	Lower managerial and professional	22.7
	higher managerial and professional	18
11 Years		
	Routine	4
	Semi routine	20
	Lower supervisory	15.4
	Small employers	14.8
	Intermediate occupations	7.6
	Lower managerial and professional	20.9
	higher managerial and professional	17.4
14 Years		
	Routine	4.6
	Semi routine	19.9
	Lower supervisory	15.2
	Small employers	15
	Intermediate occupations	7.6
	Lower managerial and professional	21.6
	higher managerial and professional	16.2

3.6.12 Covariates

Several covariates known to be associated with family SES and children's outcomes were included in the analyses. *Mothers age at birth* was measured by asking mothers to state their age in years when they gave birth. Mothers age at birth is controlled for as it is associated with family SES (Knipe 2016). Data from the Office of National Statistics show that older mothers are more likely to have higher SES than younger mothers (Knipe 2016). Similarly, younger mothers are more likely to have low SES at the time of birth while older mothers at the time of birth are likely to have higher SES (van Roode et al., 2017). In the analytic sample (N = 11,630), mothers' average age when they gave birth was 28.92 years old (SD = 5.796).

The *number of parents living in the household* was specified by the main respondent during the interview when asked whether they lived in a one or two-parent household. The number of parents living in the household is often controlled for in the SES literature as it has been found to covary with SES but has also been found to be a mediator between the between SES and child development (e.g., Sarsour et al., 2011; Flouri et al., 2016). It also has been found to have a direct association with children's cognitive ability across childhood (Harkness et al., 2019). Children also experience fewer behavioural problems when they live in a dual-parent household compared to when their father does not live with them (Booth, Scott, & King, 2009). There were 10,065 (86.54%) two-parent households and 1,565 (13.46%) one-parent households in the analytic sample.

Ethnicity is measured by asking the main respondent to indicate the child's ethnicity during the interviews. Ethnicity options in the MCS are white, black, Indian, Pakistani and Bangladeshi, Chinese, and mixed (NatCen, 2003). Ethnicity is controlled for as child

developmental differences are often seen between ethnicities (e.g., Kremer et al., 2016; Khaliq, 2020; Sullivan et al., 2021). People from ethnic minorities disproportionately live in deprived areas, have lower SES and are more likely to be poor (Noble et al., 2019). Ethnicity has also been found to influence the way in which others interact with the individual (e.g., Smith et al, 2016) meaning that their experience of their social world may be different and impact development. In the analytic sample, there were 9844 (84.64%) white, 362 (3.11%) black, 295 (2.54%) Indian, 821 (7.06%) Pakistani and Bangladeshi, 107 (0.92%) mixed, and 201 (1.73%) Chinese (or 'other') children.

Gender was measured by asking the main respondent to specify whether the child was male or female. Research indicates that there are gender differences in development are seen such as females scoring higher than males in cognitive tasks (Mous et al., 2017). It has also been found that males exert more problematic externalizing behaviours than their female counterparts (Kremer et al., 2016). Moreover, girls are reportedly more likely to present internalizing problems whereas boys have higher risks for externalizing tendencies (Rescorla et al., 2007; Zahn-Waxler et al., 2008). Therefore, gender will be controlled for in this study. There were 6027 (51.82%) males and 5603 (48.18%) females.

Data on children's *weight at birth* were measured by asking the main respondent for the information during the interview. Children's birth weight has been studied in several context and it has consistently been shown that lower birth weight can have a negative impact on children's cognitive and behavioural development (e.g., Bromley, 2009). Recent longitudinal studies (e.g., Stalnacke et al, 2019) found that low birth weight is associated with poor cognitive ability at 18 months and that the effects persisted into late childhood at age 11.

Since the present study will be using longitudinal data, it will control for children's birth weight. The average weight of children at birth was 3.355kg (or 7.4lbs) (SD = 0.587).

3.7 Data Screening

To prepare the dataset for analysis in Mplus, Tabachnick and Fidell (2007) data cleaning recommendations were followed. *Outliers* can be problematic as they can introduce bias into results estimates. Tabachnick and Fidell (2007) advise that a z-score of less than -3.29 or greater than 3.29 indicates an outlier. This criterion was applied to the data set and found 96 (<1%) outliers with extreme values on the outcome variables which were removed from the sample to avoid producing biased estimates.

It is recommended by Tabachnick and Fidell (2007) that *multicollinearity* be assessed. I ran bivariate correlations between income, education, and occupation at each of the 6 data collection points. The bivariate correlations between income, education, and occupations for all waves are presented in table 3.7 below. The largest correlation between all of the SES dimensions r=.636 and it was between wave 3 education and wave 5 income. A correlation of r=.636 indicates that there are no concerns about multicollinearity as correlations around r=.8 between SES dimensions would be needed to suggest multicollinearity may be an issue (Belinda & Peat, 2014; Young 2017). Further examination of this data was done and found that multicollinearity was not a concern (income, *Tolerance* = .625, *VIF* = 1.601; Education, *Tolerance* +.624, *VIF* = 1.603; Occupation *Tolerance* = .604, *VIF* = 1.655). For multicollinearity to be a problem the variance inflation factor (VIF) would need to be between 5 and 10 and Tolerance closer to 0.1 (Belinda & Peat, 2014). The Durbin-Watson statistic is generated to determine whether there is independence of errors. It ranges between 0 and 4 and it is desirable to have a score close to 2 to indicate independence of errors

(Kramer, 2014). In this case, the Durbin-Watson statistic = 1.982. Overall, multicollinearity was not a problem for this thesis.

Tabachnick and Fidell (2007) advise assessing *normality* with skewness and kurtosis statistics. Skewness observes the symmetry of the distribution around the mean and observes the peak of the mean. Both Skewness and Kurtosis indicate non-normality of data for age 3 and 5 language outcomes, which were left skewed, and for age 5, 7, 11, and 14 internalizing behaviour and age 14 externalizing behaviour which were right skewed. Non-normal data are to be expected due to the sample size of the MCS as real data tend to follow non-normal distributions (Blanca et al., 2013; Bono et al., 2017). It is also expected on the SDQ measures of internalizing and externalizing behaviour as the majority of responses would be expected to cluster on the lower end of the scale. Examining histograms confirms this is the case. No data transformations were carried out for two reasons. First, Central Limit Theorem (CLT: Le Cam, 1986) outlines that non-normal distributions are not problematic when using a large sample as the sample means approximate a normal distribution even when a normal distribution is not seen in the data (Anderson, 2010). CLT shows that sample means will be normally distributed irrespective of the initial distribution (Pek et al., 2017). Therefore, Pek et al. (2017) do not recommend data transformation in large samples. Second, Mplus estimations are capable of handling non-normal data in addition to missing data (Muthén & Muthén, 2017

Table 3.7. Bivariate co	prrelations	between SI	ES dimensi	ons at eac	h wave (N=	:10,683).						
	Wave 1	Wave 2	Wave 3	Wave 4	Wave 5	Wave 6	Wave 1	Wave 2	Wave 3	Wave 4	Wave 5	Wave 6
	income	income	income	income	income	income	education	education	education	education	education	education
Wave 1 education	.543	.553	.523	.525	.630	609.						
Wave 2 education	.527	.548	.522	.523	.630	.608						
Wave 3 education	.531	.549	.531	.530	.636	.616						
Wave 4 education	.496	.516	.502	.518	.629	.605						
Wave 5 education	.459	.477	.463	.478	.636	.610						
Wave 6 education	.433	.452	.436	.451	.596	.616						
Wave 1 occupation	.541	.547	.527	.514	.520	.494	.524	.522	.524	.500	.467	.446
Wave 2 occupation	.514	.533	.511	.499	.499	.476	.516	.513	.516	.496	.468	.450
Wave 3 occupation	.508	.528	.501	.489	.488	.465	.513	.512	.510	.492	.465	.448
Wave 4 occupation	.485	.506	.483	.481	.462	.441	.495	.494	.494	.479	.460	.444
Wave 5 occupation	.449	.467	.456	.458	.444	.430	.470	.473	.474	.469	.464	.451
Wave 6 occupation	.447	.469	.454	.461	.450	.440	.465	.468	.469	.464	.462	.458

 Table 3.7. Bivariate correlations between SES dimensions at each wave

3.8 Missing data

Missing data are problematic in longitudinal research with attrition often ranging between 30-70% attrition Gustavson et al. (2012). Panel attrition is problematic since too much can weaken estimates and introduce bias (Vannette et al., 2018). In the MCS there is both panel attrition and item non-response to deal with. At wave 6 the panel attrition is 39.1% with a productive sample of 60.9% of the initial sample. To account for panel attrition in the MCS, non-response weights have been generated and must be applied for analysis. The MCS nonresponse weights were generated by identifying predictors of non-response via logistic regression (Plewis, 2007; Mostafa, 2014). The largest predictors of non-response in the MCS were family income, the child's gender, mothers age at birth, whether the child was breastfed, ethnicity, housing tenure (e.g., homeowner, renter), type of accommodation (e.g., house, flat), number of parents in the household, highest level of parent's education, highest level of parent's occupation, whether the main respondent was in work (Fitzsimons, 2020). Once these predictors of non-response were identified, predicted probabilities were generated to indicate the probability of a family responding or not responding. The final step involves using inverse probability weighting (IPW; Hofler et al., 2005) to generate the inverse of the predicted probabilities. IPW assigns more weight to families who have a low probability of responding (Plewis, 2007).

A missing values analysis was run to determine the extent of item non-response from the analytic sample (N = 10,683). A stringent percentage (1%) of missing values was used to identify the smallest of omissions and found that the total missing values in the dataset were 5.016%. In terms of cases, 6,658 (62.32%) families had complete data at all waves and 4,025 (37.68%) had at least one missing value. Table 3.8 below presents the missingness for each

131

variable in the dataset in descending order. Wave 4 occupation had the most missing data at 1416 (13.3%) cases.

Variable	N Missing	% Missing	N Valid
Wave 4 Occupation	1416	13.30%	9267
Wave 3 Occupation	1246	11.70%	9437
Wave 4 Language Ability	1021	9.60%	9662
Wave 5 Occupation	989	9.30%	9694
Wave 4 Internalizing	964	9.00%	9719
Wave 4 Externalizing	962	9.00%	9721
Wave 1 Occupation	913	8.50%	9770
Wave 4 Income	884	8.30%	9799
Wave 4 Education	852	8.00%	9831
Wave 6 Language Ability	842	7.90%	9841
Wave 5 Internalizing	810	7.60%	9873
Wave 5 Externalizing	810	7.60%	9873
Wave 2 Occupation	797	7.50%	9886
Wave 6 Occupation	790	7.40%	9893
Wave 3 Externalizing	781	7.30%	9902
Wave 3 Internalizing	780	7.30%	9903
Wave 5 Language Ability	688	6.40%	9995
Wave 3 Education	652	6.10%	10031
Wave 5 Education	637	6.00%	10046

 Table 2.8. Missing values from highest to lowest

Wave 3 Income	581	5.40%	10102
Wave 1 Education	532	5.00%	10151
Wave 1 Income	524	4.90%	10159
Wave 2 Internalizing	514	4.80%	10169
Wave 2 Externalizing	513	4.80%	10170
Wave 5 Income	509	4.80%	10174
Wave 3 Language Ability	494	4.60%	10189
Birth Weight	439	4.10%	10244
Mother age	421	3.90%	10262
Wave 6 Externalizing	334	3.10%	10349
Wave 6 Internalizing	333	3.10%	10350
Wave 6 Education	169	1.60%	10514
Wave 2 Education	151	1.40%	10532
Wave 2 Income	147	1.40%	10536

Missing data can take one of three forms: missing completely at random (MCAR), missing at random (MAR), and missing not at random (MNAR) (Muthén & Muthén, 2017). MCAR means that the missing data are independent of any observed values in the dataset. MAR means that there is a systematic relationship between the missing data and the observed data in the dataset. The probability of missing values under MAR depends on the observed data rather than the unobserved data. MNAR is the most problematic type of missing data to deal with as it is predicted by unobserved variables (Muthén & Muthén, 2017). To assess the nature of the missing data in this thesis Little's (1988) Missing Completely at Random (MCAR) test was conducted. Chi square test found that the data were MAR as the results reject the null hypothesis X^2 =29140.138 (*df*=18636; *p*= <.001).

There are several methods for dealing with missing data. Listwise deletion is the default setting in most statistics software which deletes cases with missing data so that analyses are conducted on complete cases only. Listwise deletion was not used as it would have resulted in 4,025 families being removed from the analytic sample as they had at least one missing value. Other methods are able to maintain the sample size to avoid this bias. Some longitudinal panel surveys like the MCS provide longitudinal weights to account for attrition and non-response. However, recent research has indicated that using weights on their own can bias estimates due to how they are generated (Lee et al., 2019). Lee et al., (2019) compared the performance of non-response weights, multiple imputation (MI), and full information maximum likelihood (FIML). They found that MI and FIML performed comparably, and both outperformed the longitudinal non-response weights on their own by producing less biased estimates. They recommend applying the longitudinal attrition and non-response weights provided in the dataset (in this case the Millennium Cohort Study) followed by FIML or MI (Lee et al., 2019) which is a common approach in the literature (Flouri et al., 2018).

Multiple imputation involves generating several datasets and imputing missing values into each new dataset (called imputations). It is recommended that anywhere from 2-20 imputations be run (von Hippel, 2018) but more is generally better. This can become demanding due to hardware limitations and becomes less efficient than FIML (Myung, 2003). This is because FIML uses the available observed data to generate unbiased estimates in the analysis rather than imputing values. It does this by generating a likelihood function for each case based on the observed data and then accumulating the likelihood functions of each case to generate unbiased estimates (Enders & Bandalos, 2001). However, both MI and FIML assume that the missing data are missing at random (MAR) so they cannot be used in all cases, but since the missing data in this study are MAR they can be used. Research comparing the performance of MI and FIML indicates the difference between them is negligible (Lee et al., 2019; Lim & Cheung, 2022) highlighting that they are both good options for addressing missing data while retaining cases. They have both been found to be effective for missing data over 50% and up to 80% (Todd et al 2013; Kontopantelis et al., 2017; Nelson et al., 2021). However, Larsen (2011) concluded that FIML was superior to MI as it correctly estimated standard errors. Item non-response in this thesis is therefore handled using FIML because it is recommended in the literature.

3.9 Data Analysis procedures

In study 1, multiple regression models were run to examine the association between family SES dimensions and children's language ability, internalizing behaviour and externalizing behaviour during early, middle, and late childhood when children were 3, 7, and 14 years old. Model 1 regressed *early* family income, level of education, and level of occupation on children's outcomes controlling for covariates. Model 2 regressed *middle* family income, level of education, and level of occupation on children's age 7 outcomes controlling for previous income, education, and covariates. Model 3 regressed *late* family income, level of education, and level of occupation on children's age 14 outcomes. Controlling for previous income, education, and level of occupation, and occupation, and covariates. This process was repeated for language ability, internalizing behaviour, and externalizing behaviour.

To examine whether the strength of the associations between different SES dimensions and children's outcomes were significantly different I calculated t-values for each pair of variables and then the level of significance was drawn from the t statistic using the t

distribution (Clogg et al., 1995). The following formula was used to calculate the t statistic such that the difference in coefficients (b) were divided by the square root of the pooled standard errors (s) squared:

$$t = (b_1 - b_2) / \sqrt{(s_{b1}^2 + s_{b2}^2)}$$

Once the *t* statistic was calculated the value was mapped onto the *t* distribution chart. For quick reference a *t* statistic of 1.96 = p < .05, *t* statistic of 2.58 = p < .01, and *t* statistic of 3.09 = p < .001 (Wheeler, 2015).

In study 2, Latent Class Growth Analyses (LCGA) were run followed by multiple hierarchical linear regressions to examine the different SES dimensions in the data and their associations with children's language ability, internalizing behaviour, and externalizing behaviour. LCGA is a type of Growth Mixture Modelling (GMM) used to examine trajectories of SES in study two. GMM itself is an extension of Growth Curve Modelling (GCM). Growth Curve Modelling takes a person-centred approach to analysis and examines the between person (or between group) differences in within person (or within group) change (Muthén & Muthén, 2000; Ram & Grimm, 2009). GMM expands on this as it aims to describe the longitudinal patterns of change between and within *latent* subgroups (Ram & Grimm, 2009; Reinecke & Seddig, 2011). Latent groups are unobserved groups that are not directly measured in the data (Jung & Wickrama, 2008). Individuals are assigned to latent groups (sometimes referred to as latent classes) based on patterns of their observed responses (Jung & Wickrama, 2008). For example, family income trajectories are not measured directly but people can be assigned to a latent income trajectory based on the income data that they provide across several time points. The utility of GMM comes from its flexibility. Conventional growth models assume that the different intercepts and slopes from each person in the sample can be represented by a single trajectory using a single set of growth parameters (Jung & Wickrama, 2008). GMM differs in that it assumes different latent trajectories are present in the group and assigns people into different latent trajectories based on similar intercepts and slopes (Jung & Wickrama, 2008; Berlin et al., 2013). This essentially allows each latent trajectory in the sample to have its own growth parameters and estimates rather than assuming the entire sample can be described by a single trajectory (Berlin et al., 2013). GMM also allows for within class variation meaning that within each latent group there can be differences (Jung & Wickrama, 2008; Berlin et al., 2013). Latent Class Growth Analysis (LCGA) is a type of GMM that does not allow within class variation and only allows across class variation (Berlin et al., 2013). This means that it is able to identify several latent trajectories but assumes that the people categorised into the same latent trajectories are homogenous (Berlin et al., 2013). This allows for easier interpretation of classes as they will be clearer due to no within class variation but also accounts for any unobserved heterogeneity that could be missed if GMM was used (Jung & Wickrama, 2008; Berlin et al., 2013). This is useful when the goal is to identify different latent groups and compare their associations with outcomes.

To assess the latent class growth analysis model fit, goodness of fit indices were used to explain how well each specified model explained the relationship between variables (Maydeu-Olivares & Garcı[´]a-Forero, 2010). In the case of LCGA, it can help determine how many latent classes (or trajectories) best fit the observed data by comparing the model fit of two or more different models (Jung & Wickrama, 2008). The following paragraphs will discuss the goodness to fit indices and table 3.9 below will present a summary of them.

137

The two most commonly used indices are the Akaike's information criterion (AIC; Akaike, 1974) and the Bayesian information criterion (BIC; Schwarz, 1978). They both assesses model fit based on the likelihood estimates within given parameters (Burnham & Anderson 2002). The AIC and BIC estimates essentially reward a model for two reasons: being uncomplex and fitting the data well (Burnham & Anderson 2002). As more parameters are added to the model, the more complex the model becomes and the more it is penalised in the estimates (Burnham & Anderson 2002). On the other hand, the better a model fits the data the more it is rewarded in the estimates (Burnham & Anderson 2002). For example, AIC is denoted below where *k* is the number of parameters, log is the natural logarithm, and (\hat{L}) is a measure of goodness of fit:

$$AIC = 2k - 2\log(\hat{L})$$

As the number of parameters increase (k increases) the model is penalised us for becoming more complicated. As the model fits the data better (log (\hat{L}) decreases), the model is rewarded. AIC estimate is given for each model and the best model will generally have a lower estimate indicating that it strikes a better balance between complexity and model fit than other competing models (Nylund et al., 2007). BIC is essentially the same as AIC but adds an additional penalty as the sample size increases as seen in the formula: BIC = $klog(n) - 2log(\hat{L})$ where log(n) accounts for the sample size. Again, the lower the BIC the better the model fits the data (Nylund et al., 2007).

Entropy was assessed as it is a measure of the overall accuracy of the latent class separation and helps researchers determine the best model for their data by indicating the extent to which latent classes are 'separate' from each other or 'less fuzzy' (Weiss & Dardick, 2016). Entropy values range from 0-1 where 0 = no separation between classes and 1 = complete separation between classes (Weiss & Dardick, 2016). The closer the entropy value is to 1 the better as this indicates that the classes in the model are distinct from each other and indicates that the model likely fits the data well. Entropy values >.8 indicate that individuals are assigned to different groups with a good level of accuracy (Celeux & Soromenho, 1996).

Likelihood ratio tests were assessed as they are designed to compare the model fit of two models. They assess whether the current model is a better fit than a model with one less class in it. For example, it assesses whether a 4-class model is a significant improvement over a 3-class model. These tests are sometimes called K-1 tests. Two of these tests provided by Mplus are the Vong-Lo-Mendell-Rubin test (LMR; Lo, Mendell, & Rubin, 2001) and the bootstrapped likelihood ratio test (BLRT; McLachlan & Peel, 2000). In both cases if the *p*-value is significant (<.05) then it means that the current model is a significantly better fit than the previous model with one less class. The difference between LMR and BLRT is that BLRT uses bootstrapped samples to generate estimate whether the difference between class models is significant (Nylund et al., 2007).

The steps to conducting the analysis is as follows. First, latent class growth analysis models were created for family income, parents' occupation, and parents' education to examine whether different trajectories of each SES dimension could be found in the sample. This was done by using the income, education, and occupation data from each timepoint in the MCS. It is recommended that two classes be examined first so that each progressive model can be assessed to check if it offers a better fit over the previous model (Jung & Wickrama, 2008). Starting with the smallest model and working up is the most efficient way to assess whether or not increasing the complexity of the model gave a better representation of the data. Second, each model was assessed by examining the output of the abovementioned criteria (AIC, BIC, Entropy, LMR and BLRT). Third, once the appropriate latent class growth analysis model had been selected to indicate the number income, education, and occupation

139

trajectories, the latent class probabilities were exported into a separate file. Dummy variables were then created for the latent class probabilities so that multiple regressions could be performed using each of the income, education, and occupation trajectories. A series of hierarchical regressions were then performed to assess the associations between each of the generated latent class trajectories and children's age 14 language ability, internalizing behaviour, and externalizing behaviour.

Indicator	Suggested interpretations
AIC	Value less than model with one less class suggests improved model fit
BIC	Value less than model with one less class suggests improved model fit
Entropy	Value greater than 0.8 suggests participants assigned to class accurately
LMR	p < 0.05 suggests improved model fit over model with one less class
BLRT	p < 0.05 suggests improved model fit over model with one less class

Table 3.9. Summary of the goodness of fit indices

In paper 3, latent class growth analysis was run to examine the different trajectories of children's language ability, internalizing behaviour, and externalizing behaviour followed by multinomial logistic regression and the generation of predicted probabilities to examine the likelihood of children belonging to particular outcome trajectories based on their family SES trajectories. The same trajectories that were identified in the latent class growth analysis in paper two were used in paper one as the same process would have to have been carried out. Further latent class growth analysis models were then created for children's language ability, internalizing behaviour, and externalizing behaviour to examine whether different trajectories of each child outcome could be found in the sample. The same process of starting with the

smallest model and working up to more latent classes was used. Each model was assessed by examining the output of the abovementioned criteria (AIC, BIC, Entropy, LMR and BLRT). Once the appropriate latent class growth analysis models had been selected to indicate the number language ability, internalizing behaviour, and externalizing behaviour trajectories, the latent class probabilities were exported into a separate file. Dummy variables were then created for the outcome latent class probabilities so that multinomial logistic regression could be run. A series of multinomial logistic regressions were then performed to examine the likelihood that membership to SES trajectories was associated with the likelihood of belonging to a particular outcome trajectory. Multinomial logistic regressions were required as there were several levels of dependent and independent variables. The predicted probabilities of belonging to improve interpretability of the results. Multiple regression models and latent class growth analysis were conducted in Mplus version 8 and multinomial logistic regression, and predicted probabilities were conducted in R.

3.10 Summary

This chapter presented the overall methodology that was used to conduct the three studies in this thesis. Each of the studies draw on different aspects of the outlined methodology. In each of the following three results chapters (chapters 4, 5, and 6) the specific measures and analyses that were used will be highlighted. The results will then be presented, and the implications discussed.

Chapter 4 - Associations between family SES dimensions and children's language ability, internalizing and externalizing behaviours at different stages of childhood

4.1 Aim of this chapter

This chapter presents the results of study one. An introduction to the study is provided, followed by the research questions that will be addressed. The methodology used to address the research questions is presented. The findings are then presented, followed by a discussion of the findings and their implications.

4.2 Introduction

Familial socio-economic status (SES) is consistently associated with children's language ability, and internalizing and externalizing behaviours with children from lower SES families showing poorer language outcomes (Violato et al.,2011; Waldfogel 2013; Platt et al. 2014; Kiernan & Mensah, 2009, 2011; Dickerson & Popli, 2014, 2016; Zachrisson & Dearing, 2015; Hernandez-Alava & Popli, 2017), and internalizing and externalizing outcomes (Bøe et al., 2014; Hosokawa & Katsura, 2017; Hernandez-Alava & Popli, 2017; Fitzsimons et al., 2017; Tamura et al., 2020). Children with language and socioemotional problems disproportionally belong to disadvantaged families (Information Services Division, 2019).

Although the associations between family SES and children's developmental outcomes are well documented, there remain complexities about the relationship that needs to be examined. The stage of childhood during which children experience low SES and the conceptualisation of SES are two things that can impact the association between SES and children's outcomes (Schoon et al., 2012; Burchinal et al., 2018; Miller et al., 2021). Evidence indicates that family SES at specific stages of childhood may lead to different effects on children's outcomes (Schoon et al., 2012; Burchinal et al., 2018; Lai et al., 2019). For instance, both Duncan et al. (2010) and Lai et al. (2019) found that family income and poverty measures were more strongly associated with children's behaviour during late childhood than during early childhood, while other studies found that SES had a stronger association with children's cognitive outcomes during the earlier years of life (Schoon et al., 2011; 2013; Dickerson & Popli, 2012; 2016). However, there is a lack of clarity about which stage of childhood SES is most important for children's language ability, internalizing behaviour, and externalizing behaviour. This is because previous studies tend to focus on short periods of time such as early- to mid-childhood, or mid- to late childhood (Schoon et al., 2012; Burchinal et al., 2018). To better understand at which time SES is most important for children's developmental outcomes the different stages of childhood must be examined in a single study.

SES refers to an individual's social position based on their ability to access and deploy financial, human, and cultural resources (APA, 2007). SES is most commonly represented by income, education, and occupation (APA, 2007; Ryan et al., 2015). Most current literature examining the associations between SES and child development operationalise SES either as a composite measure (Dearden et al., 2011; Hanscombe et al., 2012; Poulain et al., 2019; Judd et al., 2020; Jerrim, 2020; Schoon et al., 2021; Britton et al., 2021) or use only one of the dimensions such as family income as a proxy measure of SES (Dohoon, 2014; Shaw & Shelleby, 2014; Shelleby et al., 2014; Jackson et al., 2017; Miller & Votruba-Drzal 2017; Comeau et al., 2018; Hoyt et al., 2019). However, composite measures of SES are limited by the fact that they do not differentiate between the individual associations that each SES

dimension has with children's outcomes. In other words, the associations that each SES dimension may have with children's outcomes could be masked when a combined measure of SES dimensions is used (Cowan et al., 2012; Eriksson et al., 2021; Pearson et al., 2022). On the other hand, using only one dimension such as income may overestimate the association between income and children's outcomes if other SES dimensions are not included in the model (Cowan et al., 2012). This is because income alone may capture the overlapping effect of education and occupation due to their correlations (Cowan et al., 2012). To capture more accurate associations research should include the different SES dimensions in the model.

This study aims to address these gaps in the literature by examining the unique associations that family income, parent's education, and parent's occupation have with children's language ability, externalizing and internalizing behaviours at different stages of childhood. The stages of childhood that will be examined are early, middle, and late. The goal is to find out whether there are different stages of childhood during which children are particularly sensitive to the adverse effects of low SES and whether different SES dimensions are important at different times. It will also enable us to evaluate which stage of SES in childhood is more important for language ability, externalizing behaviour, and internalizing behaviour.

4.2.1 Theoretical framework for understanding the role of SES in children's

development

Developmental theories illustrate how and why SES is associated with children's developmental outcomes. Bioecological Systems Theory (Bronfenbrenner, 1986) proposes child development takes place due to complex bidirectional interactions between children and their proximal and distal environments. Interactions with proximal environments (e.g., the
family and home) are more influential for children's development because they are most frequently exposed to these contexts (Bronfenbrenner, 1979; 1999; 2006). Family SES is important because it can determine the nature of the environments that children interact with as higher SES families are likely to have more frequent and higher quality interactions with their children and show more positive parenting behaviours than lower SES families (Rowe, 2008; Conger et al., 2010; Huttenlocher et al., 2010; Weisleder & Fernald, 2013; Zauche et al., 2016; Cote et al., 2018; Zhu et al., 2019). The specific mechanisms by which family SES can influence developmental outcomes are outlined by the family stress and investment models (Conger et al., 2010; Conger & Donnellan, 2007).

The family stress model proposes that SES is associated with children's development through parental distress and impaired family interactions (Conger et al., 2010). It states that economic pressures such as being unable to pay bills, buy nutritious food, clothing or other material needs lead to parental stress which causes increased conflict between parents (e.g., arguments and separations), harsh or inconsistent parenting, child neglect, and poorer parent-child interactions. Family stress research has found each of these impaired pathways lead to poorer developmental outcomes (Kiernan & Manesh, 2009; 2011; Avan et al., 2010; Micklewright et al., 2012; Azzi-Lessing, 2013; Shelleby et al., 2014; Sun et al., 2015; Neppl et al., 2016; Sosu & Schmidt 2017; Ahun et al., 2017). The *family investment model* states that family income is associated with children's development because it determines parents' ability to invest in the resources required for good child development such as nutritious food, educational materials, activities, and a good environment (Conger & Donnellan, 2007). Low SES families have fewer resources to invest and must concentrate on providing the basic requirements to children (e.g., food and shelter) while high SES families can provide developmental support (e.g., extra tuition and educational experiences). Family SES,

therefore, determines the environments that children have access to and interact with which ultimately shapes their development (Bronfenbrenner, 1986). Research indicates the more families invest in their children the better their language ability and behavioural outcomes are (Kalil, et al., 2012; Sohr-Preston et al., 2013; Prevoo et al., 2014; Sosu & Schmidt, 2017; Longo et al., 2017; Jeong et al., 2017; Vasilyava et al., 2018; Fomby & Musick, 2018).

Although the family stress and investment models focus primarily on the associations that family income has with children's development, the literature highlights that other dimensions of SES can also influence children's developmental outcomes (Sullivan & Brown, 2013; Coddington et al., 2014; Gonzalez et al., 2020; Miller et al., 2021). This makes sense as theory indicates that SES is a multidimensional construct (Weber, 1968; Bourdieu, 1986; Coleman, 1988). For instance, Weber's (1968) theory of social class argues that people's SES is based on their positions across several different indices and proposed three dimensions: classes, status groups, and parties. Classes refer to economic resources (e.g., income or occupation), status groups refer to the level of prestige in the community and skills (e.g., level of occupation or education), and parties refer to access to power (e.g., income and occupation) (Liberatos et al., 1988). Further theories (e.g., Bourdieu, 1986; Coleman, 1988) also argue that SES is a multidimensional construct based on one's access to, and ability to deploy, different forms of capital (e.g., economic, human, and social) to their benefit. Economic capital refers to purchasing power (e.g., income), social capital refers to formal (e.g., occupations) and informal (e.g., peer groups) social networks that individuals accrue over time, and human capital refers to the education, skills, and knowledge that people have gained. Weber argues that SES is determined by one's ability to utilise education, skills, networks, and other qualities to their advantage to navigate their social environment (Weber, 1968). Therefore, given that there are different dimensions of SES representing distinct

constructs it is likely that they have different associations with children's developmental outcomes due to shared and unique pathways as suggested by the stress and investment models.

4.2.2 Evidence of SES dimensions and children's development

Although theory suggests that SES is a multidimensional construct, relatively few studies have examined the unique associations that income, education, and occupation as individual dimensions have with children's outcomes as most of the literature focuses on income measures (Letourneau et al., 2011; Cooper & Stewart, 2017; Fitzsimons et al., 2017). However, there is evidence that the SES dimensions operate via overlapping pathways outlined in the family stress and investment models. For instance, income and occupation are both associated with parental stress and family conflict pathways which are associated with children's development (Ponnet, 2014; Shelleby et al., 2014; Dockery et al., 2016; Sosu & Schmidt, 2017; Cho, 2018; Moreira et al., 2019). Similarly, income and education are both associated with the home environment, standards of living, and parenting practices which are associated with children's development (Rijlaarsdam et al., 2013; Coddington et al., 2014; Longo et al., 2017; Jeong et al., 2017; Sosu & Schmidt, 2017; Fomby & Musick, 2018). Finally, education and occupation both operate via parental expectation pathways which are associated with children's development (Dubow et al., 2009; Kalil et al., 2012; Irwin & Elley, 2013; Plenty & Jonsson, 2020; Davis-Kean et al., 2021). There are also several unique pathways through which different SES dimensions are associated with children's developmental outcomes. Parents' education for example was found to operate via cognitive and linguistic stimulation and early childcare and education pathways, while income operated via purchasing power (Coddington et al., 2014). The different SES dimensions and the resources they represent are therefore likely to have unique associations with children's

language and behavioural outcomes via both unique and overlapping pathways (Sullivan & Brown, 2013; Coddington et al., 2014; Hosokawa & Katsura, 2018).

However, evidence comparing the strength of associations between different SES dimensions and children's development is limited. Some studies indicate parent's level of education has a stronger association with children's language outcomes than family income (Davis-Kean, 2005; Sullivan & Brown 2013; Coddington et al., 2014; Gonzaleza et al., 2020) while income has the largest association with children's internalizing and externalizing behaviour (Bøe et al., 2014; Hosokawa & Katsura, 2018). Two studies examine the associations that income, education and occupation have with children's language ability and came to different conclusions (Sullivan & Brown, 2013; Miller et al., 2021). Sullivan and Brown (2013) found that education had the strongest association with children's language ability at age 16 followed by income, but occupation was not significantly associated with children's language. Miller et al. (2021) on the other hand found that parents' occupation had larger associations with children's language ability compared to family income and parents' education for children aged 5-6 years old, 9-10 years old, and 13-14 years old. Miller et al. (2021) also found that occupation had the largest association with children's externalizing behaviour at all stages of childhood and internalizing behaviour at ages 13-14. Parents' education had the largest association with children's internalizing behaviour at ages 5-6 and 9-10. These studies indicate that associations between SES and children's language and behavioural development may differ based on the SES dimensions being considered and the stage of childhood being examined, but few studies have examined this.

4.2.3 Family SES and child development at different stages of childhood

Prior research mostly focuses on associations between early childhood SES and children's future developmental outcomes because early childhood is considered a sensitive period for children's development where they are particularly vulnerable to adverse environmental factors such as low SES (Sylva 1997; Kuhl et al., 2005). Research tends to support the assumption that early childhood is a sensitive period for children as this is when the brain has been found to develop the neural structures and functions that are responsible for future cognitive and behavioural processes (Zeanah 2011; Brummelte et al., 2017; Gilkerson et al 2018; Nelson & Gabard-Durnam 2020). Findings generally show that experiencing low SES during the earliest years of life has larger negative associations with children's developmental outcomes compared to low SES during later childhood (Schoon et al., 2012; Burchinal et al., 2018). For instance, family income at birth has a stronger association with children's midchildhood language outcomes than family income at age 3, 5, or 7 (Schoon et al., 2012; Dickerson & Popli, 2012; Burchinal et al., 2018). Similarly, family income at birth has a stronger association with children's mid-childhood internalizing and externalizing behaviour than family income at age 3 or 5 (Schoon et al., 2013). However, evidence suggests that sensitive periods can differ depending on the outcome being considered (e.g., Campbell et al., 2000; Costello et al., 2005). For example, low SES appears to have larger negative associations with cognitive outcomes during early childhood while low SES during later childhood and adolescence is important for behavioural problems (Gibb et al., 2012; Lai et al., 2019; Reynolds et al, 2019).

Evidence indicates that adolescence is a sensitive period for behavioural problems due to new experiences such as being exposed to stress such as bullying (Blakemore2014; Fuhrmann et al., 2015; Efstathopoulos et al., 2018). SES in late childhood is found to be more important

than early childhood SES, particularly for internalizing and externalizing behaviour (e.g., Duncan et al., 2010; Gibb et al., 2012; Lai et al., 2019; Reynolds et al 2019). Some studies have found that early childhood income had no association with late childhood internalizing and externalizing behaviour and only late childhood income was associated with their behaviour (Duncan et al., 2010; Lai et al., 2019). Similarly, mid-childhood income has been found to have larger associations with children's internalizing and externalizing behaviour than early childhood income. Although the literature favours early childhood SES as being the most important for children's developmental outcomes, evidence indicates that other stages of childhood can be equally or more important, but few studies look at early, middle, and late childhood in a single study.

4.2.4 The current study

The current study aims to extend the existing literature in two ways. First, the evidence indicates that SES at different stages of childhood has different associations with children's development but it is unclear which childhood stage is most important for language ability, internalizing problems, and externalizing problems. Since these outcomes are a focus of current UK policy (The Childcare Act, 2006; Well-Being of Future Generations Act, 2015; Children's Services Co-operation Act, 2015; The Education Act, 2016), this study will address this gap in the literature by examining the SES associations with these outcomes at all stages of childhood. Second, the literature largely focuses on the associations between income at sensitive periods and children's developmental outcomes and neglects to consider that children may be more sensitive to their parents' education or occupation at different stages of childhood. For example, children's language development is sensitive to parents' education during early childhood but children's behaviour problems are sensitive to parents' education during later childhood and adolescence (Sullivan & Brown, 2013; Miller et al.,

2021). This indicates that different SES dimensions can be important at different stages of childhood yet several UK policies that are designed to support family SES end after early childhood (e.g., Child Payment ends after 6 years and free early childcare ends after 4 years). Consequently, family income and parents' availability to work may be reduced which is counterintuitive to the policy goal of maximizing family income through employment (Welfare Reform Act, 2016) and could have a negative effect on children's development. Therefore, examining when children are most sensitive to different SES dimensions will help policymakers understand the effect that current policies have and indicate at what time policy could be most beneficial. This study will address these two points by examining the associations that each SES dimension (income, education, and occupation) has on children's language, internalizing, and externalizing outcomes during early, middle, and late childhood. The following research questions will be addressed:

- 1a) To what extent are dimensions of family SES (income, education, and occupation) uniquely associated with children's developmental outcomes (language development, internalizing, and externalizing behaviours)?
- 1b) Are some SES indicators more strongly associated with children's language and behavioural outcomes and is this consistent through to adolescence?
- 2) To what extent does the association between family SES and children's outcomes vary by stage of development?

4.3 Methodology

4.3.1 Sample

The data used in this study come from the first six waves of the Millennium Cohort Study (MCS). The MCS is a longitudinal national cohort study in the UK that follows the lives of 18851 families with a child that was born in the UK between 2001 and 2002 when children were 9 months old (Fitzsimons, 2017). The data were collected when children were aged, 9 months, 3, 5, 7, 11, and 14 years old and the productive samples at each wave were 18,551, 15,590, 15,246, 13,857, 13,287, and 11,726 families (Ketende 2010; Mostafa, 2014; Fitzsimons, 2020). The sample was obtained from 398 electoral wards across the UK and families from disadvantaged and ethnic minority electoral wards were oversampled to ensure that adequate sample sizes for each stratum could be achieved while accounting for attrition (Plewis, 2007; Ipsos MORI, 2017). As discussed in chapter 3, the productive sample from wave 6 (N = 11,726) was used in this study to ensure that all families included had longitudinal weights assigned to them which were used to account for attrition. 947 cases were removed for having no valid information for any of the variables of interest at any data collection wave and a further 96 cases were identified as having extreme outlier values. This process left a final analytic sample of 10,683 families.

4.3.2 Measures

The dependent variables in this study were children's language ability, internalizing behaviour, and externalizing behaviour. Data on the independent and dependent variables were obtained when children were three (MCS 2), seven (MCS 4), and fourteen (MCS 6) years old. The independent variables were family SES (i.e., family income, maternal education, and maternal occupation). The covariates used were age, gender, ethnicity, birth

weight, single or dual parent household. Data on the independent variables and covariates apart from birth weight were available at all data collection waves when children were three (MCS 2), seven (MCS 4), and fourteen (MCS 6) years old. Birth weight was recorded at wave 1.

4.3.3 Language Ability

Children's language ability was measured by testing their vocabulary knowledge when they were 3, 7, and 14 years old (Moulton et al., 2020, McElroy et al 2021). Different language ability measures were used as the children got older. British Ability Scale (BAS) subscales are used to measure language ability when children were 3 and 7 years old, and the Applied Psychology Unit (APU) Vocabulary Test (Closs, 1976) was used at age 14. To allow meaningful comparison across all of the different language measures children's scores were standardised (e.g., Pereyra-Elias et al., 2022). For each British Ability Scale subscale (at 3 and 7 years old) age-adjusted standardised T-scores are used to account for children's different ages when the tests were administered. Standardised T-scores were used at age 14 but were not age-adjusted as the Applied Psychology Unit Vocabulary test was a self-completion test conducted online when children were the same age. The following section provides a description of each language ability measure at age 3, 7, and 14.

4.3.4 Age 3 (MCS2) language ability

The BAS Naming Vocabulary subscale was used to measure children's language ability when they were 3 years old. The Naming Vocabulary subscale comprised 36 pictures of objects that children were asked to identify and name when shown. The pictures were numbered 1 through 36 with 1 being the easiest and 36 being the most difficult. Children were presented with a first tranche of 16 pictures and were only asked to go further if they made fewer than 3

mistakes. If the children progressed to the second tranche, then they saw items 17 through 30. If they made fewer than 3 mistakes, they were then presented with the third tranche containing items 31-36. The precise starting point for each child was dependent on their age and ability which meant that they were not presented with items too easy or difficult for them in order to protect their self-esteem and avoid frustration. This means that children's raw scores (summing the total correct responses) were not comparable. As a result, the raw scores were adjusted for item difficulty and age. Children scored a minimum of 0 and a maximum of 30 correct answers to the naming vocabulary test with a mean of 16.62 (SD=4.811). The MCS uses this data to generate T-scores for researchers to use. Previous analyses have found the measure to be reliable (alpha = .81; Elliott, 1997).

4.3.5 Age 7 (MCS4) language ability

The BAS Word Reading subscale was used to measure children's language ability when they were 7 years old. The BAS word reading subscale assesses children's reading vocabulary knowledge between the ages of 5 and 17 years and 11 months, so was suitable for the current study when children were 7 years old. The child is presented with a series of words on a card. There is a total of 90 words divided into 9 blocks of 10 words. Each block of words increased in difficulty as the child progressed through blocks 1 to 10. Children were asked to read each word aloud with the correct pronunciation. The number of blocks the child was asked to attempt was dependent on their performance in the previous block (e.g., if the child struggled with their current block, they would not be asked to attempt the next more difficult block. Example words from the easiest block of words (block 1) are "box", "fish", and "jump" and from the most difficult block (block 10) "catastrophe", "regurgitate", and "archaic". Children's raw scores on the word reading task were calculated by summing the total number of correct responses ranging from 0-90 with a mean score of 44.4 (SD19.278). This scale has

been shown to have good reliability in previous studies (Stuart, 1999; Hatcher, Hulme & Ellis, 1994).

4.3.6 Age 14 (MCS6) language ability

The Applied Psychology Unit Standardised Vocabulary Test was used to assess children's verbal vocabulary knowledge at age 14. The vocabulary test is designed to measure respondents understanding and meaning of words and is suitable for those aged 14 and over. The word activity involved presenting respondents with a list of 20 target words. For each target word, 5 additional words were presented. Children were asked to identify which of the 5 additional words had the same meaning as the target word. The words used were subsets of those used in the vocabulary test from the 1970 British Cohort Study which were originally from the standardised vocabulary test designed by the Applied Psychology Unit at the University of Edinburgh in 1976. Children were given 1 point for each correct answer and total scores were a sum of all correct answers meaning 20 was the maximum score. In the current sample, the scores ranged from 0-19 with a mean of 7.09 (SD= 2.625). Previous analyses have found the measure to be reliable (alpha = .83 - .95; Levy & Goldstein, 1984).

	Age 3 BAS Naming	Age 7 word	Age 14 vocabulary
	Vocabulary	reading task	test
Age 3 BAS Naming Vocabulary	-		
Age 7 word reading task	.278***	-	
Age 14 vocabulary test	.247***	.410***	-
Ν	10683	10683	10683
Μ	50.418	50.523	50.022
SD	9.81	9.84	10

 Table 4.1. Descriptive statistics and Bivariate Correlations for language ability at all

 data collection points

Note. Standardised T-scores used to generate these statistics. *** = p < .001.

4.3.7 Internalizing and Externalizing behaviour

The Strength and Difficulties Questionnaire (SDQ; Goodman, 1997) was used to measure children's internalising and externalizing behaviours at ages 3, 7, and 14 years old and was completed by the parents. The SDQ contains 25 items measuring five dimensions of behavioural development in children aged between 3 and 16. The five dimensions are emotional symptoms, conduct problems, hyperactivity, peer problems, and pro-social behaviour. Each item in the subscales is answered on a 3-point scale ("not true" = 0, "somewhat true" = 1, "certainly true" = 2). The total score on each subscale can range from 0-10. Goodman provides information and syntax for scoring the SDQ and deriving internalizing behaviour scales on his website (Goodman: SDQ Info, 2023).

The internalizing behaviour scale is derived by summing the items from the emotional subscale ("often complains of headaches, stomach aches or sickness", "many worries, often

seems worried", "often unhappy, down-hearted or tearful", "nervous or clingy in new situations, easily loses confidence", and "many fears, easily scared") and the peer problems subscale ("rather solitary, tends to play alone", "has at least one good friend", "generally liked by other children", "picked on or bullied by other children", and "gets on better with adults than with other children"). Normal internalizing scores range from 0-4, borderline problems scores range from 5-7, and abnormal scores range from 8-20 (Maurice-Stam et al., 2018). The externalizing behaviour scale is derived by summing the items from the hyperactivity subscale ("restless, overactive, cannot stay still for long", "constantly fidgeting or squirming", "thinks things out before acting", "sees tasks through to the end, good attention span", and "easily distracted, concentration wanders") and the conduct problems subscale ("often has temper tantrums or hot tempers", "generally obedient, does what adults request", "often fights with other children or bullies them", "can be spiteful to others", and "often argumentative with adults"). Normal externalizing scores range from 0-5, borderline scores range from 6-8, and abnormal scores range from 9-20 (Maurice-Stam et al 2018). The two-factor model (internalizing and externalizing behaviour) has been shown to have a good model fit (CFI = .94; TLI = .94) (Sosu &Schmidt, 2017). There is good reliability for both the internalizing behaviour scale with α ranging from .73 - .82 and the externalizing behaviour scale with a ranging from .78 - .83 (Palmieri & Smith 2007; Goodman et al., 2010; Hessel, et al., 2017). The convergent and discriminant validity of the internalizing and externalizing model has been found to be good (e.g., Goodman et al., 2010; Croft et al., 2015; Kersten et al., 2016) and the measurement invariance has shown good model fit (Sosu & Schmidt, 2017) indicating that the SDQ measures the intended constructs over time.

4.3.8 Measures of socio-economic status

Information on *family household income* was measured using net equivalised weekly income gathered at every data collection wave during the face-to-face interviews by asking respondents to state their take-home income. During early childhood, the average family income ranged from £14.39 - £1326.46 with an average of £343.89 (SD.= £223.37). During middle childhood, the average family income ranged from £14.93 - £1282.54 with an average of £398.86 (SD.= £232.38). During late childhood, the average family income ranged from £80.02 - £1153.52 with an average of £417.44 (SD.= £178.07). Parents' level of education was measured at each time point by asking parents to indicate their highest level of academic or vocational education which was mapped onto the National Vocational Qualification (NVQ). The NVQ ranges from NVQ 1 (lowest level of qualification) – NVQ 5 (highest level of qualification). Parents' occupational class was measured at every data collection wave by asking parents to indicate their occupation which was mapped onto the 7-class National Statistics Socio-economic Classification (NS-SEC). The NS-SEC ranges from 1 (routine occupations) to 7 (higher managerial and professional occupations). Table 4.2 below shows the percentage of parents in each education and occupation category during early, middle, and late childhood.

Child	Parent's level of	0/	Parent's level of	%
age	education	% 0	occupation	
3 Years				
	NVQ1	5.9	Routine	3.6
	NVQ2	25.2	Semi routine	18.7
	NVQ3	17.5	Lower supervisory	13.7
	NVQ4	42	Small employers	11.6
	NVQ5	9.4	Intermediate occupations	9.2
			Lower managerial and	22.0
			professional	22.9
			higher managerial and	20.4
			professional	20.4
7 Years				
	NVQ1	4.8	Routine	4.1
	NVQ2	23	Semi routine	20.3
	NVQ3	17.3	Lower supervisory	14
	NVQ4	41.7	Small employers	12.2
	NVQ5	13.2	Intermediate occupations	8.8
			Lower managerial and	~~~
			professional	22.7
			higher managerial and	10
			professional	18
14 Years				
	NVQ1	4.8	Routine	4.6

NVQ2	19.6	Semi routine	19.9
NVQ3	15.7	Lower supervisory	15.2
NVQ4	40.3	Small employers	15
NVQ5	19.7	Intermediate occupations	7.6
		Lower managerial and	21.6
		professional	21.0
		higher managerial and	16 2
		professional	10.2

4.3.9 Covariates

Several covariates were identified in the literature as being associated with family SES and children's outcomes so were included in the analyses. *Mothers' age at birth* (Knipe 2016) was measured by asking mothers to state their age in years when they gave birth. The *number of parents living in the household* (Flouri et al., 2016) was specified by the main respondent during the interview when asked whether they lived in a one or two-parent household. *Ethnicity* (Sullivan et al., 2021) is measured by asking the main respondent to indicate the child's ethnicity during the interviews. *Gender* (Mous et al., 2017) was measured by asking the main respondent to specify whether the child was male or female. Children's *birth weight* (Bromley, 2009) was measured by asking the main respondent for information during the interview. Table 4.3 below presents the descriptive statistics of each covariate.

4.3.10 Analytic strategy

Data were analysed using multiple regression to examine the association between family SES dimensions and children's language ability, internalizing behaviour and externalizing behaviour during early, middle, and late childhood when children were 3, 7, and 14 years old. Model 1 examined the associations between *early* family income, level of education, and level of occupation on children's outcomes controlling for covariates. Model 2 examined the associations between *middle* family income, level of education, and level of occupation on children's age 7 outcomes controlling for previous income, education, occupation, and covariates. Model 3 examined the associations between *late* family income, level of education, and level of occupation on children's age 14 outcomes. Controlling for previous income, education, occupation, and covariates. This process was repeated for language ability, internalizing behaviour, and externalizing behaviour. To examine whether the associations between different SES dimensions and children's outcomes were significantly different I calculated t-values for each pair of variables and then the level of significance was drawn from the t statistic using the t distribution (Clogg et al., 1995). The following formula was used to calculate the t statistic such that the difference in coefficients (b) was divided by the square root of the pooled standard errors (*s*) squared:

$$t = (b_1 - b_2) / \sqrt{(s_{b1}^2 + s_{b2}^2)}$$

Once the *t* statistic was calculated the value was mapped onto the *t* distribution chart. For quick reference a *t* statistic of 1.96 = p < .05, *t* statistic of 2.58 = p < .01, and *t* statistic of 3.09 = p < .001 (Wheeler, 2015). MCS non-response weights were applied to account for sample design and panel attrition (Plewis, 2007). Missing value analysis on early, middle, and late

childhood (ages 3, 7, and 14) found item level missingness of 4.008%. Item missingness was handled by full information maximum likelihood estimation (FIML) under the missing at random assumption which is a common approach in the literature (Muthén & Muthén, 2017; Flouri et al., 2018; Lee et al., 2019).

4.4 Results

4.4.1 Descriptive statistics

Table 4.3 below presents the descriptive statistics of the analytic sample. On average, children's language ability did not fluctuate much over the different stages of childhood, but children's internalizing behaviour increased over time while externalizing behaviour decreased. Suggested cut-off points for internalizing behaviour are 0-4 for normal scores, 5-7 for borderline problems, and 8-20 for abnormal scores (Maurice-Stam et al., 2018). As seen in table 4.3 below, the majority of children fall within the normal range across childhood. Suggested cut-off points for externalizing behaviour are 0-5 for normal scores, 8-6 for borderline problems, and 9-20 for abnormal scores (Maurice-Stam et al 2018). Table 4.3 shows that children on average reduced from borderline externalizing behaviour problems to a normal range as they got older. Looking at SES dimensions, income on average continues to increase over time and the average level of education increased from A levels/ Highers to teaching qualification/degree. Little change was seen with parents' level of occupation which remained in the same category on average small employers/own account workers.

	М	SD	%	Range
Language ability				
Early childhood	50.418	9.81	-	20 - 80
Middle childhood	50.523	9.84	-	20 - 80
Late childhood	50.022	10	-	20 - 80
Internalizing behaviour				
Early childhood	2.824	2.45	-	0 - 16
Middle childhood	2.64	2.69	-	0 - 18
Late childhood	3.711	3.35	-	0 - 19
Externalizing behaviour				
Early childhood	6.549	3.76	-	0 - 20
Middle childhood	4.551	3.48	-	0 - 19
Late childhood	4.323	3.51	-	0 - 19
Family income				
Early income (£)	343.89	223.37	-	14.39 - 1362.46
Middle income (£)	398.86	232.38	-	14.93- 1282.54
Late income (£)	417.44	178.07	-	80.02-1153.52
Parent's level of education				
Early childhood	2.929	1.406	-	0 - 5
Middle childhood	3.041	1.387	-	1 - 5
Late childhood	3.263	1.4	-	2 - 5
Parent's level of occupation				
Early childhood	4.614	1.945	-	1-7
Middle childhood	4.517	1.94	-	1-8

Table 4.3. Descriptive statistics of all variables in the analyses (n = 10683)

Late childhood	4.378	1.913	-	1-9
Covariates				
Mother's age at birth	29.11	5.751	-	14-51
Birth weight (lbs)	7.41	.586	-	.86 - 14.44
Parents in household				
Two parents			87.5	
One parent			12.5	
Gender				
Male			51.9	
Female			48.1	
Ethnicity				
White			86.41	
Mixed			0.09	
Indian			2.5	
Pakistani & Bangladeshi			6.5	
Black British			2.8	
Chinese/ other			1.7	

Note. Early = when children were 3 years old. Middle = when children were 7 years old.

Late = when children were 14 years old.

4.4.2 Associations between family SES and children's language ability during early, middle, and late childhood

Table 4.4 presents the results of the associations that income, education, and occupation have with children's language ability at different stages of childhood. For early childhood language ability, income, education, and occupation each had significant independent associations with children's language ability. Education had the strongest association with children's language ability ($\beta = .188$, p = <.001) and is significantly stronger than income (t = 5.755, p < .01) and occupation (t = 5.087, p < .01). For *middle* childhood language ability, all early and middle childhood SES dimensions were associated with children's language ability with parents' middle education having the strongest association ($\beta = .081$, p = <.05). However, this association was not significantly stronger than income or occupation associations. For late childhood language ability, late family income, parents' early education, and parents' early occupation were all associated with children's language ability. Parents' level of education during early childhood had the strongest association with children's late childhood language ability ($\beta = .128$, p = <.001) and was significantly stronger than early occupation (t = 2.068, p = <.05) but was not significantly stronger than late income (t = 1.51, p = n.s). Overall, for children's language ability parents' education was the most important SES dimension at all stages of childhood. Parents' level of education during the early stages of childhood in particular appears to be more important than the other SES dimensions even during later childhood stages.

	Early childhood languag	a l	Middle childhood langu		l ata childhood langus	
		D		age		age
N = 10683	β (95%CI)	SE	eta (95%CI)	SE	eta (95%CI)	SE
Early family income	0.067 (0.04, 0.093)***	.013	0.058 (0.024, 0.093)***	.018	0.007 (-0.031, 0.044)	.019
Mid family income			0.065 (0.026, 0.104)***	.020	0.045 (0.007, 0.082)*	.019
Late family income					0.074 (0.035, 0.112)***	.020
Early education	0.188 (0.155, 0.221)***	.017	0.071 (0.006, 0.136)*	.033	0.128 (0.064, 0.191)***	.032
Mid education			0.081 (0.017, 0.145)*	.033	0.009 (-0.064, 0.082)	.037
Late education					0.004 (-0.055, 0.047)	.026
Early occupation	0.077 (-0.104, -0.05)***	.014	0.068 (-0.11, -0.026)**	.022	0.054 (-0.099, -0.01)*	.023
Mid occupation			0.071 (-0.114, -0.028)***	.022	0.029 (-0.075, 0.018)	.024
Late occupation					0.03 (-0.065, 0.005)	.018
Birth Weight	0.048 (0.022, 0.074)***	.013	0.045 (0.018, 0.071)***	.014	0.006 (-0.018, 0.031)	.013
Gender	0.117 (0.095, 0.139)***	.011	0.07 (0.046, 0.094)***	.012	0.003 (-0.02, 0.027)	.012
Parents in household	0.019 (-0.008, 0.046)	.014	-0.033 (-0.067, 0.002)	.017	-0.014 (-0.044, 0.016)	.015
Mother age at birth	-0.016 (-0.043, 0.01)	.013	0.003 (-0.027, 0.032)	.015	0.041 (0.012, 0.07)**	.015
Mixed Heritage	-0.031 (-0.055, -0.006)*	.012	-0.001 (-0.027, 0.025)	.013	0.009 (-0.015, 0.033)	.012
Indian	-0.074 (-0.095, -0.052)***	.011	0.059 (0.038, 0.08)***	.011	0.001 (-0.024, 0.025)	.013
Pakistani & Bangladeshi	-0.218 (-0.243, -0.194)***	.013	0.066 (0.043, 0.089)***	.012	-0.005 (-0.025, 0.015)	.010
Black British	-0.13 (-0.165, -0.096)***	.018	0.035 (-0.002, 0.073)	.019	-0.004 (-0.029, 0.02)	.013
Other ethnic group	-0.151 (-0.186, -0.115)***	.018	0.023 (-0.003, 0.05)	.014	0.001 (-0.02, 0.022)	.011

Table 4.4 Regression models examining the associations between SES and children's language ability in early, middle, and late childhood

4.4.3 Associations between family SES and children's internalizing behaviour during early, middle, and late childhood

Table 4.5 presents the results of the associations that income, education, and occupation have with children's internalizing behaviour at different stages of childhood. For early childhood internalizing behaviour, income, education, and occupation had significant independent associations with children's internalizing behaviour. Parent's education had the strongest association with early childhood internalizing behaviour ($\beta = -.11$, p = <.001) and was significantly different from the family income (t = 2.428, p < .05), but not parent's occupation (t = .544 p = n.s). For *middle* childhood internalizing behaviour, only family income and parents' education were associated with children's internalizing behaviour. Parental education had the strongest association ($\beta = -.79$, p = <.05) but was not significantly different from family income (t = 1.102, p = n.s). For *late* childhood internalizing behaviour, family income from early, middle, and late childhood were associated with children's internalizing behaviour. Late family income had the strongest association ($\beta = -118$, p = <.001) and was significantly different from early family income (t = 2.485, p = <.05) and middle family income (t = 3.059, p = <.01). Overall, family income and education are the most important SES dimensions for children's internalizing behaviour depending on the stage of childhood. During early and middle childhood parents' education is more important however family income becomes increasingly important over time and is the only SES dimension significantly associated with children's internalizing behaviour during late childhood.

Table 4.5. Regression mc	dels examining the associat	ions be	tween SES and children's i	nternali	zing behaviour in early, midd	le,
	Early internalizing behavio	ur	Middle internalizing behavi	iour	Late internalizing behavio	our
N = 10683	β (95%CI)	SE	β (95%Cl)	SE	β (95%CI)	SE
Early family income	-0.054 (-0.082, -0.027)***	.014	-0.041 (-0.073, -0.008)*	.017	-0.053 (-0.088, -0.018)**	.018
Mid family income			-0.059 (-0.094, -0.024)***	.018	-0.038 (-0.073, -0.003)*	.018
Late family income					-0.118 (-0.156, -0.08)***	.019
Early education	-0.11 (-0.147, -0.073)***	.019	-0.016 (-0.085, 0.052)	.035	-0.003 (-0.071, 0.065)	.035
Mid education			-0.079 (-0.148, -0.01)*	.035	-0.001 (-0.077, 0.074)	.038
Late education					-0.035 (-0.092, 0.021)	.029
Early occupation	-0.097 (0.068, 0.126)***	.015	-0.034 (-0.011, 0.079)	.023	-0.01 (-0.035, 0.056)	.023
Mid occupation			-0.035 (-0.009, 0.08)	.023	-0.037 (-0.009, 0.084)	.024
Late occupation					-0.001 (-0.036, 0.038)	.019
Birth Weight	-0.026 (-0.053, 0)	.014	-0.046 (-0.073, -0.019)***	.014	-0.051 (-0.079, -0.023)***	.014
Gender	-0.049 (-0.074, -0.025)***	.012	-0.035 (-0.06, -0.01)**	.013	0.049 (0.024, 0.074)***	.013
Parents in household	0.023 (-0.011, 0.056)	.017	0.034 (-0.003, 0.071)	.019	0.022 (-0.012, 0.055)	.017
Mother age at birth	-0.059 (-0.089, -0.029)***	.015	-0.045 (-0.077, -0.013)**	.016	-0.008 (-0.039, 0.024)	.016
Mixed Heritage	-0.006 (-0.038, 0.026)	.016	0.023 (-0.004, 0.05)	.014	0.002 (-0.026, 0.03)	.014
Indian	0.036 (0.013, 0.06)**	.012	0.018 (-0.006, 0.041)	.012	-0.031 (-0.048, -0.014)***	600.
Pakistani & Bangladeshi	0.128 (0.097, 0.159)***	.016	0.07 (0.044, 0.096)***	.013	-0.005 (-0.026, 0.016)	.011
Black British	0.028 (-0.003, 0.06)	.016	0.015 (-0.017, 0.046)	.016	-0.031 (-0.062, -0.001)*	.016
Other ethnic group	0.039 (0.004, 0.074)*	.018	0.016 (-0.005, 0.037)	.011	-0.011 (-0.037, 0.016)	.013
Note. Significant estimates are in = when children were 14 years ou	bold. $p < .05 = *, p < .01 = **, p < $ ld. Gender reference category = male	.001 = *: Parents	**. Early = when children were 3 ye in household reference category = t	ears old. M two parent	tiddle = when children were 7 years o. s. Ethnicity reference category = white	d. Late

Table 4.5 Regression models examining the associations between SES and children's internalizing behaviour in early, middle, and late childhood

4.4.4 Associations between family SES and children's externalizing behaviour during early, middle, and late childhood

Table 4.6 below presents the results of the associations that income, education, and occupation have with children's externalizing behaviour at different stages of childhood. For early childhood externalizing behaviour, income, education, and occupation had significant independent associations with children's externalizing behaviour. Parents' education had the strongest association with children's early internalizing behaviour ($\beta = -.112$, p = <.001) and was significantly stronger than family income (t = 2.254, p < .05), but not parents' occupation (t = .335 p = n.s). For *middle* childhood externalizing behaviour, parents' middle education, early family income, and early parent occupation were associated with children's externalizing behaviour. Middle education had the strongest association ($\beta = -.116$, p = <.001) but was not significantly stronger than early family income (t = 1.848, p = n.s) or early occupation (t = 1.549, p = n.s). For *late* childhood externalizing behaviour, parents' middle education, late family income, and late parent occupation were associated with children's externalizing behaviour. Late family income had the strongest association ($\beta = -141$, p = <.001) and was significantly stronger than late occupation (t = 3.744, p = <.01), but not significantly stronger than middle education (t = 1.03, p = n.s). Overall, family income and education are the most important SES dimensions for children's externalizing behaviour depending on the stage of childhood. During early and middle childhood parents' education is most important while family income becomes the most important SES dimension during late childhood.

Table 4.6. Regression mo	dels examining the associat	tions be	etween SES and children's exte	ernalizin	ig behaviour at early, middle,	, and
late childhood.						
	Early externalizing behavio	our	Middle externalizing behavio	our	Late externalizing behavi	our
N = 10683	eta (95%CI)	SE	eta (95%CI)	SE	eta (95%CI)	SE
Early family income	-0.06 (-0.088, -0.032)***	.014	-0.056 (-0.09, -0.023)***	.017	-0.013 (-0.048, 0.022)	.018
Mid family income			-0.029 (-0.064, 0.005)	.017	-0.014 (-0.049, 0.021)	.018
Late family income					-0.141 (-0.175, -0.107)***	.017
Early education	-0.112 (-0.149, -0.075)***	.019	-0.01 (-0.052, 0.073)	.032	-0.006 (-0.077, 0.065)	.036
Mid education			-0.116 (-0.177, -0.055)***	.031	-0.103 (-0.181, -0.025)**	.040
Late education					-0.029 (-0.024, 0.082)	.027
Early occupation	-0.104 (0.075, 0.133)***	.015	-0.055 (0.007, 0.104)*	.025	-0.019 (-0.024, 0.062)	.022
Mid occupation			-0.017 (-0.031, 0.064)	.024	0 (-0.043, 0.044)	.022
Late occupation					-0.051 (0.018, 0.085)**	.017
Birth Weight	-0.036 (-0.062, -0.01)**	.013	-0.066 (-0.092, -0.041)***	.013	-0.056 (-0.084, -0.027)***	.015
Gender	-0.116 (-0.14, -0.091)***	.012	-0.164 (-0.188, -0.141)***	.012	-0.141 (-0.165, -0.118)***	.012
Parents in household	0.041 (0.007, 0.075)*	.017	0.033 (-0.002, 0.067)	.018	0.035 (0, 0.069)*	.018
Mother age at birth	-0.128 (-0.159, -0.097)***	.016	-0.103 (-0.133, -0.073)***	.015	-0.091 (-0.121, -0.06)***	.016
Mixed Heritage	-0.014 (-0.047, 0.019)	.017	-0.012 (-0.045, 0.021)	.017	-0.024 (-0.053, 0.004)	.015
Indian	-0.002 (-0.023, 0.019)	.011	0.004 (-0.018, 0.027)	.011	-0.034 (-0.049, -0.018)***	.008
Pakistani & Bangladeshi	0.007 (-0.019, 0.032)	.013	-0.013 (-0.033, 0.007)	.010	-0.05 (-0.07, -0.03)***	.010
Black British	-0.022 (-0.055, 0.011)	.017	-0.019 (-0.05, 0.011)	.016	-0.028 (-0.056, -0.001)*	.014
Other ethnic group	-0.005 (-0.031, 0.021)	.013	-0.027 (-0.052, -0.002)*	.013	-0.024 (-0.049, 0.001)	.013
Note. Significant estimates are in children were 14 years old. Gende	bold. p < .05 = *, p < .01 = **, p < .01 er reference category = male. Parents	01 = ***. s in house	Early = when children were 3 years old. hold reference category = two parents. E	Middle = v Ethnicity re	when children were 7 years old. Late = eference category = white	when

Table 4.6 Regression models examining the associations between SES and children's externalizing behaviour in early, middle, and late childhood

4.5 Discussion

Using data from the millennium cohort study, this study aimed to examine whether the associations between family SES and children's language ability, internalizing behaviour, and externalizing behaviour differ depending on the stage of childhood (early, middle, and late). This expanded previous literature that was largely limited to examining early and middle childhood. In addition, this study examined whether the different dimensions of SES (family income, parent's education, and parent's occupation) had unique associations with children's outcomes at each stage of childhood. This builds on the previous literature which normally focuses on composite measures of SES or a single dimension of SES. The current findings indicate that SES was important at all stages of development and differed depending on the outcome being examined. Early SES was more important for language ability, while a family's SES at the stage of development (rather than previous SES) was more important for internalizing and externalizing behaviour. The findings also indicated that the different SES dimensions had independent unique associations with children's outcomes and that the associations differed depending on the outcome being examined.

In terms of children's language ability, it was found that SES in early childhood had the largest association with children's language ability at all time points. This is in line with previous research that the earlier children experience low SES the worse their cognitive outcomes would be later in life (Schoon et al., 2011; 2013; Dickerson & Popli, 2012; 2016; Mollborn et al., 2014). However, the previous studies only examined data until middle childhood, so this study expanded on the previous findings by showing that earlier SES was still important for language ability during late childhood. A possible reason why early SES has the strongest association with children's language ability is that children are generally limited to interactions with their parents during the earliest years of life and since children's

language development occurs through exposure to language and interacting with people with better language knowledge it means their language development at this stage is reliant on their parents (Vygotsky 1978). This could be problematic as low-SES children have more negative low quality interactions with their parents compared to their peers (Conger et al., 2010). However, as children get older and interact with other people (e.g., teachers) where they are exposed to additional language knowledge they become less reliant on their parents which could be why early SES had stronger associations with children's language ability in this study. Research tends to support this as a recent meta-analysis concluded that teacherchild interactions can improve children's language development, especially for low SES children (Kharitonova et al., 2021).

For internalizing and externalizing behaviour family's most recent SES was consistently more important than their previous SES. For example, during middle childhood, SES in middle childhood had a stronger association with children's internalizing and externalizing behaviour than SES in early childhood. Similarly, during late childhood, late SES had a stronger association with children's internalizing and externalizing behaviour than SES. The results were in line with previous literature which found that SES during late childhood had stronger associations with children's internalizing and externalizing and externalizing behaviour than their previous SES (Duncan et al., 2010; Lai et al., 2019). These findings indicate that later childhood is a sensitive period for internalizing and externalizing behaviour. This possibility has been discussed in previous literature which proposed that late childhood is a particularly stressful time for children as they experience social change (Sirin et al., 2013; Fuhrmann et al., 2015). For example, low SES children are more likely to experience bullying (Blakemore2014; Fuhrmann et al., 2015;

Efstathopoulos et al., 2018) social exclusion (Hills et al., 2002) and discrimination (Walker, 2014) during late childhood and adolescence which can impact their behavioural development.

Evidence also indicates that children become more aware of their SES during late childhood (Blumenshine et al., 2010; Bannink et al., 2016; McLoyd, 2019) possibly because they cannot participate in social events, and do not have the same material possessions as their peers due to a lack of economic resources (Hjalmarsson & Mood, 2015). Evidence indicates that children's behavioural outcomes are more likely to be impacted if they become aware and perceive their SES to be low (Quon & McGrath, 2014 Rivenbark et al., 2019). Low SES children are more likely to be bullied, be bullies themselves, have poor peer relationships, be discriminated against by their peers, and excluded by their peers if they are perceived to be low SES (Jansen et al., 2011; 2012; Moore et al., 2016; Hjalmarsson; 2018; Elenbaas, 2019; Li et al 2020; Bai et al., 2021). The bioecological model highlights that peer relationships are a microsystem within which children's interactions shape development (Bronfenbrenner, 1979; 1986). If peer interactions are poor then this likely leads to poor behavioural outcomes. Evidence supports this as being bullied or excluded is associated with internalizing and externalizing behavioural problems (Taylor et al., 2018; Bai et al., 2021). It is therefore possible that children's internalizing and externalizing behaviour is particularly sensitive to SES during late childhood.

The findings indicate that family income, parents' education, and parents' occupation each have independent associations with children's language ability and externalizing behaviour at all stages of childhood. Results for internalizing behaviour were different as all SES dimensions were associated with internalizing behaviour during early childhood, but only

income and education (not occupation) were associated during middle childhood, and only family income was associated during late childhood.

Parent's level of education was found to have the largest association with children's language ability during early, middle, and late childhood. This was to be expected for children's language ability as the literature consistently finds parents' education to have the largest SES association with language outcomes when compared to family income and parent's occupation (Davis-Kean, 2005; Sullivan & Brown, 2013; Coddington et al., 2014; Gonzalez et al., 2020). A possible explanation is that parents' education operates through unique pathways in relation to children's language abilities. Harding et al. (2015) theoretical model proposes that parents' level of education is associated with children's cognitive outcomes due to the quality and quantity of language used in the household and through the quality and quantity of stimulating parenting practices (e.g., reading with the child). Better educated parents use better quality speech with their children (Huttenlocher et al., 2010; Zauche et al., 2016) and speak with their children more frequently (Rowe, 2008; Weisleder & Fernald, 2013), both of which are associated with children's language ability (Rowe, 2012; Weisleder & Fernald, 2013; Cartmill et al., 2013).

Family income was found to become increasingly important as children got older and by late childhood had the largest association with children's internalizing and externalizing behaviour. This could be due to children becoming increasingly aware of their economic situations as they get older and miss out on social events or being unable to have the same material goods as their peers due to a lack of economic resources (Hjalmarsson & Mood, 2015). This causes children to feel stress and shame (Conger et al., 2010; Walker, 2014) which could cause additional conflict and impaired interactions between parents and children leading to increased behavioural problems as outlined in the family stress model. Aside from children becoming more aware of their family's financial circumstances, the parents themselves could be feeling increased stress during later childhood as children generally become more expensive to raise as they get older (Twenge et al., 2003). This again could increase conflict in the home and impact children's behaviour outcomes (Conger et al., 2010). Similarly, as a family's costs of raising a child increase as they get older, the income available to invest in their children could decrease which the family investment model states could have a negative effect on children's outcomes. These findings highlight why it is important to examine all stages of childhood and multiple SES dimensions where possible because if this study only examined early and middle childhood as is common in the literature (Davis-Kean, 2005; Sullivan & Brown, 2013; Coddington et al., 2014; Bøe et al., 2014; Hosokawa & Katsura, 2018; Gonzalez et al., 2020) then it would have concluded that education was the most important SES dimension for internalizing and externalizing behaviour. However, the current findings show that income becomes more important during later childhood rather than earlier childhood.

These findings differ from some studies which have found that income poverty during early childhood has the largest association with children's later behavioural outcomes (Joinson et al., 2016). However, one difference between Joinson et al's study and the current study is that it did not consider the other dimensions of SES in their models meaning that the income associations could be overestimated which is likely why the current findings are different. The current findings also differ from other studies comparing the associations of family income and parents' education on children's internalizing and externalizing behaviour. For example, Miller et al. (2021) found mothers' education to be more important than family income. It is likely that the current results differ from Miller et al. (2021) due to different

measurements of SES dimensions between studies. Miller et al. (2021) focus on mothers' education in years of education completed and mothers' occupation measured in the length of time being unemployed. The current study used the highest level of household education and occupation. These measurement differences are potentially why differences are found between all studies.

4.5.1 Strengths and weaknesses of the study

The results must be interpreted in light of their limitations. This study examines the associations between SES dimensions and children's language and behavioural outcomes which means that causation cannot be determined. While there may be different underlying mechanisms between the different SES dimensions and children's outcomes, this study did not examine this. However, this study provides further evidence for future research to examine these possible mechanisms. The use of different language ability measures at each time point must be acknowledged as a potential limitation. The British Ability Scale subscales were used to measure language ability during early and middle childhood, but not during late childhood. Instead, the Applied Psychology Unit Standardised Vocabulary Test was used during late childhood. Given that the British Ability Scales were appropriate for use during late childhood, it may have been beneficial to use the same tests over time as it would provide more confidence that the desired constructs are being measured over time. Despite this, previous studies show that correlations between the British ability scales and the Applied Psychology Unit Standardised Vocabulary Test were adequate for use together (Hill, 2005; Girard et al., 2017; Sullivan et al., 2017). This study also found that the language ability measures were correlated in line with previous research.

4.5.2 Implications of the study

The results of this study have useful implications for policy. First, the findings that income, education, and occupation each have unique associations with children's developmental outcomes provide useful information for policymakers who are interested in improving children's outcomes via family SES interventions as it indicates that each SES dimension has a unique association with outcomes. This is particularly useful in the current context as UK policy is focussing on family income, level of education, and occupation (Welfare Reform Act, 2016). Current policies largely focus on increasing family income, but these findings indicate that it could be beneficial to also support parents' education and occupation. Second, considering the stage of childhood could benefit policy which currently focuses resources on reducing the SES inequalities in children's outcomes during the earliest years of life. The current results indicate that family SES is important for children's language and socioemotional behavioural development at all stages of childhood showing that children are sensitive to the adverse effects of SES beyond early childhood. It, therefore, may be beneficial for children if intervention to support family SES was provided at all stages of childhood as opposed to a lot of support being focused on the earlier years only such as Healthy Start, free childcare schemes, and the Scottish Child Payment.

Third, the findings also show that parents' education is important for children's developmental outcomes at all time points and is most important for children's language ability and early to middle childhood externalizing behaviour. Intervention to improve parents' education is therefore likely to benefit children at all stages of development but findings suggest that the earlier education is improved the better it will be for children's outcomes. While policy exists to improve parents' education such as the Fair Start programs, the funding and attention paid to parents' education is relatively low compared to family

income. If the goal of policymakers is to improve children's outcomes, then it would be beneficial to support parents in improving their education at all stages of childhood. Combined, the findings indicate that the socio-economic gap in children's language ability and socioemotional behaviour would be benefitted by a prolonged multidimensional approach.

In summary, this study extended prior research on the timing of SES which has previously focused on early and middle childhood. It has done this by examining the associations between SES and children's language ability, internalizing behaviour, and externalizing behaviour during late childhood. Further, in line with theory, this study conceptualized SES as a multidimensional construct and is one of the few to examine the unique associations that family income, parents' education, and parents' occupation had with children's outcomes at each stage of childhood. These findings may provide useful insights to policymakers and researchers concerned with reducing SES-related developmental inequalities.

4.6 Summary

This chapter examined the importance of different stages of childhood in the association between family income, parents' education, parents' occupation and children's language ability, internalizing behaviour, and externalizing behaviour. The next chapter will examine the associations between SES trajectories and children's outcomes. Chapter 5 – Associations between family SES trajectories and children's language ability, internalizing behaviour, and externalizing behaviour

5.1 Aims of this chapter

This chapter builds on the findings from chapter 4 by examining the associations between family income, parents' education, and parents' occupation trajectories and children's language ability, internalizing behaviour, and externalizing behaviour. The chapter begins by presenting an introduction to the study, a brief overview of the literature based on chapter 2, and the research questions being addressed. The methodology used is then presented followed by the findings and a discussion of their implications.

5.2 Introduction

Associations between family socioeconomic status (SES) and children's language ability, internalizing behaviour, and externalizing behaviour are well documented (Sirin, 2005; Letourneau et al., 2011; Cooper & Stewart, 2013; 2017). However, there remain gaps in our understanding of how different dynamics of SES, such as family SES trajectories, are associated with children's developmental outcomes. Much of the previous literature examining the associations between SES and children's development is cross-sectional (Berger et al., 2009; Violato et al., 2011; Buckingham et al., 2013; Flouri & Midouhas, 2016; Fuller et al., 2022) and most longitudinal studies treat SES as a static construct measured at one time point (Kiernan & Mensah, 2009, 2011; Dickerson & Popli, 2014, 2016; Hernandez-Alava & Popli, 2017; Fitzsimons et al., 2017; Hosokawa & Katsura, 2018; Tamura et al., 2020). However, SES is not static, and each dimension of SES has the capacity to change

over time (Liu & Guo, 2015; van Roode et al., 2017). It is important to examine these SES trajectories as evidence shows that when children enter or exit poverty their developmental outcomes improve or worsen in line with their new SES (Kiernan & Mensah, 2011; Dickerson & Popli, 2012; 2016), therefore SES trajectories are likely differentially associated with children's development.

Aside from poverty status and composite measures of SES, income changes have also been found to influence changes in children's developmental outcomes (Milligan & Stabile, 2011; Dahl & Lochner, 2012). Natural experiments indicate that family income increases improve children's cognitive outcomes, particularly for low-income families (Loken, Mogstad, & Wiswall, 2010). The improved family income continues to have a positive effect on children's outcomes throughout middle and late childhood (Raffington et al, 2018). Similarly, when income is increased through policy measures (e.g., welfare and tax benefits) children's cognitive outcomes improve (Duncan et al, 2011; Milligan & Stabile, 2011; Dahl & Lochner, 2012). Evidence indicates that for every \$1000 increase in family income children's educational outcomes improve by 6-7.3%. On the other hand, when income decreases, children's behavioural problems can worsen (Choi, Shin, Cho & Park, 2017).

Despite this, limited research has examined the associations between longitudinal SES trajectories and children's language ability, internalizing behaviour, and externalizing behaviour. Some studies examine SES trajectories formed during early childhood (e.g., birth until five years; Dearing et al., 2001; Kendzor et al., 2010; Zachrisson & Dearing, 2015; Bøe et al., 2016; Jackson et al., 2017), middle to late childhood (Comeau & Boyle, 2018), or late childhood to adulthood (McLeod, 1996), but there is little evidence of the associations between family SES trajectories across the entire childhood timeline in relation to language
ability, internalizing behaviour, and externalizing behaviour. Since evidence indicates the importance of early, middle, and late childhood (Dohoon, 2014; McFarland, 2017; McFarland & Hayward, 2014; Dickerson & Popli, 2016), and that SES is not a static construct, research should examine the association between SES trajectories across the entire childhood timeline and children's outcomes.

Current SES trajectory literature focuses on family income, poverty, and composite SES measures but neglects to examine trajectories of other SES dimensions of parent's education and occupation (Kendzor et al., 2012; Jackson et al., 2017; Björkenstam et al., 2017). This is problematic as SES is a multifaceted construct comprised of different dimensions such as income, wealth, poverty, education, and occupation (APA, 2007; Ryan et al., 2014). Research shows that each SES dimension has unique associations with children's language ability, internalizing behaviour, and externalizing behaviour (Sullivan & Brown, 2013; Miller et al., 2021) and this was also established in chapter 4 (study 1). Like family income, evidence also shows that education and occupation can change over time (Magnuson et al., 2009; Harding, 2015; Jonsson et al., 2017), but it is unclear whether different education and occupation trajectories can be identified and whether they have differential associations with children's language ability and behavioural outcomes.

Current UK policy aims to reduce socioeconomic differences in children's language ability and socioemotional behaviour outcomes by improving family income, levels of education, and levels of occupation (Welfare Reform Act, 2016). Research examining the long-term association between different SES trajectories and children's outcomes can provide information about the potential long-term effects that policy decisions could have on children's outcomes. This study aims to address these research gaps by examining whether

different income, education, and occupation trajectories can be identified over early, middle, and late childhood and then testing whether there are differential associations between each SES trajectory and children's language ability, internalizing behaviour, and externalizing behaviour.

5.2.1 Theoretical framework for understanding the role of SES trajectories in children's development

Bioecological Systems Theory explains child development takes place due to complex bidirectional interactions between children and their proximal and distal environments (Bronfenbrenner, 1986). Interactions in proximal environments (e.g., the family and home) are more influential for children's development because they are exposed to them most frequently (Bronfenbrenner, 1979; 1999; 2006). Family SES is associated with children's development because it can determine the nature of the environments that children interact with via several pathways (Conger et al., 2010). The pathways through which this process occurs are outlined by the family stress and investment models (Conger et al., 2010; Conger & Donnellan, 2007). Family stress model proposes that SES is associated with children's development because economic pressure leads to parental stress which causes increased conflict between parents, poor parenting practices, and poorer parent-child interactions. Evidence indicates each of these impaired pathways lead to poorer developmental outcomes (Kiernan & Manesh, 2009; 2011; Avan et al., 2010; Micklewright et al., 2012; Azzi-Lessing, 2013; Shelleby et al., 2014; Sun et al., 2015; Neppl et al., 2016; Sosu & Schmidt 2017; Ahun et al., 2017). Family investment model states that SES is associated with children's development because it determines parents' ability to invest in the resources required for good child development (e.g., nutritious food, educational materials and activities, safe environment) (Conger & Donnellan, 2007). Low SES families are less able to invest as much in their children's development while high SES families can invest more. Research indicates the more families invest in their children the better their language ability and behavioural outcomes are (Kalil, et al., 2012; Sohr-Preston et al., 2013; Prevoo et al., 2014; Sosu & Schmidt, 2017; Longo et al., 2017; Jeong et al., 2017; Vasilyava et al., 2018; Fomby & Musick, 2018).

Life Course Theory explains that SES is not a static construct and can remain stable or continuously move up or down forming trajectories from birth to death (Elder, 2003). People have several different trajectories simultaneously that can be independent or associated with each other (e.g., income, educational, occupational trajectories). Life course research proposes the accumulation of risk and social mobility models to explain how different SES trajectories are associated with children's development. Social mobility models hypothesise that the stable, upward, and downward trajectories have differential associations with children's developmental outcomes such that an upward trajectory leads to better outcomes, and downward trajectories lead to worse outcomes (Duncan, 1988; Lynch et al., 1994). Family stress model explains this occurs is because as family SES changes, then family conflict and parents' interactions with their children are likely to change in the same direction which can lead to improvement or worsening of children's outcomes. The investment model states that changes in family SES will be reflected in their ability to invest in their children's development and therefore impact the association between family SES and children's outcomes. The accumulation of risk models hypothesise that the longer children remain in a low SES trajectory, the more detrimental it will be for their development because continued exposure to low SES at several time points produces a cumulative adverse effect (Kuh et al., 2003). According to the stress and investment models, this is because they will have more

cumulative impaired bidirectional child-parent interactions and go longer periods with less investment in their development.

Combining life course trajectory models with the family stress model it is likely that SES trajectories are differentially associated with children's development because each trajectory exerts its own level of economic pressure on the family. Therefore, parental distress, family conflict, and parents' interactions with their children will be higher or lower depending on the SES trajectory which ultimately influences children's development. Similarly, the investment model states that changes in family SES will be reflected in their ability to invest in their children's development which will impact their development.

5.2.2 Evidence of SES trajectories and children's development

The literature examining associations between family SES trajectories and children's language and behavioural outcomes tends to focus on income trajectories. Evidence supports the theory that families have different income trajectories, and that each trajectory has a differential association with children's development. The literature examining trajectories is relatively small but has identified between three- and five-income trajectories (e.g., Jackson et al., 2017; Björkenstam et al., 2017; Kendzor et al., 2012). Three trajectory studies find a variation of stable high, stable medium, and stable low income (Jackson et al., 2017). Four trajectory studies find a variation of high, low, increasing and decreasing income trajectories (Kendzor et al., 2012), and five trajectory studies find a variation of highst increasing, second highest increasing, lowest, second lowest, and decreasing income trajectories (Björkenstam et al., 2017). Research examining the associations between family SES trajectories and children's developmental outcomes is relatively small and is mostly associated with children's health outcomes such as children's weight (Demment et al., 2014),

obesity (Kendzor et al., 2012), asthma (Kozyrsky et al., 2010), and oral health (Peres et al., 2011). However, some researchers have examined the associations between family income trajectories and children's language ability (Dearing et al., 2001; Schoon et al., 2011; Jackson et al. (2017) and internalizing and externalizing behaviour (Poonawalla et al., 2014; Björkenstam et al., 2017; Mok et al., 2018).

Overall, findings indicate that children belonging to persistently low-income trajectories almost always fare worse than those from high-income, increasing-income, or decreasingincome trajectories (Chen et al., 2007; Peres et al., 2007; Séguin et al., 2012; Björkenstam et al., 2017; Boe et al., 2017; Mok et al., 2018) indicating that the longer a child experiences low income, the worse their outcomes will be. Stable low income trajectories are associated with the leads to the poorest language ability outcomes (Dearing et al., 2001; Schoon et al., 2011; Jackson et al., 2017), and internalizing and externalizing behavioural problems (Zachrisson & Dearing, 2015; Comeau et al., 2018). Belonging to an increasing income trajectory has been associated with improved receptive language and expressive language (Dearing et al., 2001), improved vocabulary (Jackson et al., 2017), and leads to improvements in children's internalizing and externalizing problems (Hoyt et al., 2019). Belonging to a *decreasing* income trajectory is associated with poorer language ability outcomes (Jackson et al., 2017), and a worsening of internalizing and externalizing problems (Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Hoyt et al., 2019) compared to their stable high- and increasing-income peers, but still not as problematic as stable low-income children (Hoyt et al., 2019).

The literature examining associations between family SES trajectories and children's developmental outcomes tends to focus on SES trajectories during limited timeframes such as early childhood (Dearing et al., 2001; Zachrisson & Dearing, 2015; Jackson et al., 2017) or

from mid-childhood to late-childhood (e.g., Miller & Votruba-Drzal 2017; Comeau & Boyle, 2018). One study examining income trajectories over a longer timeframe between 3-14 years old (Björkenstam et al., 2017) and focused on internalizing behaviour outcomes found that children belonging to low- and decreasing-income trajectories had the most behavioural problems. Björkenstam et al. (2017) also found an additional increasing income trajectory that was not identified in other studies that examined shorter time frames (e.g., Miller & Votruba-Drzal, 2017) which suggests that examining longer timeframes may identify different trajectories to shorter timeframes. It would therefore be beneficial to examine SES trajectories from birth through to late childhood as it is possible that more family SES variation will be captured if data are used over a longer duration which could identify additional trajectories over and above what Björkenstam et al. (2017) found as their SES trajectories began when children were 3 years old. Doing this would provide insight into the different SES trajectories that exist over the entire childhood timeline rather than during specific stages of childhood.

Importantly, the SES trajectory literature primarily focuses on income trajectories and studies have not considered that education and occupation trajectories can equally have unique associations with children's outcomes. Although education and occupation are more stable than family income (Duncan & Magnuson, 2012), life course theory argues that several different education and occupation trajectories are likely to be present over the life course (Elder, 2003). This is supported by evidence that parents' levels of education and occupation are known to change over time (Magnuson et al., 2009; Harding, 2015; Jonsson et al., 2017). There is also evidence that different levels of parent education and occupation are associated with parental stress, family conflict, and parent-child interactions (Brooks-Gunn, et al., 2010; Kalil, et al., 2012; Hadzic et al, 2013; Augustine et al., 2014; Harding et al., 2015; Dockery et

al., 2016; Cho, 2018; Moreira et al., 2019) which could impact children's language ability and behavioural outcomes as outlined in the family stress model (Conger et al., 2010). It is therefore likely that different education and occupation trajectories will have differential associations with children's outcomes. Since policy is concerned with improving family income, education and occupation (Welfare Reform Act, 2012), research needs to examine the trajectories of each SES dimension and examine their unique associations with children's developmental outcomes.

5.2.3 The current study

The current study aims to extend the existing SES trajectory literature in two ways. First, it will use secondary data from birth to late childhood to examine different SES trajectories and examine whether the different SES trajectories are associated with children's rather than language ability, internalizing behaviour and externalizing behaviour. This builds upon the existing literature which is limited to shorter durations when looking at language and behaviour outcomes and therefore includes less information in their trajectory models. While some studies have used data from birth to late childhood, they almost exclusively focus on child health outcomes (Kozyrsky et al., 2010; Peres et al., 2011; Kendzor et al., 2012; Demment et al., 2014) which do not provide information on language or behavioural outcomes. Second, the current study will examine trajectories of family income, parent's education, and parent's occupation since they are each important to current policy but have yet to be examined in relation to children's language ability and behavioural outcomes. Examining the associations between trajectories of different SES dimensions and children's developmental outcomes in later life will benefit policymakers as it will give them a better understanding of the long-term impact that policy decisions could have on children's outcomes. To achieve the aims of this study the following questions will be answered:

- 1. To what extent does family income, parents' education, and parents' occupation change or remain stable from childbirth through late childhood?
- 2. To what extent are family income, parent's education, and parent's occupation trajectories associated with children's language and behavioural outcomes during late childhood?

5.3 Methodology

5.3.1 Sample

The data used in this study comes from the first six waves of the Millennium Cohort Study (MCS). The MCS is a longitudinal national cohort study in the UK that follows the lives of 18851 families with a child that was born in the UK between 2001 and 2002 when children were 9 months old (Fitzsimons, 2017). The data were collected when children were aged, 9 months, 3, 5, 7, 11, and 14 years old and the productive samples at each wave were 18,551, 15,590, 15,246, 13,857, 13,287, and 11,726 families (Ketende 2010; Mostafa, 2014; Fitzsimons, 2020). The sample was obtained from 398 electoral wards across the UK and families from disadvantaged and ethnic minority electoral wards were oversampled to ensure that adequate sample sizes for each stratum could be achieved while accounting for attrition (Plewis, 2007; Ipsos MORI, 2017). As discussed in chapter 3, the productive sample from wave 6 (N = 11,726) was used in this study to ensure that all families included had longitudinal weights assigned to them which were used to account for attrition. 947 cases were removed for having no valid information for any of the variables of interest at any data collection wave and a further 96 cases were identified as having extreme outlier values. This process left a final analytic sample of 10,683 families. To account for panel attrition the non-

response weights provided in the MCS were applied (Plewis, 2007; Mostafa, 2014) and missing data were handled using Full Information Maximum Likelihood estimation (Muthén & Muthén, 2017).

5.3.2 Measures

The dependent variables in this study were children's language ability, internalizing behaviour, and externalizing behaviour. Data on the dependent variables were obtained when children were fourteen (MCS 6) years old. The independent variables were family SES (i.e., family income, maternal education, and maternal occupation). The covariates used were age, gender, ethnicity, birth weight, single or dual parent household. Data on the independent variables and covariates apart from birth weight were available at all six data collection waves when children were 9 months old (MCS1), three (MCS 2), five (MCS 3), seven (MCS 4), eleven (MCS 5), and fourteen (MCS 6) years old. Birth weight was recorded at wave 1.

5.3.3 Language Ability

The Applied Psychology Unit Standardised Vocabulary Test was used to assess children's verbal vocabulary knowledge at age 14. The vocabulary test is designed to measure respondents understanding and meaning of words and is suitable for those aged 14 and over. The word activity involved presenting respondents with a list of 20 target words. For each target word, 5 additional words were presented. Children were asked to identify which of the 5 additional words had the same meaning as the target word. The words used were subsets of those used in the vocabulary test from the 1970 British Cohort Study which were originally from the standardised vocabulary test designed by the Applied Psychology Unit at the University of Edinburgh in 1976. Children were given 1 point for each correct answer and total scores were a sum of all correct answers meaning 20 was the maximum score. In the

current sample, the scores ranged from 0-19 with a mean of 7.09 (SD= 2.625). Previous analyses have found the measure to be reliable (alpha = .83 - .95; Levy & Goldstein, 1984).

5.3.4 Internalizing and Externalizing behaviour

The Strength and Difficulties Questionnaire (SDQ; Goodman, 1997) was used to measure children's internalising and externalizing behaviours at ages 3, 7, and 14 years old and was completed by the parents. The SDQ contains 25 items measuring five dimensions of behavioural development in children aged between 3 and 16. The five dimensions are emotional symptoms, conduct problems, hyperactivity, peer problems, and pro-social behaviour. Each item in the subscales is answered on a 3-point scale ("not true" = 0, "somewhat true" = 1, "certainly true" = 2). The total score on each subscale can range from 0-10. Goodman provides information and syntax for scoring the SDQ and deriving internalizing behaviour scales on his website (Goodman: SDQ Info, 2023).

The internalizing behaviour scale is derived by summing the items from the emotional subscale ("often complains of headaches, stomach aches or sickness", "many worries, often seems worried", "often unhappy, down-hearted or tearful", "nervous or clingy in new situations, easily loses confidence", and "many fears, easily scared") and the peer problems subscale ("rather solitary, tends to play alone", "has at least one good friend", "generally liked by other children", "picked on or bullied by other children", and "gets on better with adults than with other children"). Normal internalizing scores range from 0-4, borderline problems scores range from 5-7, and abnormal scores range from 8-20 (Maurice-Stam et al., 2018). The externalizing behaviour scale is derived by summing the items from the hyperactivity subscale ("restless, overactive, cannot stay still for long", "constantly fidgeting or squirming", "thinks things out before acting", "sees tasks through to the end, good

attention span", and "easily distracted, concentration wanders") and the conduct problems subscale ("often has temper tantrums or hot tempers", "generally obedient, does what adults request", "often fights with other children or bullies them", "can be spiteful to others", and "often argumentative with adults"). Normal externalizing scores range from 0-5, borderline scores range from 6-8, and abnormal scores range from 9-20 (Maurice-Stam et al 2018). The two-factor model (internalizing and externalizing behaviour) has been shown to have a good model fit (CFI = .94; TLI = .94) (Sosu &Schmidt, 2017). There is good reliability for both the internalizing behaviour scale with α ranging from .73 - .82 and the externalizing behaviour scale with α ranging from .78 - .83 (Palmieri & Smith 2007; Goodman et al., 2010; Hessel, et al., 2017). The convergent and discriminant validity of the internalizing and externalizing model has been found to be good (e.g., Goodman et al., 2010; Croft et al., 2015; Kersten et al., 2016) and the measurement invariance has shown good model fit (Sosu & Schmidt, 2017) indicating that the SDQ measures the intended constructs over time.

5.3.5 Measures of socio-economic status

Information on *family household income* was measured using net equivalised weekly income gathered at every data collection wave during the face-to-face interviews by asking respondents to state their take-home income. The average income and standard deviations are presented in table 5.1 below. Parents' *level of education* was measured at each time point by asking parents to indicate their highest level of academic or vocational education which was mapped onto the National Vocational Qualification (NVQ). The NVQ ranges from NVQ 1 (lowest level of qualification) – NVQ 5 (highest level of qualification). Parents' *occupational class* was measured at every data collection wave by asking parents to indicate their occupation which was mapped onto the 7-class National Statistics Socio-economic Classification (NS-SEC). The NS-SEC ranges from 1 (routine occupations) to 7 (higher

managerial and professional occupations). Average levels of education and occupation at each time point are presented in table 5.1 below with additional breakdowns of the percentage of parents at each level of education and occupation provided in tables 5.2 and 5.3 below.

5.3.6 Covariates

Several covariates were identified in the literature as being associated with family SES and children's outcomes so were included in the analyses. *Mothers' age at birth* (Knipe 2016) was measured by asking mothers to state their age in years when they gave birth. The *number of parents living in the household* (Flouri et al., 2016) was specified by the main respondent during the interview when asked whether they lived in a one or two-parent household. *Ethnicity* (Sullivan et al., 2021) is measured by asking the main respondent to indicate the child's ethnicity during the interviews. *Gender* (Mous et al., 2017) was measured by asking the main respondent to specify whether the child was male or female. Children's *birth weight* (Bromley, 2009) was measured by asking the main respondent for information during the interview.

5.3.7 Analytic strategy

Latent Class Growth Analyses (LCGA) models were used to examine whether different trajectories of each SES dimension could be found in the sample. This was done by using the income, education, and occupation data from all 6 data collection waves in the MCS (when children were 9 months, 3, 5, 7, 11, and 14 years old). A two-class model was run first so that each progressive model can be assessed to check if it offers a better fit over the previous model (Jung & Wickrama, 2008).

To determine how many trajectories there were for each SES dimension best fit the data, several goodness of fit indices were used (Jung & Wickrama, 2008; Maydeu-Olivares & Garcı a-Forero, 2010). Akaike's information criterion (AIC; Akaike, 1974) and Bayesian information criterion (BIC; Schwarz, 1978) both assesses model fit based on the likelihood estimates within given parameters (Burnham & Anderson 2002). The AIC and BIC estimates reward a model for being simple and fitting the data well (Burnham & Anderson 2002; Nylund et al., 2007). The better a model fits the data the more it is rewarded but as the model becomes more complex with additional parameters the more it is penalised in the estimates (Burnham & Anderson 2002). Lower AIC and BIC scores indicate better model fit. Entropy was assessed to determine the extent to which latent classes are 'separate' from each other (Weiss & Dardick, 2016). Entropy values range from 0-1 and the closer entropy is to 1 the more the model distinguishes between different classes. Entropy values >.8 indicate that individuals are assigned to different groups with a good level of accuracy (Celeux & Soromenho, 1996). Likelihood ratio and bootstrapped likelihood ratio tests were assessed to determine whether the current model is a better fit than a model with one less class (Lo, Mendell, & Rubin, 2001; McLachlan & Peel, 2000). In both cases, if the *p*-value is significant (<.05) then it means that the current model is a significantly better fit than the previous model with one less class. A summary table of these goodness to fit indicies and their suggested interpretations is presented in table 5.1 below.

Indicator	Suggested interpretations
AIC	Value less than model with one less class suggests improved model fit
BIC	Value less than model with one less class suggests improved model fit
Entropy	Value greater than 0.8 suggests participants assigned to class accurately
LMR	p < 0.05 suggests improved model fit over model with one less class
BLRT	p < 0.05 suggests improved model fit over model with one less class

Tabl	le 5	5.1.	Sumn	nary	of	the	good	ness	of	fit	ind	lices
				•			0					

After the best fitting models for family income, parent's education, and parent occupation were determined using the above criteria, latent class probabilities were generated for each family and exported. A series of hierarchical regressions were then performed to examine associations between each of the generated latent class trajectories and children's language ability, internalizing behaviour, and externalizing behaviour.

5.4 Results

5.4.1 Descriptive statistics

Table 5.2 below presents the descriptive statistics of the analytic sample. On average, children were observed to have normal internalizing and externalizing scores. Normal internalizing scores range from 0-4, and normal externalizing scores range from 0-5 with higher scores becoming problematic (Maurice-Stam et al 2018). Looking at SES dimensions, income on average continues to increase over time and the average level of education increased from A levels/ Highers to teaching qualification/degree. Little change was seen with the average parent level of occupation which remained in the same category on average small employers/own account workers. With income and education increasing on average over the 14 years it would suggest that increasing trajectories could be found. On the other

hand, with occupation remaining stable on average it suggests that only stable trajectories will be found. Additional breakdowns of the percentage of parents at each level of education and occupation are provided in tables 5.3 and 5.4 below.

	М	SD	%	Range
Language ability	50	10	-	20 - 80
Internalizing behaviour	3.711	3.35	-	0 - 19
Externalizing behaviour	4.323	3.51	-	0 - 19
Family income				
9 months	317.52	204.61	-	14.45 - 1282.61
3 years	343.89	223.37	-	14.39 - 1362.46
5 years	364.98	222.88	-	17.09 - 1282.94
7years	398.86	232.38	-	14.93 - 1282.54
11 years	418.07	178.66	-	65.85 - 1162.80
14 years	417.44	178.07	-	80.02 - 1153.52
Parent's level of education				
9 months	2.884	1.403	-	0 - 5
3 years	2.929	1.406	-	0 - 5
5 years	2.909	1.4	-	0 - 5
7years	3.041	1.387	-	0 - 5
11 years	3.181	1.388	-	0 - 5
14 years	3.263	1.4	-	0 - 5
Parent's level of occupation				
9 months	4.639	1.975	-	1-7

 Table 5.2 Descriptive statistics of all variables in the analyses (n = 10683)

3 years	4.614	1.945	-	1-7
5 years	4,577	1.941	-	1-7
7years	4.517	1.94	-	1-7
11 years	4.423	1.916	-	1-7
14 years	4.378	1.913	-	1-7
Mother's age at birth	29.11	5.751	-	14-51
Birth weight (lbs)	7.41	.586	-	.86 - 14.44
Parents in household				
Two parents			87.5	
One parent			12.5	
Gender				
Male			51.9	
Female			48.1	
Ethnicity				
White			86.41	
Mixed			0.09	
Indian			2.5	
Pakistani & Bangladeshi			6.5	
Black British			2.8	
Chinese/ other			1.7	

Child age	Parent's level of education	%
9 months		
	NVQ1	6.1
	NVQ2	25.5
	NVQ3	17.9
	NVQ4	41.9
	NVQ5	8.6
3 Years		
	NVQ1	5.9
	NVQ2	25.2
	NVQ3	17.5
	NVQ4	42
	NVQ5	9.4
5 Years		
	NVQ1	5.8
	NVQ2	24.9
	NVQ3	17.4
	NVQ4	42.4
	NVQ5	9.4
7 Years		
	NVQ1	4.8
	NVQ2	23
	NVQ3	17.3
	NVQ4	41.7

Table 5.3 Parent's level of education at each wave (N=10,683).

	NVQ5	13.2
11 Years		
	NVQ1	4.8
	NVQ2	20.6
	NVQ3	16.2
	NVQ4	40.9
	NVQ5	17.6
14 Years		
	NVQ1	4.8
	NVQ2	19.6
	NVQ3	15.7
	NVQ4	40.3
	NVQ5	19.7

Table 5.4 Parent's level of household occupation at each wave (N=10,683)

Child age	Parent's level of occupation	%
9 months		
	Routine	4.3
	Semi routine	19.2
	Lower supervisory	12.6
	Small employers	9.2
	Intermediate occupations	10.6
	Lower managerial and professional	22.7
	higher managerial and professional	21.3

3 Years		
	Routine	3.6
	Semi routine	18.7
	Lower supervisory	13.7
	Small employers	11.6
	Intermediate occupations	9.2
	Lower managerial and professional	22.9
	higher managerial and professional	20.4
5 Years		
	Routine	3.8
	Semi routine	19.4
	Lower supervisory	14.4
	Small employers	11.5
	Intermediate occupations	9
	Lower managerial and professional	22.5
	higher managerial and professional	19.3
7 Years		
	Routine	4.1
	Semi routine	20.3
	Lower supervisory	14
	Small employers	12.2
	Intermediate occupations	8.8
	Lower managerial and professional	22.7
	higher managerial and professional	18

11 Years

5.4.2 Trajectories of SES dimensions

LCGA was undertaken to determine the number of family income, parent education, and parent occupation trajectories in the analytic sample (n = 10683) from 9 months to age 14. A six-class model was selected as best fitting for family income, a six-class model was best fitting for the level of parent's education, and a five-class model was best fitting for parent's level of occupation. The trajectories for family income, parents' education, and parents' occupation are presented in figures 5.1, 5.2, and 5.3 below. The reason these models were selected over alternative models will now be discussed.

5.4.3 Trajectories of family income

The six-class family income model was selected over alternate models for the following reasons. The AIC and BIC were lower for the six-class than the five-class model while the entropy remained high. The likelihood ratio test and the bootstrapped likelihood ratio test both indicated that the six-class model was a significant improvement over the five-class model (p<.001). When examining the trajectory plots it was evident that the five-class model failed to identify a distinct stable-high income group while the six-class model di didentify this group. A seven-class model was run to test whether it would be a better fit than the six-class model. The likelihood ratio test was verging on non-significance (p = .046) indicating that it was a slightly better model fit than the six-class model. However, examination of the trajectory plots indicated that the seven-class model had two very similar low-income trajectory identified by the 6-class model. The six-class model was therefore the best fitting model for the data that allowed for meaningful interpretation of the results. Table 5.5 presents a full breakdown of the model fit indices for two- to seven-class models.

Model	AIC	BIC	Entropy	% of sample in each class	Posterior values	LMR-LRT	BLRT
Two-class	784956.236	785036.277	0.904	-	-	<.001	<.001
1				33.3	0.958		
2				66.7	0.978		
Three-class	773672.545	773774.415	0.903	-	-	<.001	<.001
1				51.33	0.941		
2				12.86	0.945		
3				35.81	0.909		
Four-class	769508.977	769632.676	0.885	-	-	<.001	<.001
1				17.38	0.913		
2				5.99	0.948		
3				31.88	0.909		
4				44.75	0.957		
Five-class	767826.813	767972.341	0.89	-	-	<.001	<.001
1				5.8	0.945		
2				4.78	0.84		
3				14.73	0.899		
4				43.77	0.96		
5				30.92	0.903		
Six-class	766131.946	766299.304	0.878	-	-	<.001	<.001
1				4.53	0.855		
2				15.68	0.868		
3				2.77	0.942		
4				41.05	0.954		
5				28.25	0.881		
6				7.72	0.891		
Seven-class	765334.639	765523.825	0.84	-	-	0.046	<.05
1				36.49	0.934		
2				22.78	0.812		
3				2.17	0.937		
4				9.87	0.848		
5				5.1	0.854		
6				5.88	0.878		
7				17.71	0.821		

Table 5.5 Indicators of model fit for family income

The following income trajectories were identified in the six-class model: Very highdecreasing (n = 290; 2.77%), stable-high (n = 828; 7.72%), high-decreasing (n = 450; 4.53%), average-increasing (n = 1666; 15.68%), low-increasing (n = 3043; 28.25%), and stable-low (n = 4406; 41.05%).



Figure 5.1. Six-class model of income trajectories

Note. Green = Very-high-decreasing. Black = Stable-high. Red = High-decreasing. Blue = Average-increasing. Yellow = low-increasing. Pink = Stable Low.

5.4.4 Associations between income trajectories and children's outcomes

Multiple regression was carried out to observe the associations that each income trajectory had on children's language ability, externalizing behaviour, and internalizing behaviour. The findings are presented in table 5.6 below. The stable-high income trajectory was used as the reference category. Results show that belonging to the stable-low-income trajectory was associated with the lowest language ability ($\beta = -0.317$, p<.001), and highest internalizing behaviour problems ($\beta = 0.325$) and externalizing behaviour problems (0.313, p<.001) followed by the low-increasing, average-increasing, high-decreasing trajectories. There was no significant association between the very-high-decreasing income trajectory and children's outcomes which suggests a cushioning effect at the highest levels of income. Children belonging to increasing income trajectories had better associations with all outcomes as children from the low-increasing trajectory had a significantly better association with language ability ($\beta = -0.248$, p<.01), internalizing behaviour ($\beta = 0.161$, p<.01) and externalizing behaviour ($\beta = 0.122$, p<.01) compared to those from the stable low income trajectory. Those in the average-increasing trajectory saw further improvements (language ability: $\beta = -0.119$, p<.01; internalizing behaviour: $\beta = -0.046$, p<.01; externalizing behaviour: $\beta = -0.019$, p = n.s.). The findings also show that falls in income even for high earners can have a negative impact on children's outcomes as the high-decreasing income trajectory has a significant negative association with children's language ability ($\beta = -0.078$, p<.001), and a significant association with children's internalizing ($\beta = 0.04$, p<.001) and externalizing ($\beta = 0.035$, p<.001) behaviour problems.

Table 5.6. Associations behaviour	between family income trajec	tories a	nd children's language ability	y, internalizi	ing behaviour, and externaliz	zing
	Age 14 Language Abilit	h	Age 14 Internalizing Be	chaviour	Age 14 Externalizing Beh	haviour
Trajectory	B (95% CI)	S.E.	B (95% CI)	S.E.	B (95% CI)	S.E.
Stable-high (ref)						
Very-high-decreasing	-0.007 (-0.033, 0.018)	0.013	-0.004 (-0.021 , 0.013)	0.009	0.006 (-0.013, 0.025)	0.01
High-decreasing	-0.078 (-0.105, -0.05)***	0.014	$0.04 (0.02, 0.061)^{***}$	0.01	0.035 (0.015, 0.056)***	0.01
Average-high						
increasing	-0.119 (-0.153, -0.085)***	0.017	$0.046 \ (0.02, \ 0.073)^{***}$	0.013	0.019 (- 0.005 , 0.044)	0.013
Low-increasing	-0.248 (-0.289, -0.208)***	0.021	$0.161 (0.128, 0.194)^{***}$	0.017	0.122 (0.091, 0.154) * * *	0.016
Stable-low	-0.371 (-0.418, -0.324)***	0.024	0.325 (0.285, 0.366)***	0.021	0.313 (0.275, 0.352)***	0.02
Birth weight	0.015 (-0.01, 0.04)	0.013	-0.053 (-0.081, -0.025)***	0.014	-0.059 (-0.088, -0.03)***	0.015
Gender	0.004 (-0.02, 0.029)	0.012	0.049 (0.025, 0.074)***	0.013	-0.14 (-0.164, -0.116)***	0.012
Parents in home	-0.03 $(-0.06, 0.001)$	0.015	0.015 (-0.019, 0.049)	0.017	0.027 (-0.007, 0.061)	0.017
Mother age at birth	$0.053 (0.024, 0.083)^{***}$	0.015	0 (-0.031, 0.032)	0.016	-0.088 (-0.119, -0.057)***	0.016
Mixed ethnicity	0.013 (-0.012, 0.037)	0.012	0.001 (-0.027, 0.029)	0.014	-0.028 (-0.057, 0.001)	0.015
Indian	0.006 (-0.019, 0.031)	0.013	-0.034 (-0.051, -0.017)***	0.009	-0.039 (-0.055, -0.023)***	0.008
Pakistani &						
Bangladeshi	-0.015 (-0.036, 0.007)	0.011	-0.005 (-0.026, 0.016)	0.011	-0.052 (-0.071, -0.032)***	0.01
Black British	0 (-0.026, 0.025)	0.013	-0.033 (-0.065, -0.002)*	0.016	-0.031 (-0.058, -0.004)*	0.014
Chinese and other	0 (-0.021, 0.021)	0.011	-0.013 (-0.039, 0.012)	0.013	-0.028 (-0.052, -0.003)*	0.013
<i>Note.</i> $p < .05 = *, p < .01$	= **, p < .00I = ***, N = 10,683.					

 Table 5.6. Associations between family income trajectories and children's language
 ability, internalizing and externalizing behaviour

5.4.5 Trajectories of parent's level of education

The six-class trajectory model for parent's level of education was selected over alternate models because the AIC and BIC were lower for the six-class than all previous models, entropy remained high, and the likelihood ratio test and the bootstrapped likelihood ratio test both indicated that the six-class model was a significant improvement over the five-class model (p<.001). When examining the trajectory plots the six-class model identified parents that improved their education over time while the five-class model did not. A seven-class model was not a significantly better model fit than the six-class model. The six-class model was therefore the best fitting model for the data. Table 5.7 below presents a full breakdown of the model fit indices for two- to seven-class models.

Model	AIC	BIC	Entropy	% of sample in each class	Posterior values	LMR-LRT	BLRT
Two-class	168351.863	168431.851	0.964	-	-	<.001	<.001
1				54.67	0.992		
2				45.33	0.989		
Three-class	142056.218	142158.021	0.978	-	-	<.001	<.001
1				19.06	0.992		
2				38.73	0.995		
3				42.21	0.988		
Four-class	130708.075	130831.692	0.977	-	-	<.001	<.001
1				36.47	0.997		
2				18.1	0.987		
3				16.72	0.976		
4				28.71	0.982		
Five-class	123459.268	123604.7	0.989	-	-	<.001	<.001
1				36.83	0.998		
2				27.11	0.992		
3				12.71	0.996		
4				15.74	0.988		
5				7.61	0.972		
Six-class	121092.039	121259.286	0.978	-	-	<.001	<.001
1				12.25	0.986		
2				15.52	0.989		
3				36.71	0.997		
4				7.86	0.98		
5				23.3	0.977		
6				4.36	0.922		
Seven-class	120450.572	120639.634	0.978	-	-	n.s.	n.s
1				0.93	0.911		
2				36.22	0.997		
3				23.13	0.976		
4				8.38	0.968		
5				15.07	0.989		
6				4.67	0.942		
7				11.6	0.984		

Table 5.7. Indicators of model fit for parent's level of education

The following parent level of education trajectories were identified in the six-class model: Stable high (n = 3912, 36.71%), stable average (n = 1644, 15.52%), stable low (n = 2492, 23.3%), no qualifications (n = 1313, 12.25%), increasing no qualification-low (n = 829, 7.86%), and increasing low-high (n = 433, 4.36%).



Figure 5.2. Parent level of education trajectories

Note. Green = Stable-high. Blue = Stable-average. Yellow = Stable-low. Red = no qualifications. Pink = Increasing no qualification-low. Black = Increasing low-high.

5.4.6 Associations between parent education trajectories and children's outcomes

Multiple regression was carried out to observe the associations that each education trajectory had on children's language ability, externalizing behaviour, and internalizing behaviour. The findings are presented in table 5.8 below. The stable-high education trajectory was used as the reference category. For language ability results show that belonging to the noqualifications trajectory was associated with the lowest language ability ($\beta = -0.213$, p<.001), followed by the stable-low, increasing no-qualifications-low, stable average, and increasing low-high trajectories. Children belonging to an increasing education trajectory had better language ability compared to their peers in non-increasing trajectories. For example, those in the increasing no-qualifications-low trajectory had better language ability ($\beta = -0.145$, p<.01) than those remaining in the no-qualifications trajectory ($\beta = -0.213$, p<.001) and the difference was significant (t = 6.15, p <.001). Similarly, those in the increasing low-high trajectory ($\beta = -0.077$, p<.01) had better language ability than their peers that remained in the stable-low trajectory ($\beta = -.19$, p<.001) and the difference between the trajectories was also significant (t = 3.01, p <.001).

For both internalizing and externalizing behaviour results show the no qualifications trajectory was associated with more internalizing ($\beta = 0.132$, p<.001) and externalizing ($\beta = 0.171$, p<.001) behaviour problems followed by the stable-low, increasing no qualificationslow, increasing low-high, and stable average. Children belonging to an increasing education trajectory also had fewer behavioural problems compared to their peers in non-increasing trajectories. Belonging to the increase no qualifications-low trajectory was associated with fewer internalizing ($\beta = 0.099$, p<.001) and externalizing ($\beta = 0.124$, p<.001) behaviour problems compared to their peers that remained in the stable-low trajectory (internalizing: $\beta = 0.132$, p<.001; externalizing: $\beta = 0.171$, p<.001). Similarly, those belonging to the increasing low-high trajectory had fewer internalizing ($\beta = 0.059$, p<.001) and externalizing ($\beta = 0.063$, p<.001) behaviour problems compared to their peers that remained in the stable-low trajectory (internalizing: $\beta = 0.11$, p<.001; externalizing ($\beta = 0.127$, p<.001).

	Age 14 Language Abil	lity	Age 14 Internalizing Be	ehaviour	Age 14 Externalizing Bel	ehaviour
Trajectory	B (95% CI)	S.E.	B (95% CI)	S.E.	B (95% CI)	S.E.
Stable-high (ref)						
Increasing low-high	-0.077 (-0.103, -0.051)***	0.013	0.059 (0.03, 0.087)***	0.015	$0.063 (0.033, 0.093)^{***}$	0.015
Stable-average	-0.113 (-0.138, -0.088)***	0.013	0.036 (0.011, 0.061)**	0.013	$0.037 (0.013, 0.06)^{**}$	0.012
Increasing no						
qualification-low	-0.145 (-0.177, -0.114)***	0.016	$0.099 (0.068, 0.129)^{***}$	0.016	$0.124 \ (0.092, 0.156)^{***}$	0.017
Stable-low	-0.19 (-0.216, -0.164)***	0.013	0.11 (0.082, 0.139)***	0.015	0.127 (0.098, 0.155)***	0.014
No Qualification	-0.213 (-0.243, -0.183)***	0.016	0.132 (0.098, 0.166)***	0.017	0.171 (0.136, 0.206)***	0.018
Birth weight	0.012 (-0.013, 0.037)	0.013	-0.056 (-0.084, -0.028)***	0.014	-0.06 (-0.09, -0.031)***	0.015
Gender	0.006 (-0.018, 0.03)	0.012	0.046 (0.021, 0.071)***	0.013	-0.143 (-0.167, -0.119)***	0.012
Parents in home	-0.023 (-0.052, 0.006)	0.015	0.037 (0.004, 0.07)*	0.017	0.042 (0.008, 0.077)*	0.018
Mother age at birth	0.078 (0.05, 0.106)***	0.014	-0.053 (-0.083, -0.022)***	0.016	-0.134 (-0.164, -0.105)***	0.015
Mixed ethnicity	0.007 (-0.017, 0.031)	0.012	0.008 (-0.022, 0.038)	0.015	-0.021 (-0.05, 0.008)	0.015
Indian	-0.001 (-0.026, 0.024)	0.013	-0.029 (-0.046, -0.012)***	0.009	-0.034 (-0.049, -0.018)***	0.008
Pakistani &						
Bangladeshi	-0.024 (-0.044, -0.004)*	0.01	$0.022 \ (0.002, 0.043)^{*}$	0.01	-0.027 (-0.047, -0.006)*	0.01
Black British	-0.019 (-0.045, 0.006)	0.013	-0.014(-0.044, 0.016)	0.015	-0.012 (-0.04, 0.015)	0.014
Chinese and other	-0.004 (-0.027, 0.018)	0.011	-0.005 (-0.031, 0.022)	0.014	-0.02 (-0.046, 0.005)	0.013

Table 5.8. Associations between parent education trajectories and children's language ability, internalizing behaviour, and externalizing beha

Table 5.8. Associations between parent education trajectories and children's languag
ability, internalizing and externalizing behaviour

5.4.7 Trajectories of parent's level of occupation

The five-class trajectory model for parent's level of education was selected over alternate models because the AIC and BIC were lower than previous models, the entropy remained high, and the likelihood ratio test indicated that the six-class model would not be a significant improvement over a five-class model. When examining the trajectory plots the six-class model had an additional class within the same level of occupation and therefore did not add anything over the five-class model. Therefore, the five-class model was, therefore, the best fitting model for the data whilst being interpretable. Table 5.9 below presents a full breakdown of the model fit indices for two- to six-class models.

Model	AIC	BIC	Entropy	% of sample in each class	Posterior values	LMR-LRT	BLRT
Two-class	192129.508	192208.921	0.958	-	-	<.001	<.001
1				58.77	0.988		
2				41.21	0.985		
Three-class	182091.529	182192.599	0.929	-	-	<.001	<.001
1				22.76	0.93		
2				49.71	0.981		
3				27.53	0.974		
Four-class	176414.25	176536.978	0.941	-	-	<.001	<.001
1				35.12	0.954		
2				24.38	0.976		
3				17.53	0.99		
4				22.97	0.976		
Five-class	171446.598	171590.984	0.962	-	-	<.001	<.001
1				27.92	0.975		
2				17.21	0.989		
3				18.72	0.951		
4				13.33	0.988		
5				22.82	0.978		
Six-class	162582.193	162748.237	0.96	-	-	n.s.	<.05
1				23.31	0.996		
2				26.92	0.872		
3				17.44	0.976		
4				8.74	0.962		
5				10.87	0.975		
6				12.72	0.972		

Table 5.9 Indicators of model fit for parent's level of occupation

The following parent level of occupation trajectories were identified in the five-class model: Decreasing very high-high (n = 2312, 22.82%), decreasing high-intermediate (n = 2777, 27.92%), stable-intermediate (n = 1944, 18.72%), stable-average (n = 1317, 13.33%), and increasing routine-semi-routine (n = 1740, 17.21%).



Figure 5.3. Parent level of occupation trajectories

Note. Black = decreasing very high-high. Red = decreasing high-intermediate. Green = stable-intermediate. Pink = stable-average. Blue = increasing routine-semi-routine.

5.4.8 Occupation trajectories on children's outcomes

Multiple regression was carried out to observe the associations that each occupation trajectory had on children's language ability, externalizing behaviour, and internalizing behaviour. The findings are presented in table 5.10 below. The decreasing very high-high occupation trajectory was used as the reference category. Results show that there is no significant difference between the decreasing very high-high reference category and the decreasing high-intermediate trajectory for any of the children's outcomes. For children's language ability, the increasing routine-semi-routine trajectory had the largest negative association ($\beta = -.203$, p <.001), followed by stable-average and stable-intermediate trajectories. For both internalizing and externalizing behaviour the routine-semi-routine trajectory had the largest associations with behaviour problems (internalizing: $\beta = .124$; p <.001; externalizing: $\beta = .128$, p <.001) followed by the stable-intermediate, and stable-average trajectories.

r, and	
haviou	
j be	
internalizing	
ability,	
language	
children's	
and	
trajectories a	
occupation	
parent	
between	
l su	JUL
ssociatio	behavid
V 0	zing
5.1	aliz
ble	tern

Age 14 Language Ability Trajectory B (95% CI) S.E. decreasing very high- - - - high (ref) - - - - High-intermediate -0.007 (-0.038, 0.025) 0.01 - 0.01 Accreasing -0.042 (0.014, 0.071)** 0.01 - 0.01 Stable-average -0.042 (0.014, 0.071)** 0.01 - 0.01	y S.E. 1 0.016 (0.015 (0.015 (Age 14 Internalizing Bel (95% CI)			
Trajectory B (95% CI) S.E. decreasing very high- -	S.E. 1 0.016 (0.015 (t (95% CI)	haviour	Age 14 Externalizing Bel	haviour
decreasing very high- high (ref)	0.016 (0.015 0		S.E.	B (95% CI)	S.E.
High-intermediate decreasing -0.007 (-0.038, 0.025) 0.01 Stable intermediate -0.036 (0.007, 0.065)* 0.01 Stable-average -0.042 (0.014, 0.071)** 0.01 routine-semi-routine -0.203 (0.174, 0.231)*** 0.01	0.016 (0.015 (0.0		,		
decreasing -0.007 (-0.038, 0.025) 0.01 Stable intermediate -0.036 (0.007, 0.065)* 0.01 Stable-average -0.042 (0.014, 0.071)** 0.01 routine-semi-routine -0.203 (0.174, 0.231)*** 0.01	0.016 (0.015 (0.015 (
Stable intermediate -0.036 (0.007, 0.065)* 0.01 Stable-average -0.042 (0.014, 0.071)** 0.01 routine-semi-routine -0.203 (0.174, 0.231)*** 0.01	0.015 0	.004 $(-0.039, 0.031)$	0.018	0.002 (-0.038, 0.034)	0.019
Stable-average -0.042 (0.014, 0.071)** 0.01 routine-semi-routine -0.203 (0.174, 0.231)*** 0.01 increasing -0.203 (0.174, 0.231)*** 0.01	0.015 (.063 (-0.094, -0.031)***	0.016	0.056 (-0.087, -0.025)***	0.016
routine-semi-routine increasing -0.203 (0.174, 0.231)*** 0.01		.041 (-0.072, -0.011) **	0.016	0.052 (-0.082, -0.022)***	0.015
increasing -0.203 (0.174, 0.231)*** 0.01					
	0.014 (.124 (-0.153, -0.096)***	0.014	0.128 (-0.157, -0.099)***	0.015
Birth weight 0.019 (-0.006, 0.045) 0.01	0.013 -	0.062 (-0.09, -0.033)***	0.014	-0.068 (-0.098, -0.039)***	0.015
Gender 0.008 (-0.016, 0.033) 0.01	0.012 ($.043 (0.018, 0.068)^{***}$	0.013	-0.149 (-0.173, -0.124)***	0.012
Parents in home -0.076 (-0.105, -0.046)*** 0.01	0.015 (.071 (0.039, 0.104) * * *	0.017	0.089 (0.055, 0.123)***	0.017
Mother age at birth 0.083 (0.054, 0.112)*** 0.01	0.015 -	0.05 (-0.081, -0.018)**	0.016	-0.136 (-0.167, -0.105)***	0.016
Mixed ethnicity 0.007 (-0.018, 0.032) 0.01	0.013 (.009 (-0.023, 0.04)	0.016	-0.02 (-0.05, 0.01)	0.015
Indian 0.004 (-0.021, 0.029) 0.01	0.013 -	0.032 (-0.049, -0.015)***	0.009	-0.037 (-0.052, -0.022)***	0.008
Pakistani &					
Bangladeshi -0.047 (-0.067, -0.027)*** 0.01	0.01 ($.039 (0.019, 0.059)^{***}$	0.01	-0.005 (-0.023, 0.014)	0.01
Black British -0.017 (-0.043, 0.009) 0.01	0.013 -	0.012 (-0.043, 0.019)	0.016	-0.01 (-0.037, 0.017)	0.014
Chinese and other -0.015 (-0.038, 0.008) 0.01	0.012 (.002 (-0.024, 0.028)	0.013	-0.012 (-0.037 , 0.014)	0.013

Table 5.10. Associations between parent occupation trajectories and children's language ability, internalizing and externalizing behaviour

5.5 Discussion

Using data from the millennium cohort study, this study aimed to examine whether different family income, parent education, and parent occupation trajectories could be identified and whether the trajectories had differential associations with children's language ability, internalizing behaviour, and externalizing behaviour during late childhood. The study found that six family income trajectories (Very high-decreasing, stable-high, high-decreasing, average-increasing, stable-average, and stable-low), six parent education trajectories (Stable high, stable average, stable low, no qualifications, increasing no qualification to low, and increasing low-high), and five parent occupation trajectories (Stable high, decreasing high to intermediate, stable intermediate, stable average, and increasing routine-semi-routine) fit the data best. A substantial proportion of the sample belonged to increasing or decreasing SES trajectories. For family income 7.3% of families belong to a decreasing income trajectory, 15.68% belonged to an increasing income trajectory, and 77.02% remained in a stable income trajectory. For parents' education, 12.22% belonged to an increasing education trajectory and 87.78% belonged to a stable education trajectory. For parents' occupations, 27.92% belonged to a decreasing trajectory, 17.21 % belonged to an increasing trajectory, and 64.87% remained in a stable trajectory.

In terms of the previous literature that has examined family income trajectories, the present results were as expected as studies have found several increasing and decreasing income trajectories (Dearing et al., 2001; Dohoon, 2014; Jackson et al., 2017; Kwon et al., 2022). Between three and six trajectories have been found in the literature (Kozyrskyj et al., 2010; Lee, 2014; Demment et al, 2014; Jackson et al., 2017; Kendzor et al., 2017; Comeua et al., 2018; Kwon et al., 2022) but most choose a three- or four-class model (Demmet et al., 2014; Lee, 2014). A possible reason for this study finding a higher number of trajectories than some

of the other studies is because this study examines a longer period of time. Studies that gather data over a longer period of time tend to find more trajectories than those that examine a shorter period of time (e.g., Miller & Votruba-Drzal 2017; Björkenstam et al., 2017). Another possible reason for the different number of trajectories could be due to the measurement of income. Generally, studies look at trajectories of poverty (Lee, 2014) or movements into or out of a low-income band (Demmet et al., 2014). Using categories such as poverty status inherently limits the number of trajectories to a variation of "always poor", "never poor", "leaving poverty", and "entering poverty" as additional trajectories would not make theoretical sense and would make findings more difficult to interpret (e.g., Demmet et al., 2014; Lee, 2014). However, the current study measured family income as continuous based on weekly income which is likely to vary more than poverty status since families can experience a decrease in income while not entering poverty for example. The interpretation of these additional trajectories makes sense given the way income is measured in this study.

Similarly, some studies do find six income trajectories but opt to use models with fewer trajectories because it fits the purpose of their study, even when it is not necessarily the best fitting model as per the goodness of fit indices (e.g., Demmet et al., 2014). Jung and Wickrama (2008) explain that while the goodness of fit indices should be used, they should not be the only determining factor in model selection. They stress that in addition to model fit, researchers should choose the models that make the most theoretical or empirical sense for their research questions. In the supplementary material from Demmet et als (2014) study the goodness of fit indices indicated that both a five-class and six-class model outperformed their four-class model. However, they chose to use the four-class model because the additional trajectories would unnecessarily complicate the model and would not add value to the interpretation of the results because two additional trajectories would be too similar.
However, in the current study, the six-class model was chosen as each trajectory was distinctly different as seen in figure 1 and the entropy value of 0.878. The six-class model identified the stable-high income trajectory representing 7.72% of the sample which the five-class model did not identify. Therefore, if a model with fewer classes were selected in this study, then those families would have been assigned to a trajectory that would not have best represented their income trajectories over the measured period.

Regarding family income, the results are in line with previous research in that the highest income trajectories are associated with better language and behavioural outcomes while the lowest income trajectories are associated with poorer language and behavioural outcomes (e.g., Dearing et al., 2001; Schoon et al., 2011; Zachrisson & Dearing, 2015; Björkenstam; Jackson et al., 2017; Comeau et al., 2018; Hoyt et al., 2019). However, this study extends previous studies by showing this pattern continues into late childhood. Like previous studies (e.g., Jackson et al., 2017; Hoyt et al., 2019) the increasing family income trajectory was associated with better child language ability, internalizing behaviour, and externalizing behaviour when compared to the non-increasing trajectory. This was particularly true regarding children's externalizing behaviour as the association between the increasing from average-high income trajectory and children's externalizing behaviour was not significantly different from the stable-high income trajectory showing that they were equally beneficial for children's externalizing behaviour during late childhood. It is possible that the increasing income trajectories are associated with better child outcomes as the family is increasingly able to invest more into their children's development as outlined in the family investment model. The family stress model also would suggest that an increased income would reduce the amount of financial pressure on parents which reduces parental stress and parent-child conflict and also leads to improved outcomes. Additionally, children become increasingly

aware of their economic situations as they get older and if their family's low income leads to them missing out on social activities (Hjalmarsson & Mood, 2015) and being excluded by their peers (Hills et al., 2002; Bai et al., 2021) then they can feel stress and shame (Conger et al., 2010; Walker, 2014; Hjalmarsson & Mood, 2015). The family stress model indicates that this would cause additional conflict and impaired interactions between parents and children leading to increased behavioural problems. By belonging to an increasing income trajectory it is less likely that children will experience this and they are therefore less likely to have behavioural problems. On the other hand, the decreasing from high income was associated with worse child language ability, internalizing behaviour, and externalizing behaviour when compared to the stable-high trajectory. Following the same logic as above then as income decreases then parents have fewer finances to invest in their children's development which impacts their outcomes. There is also likely to be increased financial pressure on the parents as their income decreases, particularly as the cost of raising a child increases (Twenge et al., 2003), which would result in parental stress and conflict leading to poorer child outcomes.

While these trends are similar to the literature, it should be noted that this study found that the decreasing-high income trajectory had a better association with children's language ability than the increasing average-high income trajectory. This indicates that while improving income is beneficial for children's language ability, it cannot fully compensate for experiencing high income during the earliest years of life. This is aligned with previous research which indicates that the earliest years are most important, particularly in terms of children's language (e.g., Costello et al., 2011; Gibb et al., 2012; Lai et al., 2019). However, this finding differs from Jackson et al. (2017) who found the increasing income trajectory had a better association with children's language outcomes than the decreasing trajectory. A possible reason for the different findings could be the measure of income used and therefore

what the income trajectories represent. Jackson et al (2017) measured family income using the household poverty ratio to determine the probability that a family would be poor. Their trajectories, therefore, reflected the increasing and decreasing probability that a family will be poor rather than describing the extent to which family income increased or decreased. In the current study, income was measured by weekly income which allowed the income trajectories to reflect the extent to which weekly income increased or decreased.

In terms of internalizing and externalizing behaviour, the current findings were in line with previous research (e.g., Hoyt et al., 2019). Hoyt et al (2019) found that increasing and decreasing income trajectories both had the same association with children's internalizing behaviour ($\beta = .04$) while this study found the same for the decreasing income trajectory ($\beta = .04$) and a similar association for the increasing income trajectory ($\beta = .046$). For externalizing behaviour, Hoyt et al (2019) found the increasing income trajectory was associated with better externalizing behaviour than the decreasing income trajectory which was also the case in the current study. In fact, the association between the increasing income trajectory and children's externalizing behaviour was not significantly different from the stable high-income trajectory indicating that increasing income is particularly beneficial for children's externalizing behaviour.

Regarding parent's education trajectories, the stable-high education trajectory had the best association with children's late childhood language and behavioural outcomes and the no qualification trajectory had the worst association with all outcomes which was expected since previous studies show that higher levels of parent education are associated with better child outcomes (Magnuson et al., 2009; Harding, 2015; Monaghan, 2017). Increasing parent education trajectories were also associated with better language ability, internalizing

behaviour, and externalizing behaviour. However, while increasing from no qualifications to low education had a better association with children's language and behavioural outcomes when compared to their peers that did not improve, the improvement was not significant. On the other hand, increasing from low-to-high education had a significantly better association with all child outcomes when compared to their peers that did not improve their education.

While this study is in line with the majority of previous studies that have shown increased parent education improves children's language and behavioural outcomes (e.g., Magnuson et al., 2009; Harding, 2015; Monaghan, 2017; Awada & Shelleby, 2021), some studies have shown that increased parent education was not associated with improved externalizing behaviour (Magnuson, 2003; Harding, 2015). A possible reason for the different findings could be the limited timeframe examined in those studies as they are limited to early childhood (Magnuson, 2003; Harding, 2015). It has been suggested that examining education increases during short periods of time may not show improved child outcomes because it takes time for improved education to filter into changes in children's environment to benefit the child (Awada & Shelleby, 2021) and therefore a longer timeframe should be examined. A further reason for the different findings could be the way in which increased education is defined. Previous studies have defined increased education as enrolment in an education program even though the parents have not necessarily completed the course (Magnuson, 2003; Magnuson et al., 2009). This is problematic as it does not reflect parents' educational attainment and assumes all parents would complete the program.

Further studies that have defined educational increases as the successful completion of a program of study tend to group parents into an increasing category without differentiating between levels of educational increase (e.g., Magnuson et al., 2009; Harding, 2015; Awada &

Shelleby, 2021). Awada and Shelleby (2021), for example, categorised parents into the increasing education group if their current level of education was higher than their baseline level of education. Similarly, Magnuson et al (2009) used a dichotomous variable to indicate whether or not parents' education increased, despite parents reporting different levels of education increase. While both Awada and Shelleby (2021) and Magnuson et al. (2009) found that increasing education had a better association with children's outcomes, they do not provide any indication about what level of education increase significantly improves children's outcomes. The current study addressed the above limitations by examining different education trajectories based on parents' attainment over 14 years and identified different increasing trajectories. Results indicate that increasing to a low level of education is not likely to significantly improve children's outcomes and suggest that parents must achieve a certain level of education to significantly improve their children's outcomes.

Regarding parents' occupation trajectories, the results found very little variation over the childhood timeline. The results largely reinforce previous research in that the higher the occupation trajectory the better the association with children's language ability and behavioural outcomes. Although an increasing trajectory was found, it only increased from routine to semi-routine occupation and there was no stable routine trajectory to compare with. Therefore, it cannot be determined whether increasing occupation is associated with better child language and behavioural outcomes in the current sample. The decreasing high-intermediate occupation trajectory showed no difference when compared to the stable-high trajectory. However, since the stable-intermediate trajectory had significantly poorer associations with children's outcomes compared to the decreasing from high-intermediate trajectory, it suggests that high occupation during early childhood could be a protective factor for children's outcomes. Since there was no difference between the stable-high occupation

trajectory and the decreasing from high- intermediate trajectory, future research may benefit from using the more efficient four-class model which combines both trajectories.

5.5.1 Limitations of the study

The limitations of this study should be considered when interpreting the results. The family income, parent education, and parent occupation trajectories were identified using latent classs growth analysis which requires researchers to select the appropriate number of latent classes at their discretion. While goodness of fit indices are used to guide researchers to the best fitting model, they do not dictate the final model selection. Researchers must consider the research question, theory, and the data alongside the goodness to fit indices when selecting the appropriate model (Jung & Wickrama, 2008). This is inherently problematic as researchers may draw different conclusions when deciding on the best model to use for their studies. Still, since the family income trajectories found in the study are similar to those found in previous studies, it provides confidence that researchers are following similar methods when selecting trajectories. However, there is no frame of reference to draw upon regarding education and occupation trajectories and this should be considered when drawing conclusions.

Second, the highest household level of parent education and occupation was measured at each time point and used to generate trajectories. This method does not provide information about which parent has the highest education and occupation at each time point or whether the parent with the highest level of education changed over time. While both maternal and paternal education and occupation have significant associations with children's developmental outcomes (Erola et al., 2016; Jeong et al., 2017), some studies indicate that maternal education is more important (Jeong et al., 2017). On the other hand, evidence

indicates that maternal education is most important for children's outcomes during early childhood and decreases over time while the association between fathers' education and children's outcomes increases over time (Erola et al., 2016). However, since both maternal and paternal education and occupation are significantly associated with children's development, measuring the highest level of household education and occupation was appropriate for this study. Attrition in the millennium cohort study was more likely to occur in low-income families (Plewis, 2007). This could potentially lead to a biased sample resulting in the SES trajectories and outcome trajectories identified in this study being less likely to represent lower income families. However, the sample design of the millennium cohort study meant that deprived families were oversampled, and attrition weights were generated and applied to account for potential bias.

5.5.2 Implications of the study

First, despite comprehensive income support for low-income families during early childhood, greater prolonged income support to improve family income over middle and late childhood is likely to be beneficial for children's language ability and behavioural outcomes. Although policy prioritises low-income families, improving income for higher earners over early to late childhood is also likely to improve children's developmental outcomes. Prolonged support is particularly important as this study found that a decrease in income can negatively impact children's outcomes. It may therefore be beneficial to consider extending the age limit for some age-related policies such as the Child Payment in Scotland otherwise the decrease in family income is likely to impair children's outcomes. Second, policy aimed at improving parents' level of education is likely to somewhat improve children's language ability and behavioural outcomes in late childhood. To achieve improvementments in children's outcomes parents should be supported to improve their education. Further research is needed

to determine the extent to which parents' education needs to be improved to significantly benefit children's outcomes as this study was only able to examine the effect of increasing to NVQ 1 level education and NVQ 4 level education. Further research may indicate that parents who achieve NVQ level 2 or NVQ level 3 qualifications significantly benefit children's outcomes. This would be easier for policy to facilitate and would cost the parents less time and money to achieve.

To conclude, this study adds to current SES trajectory literature by examining whether income, education, and occupation trajectories over the entire childhood timeline are associated with children's language ability and behavioural outcomes. It presents novel findings by examining education and occupation trajectories that have not previously been studied and examines a longer timeframe than previous literature. It highlights the potential usefulness of the findings for policy given that the SES dimensions and the child outcomes are currently of interest to policymakers.

5.6 Summary

This chapter examined whether different dimensions of family SES had unique trajectories and whether the SES trajectories were associated with children's language ability, internalizing behaviour, and externalizing behaviour and discussed the implications of the findings. In the next chapter, this study will be expanded by examining whether SES trajectories are associated with children's developmental trajectories.

Chapter 6 – Associations between family SES trajectories and children's language, internalizing, and externalizing trajectories

6.1 Aim of this chapter

This chapter aims to build upon the findings of chapter 5 by examining whether children have different developmental outcome trajectories and whether the likelihood of belonging to outcome trajectories varies by their family socioeconomic status (SES) trajectories. An introduction and a brief overview of the literature are presented followed by the methodology used to conduct this study. The findings are then presented, and their implications are discussed.

6.2 Introduction

Associations between SES and children's language ability, internalizing behaviour, and externalizing behaviour are well documented (Sirin, 2005; Letourneau et al., 2011; Cooper & Stewart, 2013; 2017), however recent research has become increasingly interested in the associations between different dynamics of SES and children's outcomes (Chen et al., 2007; Peres et al., 2007; Séguin et al., 2012; Björkenstam et al., 2016; Boe et al., 2016; Mok et al., 2018). A small body of literature has shown that different family income trajectories are differentially associated with children's language and behaviour outcomes (Dohoon, 2014; Jackson et al., 2017; Zachrisson and Dearing, 2015; Comeau et al., 2018; Hoyt et al., 2019). Chapter 5 expanded this literature by showing that trajectories of family income, parents' level of education, and parents' level of occupation from birth to late childhood are each associated with children's late childhood language ability, internalizing behaviour and externalizing behaviour. However, like SES dimensions, children's developmental outcomes are not static constructs and can also change over childhood forming their own trajectories (Elder, 2003). This can be seen in the child developmental literature that has identified several language ability and behavioural trajectories (Miller & Votruba-Drzal, 2016; Mangin et al., 2017; Yaari et al., 2018; Comeau, 2018; Gutman et al, 2019). Evidence indicates that these different language and behavioural trajectories are associated with children's future health, educational attainment, economic outcomes, cognitive functioning, and socioemotional functioning later in life (Reef et al., 2011; McKean et al., 2017; Bevilacqua et al., 2018; Zhu et al., 2019; Oerlemans et al., 2020; Jin et al., 2020; Ramírez-Luzuriaga et al., 2020; Casanova et al., 2021). Despite evidence of different family SES trajectories, and children's different developmental trajectories, no studies have examined whether the SES trajectories are differentially associated with children's language and behaviour trajectories. Since language and behavioural trajectories have long-lasting effects extending into adolescence and adulthood, it is important to understand whether membership to different SES trajectories is associated with children's membership to language and behaviour trajectories.

The previous trajectory literature examining the relationship between SES and children's language ability and behavioural outcomes takes one of two forms. First, as chapter 5 discussed, research has examined the relationship between SES trajectories and children's later life language and behaviour outcomes measured by a static outcome measure at a single time point (e.g., Kendzor et al., 2012; Zachrisson & Dearing, 2015; Jackson et al., 2017; Björkenstam et al., 2017; Miller & Votruba-Drzal, 2017). While this provides information about which SES trajectories are associated with better or worse outcomes, it does not differentiate between the nature of the outcome trajectories. This means that it is unclear

whether different SES trajectories are associated with stable, improved, or worsening outcome trajectories. Second, research has examined the associations between SES measured as a static construct at one time point and children's membership to outcome trajectories (Ukoumunne et al., 2012; Flouri & Sarmadi, 2015; Armstrong et al.,2017; Flouri & Midouhas, 2017; Miller & Votruba-Drzal, 2017; Hentges et al., 2019). Membership to different language development trajectories, for example, has been associated with static SES measures that were measured during earlier childhood (Ukoumunne et al., 2011; McKean et al, 2015; Armstrong et al., 2017; McKean et al, 2017; Hentges et al., 2019). Static measures of SES are also associated with membership to different internalizing and externalizing behaviour trajectories (Leve et al, 2005; Silver et al., 2005; Fanti & Henrich, 2010; Flouri & Sarmadi, 2015; Gutman et al, 2019; Lansford et al., 2019). This research shows that higher levels of SES are associated with better outcome trajectories but does not indicate whether children from different family SES trajectories are more likely to be in better or worse outcome trajectories.

Policy aims to reduce the inequality gaps in children's language and behaviour outcomes (The Childcare Act, 2006; The Education (Scotland) Act, 2016) by improving family income, education, and occupation (The Welfare Reform and Work Act, 2016). Examining the associations between family SES trajectories and children's outcome trajectories would allow policy makers to see whether policies increasing family SES increase the likelihood of children belonging to better outcome trajectories. Therefore, this study will address this gap in the literature by examining whether children's likelihood of belonging to different language ability, internalizing behaviour, and externalizing behaviour trajectories differs depending on their family income, parent education, and parent occupation trajectories.

6.2.1 Theoretical framework for understanding the role of family SES trajectories in children's developmental trajectories

Bioecological systems theory explains that child development takes place due to complex bidirectional interactions between children and their proximal and distal environments (Bronfenbrenner, 1986). Interactions in proximal environments (e.g., the family and home) are more influential for children's development because they are exposed to them most frequently (Bronfenbrenner, 1979; 1999; 2006). Family SES is associated with children's development because it can determine the nature of the environments that children interact with via several pathways (Conger et al., 2010). The pathways through which this process occurs are outlined by the family stress and investment models (Conger et al., 2010; Conger & Donnellan, 2007). Family stress model proposes that SES is associated with children's development because economic pressure leads to parental stress which causes increased conflict between parents, poor parenting practices, and poorer parent-child interactions. Evidence indicates each of these impaired pathways lead to poorer developmental outcomes (Kiernan & Manesh, 2009; 2011; Avan et al., 2010; Micklewright et al., 2012; Azzi-Lessing, 2013; Shelleby et al., 2014; Sun et al., 2015; Neppl et al., 2016; Sosu & Schmidt 2017; Ahun et al., 2017). Family investment model states that SES is associated with children's development because it determines parents' ability to invest in the resources required for good child development (e.g., nutritious food, educational materials and activities, safe environment) (Conger & Donnellan, 2007). Low-SES families are less able to invest as much in their children's development while high-SES families can invest more. Research indicates the more families invest in their children the better their language ability and behavioural outcomes are (Kalil, et al., 2012; Sohr-Preston et al., 2013; Prevoo et al., 2014; Sosu &

Schmidt, 2017; Longo et al., 2017; Jeong et al., 2017; Vasilyava et al., 2018; Fomby & Musick, 2018).

Life course theory explains that people belong to several different trajectories simultaneously across different domains other (e.g., family SES trajectories and children's developmental trajectories) which can be independent or associated with each (Elder, 1974; Elder, 2003). Family SES, children's language ability, and behaviour are not static constructs and can remain stable or continuously move up or down forming trajectories from birth to death (Elder, 2003). Life course accumulation of risk and social mobility models offer hypotheses about the relationship between family SES trajectories and children's developmental trajectories. Social mobility models hypothesise that the stable, upward, and downward trajectories will be mirrored in children's developmental trajectories such that children with upward SES trajectories are more likely to belong to better developmental trajectories, while children with downward SES trajectories are more likely to belong to worse developmental trajectories (Duncan, 1988; Lynch et al., 1994). The accumulation of risk models hypothesise that the longer children remain in a low SES trajectory, the more likely children will belong to poorer developmental trajectories because continued exposure to low SES produces a cumulative adverse effect (Kuh et al., 2003). The stress and investment models explain why the life course hypotheses are likely to be correct as the SES trajectories will reflect the longitudinal exposure to family stress and investment pathways. Parents from increasing SES trajectories will likely experience decreasing stress and have improved interactions with their children making it more likely that children will belong to positive outcome trajectories. Following the investment model, it is more likely that children will belong to positive outcome trajectories if they also belong to a high or increasing SES trajectory because their parents will be more able to invest in their children's development.

6.2.2 Evidence of SES associations with children's developmental outcome trajectories There is a growing body of literature examining children's developmental outcome trajectories. Studies examining children's language ability trajectories have identified between three and five trajectories (Landa et al., 2012; Ukoumunne et al., 2012; Brian et al., 2014; Song et al., 2015; Longard et al., 2017; McKean et al., 2017; Nguyen et al., 2018; Hentges et al., 2019; Jin et al., 2020; Riva et al., 2021). Studies normally find a variant of low-decreasing, stable low, stable mid, high-decreasing, and stable high language trajectories (Nguyen et al., 2018). Researchers have also identified between three and six internalizing behaviour trajectories, with most presenting a variant of stable low, stable high, increasing, and decreasing (Melchior et al., 2014; Korhonen et al., 2014; Hauser-Cram & Woodman; 2016; Miller and Votruba-Drzal, 2017; Gutman & McMaster, 2020; Kuang & Flouri, 2020; Tamura et al., 2020). Between three and five externalizing behaviour trajectories have identified and typically present a variant of stable low, stable high, stable mid, increasing, and decreasing trajectories in children (Thompson et al., 2011; Hauser-Cram & Woodman, 2016; Kjeldsen et al., 2016; Lee et al., 2016; Miller and Votruba-Drzal 2017; Gerstein et al., 2017; Olsen et al., 2017; Figge, 2018; Min et al., 2018; Perry et al., 2018; Gutman, 2019; Tamura et al., 2020).

Evidence shows that family SES dimensions are associated with children's developmental trajectories. Low family income is associated with children's membership to lower language trajectories than their higher income peers (McKean et al., 2017; Hentges et al., 2019). Children are also twice as likely to belong to a worsening language ability trajectory if their family is poor compared to their non-poor peers (Armstrong et al., 2017). Research has also

found that children are more likely to belong to low or decreasing language ability trajectories if their parents have low levels of education (Ukoumunne et al., 2012; Armstrong et al.,2017; Hentges et al., 2019) or spent less time in education (McKean et al., 2017). Regarding children's internalizing and externalizing behaviour trajectories, evidence shows that children from low income and deprived families are most likely to belong to the highest behavioural problem trajectories compared to their more affluent peers (Flouri & Sarmadi, 2015). Children who belong to poor families are more likely to belong to problematic behaviour trajectories than their non-poor peers (Flouri & Midouhas, 2017; Comeau, 2018) and those that experience intermittent poverty are most likely to belong to the middle internalizing and externalizing problem trajectory between highest and lowest behaviour trajectories (Comeau & Boyle, 2018). When families experience income loss or job loss, children are more likely to belong to a high problem or worsening behaviour trajectory (Flouri & Midouhas, 2017; Miller & Votruba-Drzal, 2017).

The above studies use static measures of SES from a single time point to examine their associations with children's developmental trajectories. Since SES is not static and has its own trajectories as found in chapter 5 (study 2) and previous literature (Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Hoyt et al., 2019), it raises the question of whether different SES trajectories are associated with children's different outcome trajectories. Addressing this question will provide useful information in terms of policy about whether short or long-term intervention and support increases children's chances of belonging to positive language and behavioural trajectories. If, for example, membership to the best outcome trajectories is most likely for stable high-income families, then it is unlikely that short term income support policies will help reduce the developmental inequality gap, but longer-term support increasing SES trajectories may be beneficial. The previous studies generally examine children's outcome trajectories during limited timeframes from early childhood (Landa et al., 2012; Ukoumunne et al., 2012; Brian et al., 2014; Song et al., 2015; Longard et al., 2017; Hentges et al., 2019), early to middle childhood (Flouri & Sarmadi, 2015; McKean et al., 2017; Armstrong et al., 2017; Flouri & Midouhas, 2017), or middle childhood to late childhood (e.g., Miller & Votruba-Drzal, 2017; Comeau & Boyle, 2018). Using short timeframes to generate trajectories is a limitation as it incorporates less data and may miss key stages of childhood. For instance, trajectories formed between middle and late childhood neglect early childhood which is a sensitive period (Kuh et al., 2003; Miller & Votruba-Drzal 2017). It would therefore be unknown whether children's outcomes improved, worsened, or were stable over throughout childhood. The more data that is included in the analysis, the more accurate the trajectories will be (Cavanagh et al., 2016; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Oshio et al., 2018).

6.2.3 The current study

This study aims to address the above gaps in existing SES trajectory literature by examining which language ability, internalizing behaviour, and externalizing behaviour trajectories children are most likely to belong to based on their family income, parent education, and parent occupation trajectories. Since chapter 5 found different income, education, and occupation trajectories had associations with children's language and behavioural outcomes, it is likely that those SES trajectories will also be associated with children's membership to language and behavioural trajectories (Elder, 2003). This will provide evidence about whether long-term support for improving family income, education, or occupation is likely to benefit children's developmental trajectories. This study will also use data from early through late childhood to form language and behaviour trajectories. This will allow policymakers to determine the impact of their SES policy decisions over a longer timeframe than is currently

available in the literature for language and behavioural outcomes. To achieve these aims the following research questions will be addressed:

1. To what extent do children belong to different developmental (language, internalizing and externalizing behaviour) trajectories?

2. To what extent does the likelihood of belonging to different developmental (language ability, internalizing and externalizing behaviour) trajectories differ by family SES trajectory (stable, increasing, or decreasing SES)?

6.3 Methodology

6.3.1 Sample

The data used in this study comes from the first six waves of the Millennium Cohort Study (MCS). The MCS is a longitudinal national cohort study in the UK that follows the lives of 18851 families with a child that was born in the UK between 2001 and 2002 when children were 9 months old (Fitzsimons, 2017). The data were collected when children were aged, 9 months, 3, 5, 7, 11, and 14 years old and the productive samples at each wave were 18,551, 15,590, 15,246, 13,857, 13,287, and 11,726 families (Ketende 2010; Mostafa, 2014; Fitzsimons, 2020). The sample was obtained from 398 electoral wards across the UK and families from disadvantaged and ethnic minority electoral wards were oversampled to ensure that adequate sample sizes for each stratum could be achieved while accounting for attrition (Plewis, 2007; Ipsos MORI, 2017). As discussed in chapter 3, the productive sample from wave 6 (N = 11,726) was used in this study to ensure that all families included had longitudinal weights assigned to them which were used to account for attrition. 947 cases were removed for having no valid information for any of the variables of interest at any data

collection wave and a further 96 cases were identified as having extreme outlier values. This process left a final analytic sample of 10,683 families. To account for panel attrition the non-response weights provided in the MCS were applied (Plewis, 2007; Mostafa, 2014) and missing data were handled using Full Information Maximum Likelihood estimation (Muthén & Muthén, 2017).

6.3.2 Measures

The dependent variables in this study were children's language ability, internalizing behaviour, and externalizing behaviour. Data on the dependent variables were obtained when children were three (MCS 2), five (MCS 3), seven (MCS 4), eleven (MCS 5), and fourteen (MCS 6) years old. The independent variables were family SES (i.e., family income, maternal education, and maternal occupation). The covariates used were age, gender, ethnicity, birth weight, single or dual parent household. Data on the independent variables and covariates apart from birth weight were available at all six data collection waves when children were 9 months old (MCS1), three (MCS 2), five (MCS 3), seven (MCS 4), eleven (MCS 5), and fourteen (MCS 6) years old. Birth weight was recorded at wave 1.

6.3.3 Language Ability

Children's language ability was measured by testing their vocabulary knowledge when they were 3, 5, 7, 11, and 14 years old (Moulton et al., 2020; McElroy et al 2021). Different measures were used to measure language ability as the children got older. At ages 3, 5, 7, and 11, British Ability Scale (BAS) subscales are used to measure language ability and at age 14 Applied Psychology Unit (APU) Vocabulary Test (Closs, 1976) was used. To allow meaningful comparison across all of the different language measures children's scores were standardised (e.g., Pereyra-Elias et al., 2022). For each British Ability Scale subscale (at 3, 5, 5, 5) and 5 a

7, and 11 years old) age-adjusted standardised T-scores are used to account for children's different ages when the tests were administered. Standardised T-scores were used at age 14 but were not age-adjusted as the Applied Psychology Unit Vocabulary test was a self-completion test conducted online when children were the same age. The following section provides a description of each language ability measure at age 3, 5, 7, 11, and 14.

6.3.4 Age 3 (MCS2) language ability

The BAS Naming Vocabulary subscale was used to measure children's language ability when they were 3 years old. The Naming Vocabulary subscale comprised 36 pictures of objects that children were asked to identify and name when shown. The pictures were numbered 1 through 36 with 1 being the easiest and 36 being the most difficult. Children were presented with a first tranche of 16 pictures and were only asked to go further if they made fewer than 3 mistakes. If the children progressed to the second tranche, then they saw items 17 through 30. If they made fewer than 3 mistakes, they were then presented with the third tranche containing items 31-36. The precise starting point for each child was dependent on their age and ability which meant that they were not presented with items too easy or difficult for them in order to protect their self-esteem and avoid frustration. This means that children's raw scores (summing the total correct responses) were not comparable. As a result, the raw scores were adjusted for item difficulty and age. Children scored a minimum of 0 and a maximum of 30 correct answers to the naming vocabulary test with a mean of 16.62 (SD=4.811). The MCS uses this data to generate T-scores for researchers to use. Previous analyses have found the measure to be reliable (alpha = .81; Elliott, 1997).

6.3.5 Age 5 (MCS3) language ability

The BAS Naming Vocabulary subscale was used again to measure language ability children were 5 years old, so the procedure was the same as above when children were age 3. Children scored a minimum of 0 and a maximum of 25 correct answers to the naming vocabulary test with a mean of 14.14 (SD=3.826).

6.3.6 Age 7 (MCS4) language ability

The BAS Word Reading subscale was used to measure children's language ability when they were 7 years old. The BAS word reading subscale assesses children's reading vocabulary knowledge between the ages of 5 and 17 years and 11 months, so was suitable for the current study when children were 7 years old. The child is presented with a series of words on a card. There is a total of 90 words divided into 9 blocks of 10 words. Each block of words increased in difficulty as the child progressed through blocks 1 to 10. Children were asked to read each word aloud with the correct pronunciation. The number of blocks the child was asked to attempt was dependent on their performance in the previous block (e.g., if the child struggled with their current block, they would not be asked to attempt the next more difficult block. Example words from the easiest block of words (block 1) are "box", "fish", and "jump" and from the most difficult block (block 10) "catastrophe", "regurgitate", and "archaic". Children's raw scores on the word reading task were calculated by summing the total number of correct responses ranging from 0-90 with a mean score of 44.4 (SD19.278). This scale has been shown to have good reliability in previous studies (Stuart, 1999; Hatcher, Hulme & Ellis, 1994).

6.3.7 Age 11 (MCS5) language ability

The BAS Word Similarities subscale assesses children's verbal vocabulary knowledge at age 11. There are a total of 21 items with each item containing 3 words that shared a common theme. The task was that upon hearing the 3 words, the child would be asked to provide a 4^{th} word that matched the theme and then state what the theme was. For example, the interviewer would say "apple, orange, banana" and the child could say "strawberry" as the 4^{th} item and "fruit" as the theme. Children only score one point if they provide both the correct word and the correct theme. Therefore, children could get a maximum of 21 points. The data show that the raw scores range from 0- 20 with a mean of 8.69 (SD=3.637).

6.3.8 Age 14 (MCS6) language ability

The Applied Psychology Unit Standardised Vocabulary Test was used to assess children's verbal vocabulary knowledge at age 14. The vocabulary test is designed to measure respondents understanding and meaning of words and is suitable for those aged 14 and over. The word activity involved presenting respondents with a list of 20 target words. For each target word, 5 additional words were presented. Children were asked to identify which of the 5 additional words had the same meaning as the target word. The words used were subsets of those used in the vocabulary test from the 1970 British Cohort Study which were originally from the standardised vocabulary test designed by the Applied Psychology Unit at the University of Edinburgh in 1976. Children were given 1 point for each correct answer and total scores were a sum of all correct answers meaning 20 was the maximum score. In the current sample, the scores ranged from 0-19 with a mean of 7.09 (SD= 2.625). Previous analyses have found the measure to be reliable (alpha = .83 - .95; Levy & Goldstein, 1984).

Previous studies using the abovementioned measures of language ability have found small to moderate correlations (r = .2 - .56) indicating a small to moderate degree of association

between the measures which provides confidence that the different measures are recording the same underlying construct (Girard et al., 2017; Hernández-Alava & Popli 2017; Forrest et al 2018; Toseeb & St Clair 2020; Kromydas et al.,2022; Pereyra-Elias et al., 2022). The correlation between the APU vocabulary test and the BAS naming vocabulary test is similar to the correlations observed between the BAS subscales at r = .36 (Hill, 2005; Girard et al., 2017; Sullivan et al., (2017). Correlations were run between the measures to ensure that the data used in this thesis had similar correlations with the literature. This was found to be the case as correlations ranged between .247 and .496 which is within the expected range (see table 6.1).

	Age 3 BAS	Age 5 BAS	Age 7 word	Age 11 word	Age 14
	Naming	Naming	reading task	similarities	vocabulary test
	Vocabulary	Vocabulary		task	
Age 3 BAS	-				
Naming					
Vocabulary					
Age 5 BAS	.496***	-			
Naming					
Vocabulary					
Age 7 word	.278***	.386***	-		
reading task					
Age 11 word	.271***	.378***	.358***	-	
similarities task					
Age 14	.247***	.317***	.410***	.354***	-
vocabulary test					
N	10683	10683	10683	10683	10683
М	49.971	49.894	49.767	49.431	49.248
SD	9.766	9.840	9.841	9.536	9.831

Table 6.1. Correlations between language ability measures at all data collection points

Note. Standardised T-scores used to generate these statistics. *** = p < .001.

6.3.9 Internalizing and externalizing behaviour

The Strength and Difficulties Questionnaire (SDQ; Goodman, 1997) was used to measure children's internalising and externalizing behaviours at ages 3, 5, 7, 11, and 14 years old and was completed by the parents. The SDQ contains 25 items measuring five dimensions of behavioural development in children aged between 3 and 16. The five dimensions are

emotion symptoms, conduct problems, hyperactivity, peer problems, and pro-social behaviour. Each item in the subscales is answered on a 3-point scale ("not true" = 0, "somewhat true" = 1, "certainly true" = 2). Total score on each subscale can range from 0-10. Goodman provides information and syntax for scoring the SDQ and deriving internalizing and externalizing behaviour scales on his website (Goodman: SDQ Info, 2023).

The internalizing behaviour scale is derived by summing the items from the emotional subscale ("often complains of headaches, stomach aches or sickness", "many worries, often seems worried", "often unhappy, down-hearted or tearful", "nervous or clingy in new situations, easily loses confidence", and "many fears, easily scared") and the peer problems subscale ("rather solitary, tends to play alone", "has at least one good friend", "generally liked by other children", "picked on or bullied by other children", and "gets on better with adults than with other children"). Normal internalizing scores range from 0-4, borderline problems scores range from 5-7, and abnormal scores range from 8-20 (Maurice-Stam et al., 2018). The externalizing behaviour scale is derived by summing the items from the hyperactivity subscale ("restless, overactive, cannot stay still for long", "constantly fidgeting or squirming", "thinks things out before acting", "sees tasks through to the end, good attention span", and "easily distracted, concentration wanders") and the conduct problems subscale ("often has temper tantrums or hot tempers", "generally obedient, does what adults request", "often fights with other children or bullies them", "can be spiteful to others", and "often argumentative with adults"). Normal externalizing scores range from 0-5, borderline scores range from 6-8, and abnormal scores range from 9-20 (Maurice-Stam et al 2018). The two-factor model (internalizing and externalizing behaviour) has been shown to have good model fit (CFI = .94; TLI = .94) (Sosu &Schmidt, 2017). There is good reliability for both the internalizing behaviour scale with α ranging from .73 - .82, and the externalizing behaviour

scale with α ranging from .78 - .83 (Palmieri & Smith 2007; Goodman et al., 2010; Hessel, et al., 2017). The convergent and discriminant validity of the internalizing and externalizing model has been found to be good (e.g., Goodman et al., 2010; Croft et al., 2015; Kersten et al., 2016) and the measurement invariance has shown good model fit (Sosu & Schmidt, 2017) indicating that the SDQ measures the intended constructs over time.

6.3.10 Measures of socio-economic status

Information on *family household income* was measured using net equivalised weekly income gathered at every data collection wave during the face-to-face interviews by asking respondents to state their take-home income. The average income and standard deviations are presented in table 6.2 below. Parents' *level of education* was measured at each time point by asking parents to indicate their highest level of academic or vocational education which was mapped onto the National Vocational Qualification (NVQ). The NVQ ranges from NVQ 1 (lowest level of qualification) – NVQ 5 (highest level of qualification). Parents' *occupational class* was measured at every data collection wave by asking parents to indicate their occupation which was mapped onto the 7-class National Statistics Socio-economic Classification (NS-SEC). The NS-SEC ranges from 1 (routine occupations) to 7 (higher managerial and professional occupations). Table 6.3 below shows the percentage of parents in each education and occupation category during early, middle, and late childhood.

6.3.11 Covariates

Several covariates were identified in the literature as being associated with family SES and children's outcomes so were included in the analyses. *Mother's age at birth* (Knipe 2016) was measured by asking mothers to state their age in years when they gave birth. The *number of parents living in the household* (Flouri et al., 2016) was specified by the main respondent

during the interview when asked whether they lived in a one or two-parent household. *Ethnicity* (Sullivan et al., 2021) is measured by asking the main respondent to indicate the child's ethnicity during the interviews. *Gender* (Mous et al., 2017) was measured by asking the main respondent to specify whether the child was male or female. Children's *birth weight* (Bromley, 2009) was measured by asking the main respondent for information during the interview. Table 4.3 below presents the descriptive statistics of each covariate.

6.3.12 Analytic strategy

The same family income, parent education, and parent occupation trajectories that were identified in study 2 were used in this study. Further latent class growth analysis (LCGA) models were used to examine whether different trajectories of language ability, internalizing behaviour, and externalizing behaviour could be found in the sample. This was done by using the language ability, internalizing behaviour, and externalizing behaviour data from all 6 data collection waves in the MCS (when children were 9 months, 3, 5, 7, 11, and 14 years old). A two-class model was run first so that each progressive model can be assessed to check if it offers a better fit over the previous model (Jung & Wickrama, 2008).

To determine how many trajectories there were for each SES dimension best fit the data, several goodness of fit indices were used (Jung & Wickrama, 2008; Maydeu-Olivares & Garcı´a-Forero, 2010). Akaike's information criterion (AIC; Akaike, 1974) and Bayesian information criterion (BIC; Schwarz, 1978) both assesses model fit based on the likelihood estimates within given parameters (Burnham & Anderson 2002). The AIC and BIC estimates reward a model for being simple and fitting the data well (Burnham & Anderson 2002; Nylund et al., 2007). The better a model fits the data the more it is rewarded but as the model becomes more complex with additional parameters the more it is penalised in the estimates

(Burnham & Anderson 2002). Lower AIC and BIC scores indicate better model fit. Entropy was assessed to determine the extent to which latent classes are 'separate' from each other (Weiss & Dardick, 2016). Entropy values range from 0-1 and the closer entropy is to 1 the more the model distinguishes between different classes. Entropy values >.8 indicate that individuals are assigned to different groups with a good level of accuracy (Celeux & Soromenho, 1996). Posterior probability values range from 0-1 and represent the accuracy of latent group classification with values closer to 1 indicating greater accuracy (Geiser, 2013). Likelihood ratio and bootstrapped likelihood ratio tests were assessed to determine whether the current model is a better fit than a model with one less class (Lo, Mendell, & Rubin, 2001; McLachlan & Peel, 2000). In both cases, if the *p*-value is significant (<.05) then it means that the current model is a significantly better fit than the previous model with one less class.

After the best fitting models for children's language ability, internalizing behaviour, and externalizing behaviour were determined using the above criteria, latent class probabilities were generated for each family and exported and merged with the income, education, and occupation probabilities from study 2. A series of multinomial logistic regressions were run to generate the predicted probabilities which were used to determine the likelihood of children belonging to language and behavioural trajectories based on their membership to SES trajectories. The predicted probabilities were plotted on graphs to improve the interpretability of the results.

6.4 Results

6.4.1 Descriptive statistics

Table 6.2 below presents the descriptive statistics of the analytic sample. Suggested cut-off points for internalizing behaviour are 0-4 for normal scores, 5-7 for borderline problems, and 8-20 for abnormal scores (Maurice-Stam et al., 2018). As seen in table 6.2 below, the majority of children fall within the normal range across childhood. Suggested cut-off points for externalizing behaviour are 0-5 for normal scores, 8-6 for borderline problems, and 9-20 for abnormal scores (Maurice-Stam et al 2018). Table 6.2 shows that children on average reduced from borderline externalizing behaviour problems to a normal range as they got older. This represents a decrease from borderline problematic scores to normal scores. Average family income continued to increase over time and the average level of education increased from A levels/ Highers to teaching qualification/degree. Little change was seen with parents' level of occupation which remained in the same category on average (small employers/own account workers). Table 6.3 below shows the percentage of parents in each education and occupation category during early, middle, and late childhood.

	М	SD	%	Range
Language ability				
3 years	50.418	9.812	-	20 - 80
5 years	50.544	9.525	-	20 - 80
7 years	50.523	9.840	-	20 - 80
11 years	50.412	9.662	-	20 - 80
14 years	50.022	10	-	20 - 80
Internalizing behaviour			-	
3 years	2.824	2.450	-	0-16
5 years	2.445	2.444	-	0 – 18
7 years	2.64	2.686	-	0-18
11 years	3.082	3.010	-	0 – 19
14 years	3.711	3.346	-	0 – 19
Externalizing behaviour				
3 years	6.549	4.021	-	0-20
5 years	4.614	3.333	-	0-20
7 years	4.551	3.484	-	0 – 19
11 years	4.294	3.436	-	0 – 19
14 years	4.323	3.506	-	0 - 19
Family income				
9 months	317.52	204.61	-	14.45 - 1282.61
3 years	343.89	223.37	-	14.39 - 1362.46
5 years	364.98	222.88	-	17.09 - 1282.94
7 years	398.86	232.38	-	14.93 - 1282.54

Table 6.2. Descriptive statistics of all variables in the analyses (n = 10683)

11 years	418.07	178.66	-	65.85 - 1162.80
14 years	417.44	178.07	-	80.02 - 1153.52
Parent's level of education				
9 months	2.884	1.403	-	0 - 5
3 years	2.929	1.406	-	0 - 5
5 years	2.909	1.4	-	0 - 5
7 years	3.041	1.387	-	0 - 5
11 years	3.181	1.388	-	0 - 5
14 years	3.263	1.4	-	0 - 5
Parent's level of occupation				
9 months	4.639	1.975	-	1-7
3 years	4.614	1.945	-	1-7
5 years	4,577	1.941	-	1-7
7 years	4.517	1.94	-	1-7
11 years	4.423	1.916	-	1-7
14 years	4.378	1.913	-	1-7
Mother's age at birth	29.11	5.751	-	14-51
Birth weight (lbs)	7.41	.586	-	.86 - 14.44
Parents in household				
Two parents			87.5	
One parent			12.5	
Gender				
Male			51.9	
Female			48.1	
Ethnicity				

White	86.41
Mixed	0.09
Indian	2.5
Pakistani & Bangladeshi	6.5
Black British	2.8
Chinese/ other	1.7

Child	Parent's level of	0/	Parent's level of	%
age	education	% 0	occupation	
3 Years				
	NVQ1	5.9	Routine	3.6
	NVQ2	25.2	Semi routine	18.7
	NVQ3	17.5	Lower supervisory	13.7
	NVQ4	42	Small employers	11.6
	NVQ5	9.4	Intermediate occupations	9.2
			Lower managerial and	22 0
			professional	22.9
			higher managerial and	20.4
			professional	20.4
7 Years				
	NVQ1	4.8	Routine	4.1
	NVQ2	23	Semi routine	20.3
	NVQ3	17.3	Lower supervisory	14
	NVQ4	41.7	Small employers	12.2
	NVQ5	13.2	Intermediate occupations	8.8
			Lower managerial and	
			professional	22.7
			higher managerial and	10
			professional	18
14 Years				
	NVQ1	4.8	Routine	4.6

Table 6.3. Parent's level of education and occupation at each time point (N=10,683)

	NVQ2	19.6	Semi routine	19.9	
	NVQ3	15.7	Lower supervisory	15.2	
	NVQ4	40.3	Small employers	15	
	NVQ5	19.7	Intermediate occupations	7.6	
	Lower managerial and professional		Lower managerial and	21.6	
			professional	21.0	
higher managerial and professional		higher managerial and	16 2		
		professional	10.2		

6.4.2 Trajectories of children's developmental outcomes

LCGA was run to determine the number of language ability, internalizing behaviour, and externalizing behaviour trajectories in the analytic sample (n = 10683) from 3 years to 14 years of age. A five-class model was selected as best fitting for all outcomes. The trajectories for language ability, internalizing behaviour, and externalizing behaviour are presented in figures 6.1, 6.2, and 6.3 below. The reason these models were selected over alternative models will now be discussed.

6.4.3 Trajectories of language ability

The five-class language ability model was selected over alternate models for the following reasons. The AIC and BIC were lower for the five-class than the four-class model and the entropy improved. The likelihood ratio test and the bootstrapped likelihood ratio test both indicated that the five-class model was a significant improvement over the four-class model (P<.001). A six-class model was run to test if it fits better than the five-class model, but the likelihood ratio test and bootstrapped likelihood ratio test were both non-significant

indicating that it was not a better model fit. Table 6.4 below presents a full breakdown of the model fit indices for two- to six-class models. The following language trajectories were identified in the five-class model: Always-high (n = 1130; 10.6%), always average (n = 6427; 60.16%), average-decreasing (n = 2197; 20.57%), low-high increasing (n = 640; 6%), always low (n = 287; 2.67%).



Figure 6.1. Five-class model of language ability trajectories

Note. Blue = Always-high. Black = Always-average. Pink = Average-decreasing. Green = lowhigh increasing. Red = Always-low.

Model	AIC	BIC	Entropy	%	Posterior	LMR-	BLRT
				Sample	values	LRT	
				in class			
Two-class	359731.929	359804.693	0.916	-	-	<.001	<.001
1				12.52	0.928		
2				87.48	0.984		
Three-class	356662.075	356756.668	0.709	-	-	<.001	<.001
1				42.6	0.836		
2				8.9	0.96		
3				48.5	0.856		
Four-class	355763.603	355879.629	0.734	-	-	<.01	<.01
1				8.4	0.966		
2				10.35	0.799		
3				20.92	0.788		
4				60.33	0.853		
Five-class	354986.687	355124.939	0.766	-	-	<.05	<.01
1				2.67	0.86		
2				10.6	0.807		
3				6	0.893		
4				20.57	0.793		
5				60.16	0.857		
Six-class	354820.359	354980.44		-	-	n.s.	n.s.
1				32.05	0.721		
2				6.1	0.896		

Table 6.4. Indicators of model fit for children's language ability

3	5.06	0.718
4	7	0.798
5	2.41	0.866
6	47.38	0.788

6.4.4 Trajectories of internalizing behaviour

The five-class internalizing behaviour model was selected over alternate models because the AIC and BIC were lower for the five-class than the four-class model and the entropy remained high. The likelihood ratio test and the bootstrapped likelihood ratio test both indicated that the five-class model was a significant improvement over the four-class model (P<.001). A six-class model was run, and the likelihood ratio test and bootstrapped likelihood ratio test indicated that it was a better model fit than the five-class model. The trajectory plots were then examined to determine which model was most appropriate and found the six-class model had two 'always-low' trajectories. Including the extra trajectory would not benefit interpretation and add would add unnecessary complexity to the model. The five-class model was selected as it identified the same trajectories as the six-class model without duplicating the 'always-low' trajectory. Table 6.5 below presents a full breakdown of the model fit indices for two- to six-class model: Always-high (n = 225; 2.11%), borderline-low decreasing (n = 1444; 13.52%), low-high increasing (n =; 581; 5.45%), low-borderline increasing (n =; 1867; 17.48%), always low (n = 6560; 61.43%).


Figure 6.2. Five-class model of internalizing behaviour trajectories

Note. Black = Always-high. Pink = Low-high increasing. Red = Borderline-low decreasing.

Blue = Low-borderline increasing. Green = Always-low.

Model	AIC	BIC	Entropy	%	Posterior	LMR-	BLRT
				Sample	values	LRT	
				in class			
Two-class	231218.744	231291.505	0.846	-	-	<.001	<.001
1				76.66	0.967		
2				23.34	0.915		
Three-class	228697.699	228792.287	0.815	-	-	<.001	<.001
1				6.64	0.903		
2				28.55	0.855		
3				64.81	0.94		
Four-class	227153.193	227269.61	0.818	-	-	<.001	<.001
1				5.61	0.896		
2				16.12	0.794		
3				64.82	0.943		
4				13.45	0.802		
Five-class	226340.445	226478.689	0.815	-	-	<.001	<.001
1				13.52	0.814		
2				17.49	0.759		
3				61.43	0.93		
4				5.45	0.853		
5				2.11	0.87		
Six-class	225839.162	225999.235	0.79	-	-	<.05	<.01
1				7.07	0.792		
2				14.34	0.698		

Table 6.5 Indicators of model fit for children's internalizing behaviour

3	56.92	0.916	
4	15.5	0.76	
5	1.46	0.885	
6	4.71	0.849	

6.4.5 Trajectories of externalizing behaviour

The five-class externalizing behaviour model was selected over alternate models as the AIC and BIC were lower for the four-class model and the entropy improved. The likelihood ratio test and the bootstrapped likelihood ratio test both indicated that the five-class model was a significant improvement over the four-class model (P<.001). A six-class model was run to test if it fits better than the five-class model, but the likelihood ratio test and bootstrapped likelihood ratio test were both non-significant indicating that it was not a better model fit. Table 6.6 below presents a full breakdown of the model fit indices for two- to six-class models. The following externalizing behaviour trajectories were identified in the five-class model: Always-high (n = 441; 4.13%), high-borderline decreasing (n = 1250; 11.71%), borderline-high increasing (n = 847; 7.93%), borderline-low decreasing (n = 3808; 35.67%), and always low (n = 4331; 40.56%).



Figure 6.3. Five-class model of externalizing behaviour trajectories

Note. Blue = Always-high. Red = High-borderline decreasing. Pink = Borderline-high increasing. Black = Borderline-low decreasing. Green = Always-low.

Model	AIC	BIC	Entropy	%	Posterior	LMR-	BLRT
				Sample	values	LRT	
				in class			
Two-class	251876.493	251949.253	0.854	-	-	<.001	<.001
1				70.13	0.966		
2				29.87	0.934		
Three-class	247360.903	247455.492	0.814	-	-	<.001	<.001
1				52.81	0.936		
2				35.7	0.879		
3				11.49	0.915		
Four-class	245718.736	245835.153	0.786	-	-	<.001	<.001
1				35.79	0.833		
2				4.22	0.926		
3				41.72	0.903		
4				18.28	0.875		
Five-class	244567.472	244705.717	0.793	-	-	<.001	<.001
1				11.71	0.812		
2				4.13	0.93		
3				40.56	0.905		
4				7.93	0.811		
5				35.67	0.829		
Six-class	244169.359	244329.432	0.745	-	-	n.s.	n.s
1				6.92	0.799		
2				32.76	0.768		

Table 6.6 Indicators of model fit for children's externalizing behaviour

3	34.46	0.876
4	14.36	0.721
5	3.37	0.929
6	8.13	0.798

6.4.6 Probabilities of children's membership to outcome trajectories

Multinomial logistic regression was performed to generate the predicted probabilities of children belonging to different trajectories of language ability, internalizing behaviour, and externalizing behaviour based on their family income, parent education, and parent occupation trajectories. The figures in the following sections present the likelihood of children belonging to each outcome trajectory. The findings discuss which income, education, and occupation trajectories were most likely to lead to the best and worst child outcome trajectories, whether those who improve their SES were more likely to belong to positive outcome trajectories than those who do not improve their SES, and whether those who did not decrease their SES.

6.4.7 Probability of belonging to outcome trajectories based on family income

Chapter 5 (study 2) found the following income trajectories which will be used in this study: Very high-decreasing (n = 290; 2.77%), stable-high (n = 828; 7.72%), high-decreasing (n = 450; 4.53%), average-increasing (n = 1666; 15.68%), low-increasing (n = 3043; 28.25%), and stable-low (n = 4406; 41.05%). The probability of children belonging to language ability trajectories based on family income trajectories is presented in figure 6.4 below. Children from the very high-decreasing income group and the stable-high income group showed similar probabilities of belonging to each language trajectory. Children from stable-lowincome families are most likely to belong to the always-low language trajectory, averagedecreasing language trajectory, and increasing low-high language trajectory. They are also least likely to belong to the always-high language trajectory. Children with stable high income were 9.3 times more likely to belong to the always-high language trajectory than those with stable-low income.

Decreasing income was shown to have a negative impact on language trajectories as compared to those with stable-high-income, children from the high-decreasing income group were 1.5 times less likely to belong to the always-high language trajectory and 3.1 times more likely to belong to the average-decreasing language trajectory. Increasing income was shown to have a positive association with language trajectories as those in the low-increasing income trajectory are 3.4 times more likely to be in the always high language trajectory than those who remain in the stable-low income trajectory. Similarly, they are 5.5 times less likely to belong to the always-low language trajectory.



Figure 6.4. Children's likelihood of belonging to language trajectories based on family income trajectories

The probability of children belonging to internalizing behaviour trajectories based on family income trajectories is presented in figure 6.5 below. Belonging to the always-low internalizing trajectory is the most desirable trajectory as it represents the least problems. Children from the decreasing very high-decreasing income group and the stable-high income group showed similar probabilities of belonging to each internalizing trajectory. Children with stable-low income are most likely to belong to the always-high internalizing problems trajectory, increasing internalizing problem trajectories, and the decreasing borderline-low internalizing trajectory. They are also the least likely to belong to the always-low internalizing trajectory. Children with stable-high income were 1.7 times more likely to belong to the always-low internalizing problems trajectory than those with stable-low income.

Decreasing income was shown to have a negative impact on internalizing trajectories as compared to those with stable-high income, children in the high-decreasing income group were 1.1 times less likely to belong to the always-low internalizing trajectory. Increasing income was shown to have a positive association with internalizing behaviour trajectories as those in the low-increasing income trajectory are 1.4 times more likely to be in the always-low internalizing problems trajectory than those who remain in the stable-low income trajectory. Similarly, they are 2 times less likely to belong to the always-high internalizing trajectory.



Figure 6.5. Children's likelihood of belonging to internalizing behaviour trajectories based on family income trajectories

A similar pattern was found for externalizing behaviour. The probability of children belonging to internalizing behaviour trajectories based on family income trajectories is presented in figure 6.6 below. Children from the decreasing very high-decreasing income group and the stable-high income group showed similar probabilities of belonging to each externalizing trajectory. Children with stable-low-income are most likely to belong to the always-high externalizing trajectory, increasingly problematic externalizing trajectories, and the decreasing from problematic to low externalizing trajectory. They are also the least likely to belong to the always-low externalizing trajectory. Children with stable-high income were 2.2 times more likely to belong to the always-low externalizing trajectory than those with stable-low income.

Decreasing income was shown to have a negative impact on externalizing trajectories as compared to the stable-high-income group, those in the high-decreasing income group were 1.2 times less likely to belong to the always-low externalizing trajectory. Increasing income was shown to have a positive association with externalizing behaviour trajectories as those in the low-increasing income trajectory are 1.6 times more likely to be in the always-low externalizing problems trajectory than those who remain in the stable-low income trajectory. Similarly, they are 28 times less likely to belong to the always-high internalizing trajectory.



Figure 6.6. Children's likelihood of belonging to externalizing behaviour trajectories based on family income trajectories

6.4.8 Probability of belonging outcome trajectories based on parent education

Chapter 5 (study 2) found the following income trajectories which will be used in this study: Stable high (n = 3912, 36.71%), stable average (n = 1644, 15.52%), stable low (n = 2492, 23.3%), no qualifications (n = 1313, 12.25%), increasing no qualification-low (n = 829, 7.86%), and increasing low-high (n = 433, 4.36%). The probability of children belonging to language ability trajectories based on parents' education trajectories is presented in figure 6.7 below. The stable-high education group were most likely to be in the always-high language trajectory and least likely to be in the always-low language trajectory. The no qualifications group were most likely to belong to the always-low language trajectory, low-high-increasing language trajectory, and average-decreasing language trajectory. They were also 11.5 times more likely than the stable-high education group to belong to the always-low language trajectory. Increasing education was shown to have positive associations with children's language ability as compared to no qualifications, those in the increasing form no qualification-low education trajectory were 2.75 times less likely to be in the always-low language trajectory and 5.2 times more likely to be in the always-high language trajectory. Those in the no qualifications group were 1.1 times more likely to be in the average-decreasing language trajectory compared to those in the increasing from no qualification-low education group. Similarly, compared to stable-low education, those who increased from low-high education were 1.5 times less likely to be in the always-low language trajectory and 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be in the always-low education group were 1.25 times more likely to be long to the average-decreasing language trajectory compared to those in the increasing low-high education group.



Figure 6.7. Children's likelihood of belonging to language trajectories based on parent's education trajectories

The probability of children belonging to internalizing behaviour trajectories based on parents' education trajectories is presented in figure 6.8 below. The stable-high education group were most likely to be in the always-low internalizing trajectory and least likely to be in the always-high internalizing trajectory. The no qualifications group were most likely to belong to the always-high internalizing problems trajectory, increasing internalizing trajectories, and borderline-low-decreasing internalizing trajectory. Compared to stable-high education, those with no qualifications were 1.7 times less likely to be in the always-low internalizing trajectory. They are also 1.5 times more likely to belong to the increasing low-borderline internalizing trajectory, and 2.75 times more likely to belong to the increasing low-high internalizing trajectory.

Increasing education was shown to have a positive association with children's internalizing behaviour as compared to the no qualifications group, those who increased from no qualification-low education were 1.2 times more likely to be in the always-low internalizing trajectory, and 1.6 times less likely to be in the always-high internalizing trajectory. The increasing no qualifications-low education group were 1.3 times more likely to be in the decreasing borderline-low internalizing trajectory than those with no qualifications. They were both equally likely to belong to the increasing low-borderline internalizing trajectory and increasing low-high internalizing trajectory. Compared to stable-low education, those increasing low-high education were 1.6 times less likely to belong to the increasing low to the increasing low-borderline internalizing trajectory and 1.2 times less likely to belong to all other internalizing trajectory.



Figure 6.8. Children's likelihood of belonging to internalizing behaviour trajectories based on parent's education trajectories

The probability of children belonging to externalizing behaviour trajectories based on parents' education trajectories is presented in figure 6.9 below. Those in the stable-high education trajectory were most likely to be in the always-low externalizing trajectory and least likely to be in the always-high externalizing trajectory. Both the no qualification trajectory and the increasing from no qualification-low education trajectory were almost equally likely to belong to all externalizing behaviour trajectories. Compared to stable-high education, those with no qualification group were 2.2 times less likely to be in the always-low externalizing trajectory, and 5.8 times more likely to be in the always-high externalizing trajectory. They are also 1.5 times more likely to belong to the increasing low-borderline externalizing trajectory, 2.75 times more likely to belong to the increasing low-high

externalizing trajectory, and 2.8 times more likely to belong to the decreasing borderline-low externalizing trajectory.

Compared to stable-low education, those increasing from low-high education were 1.2 times more likely to be in the always-low externalizing trajectory and the increasing borderlinehigh externalizing trajectory. They were also 1.2 times less likely to belong to the alwayshigh externalizing trajectory, and 1.15 times less likely to belong to the increasing borderlinehigh externalizing trajectory and the decreasing borderline-low externalizing trajectory.



Figure 1.9. Children's likelihood of belonging to externalizing behaviour trajectories based

on parent's education trajectories

6.4.9 Probability of belonging outcome trajectories based on parent occupation

Chapter 5 (study 2) found the following income trajectories which will be used in this study: Decreasing very high-high (n = 2312, 22.82%), decreasing high-intermediate (n = 2777, 27.92%), stable-intermediate (n = 1944, 18.72%), stable-average (n = 1317, 13.33%), and increasing routine-semi-routine (n = 1740, 17.21%). The probabilities of children belonging to language ability trajectories based on parents' occupation trajectories are presented in figure 6.10 below. The decreasing from very high-high occupation group were most likely to belong to the always-high language trajectory and least likely to be in the always-low language trajectory. The increasing routine-semi-routine occupation group were most likely to belong to the always-low language trajectory, increasing low-high language trajectory, and average-decreasing language trajectory.

Those in the decreasing from very high-high occupation group occupation were 5.4 times more likely to be in the aways-high language ability trajectory and 5.5 times less likely to be in the always low language trajectory compared to those in the increasing routine-semi-routine occupation group. Decreasing occupation was shown to have a negative association with children's language ability as those in the decreasing high-intermediate trajectory were 1.78 times less likely to be in the always high language trajectory and 1.4 times more likely to be in the always low language trajectory compared to the decreasing very high-high occupation group.



Figure 6.10. Children's likelihood of belonging to language trajectories based on parent's occupation trajectories

The probability of children belonging to internalizing behaviour trajectories based on parents' occupation trajectories is presented in figure 6.11 below. The decreasing from very high-high occupation group was most likely to belong to the always-low internalizing trajectory and least likely to belong to the always-high internalizing trajectory. The increasing from routine-semi-routine occupation group was most likely to belong to the always-high internalizing trajectory. The increasing from routine-semi-routine occupation group was most likely to belong to the always-high internalizing trajectory.

Those in the decreasing from very high-high occupation group occupation were 1.38 times more likely to be in the aways-low internalizing trajectory and 4.7 times less likely to be in the always-high internalizing trajectory compared to those in the increasing routine-semiroutine occupation group. Decreasing occupation was shown to have a negative association with children's internalizing behaviour as those in the decreasing high-intermediate trajectory were 1.1 times less likely to be in the always-low internalizing trajectory and 2.6 times more likely to be in the always-high internalizing trajectory compared to the decreasing very high-high occupation group.



Figure 6.11. Children's likelihood of belonging to internalizing behaviour trajectories based on parent's occupation trajectories

The probability of children belonging to externalizing behaviour trajectories based on parents' occupation trajectories is presented in figure 6.12 below.

The decreasing from very high-high occupation group was most likely to belong to the always-low externalizing trajectory and least likely to belong to the always-high externalizing trajectory. The increasing from routine-semi-routine occupation group was most likely to belong to the always-high externalizing trajectory and least likely to be in the always-low externalizing trajectory.

Those in the decreasing from very high-high occupation group occupation were 1.83 times more likely to be in the aways-low externalizing trajectory and 3.87 times less likely to be in the always-high externalizing trajectory compared to those in the increasing routine-semi-routine occupation group. Decreasing occupation was shown to have a negative association with children's externalizing behaviour as those in the decreasing high-intermediate trajectory were 1.22 times less likely to be in the always-low externalizing trajectory and twice as likely to be in the always-high externalizing trajectory compared to the decreasing very high-high occupation group.



Figure 6.12. Children's likelihood of belonging to externalizing behaviour trajectories based on parent's occupation trajectories

6.5 Discussion

This study examined whether children followed different language ability, internalizing behaviour, and externalizing behaviour trajectories from age 3 to 14 years old and whether membership to these outcome trajectories was more or less likely depending on their family's income trajectory, parent's education trajectory, and parent's occupation trajectory. As in previous studies (e.g., Ukoumunne et al., 2012; Weeks et al., 2014; Miller and Votruba-Drzal, 2017), five trajectories of language ability (Always-high, always average, average-decreasing, low-high increasing, and always low), internalizing behaviour (Always-high, borderline-low decreasing, low-high increasing, low-borderline increasing, always low), and externalizing behaviour (always-high, high-borderline decreasing, borderline-high increasing, borderline-low decreasing, and always low) were found. In addition to finding the same number of trajectories as previous studies, the trajectories were also characterised similarly for both internalizing and externalizing trajectories (Weeks et al., 2014; Miller & Votruba-Drzal, 2017).

The characteristics of the language trajectories found in this study are also somewhat similar to previous studies but with some differences. Research by Ukoumunne et al., (2012) found three similar trajectories as the current study: low-increasing, average decreasing, stable average. However, they also found average-increasing and high-decreasing language trajectories which this study did not find. Instead, this study found an always-high trajectory and an average-decreasing trajectory. Two methodological reasons could explain the different trajectories between studies. First, Ukoumunne et al., (2012) focussed on early childhood measuring language ability at 8 months, 2 years, 3 years, and 4 years. The current study measured language ability from early childhood through to late childhood meaning that more language ability data were used to produce trajectories. This additional data over a longer

time period is more likely to capture change over time (Cavanagh et al., 2016; Oshio et al., 2018) and could explain why different trajectories were found between the studies. Second, Ukoumunne et al., (2012) used parent-reported language ability (e.g., asking parents about children's vocabulary and whether they know certain words) at 8 months, 2 years and 3 years and then administered a language test to children at 4 years old. The current study administered language tests at all time points rather than using parent-reported language ability. The difference between objective and subjective measures might explain why this study found different trajectories.

In addition to examining the extent to which children belonged to different outcome trajectories, this study found that the likelihood of children belonging to each outcome trajectory was different depending on their family SES trajectories. Regarding family income, children were most likely to belong to the best language and behaviour trajectories if they belonged to the highest income trajectories, while children were most likely to belong to the worst language and behaviour trajectories if they belonged to the lowest income trajectory. Compared to families with stable-average income, children from families that increased their income were more likely to be in the always-low internalizing and externalizing trajectories, and the always-high language ability trajectory. Children that remain stable-average income were almost twice as likely to experience a decrease in their language ability over time. On the other hand, compared to families with stable-high income, children from families that experience a decrease in their income were less likely to be in the always-low internalizing and externalizing trajectories and less likely to be in the always-low internalizing trajectory. They are also more likely to experience a decrease in their language ability trajectory. They are also more likely to experience a decrease in their language ability and an increase in their internalizing and externalizing behaviour problems.

While these findings are consistent with previous studies in that higher income trajectories have better associations with children's outcomes than lower income trajectories (e.g., Miller & Votruba-Drzal 2017; Björkenstam et al., 2017), this study provides additional unique information. The findings show that improving family income is likely to have a positive impact on children's language and behavioural trajectories throughout the entirety of childhood rather than only having a better association with late childhood outcomes as presented in previous literature (Zachrisson & Dearing, 2015; Jackson et al., 2017; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Oshio et al., 2018). This provides context about how income trajectories are likely to impact children's developmental experiences over the life course and the direction of their developmental trajectories as opposed to the result at a later date. This is useful because income trajectories may have better, worse, or similar associations with children's later outcomes, but this does not explain the direction that their outcomes are likely heading in. For example, Jackson et als. (2017) study found that increasing and decreasing income trajectories both similar associations with children's language ability while Comeua and Boyle (2018) found that their moving into and out of poverty trajectories had similar associations with children's internalizing and externalizing outcomes. However, although the associations that the increasing and decreasing trajectories had with children's outcomes were similar, it does not mean they are equally beneficial as one could be on a decreasing outcome trajectory and the other in an increasing outcome trajectory. Results of the current study, therefore, add insight into the likely direction of children's outcomes.

Regarding parents' education, results found that increasing parents' education trajectories had the most benefit for children's language ability. Children of parents that increased their education from no qualifications to low-education or from low-high education were more

likely to be in the always-high language ability trajectory and less likely to decrease their language ability compared to those that did not increase their level of education. While children were also more likely to belong to the always-low internalizing and externalizing trajectories if their parents increased their level of education, they were also more likely to belong to an increasing behaviour problem trajectory. Specifically, compared to stable-low education, children of parents that increased from low-high education were more likely to belong to the increasing from low-borderline internalizing trajectory and the increasing borderline-high externalizing trajectory. In addition, children were most likely to belong to the always-high externalizing trajectory if their parents increased from no qualifications to low-education.

Previous research has shown that higher levels of education are associated with better language and behavioural outcomes (e.g., Magnuson et al., 2009; Harding, 2015; Monaghan, 2017), and that parent education is more beneficial for children's cognitive outcomes than behavioural outcome (Coddington et al., 2014; Gonzáleza et al., 2018). Chapter 5 (study 2) also indicated that compared to parents that did not improve their education, parents from increasing education trajectories had better associations with children's language and behavioural outcomes during late childhood. However, this study adds further insight showing that while increasing parents' education increases children's odds of belonging to positive outcome trajectories. One possible explanation for this finding may be that parents' level of stress has increased due to their studies. Research shows that education can be particularly stressful for parents as they have to balance childcare with education and they have more financial worries due to being in education (Lin, 2016). The family stress model

indicates that increased parental stress can impair child-parent interactions and ultimately impact their developmental outcomes (Conger et al., 2010).

Regarding parents' occupation, the results showed that children were most likely to be in the always-high language trajectory and always-low internalizing and externalizing trajectories if their parent's belonged to the decreasing very high-high occupation trajectory which was the highest level occupation trajectory found in this study. They were also most likely to be in the always-low language trajectory and always-high internalizing and externalizing behaviour trajectories if their parents were in the increasing routine-semi-routine occupation (the lowest occupation trajectory found in this study). Since no 'stable-routine' occupation trajectory was found in the analysis it is not possible to determine whether increasing from routine to semi-routine occupations is more beneficial than staying in a routine occupation.

The findings also showed that children were less likely to be in positive language and behavioural trajectories if their parents were in the decreasing high-intermediate occupation trajectory compared to the decreasing very high-high occupation trajectory. It is possible that children are more likely to be in poorer language and behavioural trajectories if their parents' occupations declined as it would cause stress in the household (Bradley, 2016). Research looking at work-related family stress shows that demanding job requirements such as moving jobs, the difficulty of work, and working long hours are associated with increased parental stress, mental health problems, poor parenting practices, and increased family conflict within the household (Brooks-Gunn, Han, & Waldfogel, 2010; Hadzic et al, 2013; Dockery et al., 2016; Cho, 2018; Moreira et al., 2019). As stated by the family stress model, parental stress, poor parenting, and family conflict are all likely to negatively impact children's developmental outcomes (Conger et al., 2010). Indeed, Moreira

et al (2019) found that work-family conflict was associated with parental anxiety, depression, and parenting stress which all had negative associations with parents' emotional awareness of their child, compassion for their child, and attention paid to their child, and each of these pathways was associated with children's externalizing (Johnson, Li, Kendall, Strazdins, & Jacoby, 2013) and internalizing (Dockery et al., 2016) behaviour problems.

6.5.1 Limitations of the study

The limitations of this study should be considered when interpreting the results. The family SES trajectories, language trajectories, and behavioural trajectories examined in this study were selected based on the result of latent class growth analysis. While this approach provides model fit indices to help researcher determine the appropriate number of latent classes in a sample, they are only guidelines and ultimately the final decision on the number of latent classes is at the researcher's discretion depending on the research questions, theory, or measures used (Jung & Wickrama, 2008). This means that the trajectories identified between studies may be different making it difficult to compare findings across studies (Demment et al., 2014). However, the language and behaviour trajectories found in this study are similar to those identified in previous literature indicating that the trajectories seem to be consistent across different samples, particularly for internalizing and externalizing behaviour (Ukoumunne et al., 2012; Weeks et al., 2014; Miller and Votruba-Drzal, 2017). Sample attrition over time is another limitation most longitudinal studies have to face. Attrition in the millennium cohort study was more likely to occur in low-income families (Plewis, 2007). This could potentially lead to a biased sample resulting in the SES trajectories and outcome trajectories identified in this study being less likely to represent lower income families. However, the sample design of the millennium cohort study meant that deprived families

were oversampled, and attrition weights were generated and applied to account for potential bias.

6.5.2 Implications of the study

Despite the limitations of this study, the findings have useful implications for improving the likelihood that children will belong to positive developmental trajectories. Examining associations between SES trajectories and children's developmental trajectories from early through late childhood provides information about the likely long-term impact that improving SES will have on children's development. Policies focused on increasing family income are likely to be beneficial for children's language ability, internalizing behaviour and externalizing behaviour as they are more likely to belong to more favourable outcome trajectories than those who do not increase their income. Increasing family income also reduces the likelihood of children belonging to a worsening outcome trajectory. However, policies increasing family income are likely to be most beneficial if they can be sustained across the entire childhood timeline rather than over a shorter period of time. This is because when family income decreases, children are more likely to belong to a decreasing language ability trajectory and an increasing internalizing and externalizing problem trajectory. This may be relevant for policies such as the Scottish Child Payment which is only available until the child turns 6 years old, after which the family may experience a decrease in income which could negatively impact children's developmental trajectories. Taken together it would suggest that in addition to increasing family income over childhood policymakers should also focus on implementing measures to prevent family income loss during later childhood.

Policy increasing parents' level of education is also likely to improve children's outcome trajectories. Helping parents with no qualifications gain a qualification increases the

likelihood that children will have high language ability by 148%, low internalizing problems by 16.32%, and low externalizing problems by 4.31%. Further increasing education from low-high increases the likelihood that children will have high language ability by 25.19%, low internalizing problems by 3.6%, and low externalizing problems by 19.37%. However, improving parents' education from low-high also increased the risk of children belonging to worsening outcome trajectories, so it may be prudent to examine why this happens before policy is implemented. Policy helping parents with no qualifications increase their level of education to NVQ 1 level may be the most practical approach based on the current study as this increased the likelihood of children belonging to positive language and behaviour trajectories without increasing their odds of belonging to worsening trajectories. Gaining an NVQ 1 qualification can typically be achieved in one academic year meaning the period of potential stress and financial pressure are shorter and the benefits for children's outcome trajectories can be seen sooner.

This study highlighted that children's membership to different language and behaviour trajectories differ depending on their family income, parent education, and parent occupation trajectories. While the findings regarding family income are rather clear, more research is required to better understand how parent education and occupation trajectories operate before making policy decisions. To conclude, this study built upon previous research on the associations between SES trajectories and children's language and behaviour outcomes by examining whether children's outcome trajectories are affected by their family income, parents' education, and parents' occupation trajectories. It presents novel findings that children's developmental trajectories are impacted by their family SES trajectories. It highlights that policy improving family income and helping parents with no qualifications gain a qualification is likely to improve children's developmental trajectories.

6.6 Summary

This chapter examined whether different family SES trajectories made children more or less likely to belong to language, internalizing, and externalizing trajectories and discussed the implications of the findings. The next chapter will be a general discussion of the thesis as a whole.

Chapter 7 – General Discussion

7.1 Aim of this chapter

This chapter will begin by giving a recap of the aims of this thesis before discussing how the aims were achieved. An overview of the findings from the three studies will be presented and a discussion about what they mean in terms of previous research. Limitations will be highlighted, and the policy implications of the findings will be discussed. Finally, a summary of the thesis as a whole will be provided.

7.2 Aim of this thesis

The overall aim of this thesis was to examine whether different dynamics of family socioeconomic status (SES) are associated with children's language ability, internalizing behaviour, and externalizing behaviour. The dynamics of SES refers to the change, movement, or stability of SES over time and is a useful area of research as it can provide an understanding of people's SES over the life course while static research can only provide information about SES during specific points in time. SES dynamics research is beneficial as it can help untangle some of the complexities involved in SES research such as how the timing, persistence, and transitions, of SES, are associated with specific outcomes which help policymakers understand when policy intervention might be useful and the likely long-term impact the interventions will have (Miller et al., 2021). This study examined two types of SES dynamics, the stage of childhood (timing) and trajectories of SES.

This study focused on the timing of SES because evidence indicates that children are sensitive to the negative effects of low SES during different stages of childhood (Blakemore, 2014), but most studies only examine short time frames such as early-middle childhood (e.g.,

Schoon et al., 2011; 2013; Burchinal et al., 2018) or mid-late childhood (Lai et al., 2019) which makes it difficult to assess at which stage SES is most important. This study addressed this gap in the literature by examining the associations between SES and children's language ability, internalizing behaviour, and externalizing behaviour during early, middle, and late childhood. The findings from this study could help policymakers understand the effect that current policies have which are primarily targeted at early childhood (e.g., Best Start) and indicate whether policies could benefit children at different stages of childhood.

This study also focused on the trajectories of SES because SES is not a static construct and people can transition between different social circumstances over their life course to form SES trajectories (Elder, 2003). Since trajectories encompass people's experiences and life events, they can better represent people's SES compared to static SES since the more data points there are the more accurate the trajectories become (Cavanagh et al., 2016; Oshio et al., 2018).

However, current literature tends to generate SES trajectories over short timeframes such as during early childhood (e.g., Dearing et al., 2001; Zachrisson & Dearing, 2015; Jackson et al., 2017) or from mid-childhood to early adolescence (e.g., Miller & Votruba-Drzal 2017; Comeau & Boyle, 2018). This means families are classified into SES trajectories based on short periods of time. Given that trajectories are created over the life course and influenced by life events and life changes (Elder, 2003), it is likely that more SES variation will be captured over a longer duration which could identify additional trajectories. This would be beneficial as it would give insight into the different trajectories that exist over the entire childhood timeline rather than specific stages of childhood only. This would also give longterm insight into the likely efficacy of policy long-term. Similarly, children's developmental outcomes are not static constructs and have their own trajectories over their life course (Elder, 2003). Since SES trajectories are associated with language ability, internalizing behaviour, and externalizing behaviour (Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Hoyt et al., 2019) it is likely that they would also be associated with trajectories of language ability, internalizing behaviour, and externalizing behaviour. Examining this provides policymakers with information about what is likely to happen to children's language and behavioural outcomes over their life course if they belong to different SES trajectories. For example, if belonging to a decreasing income trajectory increases the likelihood of belonging to a decreasing language ability trajectory then policymakers may consider policies to prevent income loss.

This thesis defined SES as a multidimensional construct comprising of family income, parent's level of education, and parent's level of occupation which is the most widely used definition of SES (Bradley & Corwyn, 2002; APA, 2007) and is consistent with theory (Weber, 1968; Bourdieu, 1986; Coleman, 1988). The thesis aim was achieved through three studies that examined whether different aspects of SES dynamics were associated with children's language ability and behaviour. First, in study 1, the extent to which family income, parent's education, and parent's occupation had unique associations with children's language ability and behaviour was examined. As children can be sensitive to family SES at different stages of childhood (Sylva 1997; Campbell et al., 2000; Costello et al., 2005; Kuhl et al., 2005), the associations between SES dimensions and children's outcomes were examined at different stages of childhood to determine whether the timing of each SES dimension was important for children's outcomes. Second, study 2 examined whether trajectories of SES, rather than static measurements of SES were associated with children's language and behavioural outcomes. Since study 1 showed that each of the SES dimensions

had a different association with children's outcomes across different stages of childhood it was likely that SES trajectories would also have associations with children's outcomes. This was supported by life course models which propose that different trajectories of each SES dimension could have differential associations with children's development (Duncan, 1988; Lynch et al., 1994; Elder, 2003). The findings from study 2 expanded those from study 1. Third, study 3 examined whether trajectories of SES increased or decreased the likelihood of children belonging to different language or behavioural trajectories. Since study 2 showed that SES trajectories had differential associations with children's late childhood language ability and behavioural outcomes, it was likely that the SES trajectories were also associated with different child developmental trajectories. Since life course models propose that both SES and children's developmental outcomes have different trajectories that are likely associated with each other, this was that natural progression from study 2.

The next section will summarise the research findings from each of the three studies and discuss the implications of the findings. Limitations and future research will be discussed.

7.4 Key findings and their Implications

7.4.1 SES dimensions

The findings from the thesis as a whole show that family income, parents' education, and parents' occupation were each associated with children's developmental outcomes. This supports the theoretical positions outlined in chapter 2 which define SES as a multidimensional construct with each construct representing the different economic and social resources a family can access (Weber, 1968; Bourdieu, 1986; Colemann, 1988; Jæger, 2007; Bukodi & Goldthorpe, 2013; Harding et al., 2015; Festin et al., 2017). In terms of previous research, few studies have examined the independent associations that family

income, parent education, and parent occupation have on children's language ability and behaviour. The majority of studies examine a single SES dimension or generate a composite SES measure (Coddington et al., 2014; Jackson et al., 2017; Gonzalez et al., 2020; Britton et al., 2021). Two notable studies that did examine family income, parent education, and parent occupation were Sullivan and Brown (2013) who focused on language ability, and Miller et al (2021) who focussed on internalizing and externalizing behaviour. They both examined the unique association that income, education, and occupation have with children's outcomes. The findings of this thesis support Sullivan and Brown (2013) who found that parents' education was the most important SES dimension for children's early adolescent language ability, followed by income, then occupation. This study expanded their findings by showing the same pattern was also found for early, middle, and late childhood which will be discussed in the next section.

Regarding internalizing and externalizing behaviour, the results of this thesis broadly support Miller et al (2021) as it found that each SES dimension was uniquely associated with children's outcomes. However, there were important differences in the findings for externalizing behaviour that has implications in terms of theory. For example, while Miller et al (2021) found that parents' occupation was the most important SES dimension for children's externalizing behaviour, this study found that occupation was significantly less important than family income and parents' education. It is likely due to methodological differences between studies as Miller et al., (2021) dichotomised occupation into working and non-working categories rather than examining the level of occupation which was done in this study. This may have theoretical implications about how research examining the associations between SES and child development should be framed. For example, this study, and several other studies, have drawn upon Bronfenbrenner's bioecological systems theory to explain

why SES dimensions are associated with child development (e.g., Schoon et al., 2010; 2013; Gutman et al., 2019; Sosu & Schmidt, 2022). The theory states that child development takes place due to frequent bidirectional interactions between children and their environments (Bronfenbrenner, 1986; Bronfenbrenner & Morris, 2006). The microsystem, being the most proximal environment including the family and home, is proposed to have the strongest influence on children's developmental outcomes due to its proximity to children and their ability to interact with it directly (Bronfenbrenner, 1979; Bronfenbrenner & Ceci, 1993; Bronfenbrenner & Morris, 1998). Therefore, it is often considered the most significant system for child development (Biggeri & Cuesta, 2021) and why researchers often use the microsystem to explain the associations between SES and children's development (Eamon, 2001; Biggeri & Cuesta, 2021; Lee & Zhang 2022). However, the results of this research suggest that while income and education belong in the microsystem, occupation might be better suited in the macrosystem. For example, parents' occupation in this study represented a family's social class so it might therefore be appropriate to place occupation in the macrosystem which is the most distal environment in Bronfenbrenner's theory and represents societal contexts which children do not interact with directly. On the other hand, children's daily interactions are more directly impacted by family income and parents' education. For example, family income can determine the material resources at home, impact parental stress, and increase family conflict (Conger & Donnellan, 2007; Conger et al., 2010) while parents' education is associated with the quantity and quality of speech that parent's use with children (Rowe, 2012; Cartmill et al., 2013). Therefore, income and education belong in the microsystem.

7.4.2 Stage of childhood

This thesis found that SES was associated with children's language and behaviour at all stages of childhood which broadly supports the different studies that have examined shorter time frames across childhood (e.g., Schoon et al., 2011; 2013; Mazza et al., 2017; Burchinal et al., 2018; Lai et al., 2019). However, the findings could support the argument that children are most sensitive to the adverse effects of SES during early childhood as all SES dimensions were associated with all outcomes during early childhood but not during later childhood. Income was associated with all outcomes at all time points, education was associated with language ability and externalizing behaviour at all childhood stages but only during early and middle childhood for internalizing behaviour, and occupation was associated with language ability and externalizing behaviour at all childhood stages but only during early childhood for internalizing behaviour. This would be in line with the sensitive period model (Sylva 1997; Kuhl et al., 2005) which proposes children are particularly sensitive to external factors (e.g., SES) during early childhood due to heightened brain plasticity in which neural networks can be modified based on external factors (Hensch, 2018; Gabard-Durnam & McLaughlin 2020). However, while SES during early childhood appeared to be most important for children's developmental outcomes generally due to all SES dimensions being associated with all outcomes, the findings are more complex as different SES dimensions were more or less important at different stages of childhood.

Consistent with most studies, the current findings showed that family income was significantly associated with children's language ability and behavioural outcomes at all childhood stages (Mazza et al., 2017; Burchinal et al., 2018). Family income was also found to become increasingly important as children got older as the strength of the associations between income and children's outcomes was always larger during late childhood. This

finding is also consistent with the accumulation of risk model which explains that children's exposure to low SES at each stage of childhood an association with children's outcomes but also accumulates to produce a cumulative adverse effect on children's outcomes (Power et al., 1991; Kuh et al., 2003). This could lead to cumulative impaired child-parent interactions and prolonged periods with less investment in their development according to the family stress and investment models (Conger et al., 2010; Conger & Donnellan, 2007). This build-up of exposure to stressful and impaired interactions is associated with worse developmental outcomes (Conger et al., 2010; Conger & Donnellan, 2007).

This study also found that while family income was associated with all outcomes at all childhood stages it had the largest associations with children's internalizing and externalizing behaviour during late childhood which is consistent with some previous studies (Costello et al., 2011; Gibb et al., 2012; Lai et al., 2019). It is likely that this is because late childhood is a period of significant social change where children are navigating relationships with peers (Blakemore2014; Fuhrmann et al., 2015; Efstathopoulos et al., 2018) where children become more aware of their economic situation because income becomes an important factor in their ability to participate in social events with their peers (Hjalmarsson & Mood, 2015). A lack of income, therefore, could limit children's ability to take part in activities with their peers and, from a family stress perspective, cause additional conflict and impaired interactions between parents and children leading to increased behavioural problems as outlined in the family stress model (Conger et al., 2010). Additionally, from an investment model perspective, low-income parents are less able to invest finances into their children's activities as children age and they become more expensive to raise as they get older (Twenge et al., 2003; Conger & Donnellan, 2007).
Parents' education was significantly associated with children's language ability and externalizing behaviour at all childhood stages but only associated with internalizing behaviour during early and middle childhood and not during late childhood. Parents' education had the largest associations with children's language ability at all stages of childhood and for internalizing and externalizing behaviour during early and middle childhood. These findings are consistent with the literature as previous studies which, although they only examined children over short timeframes, have found parents' education to have larger associations than income with children's language ability (Davis-Kean, 2005; Sullivan & Brown., 2013; Coddington et al., 2014; Harding et al., 2015; Gonzalez et al., 2020; Miller et al., 2021).

These findings have implications for theory, for example, the family stress and investment models indicate that SES is associated with children's developmental outcomes via several different pathways (Conger & Donnellan, 2007; Conger et al., 2010). It is possible that this study found parents to have the largest association with children's language ability because parents' education operates through different theoretical pathways that are more relevant for children's language development than income pathways. For instance, while income can operate via purchasing educational materials and funding extra tuition, parents' education operates via cognitive and linguistic stimulation (Coddington et al., 2014) which is noted to be essential for children's language development (Vygotsky, 1978). Similarly, the family stress model states that parenting styles are important pathways between SES and children's development (Conger et al., 2010) and parental responsiveness is a parenting style which is crucial for children's language development (Farrant & Zubrick, 2012; Madigan, 2019). Studies have found that children have better language ability if their parents show higher responsiveness during their interactions (Madigan, 2019) and parents with higher levels of

education are known to show higher levels of responsiveness to their children during interactions (Dearing et al., 2004; 2006). These unique educational pathways are likely more strongly associated with children's language development as children's language development takes place through frequent exposure to language and practice using language during interactions with their parents (Vygotsky, 1978; Bronfenbrenner, 1986). Since higher-educated parents are more likely to have more frequent, better quality, and more encouraging interactions with their children (Rowe, 2012; Cartmill et al., 2013) it could explain why parents' education has larger associations with children's language ability than income.

In terms of internalizing and externalizing behaviour, previous studies have also found parents' education to have larger associations than income with externalizing behaviour during early childhood (Boe et al., 2014; Jackson et al., 2017) while others have also found that parents' education is associated with and internalizing behaviour during early childhood but do not compare other SES dimensions. Again, from a theoretical perspective, the reason parent's education had a stronger association with children's internalizing and externalizing behaviour during early and middle childhood, but income had a stronger association during late childhood could be due to the theoretical mechanisms through which they operate. As with language ability, parental responsiveness to children's needs is a pathway through which education is associated with children's behaviour development as responsive parents are more likely to react positively to children's needs to support their development (Ray et al., 2013; Padilla-Walker et al., 2016). Additionally, parental expectations are a unique pathway between parents' education and children's outcomes (Davis-Kean et al., 2021) and higher educated parents have higher expectations for their children's outcomes, and they are more likely to invest the time and resources to promote their desired outcomes for children (Davis-Kean et al., 2021). Applying these pathways to Bronfenbrenner's bioecological model

(Bronfenbrenner, 1979; Bronfenbrenner & Morris, 1986) might explain why parents' education is more important for children's behavioural outcomes during early childhood but not late childhood. For example, during early childhood children's microsystems are generally limited to family interactions (Bronfenbrenner, 1979) and therefore children are likely to develop internalizing and externalizing behaviours based on their parents' expectations and responsiveness as they do not have any other models to learn from. However, as children age and interact with other adults and their peers, the influence of parental inputs is likely to be diluted somewhat (Hoferichter et al., 2021). At the same time, children are becoming more aware of their financial circumstances and might be more susceptible to family income pathways. This could explain why education is more important for behavioural outcomes during the younger years and income during later childhood.

There is also the possibility that parents' education and family income are associated with children's development through shared pathways. For example, as established, parenting styles (such as parental responsiveness) are pathways through which SES is associated with children's outcomes as stated in the family stress model (Conger et al., 2010). One of the most studied parenting styles is harsh parenting which is consistently shown to be a pathway between family income and children's outcomes (Rijlaarsdam et al., 2013; Neppl et al., 2016; Sutton et al., 2017; Sosu & Schmidt, 2017). However, parental expectations are also associated with harsh parenting as parents with higher levels of education have higher expectations for their children and sometimes use harsh parenting styles to achieve the desired outcome (Ren & Edwards, 2015; Jocson & McLoyd, 2015; White et al., 2016; Pinquart, 2016; Sanvictores & Mendez, 2022). It is therefore possible that different SES dimensions influence children's development through shared pathways. This information could have implications for theory because at present most theories focus on explaining the

pathways between financial pressures and children's outcomes (e.g., the family stress model) but if other SES dimensions are also shown to operate via the same pathways, then there could be scope to build upon family stress models to include additional SES dimensions that operate via the same pathways.

Examining the impact of SES at different stages of childhood also has implications for policy. Since family income was important at all childhood stages, policies that support family income at any time point are likely to improve children's outcomes quicker than supporting the other SES dimensions (e.g., it might take longer to improve education than income). While several policies are designed to support family income during early childhood, there are fewer targeting later childhood which could be an area for improvement. Policies supporting parents' education are likely to be most beneficial if it is targeted as early as possible as parents' education continues to be important at all childhood stages for language ability and early and middle childhood for behavioural outcomes. If intervention is left until late childhood, then parents' education might not be as important for children's behavioural outcomes. Policy targeting parents' occupation is also likely to be most beneficial during early childhood to benefit all outcomes. However, while parents' occupation is important for children's language development at all stages of childhood, its importance for behavioural outcomes tends to diminish over time. Additionally, parents' occupation was not found to be as important as parents' level of education or family income so policy might be more efficient if it focussed on education and income.

7.4.3 Trajectories

This thesis found that there were several different trajectories of family income, parent education, and parent occupation. The family income trajectories that were found were consistent with the range of income trajectories in previous studies which have found between three and six trajectories (e.g., Demment et al, 2014; Jackson et al., 2017; Kendzor et al., 2017; Comeua et al., 2018; Kwon et al., 2022). The novel aspect of this thesis is that it also examined trajectories of different SES dimensions (income, education and occupation) rather than only income trajectories as in previous studies (Demment et al, 2014; Zachrisson & Dearing, 2015; Björkenstam et al., 2017; Jackson et al., 2017; Kendzor et al., 2017; Hoyt et al., 2019). This showed that different trajectories of family income, parent education, and parent occupation could be identified and followed their own unique patterns over the life course. Previous literature examining education and occupation tends to use static measures or use a dichotomous variable to indicate change over time (Magnuson et al., 2009; Harding, 2015; Awada & Shelleby, 2021). This study was able to identify different trajectories of occupation and education to examine whether the different trajectories were associated with children's outcomes.

The findings are in line with previous research showing that the highest income trajectory was associated with the best outcomes, the increasing income trajectory was associated with improved developmental outcomes, and the decreasing income trajectory was associated with worsening outcomes (Zachrisson & Dearing, 2015; Jackson et al., 2017; Hoyt et al., 2019). Similarly, results showed that increasing parents' education was also associated with improved outcomes indicating that children's language and behavioural outcomes can benefit from improvement in either family income or parent education. Parents' occupation trajectories, while moving slightly, remained relatively constant across the childhood

timeline. Although there were decreasing trajectories it could be argued that the decreasing very high-high trajectory and the decreasing high-intermediate trajectory did not differ substantially. It is also unclear whether the increase from routine to semi-routine occupation trajectory improved outcomes as there was no stable routine group to compare findings with. Broadly speaking the findings suggest that parents' occupation remains stable over time which is in line with the literature which shows that social class remains relatively constant across the life course (Goldthorpe, 2016). This has useful implications for researchers interested in the associations between SES and children's outcomes as it shows that not only do the different SES dimensions have associations with children's later outcomes but that different trajectories exist in the population, and each has different associations with children's outcomes. This would benefit researchers who are interested in studying the mechanisms between SES and children's outcomes as there may be overlapping or unique trajectories through which parents' education and occupation impact children's language and behavioural outcomes.

Study 3 expanded on study 2 and provided novel findings by examining the association between SES trajectories and children's language and behavioural trajectories rather than associations with late childhood outcomes as a static measure. The results found that the language and behaviour trajectories were similar to those found in the previous literature (e.g., Ukoumunne et al., 2012; Weeks et al., 2014; Miller and Votruba-Drzal, 2017). It was found that children in the highest income education and occupation trajectories were most likely to belong to the best language and behaviour trajectories and that increasing income and education trajectories were more likely to belong to improving language and behavioural trajectories compared to their peers that did not improve their income or education. These findings showed that children's developmental experience across childhood was impacted by their family's SES trajectories and indicated the likely pattern of outcomes trajectories that children would experience. This is in line with the life course theory which assumes that individuals have multiple trajectories across several domains of life and that they can interact with each other. In this case, it shows that children's language and behavioural trajectories are associated with family SES trajectories.

A further novel aspect of this study in addition to examining trajectories of other SES dimensions, this study also examined trajectories over a longer time frame. Previous research tended to focus on early childhood trajectories (Dearing et al., 2001; Kendzor et al., 2010; Zachrisson & Dearing, 2015; Bøe et al., 2016; Jackson et al., 2017) rather than the entire childhood timeline. By using data over a longer time point, this study was able to identify an additional income trajectory and an additional language trajectory compared to previous studies (e.g., Ukoumunne et al., 2012; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017). This ultimately leads to a better understanding of how longitudinal SES is associated with children's development. It also shows the benefit of following a life course approach when examining the associations between SES and children's developmental outcomes as it states that more life events and transitions will be captured over longer timeframes leading to a better understanding of the impact of people's life experiences.

These findings have further implications for the life course theory as the assumptions in both the accumulation of advantage/disadvantage hypothesis and the social mobility hypothesis were supported in both studies 2 and 3. The accumulation of risk model (Power et al., 1991; Kuh et al., 2003) proposes that children's exposure to low SES at several time points each has an association with children's outcomes, but also accumulates to produce a cumulative adverse effect on children's outcomes (Power et al., 1991; Kuh et al., 2003). Study 2 found

this to be the case as the lowest income, education, and occupation trajectories were associated with the lowest child language ability scores and the highest behavioural problem scores. On the other hand, the highest income, education, and occupation trajectories were associated with the highest child language ability scores and low behavioural problem scores indicating the cumulative advantage of higher SES trajectories. Study 3 showed that the lowest SES trajectories were associated with the lowest language and behavioural trajectories which shows that accumulation of risk impacts children's outcomes across the life course. These findings suggest that other SES dimensions, other than income which is normally examined (e.g., Jackson et al., 2017; Björkenstam et al., 2017; Kendzor et al., 2012), can have a cumulative effect on children's outcomes and show that the accumulation of risk model is a useful way to frame research to further examine the associations between SES dimensions and children's outcomes.

The social mobility hypothesis proposes that individuals can continuously move between different social circumstances over their life course and that these upward and downward trajectories can have an association with children's developmental outcomes (Duncan, 1988; Lynch et al., 1994). This study found this to be the case as increasing income education and occupation trajectories were found as well as a decreasing income trajectory. In line with the social mobility hypothesis, the increasing income and education trajectories were associated with better language ability and lower behavioural problems compared to those who did not increase their income and education. They were also more likely to belong to improving language ability and behaviour trajectories compared to those that did not improve their income and education. Overall, life course theory was shown to be useful when applied to the associations between SES and children's language and behaviour outcomes as it was for

identifying new SES and developmental trajectories and examining their associations between different trajectories.

7.5 Overall implications

This thesis found that a significant proportion of children belonged to problematic language and behavioural trajectories over their life course which aligns with the existing literature (e.g., McKean et al., 2017; Hentges et al., 2019; Flouri & Midouhas, 2017; Miller & Votruba-Drzal, 2017; Comeau, 2018). To improve children's language and behaviour outcomes it is essential that policy address socioeconomic inequalities as the United Nations cites poverty and inequality as the largest barrier to improving children's outcomes (United Nations, 2023). Similarly, in a UK-specific context, Cooper and Stewart concluded that any attempt to improve children's outcomes is unlikely to be successful unless it addresses the socioeconomic inequalities associated with children's development (Cooper & Stewart, 2013; 2017). This is especially true as child poverty rates are rising in the UK and inequality is one of the worst in the Organization for Economic Cooperation and Development countries (OECD) (OECD, 2022). The results of this thesis have implications for policymakers to consider if they are concerned with improving children's developmental outcomes by reducing economic inequalities.

7.5.1 Multidimensional support

First, with all SES dimensions (family income, parents' education, and parents' occupation) being associated with all children's language ability and externalizing behaviour at all times, and with internalizing behaviour during early childhood, policy that takes a multidimensional approach to improve family SES is likely going to be most beneficial for children's overall development which would be in line with UK policy that aims to reduce child poverty by

improving family income through work and education as outlined in the Welfare Act (2016). However, most policy support currently focuses on improving family income directly or indirectly (Northern Ireland Executive, 2016; Welsh Government, 2015; Scottish Government, 2018). While this is beneficial for children's outcomes as the results show that family income is important at all stages of childhood, to maximise children's outcomes this thesis suggests that it would be beneficial to place additional emphasis on supporting parents' education and occupation as they were also important at all stages of childhood (although education was not associated with internalizing behaviour during late childhood and occupation was not associated with internalizing behaviour during middle and late childhood, they were still important for language ability and externalizing behaviour at those times).

Since parents' education has the strongest association with children's language ability and externalizing behaviour during early and middle childhood it is likely that intervention here would be beneficial. Although some previous studies found no benefit of increases to parent education (Magnuson, 2003; Harding, 2015) they have only examined education across short time frames (e.g., early childhood). This study examined parents' education trajectories over a longer timeframe and used more data to show that increasing parents' education will likely benefit children in the long term. Evidence of education interventions tends to support the findings of this study as parental education programs have been found to improve parenting practices and reduce children's behaviour problems (Yap et al., 2016; Buchanan et al., 2018). In terms of parents' occupation, this thesis found that higher levels of occupation were associated with better language ability and behavioural outcomes in children which is in line with previous research (Cano, 2022). However, there was very little variation in parents' occupation timeline meaning that it could not be determined whether

increasing parent's level of occupation over time would lead to significantly improved language and behaviour outcomes for children.

While a multi-dimensional approach to improving family SES dimensions is likely to benefit children, policymakers will have to carefully consider the feasibility of implementing such a large-scale policy to improve the education and occupation of parents. For example, logistically it might be challenging to implement policies to increase education and occupation as it would require collaboration with several parties (e.g., to identify eligible people, provide the services, and monitor progress), however, it could be possible by using the existing policy infrastructure. A recent example of how this could be implemented can be seen in Scotland where the Fair Start Scotland (FSS) and No One Left Behind (NOLB) policies share the same infrastructure to enable both policies to run in parallel. The Fair Start Scotland program was launched first and developed a network of local authorities, local employability service providers, and third-party service providers to identify eligible people and support them into employment (Chief Economist Directorate, 2019). When No One Left Behind was launched, it was able to utilise the same existing network of partners from FSS to minimise costs and speed up the launch (Scottish Government, 2020). In addition to providing employability support to low-income families, No One Left Behind also provides training and education to eligible people through their service provider partners (Scottish Government, 2020). If future policymakers wished to implement a policy to improve the level of education and occupation of parents, then they would likely be able to use the existing partnerships and extend the eligibility criteria to specifically target parents rather than only low-income families.

The other issue policymakers will likely come across is the cost of such a wide-reaching policy as all policies require a cost-benefit analysis to support the business case for funding.

A recent example of a cost-benefit evaluation of an employability policy in the UK comes from Fair Start Scotland (FSS) (Scottish Government, 2022). The policy has cost £82 million to run but has an estimated benefit to society of £298 million (or for every £1 spent has a ± 3.60 benefit to society) since the eligible people are earning more, paying taxes, receiving less welfare, and using less services (Scottish Government, 2022). In terms of the eligible peoples' take-home income, for every £1 spent running FSS, the person was £1.40 better off once they gained employment showing that FSS is a beneficial policy. However, the target people for FSS were unemployed to begin with so it is likely that clients were better off because they were initially unemployed and entering employment through FSS. It is unlikely that a new policy targeting parents who are already employed would yield a cost-benefit ratio as high as that seen in FSS because the benefit of supporting people who are already in employment into a higher-paying job will have less of an impact compared to getting unemployed people into work in the first place. Policymakers interested in proposing such a policy might wish to be selective in the eligibility criteria to best target parents that might benefit most. For example, study 2 indicated that children's language ability and behavioural outcomes were similar in both the stable-average and stable-intermediate occupation trajectories which would suggest that improving from average to intermediate occupation might not lead to a large improvement in terms of children's outcomes. However, there was a larger difference in children's outcomes between the routine-semi-routine and stable-average occupation trajectories which would suggest it might be more beneficial to target those with lower-level occupations.

Similar considerations would also have to be made in terms of implementing policy to increase parents' level of education. The results indicated that increasing parents' level of

education from low to high led to significantly better language and behavioural outcomes compared to those that remained in the stable-low education trajectory. However, this represents an increase from standard grades or GCSEs to bachelor's degree level (or equivalent) which would be an unrealistic goal for policymakers to have. For instance, as of 2021 only 26% of 25–64-year-olds held a bachelor's degree in the United Kingdom meaning that the number of parents that would be eligible for education under this policy would be unmanageable and likely unaffordable. As an example of costs, in 2021 Student Awards Agency Scotland (SAAS) spent almost £230 million on tuition fees for approximately 146,000 students in Scotland with overall costs of almost £1 billion when all other support such as bursaries and loans were accounted for (Student Awards Agency Scotland, 2021). The funding required to finance all parents would be unrealistic especially since the tuition costs in the rest of the UK is £9,250 per person per year (OEDC, 2021) compared to Scotland's £1,820. There are also practicalities to consider when supporting parents in education. For example, they will have childcare costs, higher costs of living than a student with no children, and less time to commit to education when raising children. Additionally, achieving a bachelor's degree typically involves at least 3 years in higher education which could also increase the financial burden on families making such courses less likely to attract parents. It is therefore likely that policymakers would have to be selective with the eligibility criteria for parents to ensure that it is affordable. While not as good as increasing parent education from low to high, the results of this thesis also found that increasing parents' level of education from no qualifications to low education also improved children's language ability and behavioural outcomes so policymakers might consider introducing basic education for parents with no qualifications.

Alternatively, further research could examine the pathways between parents' education and children's developmental outcomes to identify why parents' education is important for children. If research can identify what parents with high levels of education do to improve their children's outcomes, then perhaps policy can address those factors rather than directly increasing parents' education through formal education. For example, parents' education is often associated with children's outcomes via linguistic stimulation during parent-child interactions (Coddinton et al., 2014) so policy might be more effective if they taught parents with lower education how to have positive and encouraging interactions with their children to promote positive development. This could provide some of the benefits of having higher levels of education without the large time and financial commitment. Overall, children are likely to benefit from policies that take a multidimensional approach to improving their parents' SES, but policymakers would have to be selective with eligibility criteria to ensure the families that would benefit most get access to support.

7.5.2 Support over the childhood timeline

Second, the results of this thesis also indicate that the dimensions of family SES continued to be significantly associated with children's language ability, internalizing behaviour, and externalizing behaviour from early childhood through to late childhood and that lower SES trajectories were associated with more problematic child outcome trajectories. This is in line with the accumulation of risk theory that proposes the duration of exposure to adverse SES circumstances will lead to worsening developmental outcomes over time (Kuh et al., 2003), and has also consistently been the case in the literature focussing on the persistence of poverty which almost invariably shows that persistent poverty over a longer duration is worse for children's outcomes compared to episodic poverty (Kiernan & Mensah, 2009; 2011; Schoon et al., 2010; 2012; Holmes & Kiernan, 2013; Dickerson & Popli, 2012; 2014; 2016). These findings have implications for policy as they suggest that support for family income, parents' education, and parents' occupation would continue to be beneficial if it was prolonged throughout childhood. Currently, most policies tend to target family income during early childhood (e.g., free school meals, free childcare, Scottish Child payment, and the Healthy Start scheme) which is beneficial as early childhood is considered a sensitive period where children develop the neuropathways responsible for facilitating cognitive and behavioural processes so policies minimising the adverse effects during this stage of childhood are important (Brummelte et al., 2017). However, the results of this thesis suggest that children are also sensitive to the different SES dimensions during middle and late childhood. In addition, the results also show that when family income decreases, there are negative consequences for children's outcomes and their outcome trajectories over the life course.

These findings present difficulties in terms of how policymakers can use this information as it is likely unfeasible for policies to support family income, parents' education, and parents' occupation continuously over the entire childhood timeline, especially when resources available to policymakers have been reduced (Cooper & Stewart, 2017; Joseph Rowntree Foundation, 2021). Therefore, policymakers would likely have to be selective in how they offer support for family income, parent's education, and parent's occupation over time. For example, this thesis found that parents' education had the largest association with children's language ability, internalizing behaviour, and externalizing behaviour during early and middle childhood and that children were more likely to belong to improving language and behaviour trajectories if their parents belonged to an increasing education trajectory. These

findings suggest that any intervention aimed at increasing parents' level of education would benefit from being administered as early as possible as increasing education earlier would result in children being exposed to more positive interactions with higher-educated parents for longer than if parents' education was increased during late childhood. This would likely have a cumulative positive impact on children's outcomes in line with the accumulation of risk model (Kuh et al., 2003) and previous research on cumulative poverty (Dickerson & Popli, 2012; 2014; 2016). This would also make sense as it is during the earlier years that children's interactions are primarily with their parents and as they get older, they are exposed to other people (e.g., peers and teachers) (Bronfenbrenner, 1986). Therefore, although the results also indicate that parents' level of education is important during late childhood, it would not be the most efficient use of resources to offer educational support to parents at this time.

Another key finding from this thesis is that children are more likely to belong to problematic language and behavioural trajectories if their families belong to a decreasing income trajectory. This is particularly problematic as many of the income support policies are limited to early childhood and end when children reach a certain age (e.g., Free school meals, free childcare, Scottish child payment, and the Healthy Start scheme). When families are no longer eligible for these payments then their income will decrease which the current results indicate could have negative consequences for children's language and behavioural outcomes. This is in line with previous research that also shows decreases in income are associated with poorer child developmental outcomes (Jackson et al., 2017; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Hoyt et al., 2019). A recent example of policy decisions that resulted in reduced family income can be seen when the UK government rescinded the £20 per week Universal Credit and Working Tax Credit for low-income families, the consequences of which put 500,000 families and 200,000 children into poverty (JRF, 2021). So, while it may not be feasible to continually provide increases to income across the childhood timeline, policymakers should attempt to avoid income reductions as stable income is less problematic than decreasing income for children's outcomes. Reducing costs for lowincome families in other areas might be more feasible than policies that directly give money to people (e.g., the Scottish child payment) and can have long-lasting positive effects on children's outcomes. For example, when families' income is increased due to tax credits rather than receiving money, there are improvements in children's reading ability and behavioural outcomes (Dahl & Lochner, 2012; Hamad & Rehkopf, 2016) with improvement lasting into late childhood and adulthood (Duncan et al., 2010).

Additionally, interventions such as childcare support are also an effective way to reduce the economic strain on families (e.g., reducing the financial cost of childcare and allowing parents to be more flexible with working patterns) while improving children's outcome trajectories. At present, children in England aged 3-4 can get 1,140 hours of childcare per year however it can only be used during term time meaning they get 30 hours of free childcare per week during term time (Department for Education, 2018). In Scotland, children also get 1,140 hours but the age range is increased to between 2 and 4 years old and families can distribute their 1,140 hours across the entire year which grants those families extra flexibility to distribute their childcare in a way that makes the most economical sense for them (e.g., if they work odd hours then they can still get childcare) (Department for Education, 2018; Scottish Government, 2018b). In Wales, children aged 3-4 can get 30 hours of free childcare per week for 48 weeks totalling 1,440 hours per year (Welsh Government, 2022). After the age range has been exceeded, the cost of childcare falls on the parents,

however, if childcare was extended to include later age ranges during non-term times, then this would likely have a positive effect on family financial circumstances and benefit children's developmental trajectories through both less family economic stress and increased positive interaction with the care provider's environment. As per the bioecological model (Bronfenbrenner & Morris, 1986), positive interactions between children and their childcare environments are likely to improve children's developmental outcomes and evidence tends to support this as good quality out-of-home childcare environments are associated with improved children's cognitive and behavioural development from age 6 years onwards (Melhuish, 2015; Gluschkoff et al., 2018). Overall, providing alternative support such as childcare might mitigate the negative impact that low SES trajectories have on children's language and behavioural trajectories.

7.5.3 Addressing stress as a mechanism between SES and child development

Policymakers interested in improving children's developmental outcomes might also consider addressing parent and child levels of stress as stress is considered one of the primary mechanisms through which SES can impair children's developmental outcomes (Masarik & Conger, 2017). This is because stress, when experienced for long enough durations, leads to prolonged activation of the stress response which can create changes in the neurobiological systems and consequently impair children's brain, learning, and behaviour development (Smith & Pollak, 2020; Shonkoff et al., 2021). Although the results of this thesis indicate that family income, parent education, and parent occupation are each associated with children's development over the life course, the Family Stress Model (FSM) would suggest that each of the SES dimensions contributes to parental stress which in turn impacts children's via parental conflict and disrupted parenting (Conger et al., 2010). This is in line with Masarik and Conger (2017) who contend that the FSM also applies to broader environmental factors beyond purely financial pressures.

Indeed, research shows that family income, parent education, and parent occupation are associated with parental stress which impacts children's outcomes via family conflict, parenting practices, and child stress (Shaw & Shelleby, 2014; Shelleby et al., 2014; Parkes et al., 2015; Dockery et al., 2016; Vahedi et al., 2018; Moreira et al., 2019). For example, evidence indicates that low income is consistently associated with parental stress and family conflict (Shelleby et al., 2014; Simons et al., 2016; Sosu & Schmidt, 2017). Parents with lower levels of education also exhibit higher levels of stress compared to parents with higher levels of education (Parkes et al., 2015; Aizer et al., 2016). Parents with lower levels of occupation have also been shown to have increased job-related work stress that spills over into the child's home environment leading to increased family conflict, poor parenting practices, abuse, and neglect which impacts children's developmental outcomes (Li, Kendall, Strazdins, & Jacoby, 2013; Dockery et al., 2016; Cho, 2018; Moreira et al., 2019; Baldwin et al., 2020; Lawson et al., 2020).

It is also important to note that parental stress can be caused by non-economic pressures. For example, parents who are raising children with disabilities have significantly higher levels of parental stress than parents who are raising typically developing children (Cheng & Lai, 2023). This can be due to several factors such as having to manage treatment and rehabilitation plans, manage the additional costs associated with raising disabled children, and cope with the stress associated with the uncertainty about their child's future (Crnic et al., 2017; Hoff et al., 2019). The increased parental stress has been found to reduce the quality and frequency of parent-child interactions and reduce parent's ability to provide the necessary support for their children (Mbatha & Mokwena, 2023). Similarly, parental stress has been

shown to be higher in families with neurodiverse children (Da Paz & Wallander, 2017; Staunton et al., 2020; Chad-Friedmanet al., 2022). Children with learning difficulties (LD), autism spectrum disorder (ASD), and attention-deficit/hyperactivity disorder (ADHD) typically suffer from isolation, functional impairments, increased socioemotional behaviour problems, and academic difficulties which result in parents reporting higher levels of parental stress (Chad-Friedmanet al., 2022). Although all parents raising children with disabilities have higher stress levels than parents raising typically developing children, the literature indicates that there is also an SES gradient in parental stress levels (Cheng & Lai, 2023).

While parental stress has been associated with children's developmental outcomes, there is also evidence that child-level stress can impair their development (Franke, 2014; Hambrick et al., 2019). Children can also experience increased stress due to economic factors (e.g., food insecurity), family conflict, and impaired parental interactions (e.g., neglect, abuse, poor parenting, parental conflict) (Franke, 2014; Merwin et al., 2017; Liu & Merritt, 2018; Hambrick et al., 2019). There are also child-level stressors that do not necessarily stem from the family. For example, neurodiverse and disabled children can experience increased stress due to isolation and peer rejection (Plexousakis et al., 2019; Chad-Friedmanet al., 2022) and children who experience bullying at school also experience higher levels of stress (Smokowski & Evans, 2019; Andreou et al., 2021). Child stress from these experiences has been associated with increased internalizing and externalizing behaviour and poorer academic outcomes (Goodman et al., 2012; Hair et al., 2015; Burneo-Garcés et al., 2019; Schuurmans et al., 2022).

Another factor that leads to increased parent and child stress is sleep quality (Alotaini et al., 2020) which is particularly relevant to this thesis as sleep problems are more prevalent in

parents and children from low SES families which contributes to developmental inequalities (Boe et al., 2012). Sleep and stress have been shown to have a bidirectional relationship in that people who are stressed are more likely to have sleep disorders, and people with sleep disorders are more likely to experience stress (Merrill & Slavik, 2023). Similar to the above, parents with disabled children are more likely to have poor sleep quality which can exacerbate their levels of stress (Gallagher et al., 2010). Recent research has shown that the quality of sleep leads to higher levels of parent and child stress which impairs children's developmental outcomes (Alotaini et al., 2020).

According to the bioecological model (Bronfenbrenner, 1986), irrespective of the origin of stress, it is likely to cause impaired interactions between children and their microsystems (e.g., the family home) and impair their development. Therefore, policymakers interested in improving children's developmental outcomes should consider targeting parent and child levels of stress. This is in line with a recent literature review that concluded that interventions targeting family coping and stress regulation are likely to be beneficial methods of intervention specifically for low SES families (Mayo et al., 2022). Evidence has shown that mindfulness-based interventions focussing on stress reduction techniques were able to lower parental stress levels which resulted in improved interactions with children and improved child behaviour (Singh et al., 2020). Another stress reduction intervention focused on improving education, problem-solving, and communication between parents and their children (Prithivirajh & Edwards, 2011). The results found that parents felt empowered and confident in their abilities as a parent while children reported that their parents seemed more relaxed, approachable, loving, and inclusive (parents do more activities and engage with their children). Another intervention for specifically reducing financial-related stress also showed positive results by teaching parents coping skills for stress and communication skills for

improving relationships with their family (Wadsworth et al., 2011). The results of the previous interventions therefore give confidence that targeting parent and child stress in low SES families could have a positive impact on children's language and behavioural outcomes and offers an alternative approach to addressing SES dimensions directly.

Overall, this section discussed several ways in which policymakers might find the results of this thesis useful. The results suggested that a multidimensional approach to supporting family SES dimensions for a prolonged duration would be beneficial for children's language and behavioural development, however, it was acknowledged that in practice it would likely be unfeasible to implement such broad policies to address the SES dimensions directly. Alternative suggestions were discussed including being more selective with eligibility criteria for policy support and the timing of when support should be made available. Additional recommendations were made to target intervention toward reducing stress as the family stress model indicates that stress is an important mechanism in the SES-child development relationship (Conger et al., 2010; Masarik & Conger, 2017) and such interventions are likely more feasible than large-scale income, education, and occupation policies.

7.6 Limitations

There are several limitations in this thesis that should be acknowledged. The first limitation is missing data. Missing data are inherent in secondary longitudinal data that disproportionately affects lower SES families which could potentially lead to bias. There are two forms of missing data in this thesis: panel attrition and item-level missingness. Despite the problems that can be introduced by missing data several methods are used to mitigate potential bias in the sample. Longitudinal nonresponse weights are provided to account for attrition and sample design (Plewis, 2007; Mostafa, 2014) while item-level missingness was handled by

full information maximum likelihood which produces unbiased estimates (Enders & Bandalos, 2001; Larsen, 2011). The second limitation is that this study did not account for potential genetic-environmental confounding. Since genetic factors are associated with both SES and children's developmental outcomes (Pingault et al., 2015; Krapohl & Plomin, 2016), it is likely that biasing the extent to which family SES dimensions are associated with children's language and behavioural outcomes.

The third limitation is the measures used. The measures used in longitudinal data are restricted to those that fit the original researcher's purposes. The use of different language ability measures at each time point must be acknowledged as a potential limitation. The British Ability Scale subscales were used to measure language ability during early and middle childhood, but not during late childhood. Instead, the Applied Psychology Unit Standardised Vocabulary Test was used during late childhood, it may have been beneficial to use the same tests over time as it would provide more confidence that the desired constructs are being measured over time. Despite this, previous studies show that correlations between the British ability scales and the Applied Psychology Unit Standardised Vocabulary Test were adequate for use together (Hill, 2005; Girard et al., 2017; Sullivan et al., 2017). This study also found that language ability measures were correlated in line with previous research.

The fourth limitation is the method used to identify trajectories. Trajectories of family income, parent education, parent occupation, children's language ability, internalizing behaviour, and externalizing behaviour were all identified using latent class growth analysis which requires researchers to select the appropriate number of latent classes at their discretion. While goodness of fit indices are used to guide researchers to the best fitting

model, they do not dictate the final model selection. Researchers must consider the research question, theory, and data alongside the goodness to fit indices when selecting the appropriate model (Jung & Wickrama, 2008). This is inherently problematic as researchers may draw different conclusions when deciding on the best model to use for their studies. Still, since the family income trajectories found in the study are similar to those found in previous studies, it provides confidence that researchers are following similar methods when selecting trajectories. However, there is no frame of reference to draw upon regarding education and occupation trajectories and this should be considered when drawing conclusions.

The fifth limitation is how parents' education and occupation were measured over time. This thesis was interested in the highest household level of parent education and occupation which was measured at each time point and used to generate trajectories. This method does not provide information about which parent has the highest education and occupation at each time point or whether the parent with the highest level of education changed over time. While both maternal and paternal education and occupation have significant associations with children's developmental outcomes (Erola et al., 2016; Jeong et al., 2017), some studies indicate that maternal education is more important (Jeong et al., 2017). On the other hand, evidence indicates that maternal education is most important for children's outcomes during early childhood and decreases over time while the association between fathers' education and children's development, measuring the highest level of household education and occupation was appropriate for this study.

7.7 Future research

7.7.1 Examining pathways between SES and children's developmental outcomes

While research following the family stress model has consistently shown that economic stress is associated with children's developmental outcomes via disrupted parenting and interparental relationship problems (Conger et al., 2010), there are likely several other pathways through which parents' economic stress is associated with children's developmental outcomes. To identify new pathways, it would be beneficial to conduct qualitative research with families to better understand how they believe economic stress might affect the home environments that children interact with and their children's development. Thematic analysis in particular is likely to be beneficial because allows interview responses from participants to be collected and broad themes to be identified. Each broad theme may contain several more focused subthemes which can be examined in more detail (Braun & Clarke, 2006). For example, a recent study by Mathur et al. (2023) has shown the utility of qualitative research in identifying pathways between parental stress caused by the COVID-19 pandemic and children's outcomes. By interviewing parents about their experiences during the COVID-19 lockdown, the researchers were able to identify new stressors that parents believed had an impact on their children. A thematic analysis of parents' responses identified that "virtual schooling" was a common stressor for parents and was one of the main overarching themes in the study. The "virtual schooling" theme contained several subthemes such as parents having to learn how to properly facilitate virtual schooling, parents acting as teachers, learning new technologies, and overcoming technology issues. Parents believed each of these themes had an impact on their children. This study is an example of how qualitative research can be useful in identifying new pathways between parental stress and children's outcomes.

To help visualise how thematic analysis might expand the research in this thesis, the family stress model (FSM) is presented below. The FSM proposes that parental stress, because of economic pressure, is associated with children's developmental outcomes via two pathways: Interparental relationship problems and disrupted parenting (see boxes 4a and 4b in figure 2 below).



Figure 2. Family Stress Model (figure in Masarik & Conger, 2017)

Both of these pathways can be considered main themes with several subthemes falling under them. For example, interparental relationship problems (box 4a) is a broad theme that contains several subthemes that have been examined in the literature such as parental relationship satisfaction, relationship quality, work-related conflict, child-related conflict, family structure security, researcher-observed hostility and negativity (Cummings et al., 2015; Masarik & Conger, 2017; van Eldik et al., 2020). Similarly, disrupted parenting (box 4b) is also a broad theme which contains several subthemes that have been examined in the literature, for example, harsh parenting, unresponsive parenting, child maltreatment or abuse, and child neglect (Rijlaarsdam et al., 2013; Masarik & Conger, 2017; Liu & Merritt, 2018; Zhang et al., 2022). By conducting a thematic analysis researchers might identify additional ways in which interparental relationship problems (box 4a) and disrupted parenting (box 4b) can be operationalised or they may find new broad themes to expand the family stress model (e.g., adding a box 4c into the model to further explain the association between parental stress due to family economic pressure and children's developmental outcomes through new pathways).

On a similar note, since this study shows that family income, parents' education, and parents' occupation each have unique associations with children's developmental outcomes, future research should consider the different pathways through which each SES dimension operates. At present, income pathways have been examined primarily through a family stress and investment framework (Shelleby et al., 2014; Sun et al., 2015; Sosu & Schmidt, 2017) but the current results show that those frameworks may be too narrowly focused on income. The family stress model, for example, states that economic pressure is associated with children's developmental outcomes via several pathways. This thesis shows that the different SES dimensions could be considered economic pressures so research should examine the extent to which the different SES dimensions operate via unique or similar pathways. Similar to the above discussion, qualitative research may also be useful to identify unique education and occupation pathways.

7.7.2 Considering sex differences

Future research should also consider examining different subgroups as this thesis did not analyse the different associations that SES dimensions might have with different groups of people. There is a strong case for future research to examine whether the differences in the associations between SES and children's outcomes at different stages of childhood and the associations between SES trajectories and boys' and girls' outcome trajectories differ depending on the sex of the children. Examining sex differences at different time points might be useful as evidence shows that language ability, internalizing behaviour, and externalizing behaviour can differ between boys and girls at different stages of childhood. For instance, it is well-documented that girls have better language ability than boys from early childhood with larger vocabularies and fewer language developmental problems (Eriksson et al., 2012; Lange et al., 2016; Adani & Cepanec, 2019). There is also evidence that girls have more prevalent internalizing behavioural problems than boys from early childhood through adolescence while boys have higher externalizing behaviour problems (Keiley et al., 2010 Matos et al., 2017; Gutman & McMaster, 2020; Lau et al., 2021). Given that there are differences in boys' and girls' outcomes across childhood stages, it is likely that SES dimensions will have different associations with boys' and girls' language ability, internalizing behaviour, and externalizing behaviour at different stages of childhood.

Similarly, since boys and girls have different outcomes at different stages of childhood, it is likely that they would also have different developmental trajectories over the life course. For example, if girls have higher language ability than boys during early childhood, then it is likely that they will have better overall trajectories from early to late childhood. Therefore, there will likely be different associations between family SES and the different boys' and girls' outcome trajectories. A recent study by Houweling et al. (2022) supports this idea by examining the associations between family SES is associated with boys' and girls' language ability trajectories during early childhood (1-4 years old), and internalizing behaviour and externalizing behaviour trajectories from early-mid childhood (1.5 - 9 years old). They found that boys and girls both had different trajectories of language ability, internalizing behaviour,

and externalizing behaviour. They also found that girls from low SES families had higher problematic language ability and internalizing behaviour trajectories than boys, while boys from low SES families had higher problematic externalizing behaviour trajectories. These findings suggest that there are gender differences in the extent to which SES affects boys' and girls' different developmental outcome trajectories, but it does not examine whether the trajectories of different SES dimensions also have different associations with boys' and girls' outcome trajectories. It would be important to examine this as there is evidence that different dimensions of SES have different associations with boys' and girls' outcomes. For example, recent research found that both family income and parents' level of education had direct associations with male reading ability while only parents' level of education was associated with female reading ability (Guo et al., 2018). Future research should therefore build on this thesis and previous research by examining the extent to which the trajectories of different SES dimensions are associated with boys' and girls' language ability, internalizing behaviour, and externalizing behaviour trajectories across the different stages of childhood.

7.7.3 Considering neurodiversity

Future research should also examine the associations between the different dimensions of SES and children's language ability, internalizing behaviour and externalizing behaviour in neurodiverse children (e.g., children with autism spectrum disorders (ASD), attention deficit hyperactivity disorders (ADHD), and learning difficulties (LD)). Neurodiverse children tend to have more impaired language abilities, internalizing behaviour, and externalizing behaviour compared to their neurotypical (typically developing) peers (May et al., 2015; Howlin & Magiati, 2017; McGuinness et al., 2017; Simonoff et al., 2020; McRae et al., 2020: Vassiliu et al., 2023). For example, recent research found that children with ADHD scored 12% less in expressive and receptive vocabulary than typically developing children while

children with learning difficulties scored 27.8% less in receptive vocabulary and 40.5% less in expressive vocabulary compared to typically developing children (Vassiliu et al., 2023). Children with ASD have also been found to score significantly higher for internalizing and externalizing behaviour problems than typically developing children (Li et al., 2020). These differences in neurodiverse children lead them to have different developmental trajectories compared to typically developing children. For example, Landa et al. (2012) examined the language and behaviour trajectories of children with early ASD (diagnosis before 14 months), late ASD (diagnosis after 14 months), and neurotypical children over the first three years of life. They found that neurotypical children had better language and behaviour trajectories compared to children in both the early and late ASD groups. They also examined the extent to which children from each of the three groups exhibited diminishing language skills over time by assessing language ability at multiple time points. They found that 29% of children in the early ASD group had diminishing language skills compared to 19% in the late ASD group, and only 2% in the neurotypical group. These findings indicate that it is likely that different language ability and behaviour trajectories would be found in neurodiverse children.

There is also evidence that family SES has different associations with neurodiverse children's outcomes compared to typically developing children. For instance, research has shown that although low SES is associated with poorer language ability for all children, the negative effect is worse for children with ASD (Olson et al., 2021). Similar research has found the same pattern in children's behavioural outcomes with low SES being associated with more problematic behaviour in children with ASD compared to their typically developing peers (Olsen et al., 2022). With evidence indicating that neurodiverse children have different outcomes and outcome trajectories compared to typically developing children and that SES has different associations with neurodiverse children, it would be beneficial if future research

examined these associations in more detail to better understand the extent to which the different trajectories of SES are associated with neurodiverse children's developmental outcomes in order to better inform policy.

7.7.4 Considering different contexts

Future research should also consider the wider social context in which families live as areas with different levels of deprivation might influence the SES trajectories that are found and ultimately the associations between family SES and children's developmental outcomes. This thesis focused on national cohort data which found income, education, and occupation trajectories based on data from across the entire UK. However, if future research were to focus on particular areas such as high or low SES neighbourhoods it is likely that different trajectories would be found since research has consistently found that people from affluent areas have higher levels of income, education, and occupation than people from deprived neighbourhoods (Zuccotti, 2019; Nieuwenhuis et al., 2021; Troost et al., 2023; Kelly & Evans, 2023). It is therefore likely that the trajectories found in affluent and deprived neighbourhoods would differ. It is worth noting that even if similar SES trajectories over time moved in the same direction, their meaning could be different as a "high" SES trajectory in a deprived SES neighbourhood may not necessarily be considered "high" in a more affluent area.

For example, in Scotland, the most recent data on local-level household income estimates indicate that the average weekly income in the most deprived area was £208 per week while the most affluent areas had an average weekly income of £1,643 (Scottish Government, 2018). If income trajectories were examined independently in each of these areas, it is likely

that both areas would have a "high income" trajectory since trajectories are estimates of the data provided. However, the "high income" trajectory from the deprived area would not represent the same monetary value as the "high income" trajectory from the affluent area. Therefore, although families from both areas may categorised into a high-income trajectory, the associations that each of the "high-income" trajectories would have with children's developmental outcomes are likely to differ. It would therefore be beneficial if future research examined SES trajectories in contexts with different levels of affluence and deprivation to better inform policymakers on how to benefit people at a local level.

7.7.5 International research

Future research should also conduct similar studies in an international context. The current thesis used a nationally representative UK sample which may not reflect the associations between SES dimensions and children's outcomes in other countries with different social contexts and policies. While there is evidence that similar income trajectories are found across different countries (Dickerson & Popli, 2016; Jackson et al., 2017; Miller & Votruba-Drzal 2017; Björkenstam et al., 2017; Mok et al., 2018; Lai et al., 2019; Hoyt et al., 2019; Pryor et al., 2022), the same evidence base does not exist for education and occupation trajectories. It is possible that different trajectories of education and occupation would be found in other countries if they place different levels of importance on them. For example, the UK has prioritised policies to help people get into work or return to education (Department for Business Innovation & Skills, 2012; Northern Ireland Executive, 2021; Welsh Government, 2022; Chief Economist Directorate, 2022) but the same priorities may not be present in all countries, particularly in lower-income countries (Crawfurd et al., 2021) which could impact the education and occupation trajectories found in those countries. Even where similar levels of importance are placed on education and occupation, they may not be

as important for children's development in other countries. For example, the UK and US both tend to view education as important however Jackson et al. (2017) found that mothers' level of education was only important for children's language ability and externalizing behaviour in a UK sample, and not important in a US sample where family income was the only important SES dimension for children's development. Overall, it should not be assumed that the results found in this thesis will apply to other countries and future research should address this.

7.8 Conclusion

This thesis shows that each dimension of socioeconomic status has a unique association with children's developmental outcomes at each stage of childhood and that the importance of each SES dimension varies depending on the outcomes being examined. It also shows that the SES dimensions and child outcomes are not static constructs and can vary over the life course. The different trajectories of SES each have a unique association with children's language ability, internalizing behaviour, and externalizing behaviour. These associations underscore the utility of legislation requiring authorities to measure income, education, and occupation (Welfare Reform and Work Act, 2016) is likely to be beneficial for children's outcomes. Considering future policy, this thesis suggests that it would be beneficial for policymakers to expand current legislation to take a multidimensional approach to support family SES over the course of a child's life in order to improve their developmental outcomes.

References

- Adani S, Cepanec M. (2019). Sex differences in early communication development:
 behavioral and neurobiological indicators of more vulnerable communication system
 development in boys. *Croatian Medical Journal*, 60, 141-149.
- Adler, N. E., Epel, E. S., Castellazzo, G., & Ickovics, J. R. (2000). Relationship of subjective and objective social status with psychological and physiological functioning:
 Preliminary data in healthy, White women. *Health Psychology*, *19*, 586–592.
- Adler, N. E., Stewart, J. (2007). *The MacArthur Scale of Subjective Social Status*. Retrieved from http://www.macses.ucsf.edu/research/psychosocial/subjective.php.
- Advani, A., Bangham, G., Leslie, J. (2020). *The UKs wealth distribution and characteristics of high wealth households*. Resolution Foundation.
- Ahun, M. N., Geoffroy, M-C., Herba, C. M., Brendgen, M., Séguin, J. R., Sutter-Dallay, A-L., Boivin, M., Tremblay, R. E., & Côté, S. M. (2017). Timing and chronicity of maternal depression symptoms and children's verbal abilities. *The Journal of Pediatrics*, 190,251-257.
- Akaike, H. (1974). A new look at the statistical model identification. *IEEE Transactions on Automatic Control*, *19*(6), 716–723.

- Alan, K. M. A., Altman, Y., & Roussel, J. (2008). Employee Training Needs and Perceived Value of Training in the Pearl River Delta of China: A Human Capital Development Approach. *Journal of European Industrial Training*, 32(1), 19-31.
- Alotaibi, A. D., Alosaimi, F. M., Alajlan, A. A., & Bin Abdulrahman, K. A. (2020). The relationship between sleep quality, stress, and academic performance among medical students. *Journal of Family and Community Medicine*, 2020, 27, 23-28.
- Amir, D., Valeggia, C., Srinivasan, M., Sugiyama, L. S., & Dunham, Y. (2019). Measuring subjective social status in children of diverse societies. *PLoS One*, 14, e0226550.
- American Psychiatric Association. (2013). American Psychiatric Association, & DSM-5 Task
 Force. Diagnostic and statistical manual of mental disorders: DSM-5 (5th ed.).
 American Psychiatric Publishing.
- American Psychological Association, Task Force on Socioeconomic Status. (2007). Report of the APA Task Force on Socioeconomic Status. Washington.

Anderson, C. J. (2010). Central Limit Theorem. The Corsini Encyclopedia of Psychology.

- Andersson, M. A. (2015). How do we assign ourselves social status? A cross-cultural test of the cognitive averaging principle. Soc Sci Res, 52, 317–29.
- Andersen, S. L., & Teicher, M. H. (2008). Stress, sensitive periods and maturational events in adolescent depression. *Trends in Neurosciences*, 31, 183–191.

- Andreou, E., Tsermentseli, S., Anastasiou, O., & Kouklari, E. (2021). Retrospective
 Accounts of Bullying Victimization at School: Associations with Post-Traumatic
 Stress Disorder Symptoms and Post-Traumatic Growth among University
 Students. *Journal of Child and Adolescent Trauma, 14*, 9–18.
- Augustine, J. M., & Negraia, D. V. (2018). Can Increased Educational Attainment Among Lower-Educated Mothers Reduce Inequalities in Children's Skill Development?. *Demography*, 55, 59-82.
- Armstrong, R., Scott, J. G., Whitehouse, A. J. O., Copland, D. A., Mcmahon, K. L., & Arnott, W. (2017). Late talkers and later language outcomes: Predicting the different language trajectories. *International Journal of Speech-Language Pathology*, 19, 237– 250.
- Avan, B., Richter, L. M., Ramachandani, P. G., Norris, S. A., & Stein, A. (2010). Maternal postnatal depression and children's growth and behaviour during the early years of life: exploring the interaction between physical and mental health. *Archives of Disease in Childhood*, 95, 690-5.
- Avlund, K., Holstein, B. E., Osler, M., Damsgaard, M. T., Holm-Pedersen, P., & Rasmussen,
 N. K. (2003). Social position and health in old age: the relevance of different
 indicators of social position. *Scandinavian Journal of Public Health*, *31*, 126–36.
- Awada, S. R., & Shelleby, E. C. (2021). Increases in Maternal Education and Child
 Behavioral and Academic Outcomes. *Journal of Child and Family Studies*, 30, 1813–
 1830.
- Bai, X., Jiang, L., Zhang, Q., Wu, T., Wang, S., Zeng, X., Li, Y., Zhang, L., Li, J., Zhao, Y.,
 & Dai, J. (2021). Subjective Family Socioeconomic Status and Peer Relationships:
 Mediating Roles of Self-Esteem and Perceived Stress. *Frontiers in Psychiatry*, 12, 634976.
- Baldwin, H., Biehal, N., Allgar, V., Cusworth, L., & Pickett, K. (2020). Antenatal risk factors for child maltreatment: Linkage of data from a birth cohort study to child welfare records. *Child Abuse and Neglect*, 107, 104605.
- Bandura, A. (1986). Social foundations of thought and action. Englewood Cliffs, NJ: Prentice Hall.
- Bannink, R., Pearce, A., & Hope, S. (2016). Family income and young adolescents' perceived social position: associations with self-esteem and life satisfaction in the UK Millennium Cohort Study. *Archives of Disease in Childhood*, 101, 917–921.
- Barnes, M., Chanfreau, J., & Tomaszewski, W. (2010) *Growing up in Scotland: the circumstances of persistently poor children*. Scottish Government: Edinburgh.
- Barthel, A. L., Hay, A., Doan, S. N., & Hofmann, S. G. (2018). Interpersonal Emotion Regulation: A Review of Social and Developmental Components. *Behaviour Change*, 35, 203-216.

- Belinda, B. and Peat, J. (2014). *Medical statistics: a guide to SPSS, data analysis, and critical appraisal.* Wiley: UK.
- Benner, A. D., & Kim, S. Y. (2010). Understanding Chinese American adolescents' developmental outcomes: Insights from the family stress model. *Journal of Research* on Adolescents, 20, 1-12.
- Berger, E.M., & Spiess, C.K. (2011). Maternal life satisfaction and child outcomes: Are they related? *Journal of Economic Psychology*, *32*, 142–158.
- Berger, L. M., Paxton, C., & Waldfogel, J. (2009). Income and child development. *Children* and Youth Services Review, 31, 978–989.
- Berlin, K. S., Parra, G. R., & Williams, N. A. (2013). An Introduction to Latent Variable
 Mixture Modeling (Part 2): Longitudinal Latent Class Growth Analysis and Growth
 Mixture Models. *Journal of Pediatric Psychology*, 39, 188–203.
- Berthelon, M., Kruger, D., & Sanchez, R. (2021). Maternal stress during pregnancy and Early childhood development. *Economics & Human Biology, 43*, 101047.
- Betancourt, L. M., Brodsky, N. L., & Hurt, H. (2015). Socioeconomic (SES) differences in language are evident in female infants at 7 months of age. *Early Human Development*, 91, 719–724.

- Bevilacqua, L., Hale, D., Barker, E. D., & Viner, R. (2017). Conduct problems trajectories and psychosocial outcomes: a systematic review and meta-analysis. *European Child* & Adolescent Psychiatry, 27, 1239–1260.
- Bhattacharya, A. (2010). Children and Adolescents From Poverty and Reading Development: A Research Review. *Reading & Writing Quarterly*, 26, 115–139.
- Biggeri, M., & Cuesta, J. A. (2021). An Integrated Framework for Child Poverty and Well-Being Measurement: Reconciling Theories. *Child Indicators Research*, *14*, 821–846.
- Birch, J, & Beerten, R. (2002). Accuracy of the self-coded version of the National Statistics
 Socio-economic classification (NS-SEC). *Social Survey Methodology Bulletin, 50*, 111.
- Birt, C., Drake, B., Cabula, C., Collingwood, A., Elliott, J., Leese, D., Matejic, P.,Schwendel, G., & Wenham, A. (2022). Poverty in Northern Ireland 2022. JosephRowntree Foundation.
- Bitsko, R. H., Holbrook, J. R., Robinson, L. R., Kaminski, J. W., Ghandour, R., Smith, C., & Peacock, G. (2016). Health Care, Family, and Community Factors Associated with Mental, Behavioral, and Developmental Disorders in Early Childhood United States, 2011–2012. *Morbidity and Mortality Weekly Report, 65*, 221–226.
- Björkenstam, E., Cheng, S., Burström, B., Pebley, A. R., Björkenstam, C., & Kosidou, K.(2017). Association between income trajectories in childhood and psychiatric

disorder: a Swedish population-based study. *Journal of Epidemiology and Community Health*, 71, 648-654.

- Blakemore, S-J., & Mills, K. L. (2014). Is Adolescence a Sensitive Period for Sociocultural Processing?. Annual Review of Psychology, 65, 187–207.
- Blanca, M.J., Arnau, J., López-Montiel, D., Bono, R., & Bendayan, R. (2013). Skewness and Kurtosis in Real Data Samples Skewness and kurtosis in real data samples *Methodology: European Journal of Research Methods for the Behavioral and Social Sciences*, 9, 78–84.
- Blanden, J., & Machin, S. (2010). Intergenerational inequality in early years assessments. In
 K. Hansen, H. Joshi, & S. Dex (Eds.), Children of the 21st Century–The First
 FiveYears (pp. 153–168). Bristol: The Policy Press.
- Blumenshine, P., Egerter, S., Barclay, C. J., Cubbin, C., & Braveman, P. A. (2010). Socioeconomic Disparities in Adverse Birth Outcomes. *American Journal of Preventive Medicine*, 39, 263–272.
- Bøe, T., Hysing, M., Morten, K., Lundervold, A. J., & Sivertsen, B. (2012). Sleep problems as a mediator of the association between parental education levels, perceived family economy and poor mental health in children. *Journal of Psychosomatic Research*, 73, 430-436.

- Bøe, T., Sivertsen, B., Heiervang, E., Goodman, R., Lundervold, A.J., & Hysing, M. (2014).
 Socioeconomic status and child mental health: The role of parental emotional wellbeing and parenting practices. *Journal of Abnormal Child Psychology*, 42, 705–715.
- Bono, R., Blanca, M. J., Arnau, J., & Gómez-Benito, J. (2017). Non-normal Distributions
 Commonly Used in Health, Education, and Social Sciences: A Systematic
 Review. *Frontiers in Psychology*, *8*, 1602.
- Booth, A., Scott, M. E., & King, V. (2009). Father Residence and Adolescent Problem
 Behavior: Are Youth Always Better Off in Two-Parent Families?. *Journal of Family Issues*, 31, 585-605.
- Bornstein, M. H., & Bradley, R. H. (2003). Socioeconomic Status, Parenting, and Child Development. Mahwah, NJ: Erlbaum.
- Bornstein, M. H., Hahn, C., Suwalsky, J. T. D., & Haynes, O. M. (2003). Hollingshead four factor index of social status and the socioeconomic index of occupations. In M. H.
 Bornstein & R. H. Bradley (Eds.), *Socioeconomic status, parenting, and child development* (pp. 29-82).
- Boslaugh, S. (2007). Secondary Data Sources for Public Health: A Practical Guide. Cambridge University Press.
- Bourdieu, P. (1986). *The forms of capital. In: Richardson, J., Handbook of Theory and Research for the Sociology of Education.* Westport, CT: Greenwood: 241–58.

- Bradley, R. H., Corwyn, R. F. (2002). Socioeconomic status and child development. *Annual Review of Psychology*, *53*, 371-399.
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, *3*, 77-101.
- Brian, J., Roncadin, C., Bryson, S. E., Smith, I. M., Roberts, W., Szatmari, P., & DePalma, T.
 (2014) Emerging cognitive profiles in toddlers with autism spectrum disorders. *Research in Autism Spectrum Disorders*, 8, 1557–1566.
- Britto, P. R., Lye, S. J., Proulx, K., Yousafzai, A. K., Matthews, S. G., Vai- vada, T. (2017).
 Nurturing care: promoting early childhood development. *Lancet*, 389, 91–102
- Britton, J., Dearden, L., Waltmann, B. (2021). *The returns to undergraduate degrees by socio-economic group and ethnicity*. Institute for Fiscal Studies.
- Broeren, K. A., Muris, P., Diamantopoulou, S., & Baker, J. R. (2013). The course of childhoods anxiety symptoms: Developmental trajectories and child-related factors in normal children. *Journal of Abnormal Child Psychology*, 41, 81-95.
- Bromley, C. (2009). *Growing Up in Scotland: The impact of children's early activities on cognitive development*. Edinburgh: Scottish Government.
- Bronfenbrenner, U. (1973). Social ecology of human development. In F. Richardson (Ed.),
 Brain and intelligence: The ecology of child development (pp. 113–129). Hyattsville,
 MD: National Education Press.

- Bronfenbrenner, U. (1974). Developmental research, public policy, and the ecology of childhood. *Child Development*, *45*, 1–5.
- Bronfenbrenner, U. (1976). *The experimental ecology of education*. Teachers College Record, 78, 157–204.
- Bronfenbrenner, U. (1978). The social role of the child in ecological perspective. *Zeitschrift fur Soziologie*, 7, 4–20.
- Bronfenbrenner, U. (1976). The experimental ecology of education. *Teachers College Record*, 78, 157–204.
- Bronfenbrenner, U. (1979). *The ecology of human development: Experiments by nature and design*. Cambridge, MA: Harvard University Press.
- Bronfenbrenner, U. (1979b). Contexts of child rearing: Problems and prospects. *American Psychologist*, *34*, 844–850.
- Bronfenbrenner, U. (1986). Ecology of the family as a context for human development: Research perspectives. *Developmental Psychology*, 22, 723–742.
- Bronfenbrenner, U. (1989). Ecological systems theory. *Annals of Child Development*, 6, 187–249.

- Bronfenbrenner, U., & Ceci, S. J. (1993). Heredity, environment, and the question "how?" A first approximation. In R. Plomin & G. G. McClern (Eds.), *Nature, nurture, and psychology (pp. 313–323)*. Washington, DC: American Psychological Association.
- Bronfenbrenner, U., & Evans, G. W. (2000). Developmental science in the 21st century: Emerging theoretical models, research designs, and empirical findings. *Social Development*, 9, 115–125.
- Bronfenbrenner, U., & Morris, P. A. (1998). The ecology of development processes. In W.
 Damon (Series Ed.) & R. M. Lerner (Vol. Ed.), *Handbook of child psychology: Vol. 1*. Theoretical models of human development (pp. 993–1027). New York, NY: Wiley.
- Bronfenbrenner, U., & Morris, P. A. (2006). The bioecological model of human development. In W. Damon & R. M. Lerner (Series Eds.) & R. M. Lerner (Vol. Ed.), *Handbook of child psychology: Vol. 1. Theoretical models of human development* (6th ed., pp. 793–828). New York, NY: Wiley.
- Bronfenbrenner, U. (2000). Ecological systems theory. In A. Kazdin (Ed.), *Encyclopaedia of psychology (Vol. 3, pp. 129–133)*. Washington, DC: American Psychological Association.
- Bronfenbrenner, U. (2001). The bioecological theory of human development. In N. J. Smelser
 & P. B. Baltes (Eds.), *International encyclopedia of the social and behavioral* sciences (Vol. 10, pp. 6963–6970). New York, NY: Elsevier.

- Bronfenbrenner, U. (1999). Environments in developmental perspective: Theoretical and operational models. In S. L. Friedman & T. D. Wachs (Eds.), *Measuring environment across the life span: Emerging methods and concepts (pp. 3–28)*. Washington, DC: American Psychological Association Press.
- Brooks-Gunn, J., & Duncan, G. J. (1997). The effects of poverty on children. *Future Child* 7, 55–71.
- Brooks-Gunn, J., Han, W. J., & Waldfogel, J. (2010). First-year maternal employment and child development in the first seven years: What distinguishes women who work fulltime, part-time, or not at all in the 1st year?. *Monographs of the Society for Research in Child Development*, 75, 35–49.
- Brook, J. S., Zhang, C., Balka, E. B., & Brook, D. W. (2012). Pathways to Children's Externalizing Behavior: A Three-Generation Study. *The Journal of Genetic Psychology*, 173, 175–197.
- Brownell, M. D., Chartier, M. J., Nickel, N. C., Chateau, D., Martens, P. J., Sarkar, J.,
 Burland, E., Jutte, D. P., Taylor, C., Santos, R. G., & Katz, A. (2016). Unconditional
 Prenatal Income Supplement and Birth Outcomes. *Pediatrics*, *137*, e20152992.
- Brummelte, S., Mc Glanaghy, E., Bonnin, A., & Oberlander, T. F. (2017). Developmental changes in serotonin signaling: Implications for early brain function, behavior and adaptation. *Neuroscience*, *342*, 212–231.

- Buchanan, P. S., Gray, K. M., Gordon, M., & Melvin, G. A. (2018). Systematic review and meta-analysis of parent group interventions for primary school children aged 4–12 years with externalizing and/or internalizing problems. *Child Psychiatry and Human Development*, 49, 244–267.
- Buckingham, J., Wheldall, K., & Beaman-Wheldall, R. (2013). Why poor children are more likely to become poor readers: The school years. *Australian Journal of Education*, 57, 190-213.
- Bukodi,E., & Goldthorpe, J. (2013). Decomposing 'Social Origins': The Effects of Parents' Class, Status, and Education on the Educational Attainment of Their Children. *European Sociological Review*, 29, 1024-1039.
- Burchinal, M., Robert C. Carra, R. C., Vernon-Feagansa, L., Blair, C., Coxa, M., & The Family Life Project Key Investigators. (2018). Depth, persistence, and timing of poverty and the development of school readiness skills in rural low-income regions:
 Results from the family life project. *Early Childhood Research Quarterly*, 45, 115–130.
- Burneo-Garcés, C., Cruz-Quintana, F., Pérez-García, M., Fernández-Alcántara, M., Fasfous,
 A., & Pérez-Marfil, M. N. (2019). Interaction between Socioeconomic Status and
 Cognitive Development in Children Aged 7, 9, and 11 Years: A Cross-Sectional
 Study. *Developmental and Neuropsychology*, 44, 1-16.

- Burnett, K., & Farkas, G. (2009). Poverty and family structure effects on children's mathematics achievement: Estimates from random and fixed effects models. *The Social Science Journal*, *46*, 297–318.
- Burnham, K. P., & Anderson, D. R. (2002). Model selection and multinomial inference. New York: Springer.
- Campbell, S. B., Shaw, D. S., & Gilliom, M. (2000). Early externalizing behavior problems: toddlers and preschoolers at risk for later maladjustment. *Development and Psychopathology*, 12, 467–488.
- Cano, T. (2022). Social class, parenting, and child development: A multidimensional approach. *Research in Social Stratification and Mobility*, 77, 100648.
- Carneiro, P., Meghir, C., & Parey, M. (2013). Maternal education, home environments, and the development of children and adolescents. *Journal of the European Economic Association*, 11, 123–160.
- Carstairs, V., & Morris, R. (1989). Deprivation, mortality and resource allocation. *Community Medicine*, *11*, 364–372.
- Caruana, E. J., Roman, M., Hernandez-Sanchez, J., & Solli, P. (2015). Longitudinal studies. *Journal of Thoracic Disease*, 7, 537-540.

- Carr, A., & Pike, A. (2012). Maternal scaffolding behavior: Links with parenting style and maternal education. *Developmental Psychology*, 48, 543–551.
- Cartmill, E. A., Armstrong, B. F., Gleitman, L. R., Goldin-Meadow, S., Medina, T. N., & Trueswell, J. C. (2013). Quality of early parent input predicts child vocabulary 3 years later. *Proceedings of the National Academy of Sciences*, *110*, 11278–11283.
- Cavanagh, S. E., Schiller, K. S., Huynh, V. W., & Riegle-Crumb, C. (2016). A latent class analysis of socioeconomic status and academic achievement: Evidence from young children. *Social Science Research*, 57, 138-151.
- Celeux, G., & Soromenho, G. (1996). An entropy criterion for assessing the number of clusters in a mixture model. *Journal of Classification*, *2*, 195-212.
- Cesarini, D., Lindqvist, E., Notowidigdo, M. J., & Östling, R. (2017). The Effect of Wealth on Individual and Household Labor Supply: Evidence from Swedish Lotteries. *American Economic Review*, 107, 3917–3946.
- Chad-Friedman, E., Kuhlthau, K. A., Millstein, R. A., Perez, G. K., Luberto, C. M., Traeger,
 L., Proszynski, J., & Park, E. (2022). Characteristics and Experiences of Parents of
 Children with Learning and Attention Disabilities and Autism Spectrum Disorder: A
 Mixed Methods Study. *The Family Journal, 30*, 427–436.

Chandola, T. (2001). Ethnic and class differences in health in relation to British South Asians: using the new National Statistics Socio-Economic Classification. *Social Science & Medicine 52*, 1285–1296.

- Chen, E., Martin, A. D., & Matthews, K. A. (2007). Trajectories of Socioeconomic Status Across Children's Lifetime Predict Health. *Pediatrics*, *120*, 297–303.
- Cheng, A. W. Y., & Lai, C. Y. Y. (2023). Parental stress in families of children with special educational needs: a systematic review. *Frontiers in Psychiatry*, *14*, 1198302.
- Cheng, H., & Furnham, A. (2012). Childhood cognitive ability, education, and personality traits predict attainment in adult occupational prestige over 17 years. *Journal of Vocational Behavior*, 81, 218–226.
- Chetty, R., Friedman, J. N., Hilger, N., Saez, E., Schanzenbach, D. W., & Yagan, D. (2011). How does your kindergarten classroom affect your earnings? Evidence from Project STAR. *The Quarterly Journal of Economics*, *126*, 1593-1660.
- Chief Economist Directorate. (2022). Scotland's Devolved Employment Services: Statistical Summary. Scottish Government.

Chief Economist Directorate. (2019). Fair Start Scotland Evaluation Report 1: Implementation and Early Delivery Review (June 2019). Scottish Government.

Childcare Act. (2006). C.21

Child Poverty Act. (2010). C.9.

Children's Services Co-operation Act. (2015). C. 10.

- Cho, Y. (2018). The effects of nonstandard work schedules on workers' health: a mediating role of work-to-family conflict. *International Journal of Social Welfare*, 27, 74–87.
- Choi, Y., Shin, J., Cho, K. H., & Park, E.-C. (2017). Change in household income and risk for attention deficit hyperactivity disorder during childhood: A nationwide population-based cohort study. *Journal of Epidemiology*, 27, 56-62.
- Chow, J. C., & Jacobs, M. (2016). The role of language in fraction performance: A synthesis of literature. *Learning and Individual Differences*, 47, 252–257.
- Civil Service. (2018). Measuring socio-economic background in your workforce: recommended measures for use by employers. Civil Service.
- Clarke, C., & O. Thévenon. (2022). *Starting unequal: How's life for disadvantaged children?*. *OECD Papers on Well-being and Inequalities*. OECD Publishing: Paris.
- Clogg, C. C., Petkova, E., & Haritou, A. (1995). Statistical Methods for Comparing
 Regression Coefficients Between Models. *American Journal of Sociology*, 100, 1261–1293.

- Closs, S. J. (1976). *APU vocabulary test (multiple choice format, 1986)*. Kent: Hodder and Stoughton Educational Ltd.
- Coddington, C. H., Mistry, R. S., & Bailey, A. L. (2014). Socioeconomic status and receptive vocabulary development: replication of the parental investment model with Chilean preschoolers and their families. *Early Childhood Research Quarterly* 29, 538-549.
- Coleman, J. S. (1988). Social Capital in the Creation of Human Capital. *American Journal of Sociology*, *94*, 95–120.
- Coley, R. L., Spielvogel, B., Kruzik, C., Miller, P., Betancur, L., & Votruba-Drzal, E. (2021).
 Explaining income disparities in young children's development: The role of community contexts and family processes. *Early Childhood Research Quarterly*, 55, 295, 311.
- Comeau, J., & Boyle, M. H. (2018). Patterns of poverty exposure and children's trajectories of externalizing and internalizing behaviors. *SSM Population Health*, *4*, 86–94.
- Conger, R. D., & Donnellan, M. B. (2007). An interactionist perspective on the socioeconomic context of human development. *Soc. Sci.* 58, 175–199.
- Conger, R. D., Conger, K. J., Martin, M. J. (2010). Socioeconomic status, family processes, and individual development. *Journal of Marriage Family*, 72, 685-704.

- Connelly, R., & Platt, L. (2014). Cohort Profile: UK Millennium Cohort Study (MCS). International Journal of Epidemiology, 43, 1719–1725
- Connelly, R. (2013) *Interpreting test scores. Millennium Cohort Study Data Note 2013/1*. Centre for Longitudinal Studies, Institute of Education: London.
- Cooper, K. and Stewart, K. (2013). *Does money affect children's outcomes? A systematic review*. York: Joseph Rowntree Foundation.
- Cooper, K. and Stewart, K. (2017). *Does money affect children's outcomes? An update.* London: Centre for Analysis of Social Exclusion.
- Costello, E. J, Copeland, W., & Angold, A. (2011). Trends in psychopathology across the adolescent years: what changes when children become adolescents, and when adolescents become adults? *Journal of Child Psychol Psychiatry*, *52*, 1015-25.
- Costello, E. J., Egger, H. L., & Angold, A. (2005). The developmental epidemiology of anxiety disorders: phenomenology, prevalence, and comorbidity. *Child and Adolescent Psychiatric Clinics of North America*, 14, 631–648.
- Côté, S. M., Orri, M., Tremblay, R. E., & Doyle, O. (2018). A Multicomponent Early Intervention Program and Trajectories of Behavior, Cognition, and Health. *Pediatrics*, 141, e20173174.

Cowan, C. D., Hauser, R. M., Kominski, R. A., Levin, H. M., Lucas, S. R., Morgan, S. L.,
 Spencer, M. B., & Chapman, C. (2012). *Improving the Measurement of* Socioeconomic Status for the National Assessment of Educational Progress: A
 Theoretical Foundation--Recommendations to the National Center for Education
 Statistics. National Center for Education Statistics.

- Crawfurd, L., Hares, S., Minardi, A., & Sandefur, J. (2021). Understanding Education Policy Preferences: Survey Experiments with Policymakers in 35 Developing Countries.
 Center for Global Development.
- Cribb, J., Waters, T., Wernham, T., & Xu, X. (2022). *Living standards, poverty and inequality in the UK: 2022*. Institute of Fiscal Studies.
- Crnic, K., & Ross, E. (2017). Parenting stress and parental efficacy. In K. Deater-Deckard &
 R. Panneton (Eds.), *Parental stress and early child development: Adaptive and maladaptive outcomes* (pp. 263–284). Springer International Publishing.
- Cummings, E. M., Koss, K. J., & Davies, P. T. (2015). Prospective Relations between Family
 Conflict and Adolescent Maladjustment: Security in the Family System as a
 Mediating Process. *Journal of Abnormal Child Psychology*, 43, 503-515.
- Dahl, G. B., & Lochner, L. (2012). The Impact of Family Income on Child Achievement:
 Evidence from the Earned Income Tax Credit. *American Economic Review*, 102, 1927-56.

- Dalstra, J. A. A., Kunst, A. E., Mackenbach, J. P., & The EU Working Group on Socioeconomic Inequalities in Health. (2006). A comparative appraisal of the relationship of education, income and housing tenure with less than good health among the elderly in Europe. *Social Science & Medicine*, 62, 2046–2060.
- Da Paz, N. S., & Wallander, J. L. (2017). Interventions that target improvements in mental health for parents of children with autism spectrum disorders: A narrative review. *Clinical Psychology Review*, *51*, 1–14.
- Darin-Mattsson, A., Fors, S., Kåreholt, I. (2017). Different indicators of socioeconomic status and their relative importance as determinants of health in old age. *International Journal of Equity Health, 16*, 173.
- Davis-Kean, P.E. (2005). The influence of parent education and family income on child achievement: the indirect role of parental expectations and the home environment.
 Journal of the Division of Family Psychology of the American Psychological Association, 19, 294-304.
- Davis-Kean, P. E., Tighe, L. A., Waters, N. E. (2021). The Role of Parent Educational Attainment in Parenting and Children's Development. Association for Psychological Sciences, 30, 186-192.
- Dearden, L., Sibieta, L., & Sylva, K. (2011). *The socio-economic gradient in early child outcomes: evidence from the Millennium Cohort Study*. Institute for Fiscal Studies.

- Dearing, E., McCartney, K., & Taylor, B. A. (2001). Change in family income-to-needs matters more for children with less. *Child Development*, 72, 1779-93
- Dearing, E., McCartney, K., & Taylor, B. A. (2006). Within-child associations between family income and externalizing and internalizing problems. *Developmental Psychology*, 42, 237-252.
- Dearing, E., McCartney, K., Weiss, H. B., Kreider, H., & Simpkins, S. (2004). The promotive effects of family educational involvement for low-income children's literacy. *Journal of School Psychology*, *42*, 445–460.
- Dearing, E., Kreider, H., Simpkinds, S., & Weiss, H. B. (2006). Family involvement in school and low-income children's literacy: Longitudinal associations between and within families. *Journal of Educational Psychology*, 98, 653–664.
- Demment, M. M., Haas, J. D., & Olson, C. M. (2014). Changes in family income status and the development of overweight and obesity from 2 to 15 years: a longitudinal study.
 BMC Public Health, 14, 417.
- Department for Business, Energy, & Industrial Strategy. (2022). Warm Home Discount The Government Response to the Warm Home Discount: Better targeted support from 2022 consultation. Department for Business, Energy, & Industrial Strategy.
- Department for Business Innovation & Skills. (2012). *National careers service: the right advice at the right time*. Department for Business Innovation & Skills.

- Department for Education. (2011). A New Approach to Child Poverty: Tackling the Causes of Disadvantage and Transforming Families' Lives. HM Government.
- Department for Education. (2018). *Early years entitlements: operational guidance for local authorities and providers*. Department for Education.
- Department for Education. (2019). *Early years foundation stage profile results in England,* 2019. Department for Education.
- Department for Education. (2021). *Skills for Jobs: Lifelong Learning for Opportunity and Growth*. Department for Education.
- Department for Education. (2021). *Statutory framework for the early years foundation stage*. Department for Education.
- Department for Environment Food and Rural Affairs. (2012). *Changes to WaterSure as a result of the introduction of Universal Credit*. Department for Environment Food and Rural Affairs.
- Department for Work & Pensions. (2011). *The work programme*. Department for Work & Pensions.

- Department for Work & Pensions. (2017). Understanding how universal credit influences employment behaviour: Findings from qualitative and experimental research with claimants. Department for Work & Pensions.
- Department for Work and Pensions. (2017). *Improving Lives: Helping workless families, analysis and research pack.* Department for Work and Pensions.
- Department for Work and Pensions. (2020). *Restart Scheme*. Department for Work and Pensions.
- Department for Work and Pensions. (2020). *Nation's job hunt jets off*. Department for Work and Pensions.
- Department for Work & Pensions. (2022). *The Restart scheme for long-term unemployed people*. National Audit Office.
- Destin, M., Richman, S., Varner, F., & Mandara, J. (2012). "Feeling" hierarchy: The pathway from subjective social status to achievement. *Journal of Adolescence*, *35*, 1571-1579.
- Dickerson, A., & Popli, G. K. (2012). Persistent poverty and children's cognitive development: evidence from the UK Millennium Cohort Study. Institute of Education: London.
- Dickerson, A., & Popli, G. (2014). *The many dimensions of child poverty: Evidence from the UK Millennium Cohort Study*. The University of Sheffield, Department of Economics.

- Dickerson, A., & Popli, G. K. (2016). Persistent poverty and children's cognitive development: evidence from the UK Millennium Cohort Study. *Journal Royal Statistical Society Series A*, 179, 535-558.
- Diener, E., Inglehart, R., & Tay, L. (2013). Theory and validity of life satisfaction scales. *Social Indicators Research*, *112*, 497–527.
- Dockery, A. M., Li, J., & Kendall, G. (2016). Sole-parent work schedules and adolescent wellbeing: evidence from Australia. *Social Science & Medicine*, *168*, 167–174.
- Dohoon, L. (2014). Age Trajectories of Poverty During Childhood and High School Graduation. *Sociological Science*, *1*, 344-365.
- Dubow, E. F., Boxer, P., & Huesmann, L. R. (2009). Long-term Effects of Parents'
 Education on Children's Educational and Occupational Success: Mediation by Family
 Interactions, Child Aggression, and Teenage Aspirations. *Merrill Palmer Quarterly*, 55, 224–249.
- Duckworth, K., & Schoon, I. (2012). Beating the Odds: Exploring the Impact of Social Risk on Young People's School-to-Work Transitions during Recession in the UK. *National Institute Economic Review*, 222, 38–51.

- Duff, F. J., Reen, G., Plunkett, K., & Nation, K. (2015). Do infant vocabulary skills predict school-age language and literacy outcomes?. *Journal of Child Psychology and Psychiatry*, 56, 848–856.
- Duncan, G. (1988). *The volatility of family income over the life course*. In Baltes, P.,
 Featherman, D., & Lerner, R. M. Life-Span Development and Behavior. Hillsdale,
 NJ: Lawrence Erlbaum Associates, 9, 317–58.
- Duncan, O. D., Featherman, D. L. and Duncan, B. (1972). *Socioeconomic Background and Achievement*. New York: Seminar Press.
- Duncan, G. J., Daly, M. C., McDonough, P., & Williams, D. R. (2002). Optimal Indicators of Socioeconomic Status for Health Research. *Research and Practice*, 92, 1151-1157.
- Duncan, G. J., & Magnuson, K. A. (2003). Off with Hollingshead: Socioeconomic resources, parenting, and child development. In M. H. Bornstein & R. H. Bradley (Eds.), *Socioeconomic status, parenting, and child development* (pp. 83–106). Lawrence Erlbaum Associates Publishers.
- Duncan, G. J., & Magnuson, K. (2012). Socioeconomic status and cognitive functioning: moving from correlation to causation. Wiley Interdisciplinary Reviews: Cognitive Science, 3, 377–386.
- Duncan, G. J., Magnuson, K., & Votruba-Drzal, E. (2014). Boosting family income to promote child development. *Future Child*, *24*, 99-120.

- Duncan, G. J., Morris, P. A., & Rodrigues, C. (2011). Does money really matter? Estimating impacts of family income on young children's achievement with data from randomassignment experiments. *Developmental Psychology*, 47, 1263–1279.
- Duncan, G. J., Ziol-Guest, K. M., & Kalil, A. (2010). Early-childhood poverty and adult attainment, behaviour, and health. *Child Development*, *81*, 306-325.
- Duncan, O. D., Featherman, D. L., & Duncan, B. (1972). Socio-economic background and achievement. NY: Seminar Press.
- Durkin, M. S., Maenner, M. J., Meaney, F. J., Levy, S. E., DiGuiseppi, C., Nicholas, J. S., Kirby, R. S., Pinto-Martin, J. A., Schieve, L. A. (2010). Socioeconomic Inequality in the Prevalence of Autism Spectrum Disorder: Evidence from a U.S. Cross-Sectional Study. *PLoS ONE*, *5*, e11551.
- Eamon, M. K. (2001). The Effects of Poverty on Children's Socioemotional Development: An Ecological Systems Analysis. *Social Work, 46*, 256–266.

Education (Scotland) Act. (2016). Asp. 8.

Efstathopoulos, P., Andersson, F., Melas, P. A., Yang, L. L., Villaescusa, C., Ruegg, J.,
Ekstrom, T J., Forsell, Y., Galanti, M. R., & Lavebratt, C. (2018). NR3C1
hypermethylation in depressed and bullied adolescents. *Translational Psychiatry*, 8, 121.

- Eisenberg, N., Cumberland, A., Spinrad, T. L., Fabes, R. A., Shepard, S. A., Reiser, M.,
 Murphy, B. C., Losoya, S. H., Guthrie, I. K. (2001). The relations of regulation and
 emotionality to children's externalizing and internalizing problem behavior. *Child Development*, 72, 1112-1134.
- Elder, G. H., Johnson, M. K., & Crosnoe, R. (2003). The Emergence and Development of Life Course Theory. Handbooks of Sociology and Social Research, 3–19.
- Elenbaas, L. (2019). Perceptions of economic inequality are related to children's judgments about access to opportunities. *Developmental Psychology*, 55, 471–481.
- Elliott, C. D., Smith, P., & McCulloch, K. (1997). British Ability Scales Second Edition (BAS II). Technical Manual. London: Nelson.
- Employers for Childcare. (2022). *Is free or funded childcare available in northern ireland?*. Employers for Childcare. Available at: https://www.employersforchildcare.org/newsitem/is-free-or-funded-childcare-available-in-northern-ireland/

Enders, C. K. (2010). Applied missing data analysis. The Guilford Press: London.

Enders, C. K., & Bandalos, D. L. (2001). The Relative Performance of Full InformationMaximum Likelihood Estimation for Missing Data in Structural Equation Models.*Structural Equation Modeling*, *8*, 430-457.

- Ensminger, M. E., & Fothergill, K. (2003). A decade of measuring SES: What it tells us and where to go from here. In M. H. Bornstein & R. H. Bradley (Eds.), *Socioeconomic status, parenting, and child development* (pp. 13–27). Lawrence Erlbaum Associates Publishers.
- Erola, J., Jalonen, S., & Lehti, H. (2016). Parental education, class and income over early life course and children's achievement. *Research in Social Stratification and Mobility*, 44, 33-43.
- Entwisle, D. R., & Astone, N. M. (1994). Some practical guidelines for measuring youth's race/ethnicity and socioeconomic status. *Child Development*, 65, 1521–1540.
- Eriksson, K., Lindvall, J., Helenius, O., & Ryve, A. (2021). Socioeconomic Status as a Multidimensional Predictor of Student Achievement in 77 Societies. *Frontiers in Education, 6:* 731634.
- Eriksson, M., Marschik, P. B., Tulviste, T., Almgren, M., Pereira, M. P., Wehberg, S.
 Marjanovič-Umek, L., Gayraud, F., Kovacevic, M., Gallefo, C. (2012). Differences
 between girls and boys in emerging language skills: evidence from 10 language
 communities. *British Journal of Developmental Psychology*, *30*, 326 343.
- Evans, G. W., & Cassells, R. C. (2014). Childhood Poverty, Cumulative Risk Exposure, and Mental Health in Emerging Adults. *Clinical Psychological Science*, *2*, 287-296.
- Fanti, K. A., & Henrich, C. C. (2010). Trajectories of pure and co-occurring internalizing and externalizing problems from age 2 to age 12: Findings from the National Institute of

Child Health and Human Development Study of Early Child Care. *Developmental Psychology*, *46*, 1159–1175.

- Farrant & Zubrick, 2012). Early vocabulary development: The importance of joint attention and parent-child book reading. *First Language*, *32*, 343-364.
- Feinstein, J. S. (1993). The relationship between socioeconomic status and health: A review of the literature. *The Milbank Quarterly*, *71*, 279-322.
- Feinstein, L. (2003). Inequality in the Early Cognitive Development of British Children in the 1970 Cohort. *Economica*, *70*, 73–97.
- Fernald A, Marchman VA, Weisleder A. (2013). SES differences in language processing skill and vocabulary are evident at 18 months. *Developmental Science*, *16*, 1–13.
- Festin K, Thomas K, Ekberg J., & Kristenson M. (2017). Choice of measure matters: A study of the relationship between socioeconomic status and psychosocial resources in a middle-aged normal population. *PLoS ONE*, *12*, e0178929.
- Figge, C. J., Martinez-Torteya, C., & Weeks, J. E. (2017). Social–ecological predictors of externalizing behavior trajectories in at-risk youth. *Development and Psychopathology*, 30, 255–266.
- Fisher, S. D., Brock, R. L., O'Hara, M. W., Kopelman, R. S. (2015). Longitudinal contribution of maternal and paternal depression to toddler behaviors: Interparental

conflict and later depression as mediators. *Couple and Family Psychology: Research and Practice*, *4*, 61-73.

- Fitzsimons, E. (2020). *Millennium Cohort Study Sixth Survey 2015-2016: User Guide* (*Second Edition*). London: UCL Centre for Longitudinal Studies.
- Fitzsimons, E., Goodman, A., Kelly, E., & Smith, J. P. (2017). Poverty dynamics and parental mental health: Determinants of childhood mental health in the UK. Soc. Sci. Med, 175, 43–51.
- Fitzsimons, E., Haselden, L., Smith, K., Gilbert, E., Calderwood, L., Agalioti- Sgompou, V., Veeravalli, S., Silverwood, R., Ploubidis, G. (2020) *Millennium Cohort Study Age 17 Sweep (MCS7): User Guide*. London: UCL Centre for Longitudinal Studies.
- Flouri, E., & Midouhas, E. (2017). Environmental Adversity and Children's Early Trajectories of Problem Behavior: The Role of Harsh Parental Discipline. *Journal of Family Psychology*, 31, 234-243.
- Flouri, E., Midouhas, E., Charman, T., & Sarmadi, Z. (2015). Poverty and the Growth of Emotional and Conduct Problems in Children with Autism With and Without Comorbid ADHD. *Journal of Autism and Developmental Disorders*, 45, 2928–2938.
- Flouri, E., Midouhas, E., & Ruddy, A. (2016). Socio–economic status and family structure differences in early trajectories of child adjustment: Individual and neighbourhood effects. *Health & Place*, 37, 8–15.

- Flouri, E., Papachristou, E., Midouhas, E., Joshi, H., Ploubidis, G. B., & Lewis, G. (2018).
 Early adolescent outcomes of joint developmental trajectories of problem behavior and IQ in childhood. *European Child & Adolescent Psychiatry*, 27, 1595–1605.
- Flouri, E., Midouhas, E., Charman, T., & Sarmadi, Z. (2015). Poverty and the Growth of Emotional and Conduct Problems in Children with Autism With and Without Comorbid ADHD. *Journal of Autism and Developmental Disorders*, 45, 2928–2938.
- Fomby, P., & Musick, K. (2018). Mothers' Time, the Parenting Package, and Links to Healthy Child Development. *Journal of Marriage and Family*, 80, 166-181.
- Forrest, C, Gibson, J. L., Halligan, S., & St Clair, M. (2018) .A longitudinal analysis of early language difficulty and peer problems on later emotional difficulties in adolescence:
 Evidence from the Millennium Cohort Study. *Autism & Developmental Language Impairments*, *3*, 2396941518795392.
- Fuhrmann, D., Knoll, L. J., Blakemore, S. J. (2015). Adolescence as a sensitive period of brain development. *Trends in Cognitive Sciences*, 19, 558-566.
- Fujishiro, K., Xu, J., & Gong, F. (2010). What does "occupation" represent as an indicator of socioeconomic status?: Exploring occupational prestige and health. *Social Science & Medicine*, 71, 2100–2107.

- Fuller-Thomson, E., Nosrati-Inanlou, M., Sellors, A., & MacNeil, A. (2022). Flourishing mental health among adults with child welfare contact during childhood: Findings from a nationally representative Canadian survey. *Psychiatry Research*, 316, 114660.
- Furnham, A., & Cheng, H. (2019). The Big-Five personality factors, mental health, and social-demographic indicators as independent predictors of gratification delay. *Personality and Individual Differences*, 150, 109533
- Francis-Devine, B. (2022). *Poverty in the UK: Statistics*. Commons Library Research Briefing.
- Franke, H. A. (2014). Toxic Stress: Effects, Prevention and Treatment. *Children (Basel), 3,* 390-402.
- Gabard-Durnam, L., & McLaughlin, K. A. (2020). Sensitive periods in human development:
 Charting a course for the future. *Current Opinion in Behavioral Sciences*, *36*, 120–128.
- Gallo, L. C., & Matthews, K. A. (2003). Understanding the association between socioeconomic status and physical health: Do negative emotions play a role? *Psychological Bulletin*, 129, 10–51.
- Galobardes, B. (2006). Indicators of socioeconomic position (part 1). *Journal of Epidemiology & Community Health*, 60, 7–12.

- Galobardes, B. (2006). Indicators of socioeconomic position (part 2). *Journal of Epidemiology & Community Health*, 60, 95–101.
- Galvan, M. J., Payne, K., Hannay, J., Georgeson, A., & Muscatell, K. (2022). What does the MacArthur Scale of Subjective Social Status Measure? Separating Economic Circumstances and Social Status to Predict Health. *PsyArXiv Preprints*.
- Gariepy, G., Elgar, F. J., Sentenac, M., & Barrington-Leigh, C. (2017). Early-life family income and subjective well-being in adolescents. *PLOS ONE*, *12*, e0179380.
- Gay, B., Sonnenschein, S., Sun, S., & Baker, L. (2020). Poverty, Parent Involvement, and Children's Reading Skills: Testing the Compensatory Effect of the Amount of Classroom Reading Instruction. *Early Education and Development*, 32, 1–13.

Geiser, C. (2013). Data analysis with Mplus. New York: Guilford Press.

- Gerstein, E. D., Woodman, A. C., Burnson, C., Cheng, E. R., & Poehlmann-Tynan, J. (2017).
 Trajectories of Externalizing and Internalizing Behaviors in Preterm Children
 Admitted to a Neonatal Intensive Care Unit. *The Journal of Pediatrics, 187*, 111–118.
- Geyer, S., Hemström, Ö, P. R., & Vågerö, D. (2006). Education, income, and occupational class cannot be used interchangeably in social epidemiology. Empirical evidence against a common practice. *Journal of Epidemiology & Community Health*, 60, 804– 10.

- Gibb, S. J., Fergusson, D. M., & Horwood, L. J. (2012). Childhood family income and life outcomes in adulthood: Findings from a 30-year longitudinal study in New Zealand. *Social Science & Medicine*, 74, 1979-1986.
- Gilkerson, J., Richards, J. A., Warren, F. Oller, K., & Russo, R. (2018). Language Experience in the Second Year of Life and Language Outcomes in Late Childhood. *Pediatrics*, 142, e20174276
- Gillespie-Lynch, K., Sepeta, L., Wang, Y., Marshall, S., Gomez, L., Sigman, M., & Hutman, T. (2012). Early Childhood Predictors of the Social Competence of Adults with Autism. *Journal of Autism Developmental Disorder*, 42, 161–174.
- Girard, L-C., Pingault, J-B., Doyle, O., Falissard, B., & Tremblay, R. E. (2017). Expressive language and prosocial behaviour in early childhood: Longitudinal associations in the UK Millennium Cohort Study. *European Journal of Developmental Psychology*, 14, 381-398.
- Gluschkoff, K., Oksman, E., Knafo-Noam, A., Dobewall, H., Hintsa, T., Keltikangas-Järvinen, L. (2018). The early roots of compassion: from child care arrangements to dispositional compassion in adulthood. *Personality and Individual differences, 129*, 28–32.
- Goldthorpe, J.H. (1997). The "Goldthorpe" Class Schema: Some Observations on Conceptual and Operational Issues in Relation to the ESRC Review of Government

Social Classification. In D. Rose & K. O'Reilly (Eds.), Constructing Classes: Towards a New Social Classification for the UK. Swindon: ESRC/ONS.

- Goldthorpe, J. H. (2016). Social class mobility in modern Britain: changing structure, constant process. *Journal of the British Academy*, *4*, 89-111.
- Goldthorpe, J. H., & Jackson, M. (2007). Intergenerational class mobility in contemporary Britain: political concerns and empirical findings1. *The British Journal of Sociology*, 58, 525–546.
- Gonzalez-Casanova, I., DiGirolamo, A., Kroker-Lobos, M. F., Ochaeta, L., Ramirez-Zea, M., Martorell, R., & Stein, A. D. (2021). Association between early child development trajectories and adult cognitive function in a 50-year longitudinal study in Guatemala. *BMJ Open*, *11*, e044966.
- Gonzalez, L., Cortes-Sancho, R., Murcia, M., Ballester, F., Rebagliato, M., & Rodriguez-Bernal, C. L. (2020). The role of parental social class, education, and unemployment on child cognitive development. *Gaceta Sanitaria*, 34, 51-60.
- Goodman, A., & Gregg, P. (2010) *Poorer children's educational attainment: how important are attitudes and behaviour?*. Joseph Rowntree Foundation, London.
- Goodman, A., Joyce, R., & Smith, J. P. (2011). The long shadow cast by childhood physical and mental problems on adult life. *Proceedings of the National Academy of Sciences of the United States of America, 108,* 6032-6037.

- Goodman, A., Lamping, D. L., & Ploubidis, G. B. (2010). When to Use Broader Internalising and Externalising Subscales Instead of the Hypothesised Five Subscales on the Strengths and Difficulties Questionnaire (SDQ): Data from British Parents, Teachers and Children. *Journal of Abnormal Child Psychology*, 38, 1179–1191.
- Goodman, R. (1997). The Strengths and Difficulties Questionnaire: A research note. *Journal* of Child Psychology and Psychiatry, 38, 581–586.
- Goodman, R. D., Miller, M. D., & West-Olatunji, C. A. (2012). Traumatic stress, socioeconomic status, and academic achievement among primary school students. *Psychological Trauma: Theory, Research, Practice, and Policy*, 4, 252–259.
- Goodman, R., Renfrew, D., & Mullick, M. (2000). Predicting type of psychiatric disorder from strengths and difficulties questionnaire (SDQ) scores in child mental health clinics in London and Dhaka. *European Child & Adolescent Psychiatry*, 9, 129–134.

Goodman, R: SDQ Info. (2023). www.sdqinfo.org 2023. Accessed 20/07/2023.

Gorard, S. (2012). Who Is Eligible for Free School meals? Characterising Free School Meals as a Measure of Disadvantage in England. *British Educational Research Journal, 38*, 1003–1017.

- Gottfried, A. W. (1984). Home environment and early cognitive development: Integration, meta-analyses, and conclusions. In A. W. Gottfried (Ed.), Home Environment and Early Cognitive Development. New York: Academic Press. Pp 329-342.
- Gottfried, A. W. (1985). Measures of Socioeconomic Status in Child Development Research: Data and Recommendations. *Merrill-Palmer Quarterly*, *31*, 85-92.
- Green, M. J., & Popham, F. (2016). Life course models: improving interpretation by consideration of total effects. *International Journal of Epidemiology*, *46*, 1057-1062.
- Green, M. J., Stritzel, H., Smith, C., Popham, F., & Crosnoe, R. (2018). Timing of poverty in childhood and adolescent health: Evidence from the US and UK. *Social Science & Medicine*, 197, 136–143.
- Grundy, E., & Holt, G. (2001). The socioeconomic status of older adults: How should we measure it in studies of health inequalities? *Journal of Epidemiology & Community Health*, 55, 895–904.
- Guo, G. (1998). The Timing of the Influences of Cumulative Poverty on Children's Cognitive Ability and Achievement. *Social Forces*, *77*, 257-287.
- Guo, X., Lv, B., Zhou, H., Liu, C., Liu, J., Jiang, K., & Luo, L. (2018). Gender Differences in How Family Income and Parental Education Relate to Reading Achievement in China: The Mediating Role of Parental Expectation and Parental Involvement. *Frontiers in Psychology*, *9*, 103389.

- Gustafsson, J. E. (2010). Longitudinal designs. In Creemers, B. P. M., Leonidas Kyriakides,
 & Sammons, P. *Methodological advances in educational effectiveness research* (pp. 77-102). London Routledge.
- Gustafsson J-E, Allodi Westling M, Åkerman A, Eriksson C, Eriksson L, Fischbein S.
 (2010). School, learning and mental health: A systematic review. *Stockholm: Health Committee of The Royal Swedish Academy of Sciences*, 26, 102018.
- Gustavson, K., von Soest, T., Karevold, E., & Røysamb, E. (2012). Attrition and generalizability in longitudinal studies: findings from a 15-year population-based study and a Monte Carlo simulation study. *BMC Public Health*, 12, 918.
- Gutman, L. M., & Codiroli McMaster, N. (2020). Gendered Pathways of Internalizing Problems from Early Childhood to Adolescence and Associated Adolescent Outcomes. *Journal of Abnormal Child Psychology*, 48, 703–718.
- Gutman, L. M., Joshi, H., & Schoon, I. (2019) Developmental Trajectories of Conduct
 Problems and Cumulative Risk from Early Childhood to Adolescence. *Journal of Youth and Adolescence* 48, 181–198.
- Hair, N. L., Hanson, J. L., Wolfe, B. L., & Pollak, S. D. (2015). Association of child poverty, brain development, and academic achievement. *JAMA Pediatrics*, 169, 822-829.
- Harding, J. F. (2015). Increases in maternal education and low-income children's cognitive and behavioral outcomes. *Developmental Psychology*, *51*, 583-599.
- Hadzic, R., Magee, C. M., & Robinson, L. (2013). Parental employment and child behaviors:Do parenting practices underlie these relationships?. *International Journal of Behavioral Development 37*, 332–339.
- Hällsten, M., & Thaning, M. (2022). Wealth as One of the "Big Four" SES Dimensions in Intergenerational Transmissions. *Social Forces*, 100, 1533–1560.
- Hambrick, E. P., Brawner, T. W., & Perry, B. D. (2019). Timing of Early-Life Stress and the Development of Brain-Related Capacities. *Frontiers in Behavioural Neuroscience*, 13, 10.3389.
- Hanscombe, K. B., Trzaskowski, M., Haworth, C. M. A., Davis, O. S. P., Dale, P. S., &
 Plomin, R. (2012) Socioeconomic Status (SES) and Children's Intelligence (IQ): In a
 UK-Representative Sample SES Moderates the Environmental, Not Genetic, Effect
 on IQ. *PLoS ONE*, *7*, e30320.
- Hansen, K. (2014). Millennium Cohort Study A Guide to the Datasets (Eighth Edition) First, Second, Third, Fourth and Fifth Surveys. Institute of Education.
- Hansen, K., & Joshi, H. (2007). *Millennium Cohort Study Second Survey: A User's Guide to Initial Findings*. UCL Discovery.

- Harkness, S., Gregg, P., & Fernández-Salgado, M. (2019). The Rise in Single-MotherFamilies and Children's Cognitive Development: Evidence From Three British BirthCohorts. *Child Development*, 91, 1762-17855.
- Hart, B., & Risley, T. (1995). Meaningful differences in the everyday experience of young American children. Baltimore, MD: Paul H. Brookes.
- Hatcher, P. J., Hulme, C. & Ellis, A. W. (1994). Ameliorating early reading failure by integrating the teaching of reading and phonological skills: The phonological linkage hypothesis. *Child Development*, 65, 41–57.
- Hauser-Cram, P., & Woodman, A. C. (2015). Trajectories of Internalizing and Externalizing
 Behavior Problems in Children with Developmental Disabilities. *Journal of Abnormal Child Psychology*, 44, 811–821.

Health and Social Care Act (2012). C. 7.

- Hensch, T. K. (2018). Critical periods in cortical development. In R. Gibb & B. Kolb (Eds.), *The neurobiology of brain and behavioral development* (pp. 133–151). Elsevier Academic Press.
- Hentges, R. F., Madigan, S. L., Plamondon, A. R., Racine, N. M., Collisson, B. A., Tough, S. C., & Graham, S. A. (2019). Heterogeneous Trajectories of Delayed Communicative Development From 12 to 36 Months: Predictors and Consequences. *Journal of Developmental & Behavioral Pediatrics*, 40, 335-343.

- Hernandez-Alava, M., & Popli, G. (2017). Children's development and parental input: Evidence from the UK millennium cohort study. *Demography*, *54*, 485-511.
- Hertzman, C., Power, C., Matthews, S. (2001). Using an interactive framework of society and lifecourse to explain self-rated health in early adulthood. *Social Science & Medicine*, 53,1575–1585.
- Hessel, H., He, Y., & Dworkin, J. (2017). Paternal monitoring: The relationship between online and in-person solicitation and youth outcomes. *Journal of Youth and Adolescence*, 46, 288–299.
- Hill, V. (2005). Through the past darkly: A review of the british ability scales second edition. Child and Adolescent Mental Health, 10, 87-98.
- Hills, J., LeGrand, J., & Piachaud, D. (2002). Understanding social exclusion. New York: Oxford University Press.
- Hindman, A. H., & Morrison, F. J. (2012). Differential Contributions of Three Parenting
 Dimensions to Preschool Literacy and Social Skills in a Middle-Income
 Sample. *Merrill-Palmer Quarterly*, 58, 191–223.
- Hirsh-Pasek, K., Adamson, L. B., Bakeman, R., Owen, M. T., Golinkoff, R. M., Pace, A., Yust, P. K. S., & Suma, K. (2015). The contribution of early communication quality to low-income children's language success. *Psychological Science*, 26, 1071–1083.

- Hjalmarsson, S., & Mood, C. (2015). Do poorer youth have fewer friends? The role of household and child economic resources in adolescent school-class friendships. *Children and Youth Services Review*, 57, 201–211.
- Hobbs, G. and Vignoles, A. (2010) Is children's free school meal 'eligibility' a good proxy for family income? *British Educational Research Journal*, *36*, 673–690.
- Hoferichter F, Kulakow S and Hufenbach MC (2021) Support From Parents, Peers, and Teachers Is Differently Associated With Middle School Students' Well-Being. *Frontiers in Psychology 12*, 758226.
- Hofler, M., Pfister, H., Lieb, R., & Wittchen, H.-U. (2005). The use of weights to account for non-response and drop-out. *Social Psychiatry and Psychiatric Epidemiology*, 40, 291– 299.
- Hoff, E. (2003). Causes and consequences of SES-related differences in parent-to-child speech. In M. H. Bornstein & R. H. Bradley (Eds.), *Socioeconomic status, parenting, and child development* (pp. 147–160). Lawrence Erlbaum Associates Publishers.
- Hoff, E. (2013). Interpreting the early language trajectories of children from low-SES and language minority homes: Implications for closing achievement gaps. *Developmental Psychology*, 49, 4–14.
- Hoff, E., & Laursen, B. (2019). Socioeconomic status and parenting. In M. H. Bornstein (Ed.), *Handbook of parenting: Biology and ecology of parenting* (pp. 421–447).
 Routledge/Taylor & Francis Group.

- Hoffmann, R., Kröger, H., Geyer, S. (2019). Social causation versus health selection in the life course: does their relative importance differ by dimension of ses?. *Social Indicators Research*, 141, 1341–1367.
- Hofler, M., Pfister, H., Lieb, R., & Wittchen, H. (2005). The use of weights to account for non-response and drop-out. *Social Psychiatry and Psychiatric Epidemiology* 40, 291– 299.
- Holmes, J., & Kiernan, K. (2013). Persistent poverty and children's development in the early years of childhood. *Policy & Politics, 41*, 19–42.
- Hornby, G., & Blackwell, I. (2018). Barriers to parental involvement in education: An update. *Educational Review*, *70*(1), 109–119.
- Hornby, G., & Lafaele, R. (2011). Barriers to parental involvement in education: An explanatory model. *Educational Review*, 63, 37–52.
- Hosokawa, R., & Katsura, T. (2017). A longitudinal study of socioeconomic status, family processes, and child adjustment from preschool until early elementary school: the role of social competence. *Child and Adolescent Psychiatry and Mental Health*, *11*, 62.

Hosokawa, R., & Katsura, T. (2018). Effect of socioeconomic status on behavioral problems from preschool to early elementary school – A Japanese longitudinal study. *PLoS ONE*, *13*, e0197961.

House of Commons. (2022). Social security support for children. House of Commons.

- Houweling, T. A. J., Oude Groeniger, J., Jansen, P. W., van Lier, P., Horoz, N., Buil, M., van Lenthe, F. J. (2022). Trajectories of socioeconomic inequality in early child development: a cohort analysis. *International Journal of Equity Health*, 21, 101186.
- Howlin, P., & Magiati, I. (2017). Autism spectrum disorder: outcomes in adulthood. *Current Opinion in Psychiatry*, 30, 69-76.
- Hoyt, L. T., Sabol, T. J., Chaku, N., Kessler, C. L. (2019). Family income from birth through adolescence: Implications for positive youth development. *Journal of Applied Developmental Psychology* 64: 101055.
- Hukkelberg, S., Keles, S., Ogden, T., & Hammerstom, K. (2019). The relation between behavioral problems and social competence: A correlational Meta-analysis. *BMC Psychiatry 19*, 354.
- Hulme, C., Snowling, M. J., West, G., Lervåg, A., & Melby-Lervåg, M. (2020). Children's
 Language Skills Can Be Improved: Lessons From Psychological Science for
 Educational Policy. *Current Directions in Psychological Science*, 29, 372–377

- Hulme, C., Nash, H. M., Gooch, D., Lervåg, A., & Snowling, M. J. (2015). The foundations of literacy development in children at familial risk of dyslexia. *Psychological Science*, 26, 1877–1886.
- Huttenlocher, J., Waterfall, H., Vasilyeva, M., Vevea, J., & Hedges, L. V. (2010). Sources of variability in children's language growth. *Cognitive Psychology*, *61*, 343-365.
- Information Services Division. (2019). *Child health 27-30 month review statistics technical report*. NHS Scotland: Information Services Division.
- Ipsos MORI. (2017). *Millennium Cohort Study Sixth Sweep (MCS6) Technical Report*. Social Research Institute.
- Iruka, I. U., LaForett, D. R., & Odom, E. C. (2012). Examining the validity of the family investment and stress models and relationship to children's school readiness across five cultural groups. *Journal of Family Psychology*, 26, 359–370.
- Irwin, S., & Elley, S. (2013). Parents' Hopes and Expectations for Their Children's Future Occupations. The Sociological Review, 61, 111–130.
- Jackson, M. I., Kiernan, K., & McLanahan, S. (2017). Maternal education, changing family circumstances, and children's skill development in the united states and uk. *The Annals of The American Academy*, 674, 59-84.

- Jæger, M. M. (2011). Does Cultural Capital Really Affect Academic Achievement? New Evidence from Combined Sibling and Panel Data. Sociology of Education, 84, 281– 298.
- Jansen, D. E., Veenstra, R., Ormel, J., Verhulst, F. C., & Reijneveld, S. A. (2011). Early risk factors for being a bully, victim, or bully/victim in late elementary and early secondary education. The longitudinal TRAILS study. *BMC Public Health*, 11, 440.
- Jansen, P. W., Verlinden, M., Berkel, A. D., Mieloo, C., van der Ende, J., Veenstra, R., Verhulst, F. C., Jansen, W., & Tiemeier, H. (2012). Prevalence of bullying and victimization among children in early elementary school: Do family and school neighbourhood socioeconomic status matter? *BMC Public Health*, 12, 494.
- Jeong, J., McCoy, D. C., & Fink, G. (2017). Pathways between paternal and maternal education, caregivers' support for learning, and early child development in 44 lowand middle-income countries. *Early Childhood Research Quarterly*, 41, 136-148.
- Jerrim, J. (2020). Measuring Disadvantage: Research Brief. The Sutton Trust.
- Jin, F., Schjølberg, S., Wang, M. V., Eadie, P., Nes, R. B., Røysamb, E., & Tambs, K.
 (2020). Predicting Literacy Skills at 8 Years From Preschool Language Trajectories: A Population-Based Cohort Study. *Journal of Speech, Language, and Hearing Research*, 63, 2752–2762.

- Jocson, R. M., McLoyd, V. C. (2015). Neighborhood and housing disorder, parenting, and youth adjustment in low-income urban families. *American Journal of Community Psychology*, *55*, 304-13.
- Johnson, C. J., Beitchman, J. H., & Brownlie, E. B. (2010). Twenty-year follow-up of children with and without speech-language impairments: family, educational, occupational, and quality of life outcomes. *American Journal of Speech and Language Pathology*, 19, 51–65
- Johnson, S., Li, J., Kendall, G., Strazdins, L., & Jacoby, P. (2013). Mothers' and fathers' work hours, child gender, and behavior in middle childhood. *Journal Marriage & Family*, *75*, 56–74.
- Joinson, C., Kounali, D., & Lewis, G. (2016). Family socioeconomic position in early life and onset of depressive symptoms and depression: a prospective cohort study. *Social Psychiatry and Psychiatric Epidemiology*, 52, 95–103.
- Jonsson, F., Sebastian, M.S., Hammarström, A., & Gustafsson, P. E. (2017).
 Intragenerational social mobility and functional somatic symptoms in a northern Swedish context: analyses of diagonal reference models. *International Journal of Equity Health, 16*, 1.
- Joseph Rowntree Foundation. (2021). *Impact of the planned £20-per-week reduction in Universal Credit and Working Tax Credit*. Joseph Rowntree Foundation.

- Joseph Rowntree Foundation. (2021). *UK heading for the biggest overnight cut to the basic rate of social security since World War II*. Joseph Rowntree Foundation.
- Joseph Rowntree Foundation (2022). *UK poverty: The essential guide to understanding poverty in the UK*. Joseph Rowntree Foundation.
- Judd N., Sauce B., Wiedenhoeft J., Tromp J., Chaarani B., Schliep A., van Noort B., Penttilä
 J., Grimmer Y., Insensee C., Becker A., Banaschewski T., Bokde A.L.W., Quinlan
 E.B., Desrivières S., Flor H., Grigis A., Gowland P., Heinz A., & Klingberg T.
 (2020). Cognitive and brain development is independently influenced by
 socioeconomic status and polygenic scores for educational attainment. *Proceedings of the National Academy of the USA*, *117*, 12411–12418
- Jung, T., & Wickrama, K. A. S. (2008). An introduction to latent class growth analysis and growth mixture modelling. *Social and Personality Psychology Compass*, *2*, 302–317.
- Justice, L. M., Jiang, H., Purtell, K. M., Schmeer, K., Boone, K., Bates, R., & Salsberry, P. J. (2019). Conditions of Poverty, Parent–Child Interactions, and Toddlers' Early Language Skills in Low-Income Families. *Maternal and Child Health Journal, 23*, 971–978.
- Kaiser, T., Jianghong, L., Pollmann-Schult, M., & Song, A. Y. (2017). Poverty and Child Behavioral Problems: The Mediating Role of Parenting and Parental Well-Being. *International Journal of Environmental Research and Public Health*, 14, 981.

- Kail, R. V. (2015). *Children and Their Development (6th Edition)*. Englewood Cliffs, NJ:Prentice Hall.
- Kalil, A., Ryan, R., & Corey, M. (2012). Diverging destinies: Maternal education and the developmental gradient in time with children. *Demography*, 49, 1361–1383.
- Katz, J., & Nelson, R. A. (2007). Family Experiences and Self-Criticism in College Students: Testing a Model of Family Stress, Past Unfairness, and Self-Esteem. The American Journal of Family Therapy, 35, 447–457.
- Keiley, M. K., Bates, J. E., Dodge, K. A., & Pettit, G. S. (2010). A cross-domain growth analysis: Externalizing and internalizing behaviors during 8 years of childhood. *Journal of Abnormal Child Psychology*, 28, 161-179.
- Kelly, J., & Evans, M. D. R. (2023). Geographic disadvantage and quality of employment. Australian Journal of Social Issues, 00, 1-29.
- Kendzor, D., O'Caughy, M., & Owen, M. T. (2012). Family income trajectory during childhood is associated with adiposity in adolescence: a latent class growth analysis.*BMC Public Health*, 12: 611.
- Ketende, S. C. (2010). *Millennium Cohort Study: Technical Report on Response*. UCL Centre for Longitudinal Studies.

Khalatbari-Soltani, S., Stanaway, F., Sherrington, C., Blyth, F. M., Naganathan, V.,
Handelsman, D. J., ... & Cumming, R. G. (2021). The prospective association
between socioeconomic status and falls among community-dwelling older men. *The Journals of Gerontology: Series A*, 76, 1821-1828.

- Khaliq, M. (2020). *Child poverty and education outcomes by ethnicity*. Office for National Statistics.
- Khan, S., Ayaz, M., Ghazi, S. R., Khan, I. U., & Khan, N. (2017). Impact of Parents'
 Occupation on Students Self-Concept at Secondary Level. *International Journal of Academic Research in Business and Social Sciences*, 7, 46-53.
- Khanam, R., & Nghien, S. (2016). Family Income and Child Cognitive and Noncognitive Development in Australia: Does Money Matter?. *Demography*, 53, 597-621.
- Kharitonova, M., Jamieson, J., Xu, Z., & Farquharson, K. (2021). Early childhood educators' language interactions and their relation to children's language outcomes: A metaanalysis. *Early Childhood Research Quarterly*, 57, 136-146.
- Kuhl, P. K., Conboy, B. T., Padden, D., Nelson, T., & Pruitt, J. (2005). Early Speech
 Perception and Later Language Development: Implications for the Critical Period.
 Language Learning and Development, 1, 237–264.

- Kiernan, K.E., & Huerta, M.C. (2008). Economic deprivation, maternal depression, parenting and children's cognitive and emotional development in early childhood. *British Journal of Sociology*, 59, 783–806.
- Kiernan, K. E. and Mensah, F. K. (2009) Poverty, maternal depression, family status and children's cognitive and behavioural development in early childhood: a longitudinal study. *Journal of Social Policy*, 38, 569–588.
- Kiernan, K. E. and Mensah, F. K. (2011) Poverty, family resources and children's early educational attainment: the mediating role of parenting. *British Educational Research Journal*, 37, 317–336.
- King, S., & Laplante, D. P. (2005). The effects of prenatal maternal stress on children's cognitive development: Project Ice Storm. *Stress*, *8*, 35–45.
- Kjeldsen, A., Nilsen, W., Gustavson, K., Skipstein, A., Melkevik, O., & Karevold, E. B.
 (2016). Predicting Well-Being and Internalizing Symptoms in Late Adolescence From Trajectories of Externalizing Behavior Starting in Infancy. *Journal of Research on Adolescence*, 26, 991–1008.
- Knipe, E. (2016). Childbearing by socio-economic status and country of birth of mother:2014. Office for National Statistics.
- Kohl, P. L., Kagotho, J. N., & Dixon, D. (2011). Parenting Practices among DepressedMothers in the Child Welfare System. *Social Work Research*, 35, 215–225.

- Kontopantelis, E., White, I. R., Sperrin, M., & Buchan, I. (2017). Outcome-sensitive multiple imputation: a simulation study. *BMC Medical Research Methodology*, *17*, 2.
- Korhonen, M., Luoma, I., Salmelin, R. K., Helminen, M., Kaltiala-Heino, R., & Tamminen, T. (2014). The trajectories of child's internalizing and externalizing problems, social competence and adolescent self-reported problems in a Finnish normal population sample. *School Psychology International*, *35*, 561–579.
- Kousgaard, S. J., Boldsen, S. K., Mohr-Jensen, C., & Lauritsen, M. B. (2018). The effect of having a child with ADHD or ASD on family separation. *Social Psychiatry and Psychiatric Epidemiology*, 53, 1391–1399.
- Kozyrskyj, A. L., Kendall, G. E., Jacoby, P., Sly, P.D., & Zubrick, S. R. (2010). Association between socioeconomic status and the development of asthma: analyses of income trajectories. *American Journal of Public Health*, 100, 540-6.
- Kraft, P., Kraft, B., Hagen, T., & Espeseth, T. (2022). Subjective socioeconomic status, cognitive abilities, and personal control: associations with health behaviours. *Frontiers in Psychology*, 12:784758.
- Krapohl, E., & Plomin, R. (2016). Genetic link between family socioeconomic status and children's educational achievement estimated from genome-wide SNPs. *Molecular Psychiatry*, 21, 437-43.

- Kremer, K. P., Flower, A., Huang, J., & Vaughn, M. G. (2016). Behavior problems and children's academic achievement: A test of growth-curve models with gender and racial differences. *Children and Youth Services Review*, 67, 95–104.
- Kromydas, T., Campbell, M., Chambers, S., Boon, M. H., Pearce, A., Wells, V., & Craig, P. (2022). The effect of school summer holidays on inequalities in children and young people's mental health and cognitive ability in the UK using data from the millennium cohort study. *BMC Public Health*, 22, 154.
- Kuang, Y., & Flouri, E. (2020). Spatial working memory in young adolescents with different childhood trajectories of internalizing, conduct and hyperactivity/inattention problems. *British Journal of Educational Psychology*, *91*, 827-849.
- Kuh, D., Ben-Shlomo, Y., Lynch, J., Hallqvist, J., & Power, C. (2003). Life course epidemiology. *Journal of Epidemiology & Community Health*, 57, 778–783.
- Kwon, E., Park, S., & Lee, H. (2022). Early-Life Socioeconomic Disadvantage and Health in Late Middle-Age: Importance of Heterogeneous Income Trajectories. *Research on Aging*, 0, 1-17.
- Lai, E. T. C., Wickham, S., Law, C., Whitehead, M., Barr, B., & Taylor-Robinson, D. (2019).
 Poverty dynamics and health in late childhood in the UK: evidence from the
 Millennium Cohort Study. *Disease in Childhood, 104*: 316702.

- Laplante, D. P., Barr, R. G., Brunet, A., Du Fort, G. G., Meaney, M. L., Saucier, J.-F.,
 Zelazo, P. R., & King, S. (2004). Stress During Pregnancy Affects General
 Intellectual and Language Functioning in Human Toddlers. *Pediatric Research*, 56, 400–410.
- Landa, R. J., Gross, A. L., Stuart, E. A., & Bauman, M. (2012). Latent class analysis of early developmental trajectory in baby siblings of children with autism. *Journal of Child Psychology and Psychiatry*, 53, 986-96.
- Landa, R. J., Gross, A. L., Stuart, E. A., & Faherty, A. (2012). Developmental trajectories in children with and without autism spectrum disorders: the first 3 years. *Child Development*, 84, 429-442.
- Landers-Potts, M. A., Wickrama, K. A. S., Simons, L. G., Cutrona, C., Gibbons, F. X., Simons, R. L., & Conger, R. (2015). An extension and moderational analysis of the family stress model focusing on african american adolescents. *Family Relations, Interdisciplinary Journal of Applied Family Studies, 64*, 233-248.
- Lange, B. P., Euler, H. A., & Zaretsky, E. (2016). Sex differences in language competence of3- to 6-year-old children. *Applied Psycholinguistics*, 37, 1417–1438.
- Langlois, S., Zern, A., Anderson, S., Ashekun, O., Ellis, S., Graves, J., & Compton, M. T.
 (2020). Subjective Social Status, Objective Social Status, and Substance Use among Individuals with Serious Mental Illnesses. *Psychiatry Research*, 293, 113352.

Lansford, J. E., Malone, P. S., Tapanya, S., Tirado, L. M. U., Zelli, A., Alampay, L. P., Al-Hassan, S. M., Bacchini, D., Bornstein, M. H., Chang, L., Deater-Deckard, K., Giunta, L. D., Dodge, K. A., Oburu, P., Pastorelli, C., Skinner, A. T., Sorbring, E., & Steinberg, L. (2018). Household income predicts trajectories of child internalizing and externalizing behavior in high-, middle-, and low-income countries. *International Journal of Behavioral Development*, *43*, 74–79.

Larsen, R. (2011). Missing data imputation versus full information maximum likelihood with second-level dependencies. *Structural Equation Modeling*, *18*, 649–662.

Lau, T. W. I., Lim, C. G., Acharryya, S., Lim-Ashworth, N., Tan, Y. R., & Fung, S. S. D.

(2021). Gender differences in externalizing and internalizing problems in Singaporean children and adolescents with attention-deficit/hyperactivity disorder. *Child and Adolescent Psychiatry and Mental Health*, *15*, 101186.

- Lawson, M., Piel, M. H., & Simon, M. (2020). Child Maltreatment during the COVID-19 Pandemic: Consequences of Parental Job Loss on Psychological and Physical Abuse Towards Children. *Child Abuse & Neglect*, 110, 104709.
- Lazarsfeld, P. F. (1939). Interchangeability of indices in the measurement of economic influences. *Journal of Applied Psychology*, 23, 33–45.

Le Cam, L. (1986). The Central Limit Theorem Around 1935. Statistical Science, 1, 78–91.

Lee, D. (2014). Age Trajectories of Poverty During Childhood and High School Graduation. *Sociological Science*, *1*, 344–365.

- Lee, D. Y. Harring, J. R., & Stapleton, L. M. (2019). Comparing Methods for Addressing Missingness in Longitudinal Modeling of Panel Data. *Journal of Experimental Education*, 87, 596-615.
- Lee, Y., Liu, X., & Watson, M. W. (2016). The Timing Effect of Bullying in Childhood and Adolescence on Developmental Trajectories of Externalizing Behaviors. *Journal of Interpersonal Violence*, 31, 2775–2800.
- Lee, K., & Zhang, L. (2022). Cumulative Effects of Poverty on Children's Social-Emotional Development: Absolute Poverty and Relative Poverty. *Community Mental Health Journal*, 58, 930–943.
- Letourneau, N. L., Duffett-Leger, L., Levac, L., Watson, B., & Young-Morris, C. (2013). Socioeconomic status and child development: A meta-analysis. *Journal of Emotional* and Behavioral Disorders, 21, 211–224.
- Leve, L. D., Kim, H. K., & Pears, K. C. (2005). Childhood Temperament and Family Environment as Predictors of Internalizing and Externalizing Trajectories From Ages 5 to 17. *Journal of Abnormal Child Psychology*, 33, 505–520.
- Levy, P. & Goldstein, H. (1984). *Tests in Education: a book of critical reviews*. London: Academic Press.

- Li, B., Bos, M. G., Stockmann, L., & Rieffe, C. (2020). Emotional functioning and the development of internalizing and externalizing problems in young boys with and without autism spectrum disorder. *Autism*, 24, 200–210.
- Li, J., Wang, J., Li, J., Qian, S., Jia, R., Wang, Y., Liang, J., & Xu, Y. (2020). How do socioeconomic status relate to social relationships among adolescents: a school-based study in East China. *BMC Pediatrics*, 20, 271.
- Liberatos, P., Link, B. G., & Kelsey, J. L. (1988). The measurement of social class in epidemiology. *Epidemiology Review*, *10*, 87–121.
- Lim, A. J.-M. ., & Cheung, M. W.-L. . (2021). Evaluating FIML and multiple imputation in joint ordinal-continuous measurements models with missing data. *Behavior Research Methods*, 54, 1063-1077.
- Lin, X. (2016). Barriers and challenges of female adult students enrolled in higher education: A literature review. *Higher Education Studies*, *6*, 119-126.
- Liu, J. (2004). Childhood Externalizing Behavior: Theory and Implications. *Journal of Child* and Adolescent Psychiatric Nursing, 17,93-103.
- Liu, Y., & Merritt, D. H. (2018). Familial financial stress and child internalizing behaviors: The roles of caregivers' maltreating behaviors and social services. *Child Abuse & Neglect*, 86, 324–335.

- Little, R. J. A. (1988). A Test of Missing Completely at Random for Multivariate Data with Missing Values. *Journal of the American Statistical Association*, *83*, 1198–1202.
- Little, T.D., Jorgensen, T. D., Lang, M. A., & Moore, W. G. (2013). On the joys of missing data. *Journal of Pediatric Psychology*, 39, 151–162,
- Lohndorf, R. T., Vermeer, H. J., Carcamo, R. A., & Mesman, J. (2017). Preschoolers' vocabulary acquisition in Chile: the roles of socioeconomic status and quality of home environment. *Journal of Child Language*, 21, 1-22.
- Lo, Y., Mendell, N. R., & Rubin, D. B. (2001). Testing the Number of Components in a Normal Mixture. *Biometrika*, 88, 767–778.
- Loken, K. V., Mogstad, M., & Wiswall, M. (2010). What Linear Estimators Miss: The Effects of Family Income on Child Outcomes. *American Economic Journal: Applied Economics 4*, 1–35.
- Longard, J., Brian, J., Zwaigenbaum, L., Duku, E., Moore, C., Smith, I. M., Garon, N., Szatmari, P., Vaillancourt, T., & Bryson, S. (2017). Early expressive and receptive language trajectories in high-risk infant siblings of children with autism spectrum disorder. *Autism & Developmental Language Impairments*, 2.
- Longo, F., McPherran Lombardi, C., & Dearing, E. (2017). Family investments in lowincome children's achievement and socioemotional functioning. *Developmental Psychology*, 53, 2273–2289.

- Love R, Adams J, Atkin A, & van Sluijs, E. (2019). Socioeconomic and ethnic differences in children's vigorous intensity physical activity: a cross-sectional analysis of the UK Millennium Cohort Study. *BMJ Open*, 9, e027627.
- Low Pay Commission (2022). *The National Living Wage Review (2015-2020)*. Low Pay Commission.
- Löytömäki, J., Laakso, M., & Huttunen, K. (2022). Social-Emotional and Behavioural
 Difficulties in Children with Neurodevelopmental Disorders: Emotion Perception in
 Daily Life and in a Formal Assessment Context. *Journal of Autism and Developmental Disorders, 3*, 36184695.
- Lynch, J. W., Kaplan, G. A., Cohen, R. D., Kauhanen, J., Wilson T. W., Smith, N. L., & Salonen, J. T. (1994). Childhood and adulthood socioeconomic status as predictors of mortality in Finland. *Public Health*, 343, 524-527.
- Ma, J., Han, Y., Grogan-Kaylor, A., Delva, J., & Castillo, M. (2012). Corporal punishment and youth externalizing behavior in Santiago, Chile. *Child Abuse & Neglect, 36*, 481– 490.
- Madigan, S., Prime, H., Graham, S. A., Rodrigues, M., Anderson, N., Khoury, J., & Jenkins, J. M. (2019). Parenting Behavior and Child Language: A Metaanalysis. *Pediatrics*, 144, e20183556.

- Magnuson, K. (2003). The effect of increases in welfare mothers' education on their young children's academic and behavioral outcomes. University of Wisconsin. Institute for Research on Poverty.
- Magnuson, K., Sexton, H., Davis-Kean, P., & Huston, A. (2009). Increases in Maternal Education and Young Children's Language Skills. *Merrill-Palmer Quarterly*, 55, 319-350.
- Mangin, K. S., Horwood, L. J., & Woodward, L. J. (2017). Cognitive Development Trajectories of Very Preterm and Typically Developing Children. *Child Development*, 88, 282-298.
- Manley, J., Fernald, L., & Gertler, P. (2015). Wealthy, healthy and wise: does money compensate for being born into difficult conditions?. *Applied Economics Letters*, 22, 121-126.
- Marchman VA, Fernald A. (2008). Speed of word recognition and vocabulary knowledge in infancy predict cognitive and language outcomes in later childhood. *Developmental Science*, *11*, 9–16.
- Markant, J., Ackerman, L. K., Nussenbaum, K., & Amso, D. (2016). Selective attention neutralizes the adverse effects of low socioeconomic status on memory in 9-monthold infants. *Developmental Cognitive Neuroscience*, 18, 26–33.

- Martinez, W., & Polo, A. J. (2018). Neighborhood Context, Family Cultural Values, and Latinx Youth Externalizing Problems. *Journal of Youth and Adolescence*, 47, 2440-2452.
- Marryat, L., & Martin, C. (2010). *Growing up in scotland: Maternal mental health and its impact on child behaviour and development*. Scottish Government.
- Marryat, L., Thompson, L., Minnis, H., & Wilson, P. (2018). Primary scools and the amplification of social differences in child mental health: a population based cohort study. *Journal of Epidemiology & Community Health*, 72, 27-33.
- Marx, K. (1967). Capital: A critique of political economy (Vol. 1). New York: International.
- Masarik, A. S., & Conger, R. D., (2017). Stress and Child Development: A Review of the Family Stress Model. *Current Opinion in Psyhcology*, *13*, 85-90.
- Mathur, M., Robiolio, H., Ebert, L., & Kerr, B. (2023). Family stress during the COVID-19 pandemic: a qualitative analysis. *BMJ Open*, 13, e061396.
- Matos, A., Salvador, M., Costa, J.J., Pinheiro, M.R., Arnarson, E.O., Craighead, W.E. (2017).
 The relationship between internalizing and externalizing problems in adolescence:
 Does gender make a difference?. *International Journal of Social Science and Education*, 8, 45–63.
- Maurice-Stam, H., Haverman, L., Splinter, A., van Oers, H. A., Schepers, S. A., & Grootenhuis, M. A. (2018). Dutch norms for the Strengths and Difficulties

Questionnaire (SDQ) – parent form for children aged 2–18 years. *Health and Quality of Life Outcomes, 16,* 123.

- May, T., Rinehart, N. J., Wilding, J., & Cornish, K. (2015). Attention and basic literacy and numeracy in children with autism spectrum disorder: a one-year follow-up study. *Research in Autism Spectrum Disorders*, 9, 193–201.
- Maydeu-Olivares, A. & Garcia-Forero, C. (2010) Goodness-of-Fit Testing. *International Encyclopedia of Education*, 7, 190-196.
- Mayo, C. O., Pham, H., Patallo, B., Joos C. M., & Wadsworth. (2022). Coping with povertyrelated stress: A narrative review. *Developmental Review*, *64*, 101024.
- Mazza, J. R. S. E., Lambert, J., Zunzunegui, M. V., Tremblay, R. E., Boivin, M., & Cote, S.
 M. (2017). Early adolescence behavior problems and timing of poverty during childhood: A comparison of lifecourse models. *Social Science & Medicine*, *177*, 35-42.
- Mbatha, N. L., & Mokwena, K. E. (2023). Parental stress in raising a child with developmental disabilities in a rural community in south africa. *International Journal of Environmental Research and Public Health*, 20, 3969.
- McElroy, E., Richards, M., Fitzsimons, E., Conti, G., Ploubidis, G.B., Sullivan, A., Moulton,
 V. (2021). Feasibility of retrospectively harmonising cognitive measures in five
 British birth cohort studies. London, UK: CLOSER.

- McFarland, M. J. (2017). Poverty and problem behaviors across the early life course: The role of sensitive period exposure. *Population Research Policy Review*, *36*, 739-760.
- McFarland, M. J., & Hayward, M. D. (2014). Poverty and Awakening Cortisol in Adolescence: The Importance of Timing in Early Life. *Society and Mental Health, 4,* 21–37.
- McGuinness, P. S., Cook, N. E., Bush, H., Lind, H., Vuijk, P. J., Doyle, A. E., & Braaten, E.
 B. (2017). Children with autism spectrum disorder and comorbid specific learning disorder demonstrate cognitive weakness compared to autism spectrum disorder alone. *Journal of the American Academy of Child & Adolescent Psychiatry*, 56, S231.
- McKean, C., Mensah, F. K., Eadie, P., Bavin, E. L., Bretherton, L., Cini, E., & Reilly, S.
 (2015). Levers for Language Growth: Characteristics and Predictors of Language Trajectories between 4 and 7 Years. *PLOS ONE*, *10*, e0134251.
- McKean, C., Wraith, D., Eadie, P., Cook, F., Mensah, F., Reilly, S. (2017). Subgroups in language trajectories from 4 to 11 years: the nature and predictors of stable, improving and decreasing language trajectory groups. *Journal of Child Psychology and Psychiatry*, 58, 1081-1091.
- McLachlan, G., & Peel, D. (2000). *Finite Mixture Models*. *Wiley Series in Probability and Statistics*. John Wiley & Sons, Inc.

- McLennan, D., Noble, S., Noble, M., Plunkett, E., Wright, G., & Gutacker, N. (2019). The English Indices of Deprivation 2019: technical report. Digital Education Resource Archive.
- McLoyd, V. (1998). Socioeconomic disadvantage and child development. *American Psychologist*, 53, 185–204.
- McCoy, D. C., Zuilkowski, S. S., & Fink, G. (2015). Poverty, physical stature, and cognitive skills: Mechanisms underlying children's school enrollment in Zambia. *Developmental Psychology*, 51, 600–614.
- McLoyd, V. C. (1998). Socioeconomic disadvantage and child development. *American Psychologist*, 53, 185–204.
- McLoyd, V. C. (2019). How children and adolescents think about, make sense of, and respond to economic inequality: Why does it matter? *Developmental Psychology*, 53, 592-600.
- McRae, E., Stoppelbein, L., O'Kelly, S., Fite, P., & Smith, S. (2020). Comorbid Internalizing and Externalizing Symptoms Among Children with ADHD: The Influence of Parental Distress, Parenting Practices, and Child Routines. *Child Psychiatry & Human Development, 51*, 813-826.
- Meanley, S., Biernesser, C., O'Malley, T., Bear, T., & Trauth, J. (2020). Employing Position Generators to Assess Social Capital and Health: A Scoping Review of the Literature

and Recommendations in Future Population Health Surveillance. *Journal of Health Disparities Research and Practice, 13*, 16-45.

- Melchior, M., Touchette, É., Prokofyeva, E., Chollet, A., Fombonne, E., Elidemir, G., &
 Galéra, C. (2014). Negative Events in Childhood Predict Trajectories of Internalizing
 Symptoms Up to Young Adulthood: An 18-Year Longitudinal Study. *PLoS ONE*, 9, e114526.
- Melhuish, E. (2015). Early childhood environments: long-term consequences of early childhood education and parenting. In s. Hay (ed.), early years education and care.
 Oxford, UK: Routledge, 29–49.
- Merrill, R. M., & Slavik, K. R. (2023). Relating parental stress with sleep disorders in parents and children. *PLoS One*, 25, e0279476.
- Merwin, S. M., Leppert, K. A., Smith, V. C., Dougherty, L. R. (2017). Parental depression and parent and child stress physiology: Moderation by parental hostility. *Developmental Psychobiology*, 59, 997-1009.
- Micklewright, J. L., King, T. Z., O'Toole, K., Henrich, C., & Floyd, F. J. (2012). Parental Distress, Parenting Practices, and Child Adaptive Outcomes Following Traumatic Brain Injury. *Journal of the International Neuropsychological Society*, 18, 343–350.

- Miller, P., Podvysotska, T., Betancur, L., & Votruba-Drzal, E. (2021). Wealth and Child
 Development: Differences in Associations by Family Income and Developmental
 Stage. *The Russell Sage Foundation Journal of the Social Sciences* 7, 52–74.
- Miller, P., Betancur, L., Whitfield, K., & Votruba-Drzal, E. (2021). Examining income dynamics and externalizing and internalizing trajectories through a developmental psychopathology lens: A nationally representative study. *Development and Psychopathology*, 33, 1–17.
- Miller, P., & Votruba-Drzal, E. (2017). The Role of Family Income Dynamics in Predicting Trajectories of Internalizing and Externalizing Problems. *Journal of Abnormal Child Psychology*, 45, 543-556.
- Miller, P., & Votruba-Drzal, E., & Setodji, C. M. (2013). Family Income and Early Achievement Across the Urban–Rural Continuum. *Developmental Psychology*, 49, 1452-1465.
- Min, M. O., Minnes, S., Park, H., Ridenour, T., Kim, J.-Y., Yoon, M., & Singer, L. T. (2018).
 Developmental trajectories of externalizing behavior from ages 4 to 12: Prenatal cocaine exposure and adolescent correlates. *Drug and Alcohol Dependence*, *192*, 223–232.
- Ministry of Housing, Communities, and Local Government. (2020). *English Housing Survey*. Ministry of Housing, Communities, and Local Government.

- Milligan, K., & Stabile, M. (2011). Do Child Tax Benefits Affect the Well-Being of Children? Evidence from Canadian Child Benefit Expansions. *American Economic Journal: Economic Policy*, 3, 175 – 205.
- Mok, P. L. H., Antonsen, S., Pedersen, C. B., Carr, M. J., Kapur, N., Nazroo, J., & Webb, R.
 T. (2018). Family income inequalities and trajectories through childhood and selfharm and violence in young adults: a population-based, nested case-control study. *Lancet Public Health, 3*, 498-507.
- Mollborn, S., Lawrence, E., James-Hawkins, L., & Fomby, P. (2014). When do socioeconomic resources matter most in early childhood? *Advances in Life Course Research*, 20, 56–69.
- Monaghan, D. (2017). Does college enrolment and bachelor's completion by mothers impact children's education outcomes?. *Sociology of Education*, 90, 3-24.
- Moore, P. J., Adler, N. E., Williams, D. R., & Jackson, J. S. (2002). Socioeconomic Status and Health: The Role of Sleep. *Psychosomatic Medicine*, *64*, 337–344.
- Moore, G., Anthony, R., Angel, L. Hawkins, J., Morgan, K., Copeland, L., Murphy, S., Van Godwin, J., & Shenderovich, Y. (2022). Mental health and life satisfaction among 10–11-year-olds in Wales, before and one year after onset of the COVID-19 pandemic. *BMC Public Health 22*, 379.

- Moore, K. E., Tangney, J. P., & Stuewig, J. B. (2016). The self-stigma process in criminal offenders. *Stigma and Health*, *1*, 206–224.
- Moreira, H., Fonseca, A., Caiado, B., & Canavarro, M. C. (2019). Work-Family Conflict and Mindful Parenting: The Mediating Role of Parental Psychopathology Symptoms and Parenting Stress in a Sample of Portuguese Employed Parents. *Frontiers in Psychology, 10*: 635.
- Morris, P. A., & Gennetian, L. A. (2003). Identifying the Effects of Income on Children's
 Development Using Experimental Data. *Journal of Marriage and Family*, 65, 716–729.
- Moss, J. L., Johnson, N. J., Yu, M., Altekruse, S. F., & Cronin, K. A. (2021). Comparisons of individual- and area-level socioeconomic status as proxies for individual-level measures: evidence from the Mortality Disparities in American Communities study.
 Population Health Metrics, 19: 1.
- Mostafa, T. (2014). *Millennium cohort study: Technical report on response in sweep 5 (age 11)*. Centre for Longitudinal Studies.
- Moulton, V., McElroy, E., Richards, M., Fitzsimons, E., Northstone, K., Conti, G., Ploibidis, G. B., Sullivan, A., & O'Neill, D. (2020). A guide to the cognitive measures in five British birth cohort studies. London, UK:CLOSER.

Mous, S. E., Schoemaker, N. K., Blanken, L. M. E., Thijssen, S., van der Ende, J.,
Polderman, T. J. C., Jaddoe, V. W. V., Hofman, A., Verhulst, F. C., Tiemeier, H., &
White, T. (2016). The association of gender, age, and intelligence with
neuropsychological functioning in young typically developing children: The
Generation R study. *Applied Neuropsychology: Child, 6*, 22–40.

Moyi, P. (2010). Household characteristics and delayed school enrollment in Malawi.
 International Journal of Educational Development, *30*, 236–242. Mueller, C. W., & Parcel, T. L. (1981). Measures of Socioeconomic Status: Alternatives and Recommendations. *Child Development*, *52*, 13-30.

- Muthén, B. O., Muthén. (2000). Integrating person-centered and variable-centered analyses: growth mixture modeling with latent trajectory classes. (2000). *Alcoholism, Clinical and Experimental Research*, 24, 882-891.
- Muthén, B. O., Muthén, L. K., & Asparouhov, T. (2017). *Regression And Mediation Analysis* Using Mplus. Muthén & Muthén.
- Muntaner, C., Borrell, C., Vanroelen, C., Chung, H., Benach, J., Kim, I. H., & Ng, E. (2010). Employment relations, social class and health: A review and analysis of conceptual and measurement alternatives. *Social Science & Medicine*, *71*, 2130–2140.
- Myung, I. J. (2003). Tutorial on maximum likelihood estimation. *Journal of Mathematical Psychology*, 47, 90–100.

- NatCen. (2003). *Millennium Cohort Study First Survey: CAPI Questionnaire Documentation*. London: UCL Centre for Longitudinal Studies.
- National Institute of Child Health and Human Development Early Child Care Research Network. (2005). Duration and Developmental Timing of Poverty and Children's Cognitive and Social Development From Birth Through Third Grade. *Child Development*, 76, 795 – 810.
- Nelson, C. A., & Gabard-Durnam, L. J. (2020). Early Adversity and Critical Periods: Neurodevelopmental Consequences of Violating the Expectable Environment. *Trends in Neurosciences*, 43, 133–143.
- Nelson, T. D., Brock, R. L., Yokum, S., Tomaso, C. C., Savage, C. R., & Stice, E. (2021).
 Much Ado About Missingness: A Demonstration of Full Information Maximum
 Likelihood Estimation to Address Missingness in Functional Magnetic Resonance
 Imaging Data. *Frontiers in Neuroscience*, 15, 746424.
- Neppl, T. K., Senia, J. M., & Donnellan, M. B. (2016). Effects of economic hardship: Testing the family stress model over time. *Journal of Family Psychology*, 30, 12–21.
- Newland, R. P., Crnic, K. A., Cox, M. J., & Mills-Koonce, R. W. (2013). The family model stress and maternal psychological symptoms: mediated pathways from economic hardship to parenting. *Journal of Family Psychology*, 27, 96-105.

- Newman, K. (1999). *Falling from grace: Downward mobility in the age of affluence*. University of California Press: Los Angeles
- Ng-Knight, T., & Schoon, I. (2016). Disentangling the Influence of Socioeconomic Risks on Children's Early Self-Control. *Journal of Personality*, 85, 793–806.
- Nguyen, T.-N.-N., Spencer-Smith, M., Zannino, D., Burnett, A., Scratch, S. E., Pascoe, L.,
 Ellis, R., Cheong, J., Thompson, D., Inder, T., Doyle, L. W., & Anderson, P. J.
 (2018). Developmental Trajectory of Language From 2 to 13 Years in Children Born
 Very Preterm. *Pediatrics, 141*, e20172831.
- Ngware, M. W., Oketch, M., Ezeh, A. C., & Mutisya, M. (2013). The effect of free primary education policy on late school entry in urban primary schools in Kenya. *International Review of Education*, *59*, 603–625.
- NHS Digital. (2021). *Mental Health of Children and Young People in England*, 2021. National Health Service.
- Nieuwenhuis, J., Kleinepier, T., & van Ham, M. (2021). The role of exposure to neighborhood and school poverty in understanding educational attainment. *Journal of Youth and Adolescence*, *50*, 872-892.
- Nightingale, C. M., Rudnicka, A. R., Ram, B., Shankar, A., Limb, E. S., Proctor, D., Cooper,A. R., Page, A. S., Ellaway, E., Giles-Corti, B., Clary, C., Lewis, D., Cummins, S.,Whincup, P. H., Cook, D. G., & Owen, C. G. (2018). Housing, neighbourhood and

sociodemographic associations with adult levels of physical activity and adiposity: baseline findings from the ENABLE London study. *British Medical Journal Open, 8*, e021257.

Noble, S., McLennan, D., Noble, M., Plunket, E., Gutacker, N., Silk, M., Wright, G. (2019).
 The English indices of deprivation: Research report. Ministry of Housing,
 Communities and Local Government.

Northern Ireland Executive. (2015). A Fresh Start: The stormfront agreement and implementation plan. Northern Ireland Executive.

- Northern Ireland Executive (2016). *Delivering social change: The executive's child poverty strategy*. Northern Ireland Executive.
- Northern Ireland Executive. (2021). 2020/21 Report on the Northern Ireland Executive's child poverty strategy. Northern Ireland Executive.
- Nylund, K. L., Asparouhov, T., & Muthén, B. O. (2007). Deciding on the number of classes in latent class analysis and growth mixture modeling: A Monte Carlo simulation study. *Structural Equation Modeling*, *14*, 535–569.
- Oakes, & Rossi (2003). The measurement of SES in Health Research: Current practice and steps toward a new approach. *Social Science & Medicine*, *56*, 769–784.

OECD (2021). Annual average (or most common) tuition fees charged by tertiary institutions to national and foreign students (2019/20): In equivalent USD converted using PPPs, for full-time students, by type of institutions and level of education (Indicator). Doi: 10.1787/73156101-en.

OECD. (2022). Income inequality (indicator). doi: 10.1787/459aa7f1-en.

- Oerlemans, A. M., Wardenaar, K. J., Raven, D., Hartman, C. A., & Ormel, J. (2020). The association of developmental trajectories of adolescent mental health with early-adult functioning. *PLOS ONE*, *15*, e0233648.
- Okamoto, S. (2021). Parental socioeconomic status and adolescent health in Japan. *Scientific Reports, 11*, 12089.
- Office for National Statistics (2020). *Improving the measurement of household income*. Office for National Statistics.
- Office of National Statistics. (2022). *Household total wealth in great Britain: April 2018 to march 2020*. Office of National Statistics.
- Office of National Statistics. (2016). *The National Statistics Socio-economic classification* (*NS-SEC*).Office for National Statistics.
- Oliveira, I. M., Porfeli, E.J., Taveira, M. D. C., & Lee, B. (2020). Children's Career Expectations and Parents' Jobs: Intergenerational (Dis)continuities. *The Career Development Quarterly*, 68, 63-77.

- Olson, S. L., Choe, D. E., & Sameroff, A. J. (2017). Trajectories of child externalizing problems between ages 3 and 10 years: Contributions of children's early effortful control, theory of mind, and parenting experiences. *Development and Psychopathology*, 29, 1333–1351.
- Olson, L., Kinnear, M., Chen, B., Reynolds, S., Ibarra, C., Wang, T., Linke, A., & Fishman,
 I. (2021). Socioeconomic Factors Account for Variability in Language Skills in
 Preschoolers with Autism Spectrum Disorders. *Journal of Developmental & Behavioral Pediatrics*, 42, 101-108.
- Olson, L., Chen, B., Ibarra, C., Wang, T., Mash, L., Linke, A., Kinnear, M., & Fishman, I.
 (2022). Externalizing Behaviors are Associated with Increased Parenting Stress in Caregivers of Young Children with Autism. *Journal of Autism and Developmental Disorders*, 52, 975-986.
- Oshio, T., Kan, M., & Nagase, N. (2018). Changes in household income and subjective wellbeing in Japan, 2011–2014: A latent class growth analysis. *Social Indicators Research*, 136, 1033-1046.
- Padilla-Walker, L. M., Nielson, M. G., & Day, R. D. (2016). The role of parental warmth and hostility on adolescents' prosocial behavior toward multiple targets. *Journal of Family Psychology, 30*, 331–340.
Palmieri, P. A., & Smith, G.C. (2007). Examining the structural validity of the Strengths and
 Difficulties Questionnaire (SDQ) in a U.S. sample of custodial grandmothers.
 Psychological Assessment, 19, 189–198.

- Pandey, R., (1983). Max weber's theory of social stratification: controversies, contexts and correctives. *Sociological Bulletin*, 32, 171-203.
- Pardo-Crespo, M.R., Narla, N. P., Williams, A. R., Beebe, T. J., Sloan, J., Yawn, B. P.,
 Wheeler, P. H., & Juhn, Y. J. (2013). Comparison of individual-level versus arealevel socioeconomic measures in assessing health outcomes of children in Olmsted County, Minnesota. *Journal of Epidemiology and Community Health*, 67, 305-310.
- Park, J. M., Fertig, A. R., & Allison, P. D. (2011). Physical and mental health, cognitive development, and healthcare use by housing status of low-income young children in 20 american cities: A prospective cohort study. *American Journal of Public Health*, *101*, 255-261.
- Pearson, N., Griffiths, P., van Sluijs, E., Atkin, A. J., Khunti, K., & Sherar, L. B. (2022). Associations between socioeconomic position and young people's physical activity and sedentary behaviour in the UK: a scoping review. *BMJ Open*, *12*: e051736.
- Pedersen, S., Heitmann, K. A., Sagelv, E. H., Johansen, D., & Pettersen, S. A. (2019).
 Improved maximal strength is not associated with improvements in sprint time or jump height in high-level female football players: a cluster randomized controlled trial. *BMC Sports Science Medicine and Rehabilitation, 17*, 11-20.

- Pek, J., Wong, O., & Wong, A. (2019). Data Transformations for Inference with Linear Regression: Clarifications and Recommendations. *Practical Assessment, Research,* and Evaluation, 22, 9.
- Peres, M. A., Peres, K. G., de Barros, A. J. D., & Victora, C. G. (2007). The relation between family socioeconomic trajectories from childhood to adolescence and dental caries and associated oral behaviours. *Journal of Epidemiology and Community Health*, 61, 141–145.
- Peres, M. A., Peres, K. G., Thomson, W. M., Broadbent, J. M., Gigante, D. P., & Horta, B. L. (2011). The influence of family income trajectories from birth to adulthood on adult oral health: findings from the 1982 Pelotas Birth Cohort. *American Journal of Public Health*, 101, 730–736.
- Pereyra-Elias, R., Quigley, M. A., & Carson, C. (2022). To what extent does confounding explain the association between breastfeeding duration and cognitive development up to age 14? Findings from the UK Millennium Cohort Study. *PLoS, ONE 17*, e0267326.
- Perry, N. B., Calkins, S. D., Dollar, J. M., Keane, S. P., & Shanahan, L. (2017). Selfregulation as a predictor of patterns of change in externalizing behaviors from infancy to adolescence. *Development and Psychopathology*, 30, 497–510.

- Perry-Jenkins, M., Laws, H. B., Sayer, A., & Newkirk, K. (2020). Parents' Work and Children's Development: A Longitudinal Investigation of Working-Class Families. *Journal of Family Psychology*, 34, 257-268.
- Peverill, M., Dirks, M. A., Narvaja, T., Herts, K. L., Comer, J. S., & McLaughlin, K. A. (2021). Socioeconomic status and child psychopathology in the United States: A meta-analysis of population-based studies. *Clinical Psychology Review*, 83, 101933.
- Piachaud, D., & Sutherland, H. (2001). Child poverty in Britain and the New Labour government. *Journal of social policy*, *30*, 95-118.
- Pinquart, M. (2016). Associations of Parenting Styles and Dimensions with Academic Achievement in Children and Adolescents: A Meta-analysis. *Educational Psychology Review*, 28, 475-493.
- Pingault, J. B, Viding, E., Gale['], C., Greven, C. U, Zheng, Y., & Plomin, R. (2015). Genetic and environmental influences on the developmental course of attentiondeficit/hyperactivity disorder symptoms from childhood to adolescence. *JAMA Psychiatry*, 72, 651–658.
- Piotrowska, P. J., Stride, C. B., Croft, S. E., & Rowe, R. (2015). Socioeconomic status and antisocial behaviour among children and adolescents: A systematic review and metaanalysis. *Clinical Psychology Review*, 35, 47-55.

- Platt, L., Smith, K., Parsons, S., Connelly, R., Joshi, H., Rosenberg, R., Hansen, K., Brown, M., Sullivan, A., Chatzltheocharl, S., & Mostafa, T. (2014). *Millennium Cohort Study: initial findings from the Age 11 survey*. Centre for Longitudinal Studies, Institute of Education.
- Plenty, S. M., & Jonsson, J. O. (2020). Students' Occupational Aspirations: Can Family Relationships Account for Differences Between Immigrant and Socioeconomic Groups? *Child Development*, 92(1), 157–173.
- Plewis, I. (2007). Non-Response in a Birth Cohort Study: The Case of the Millennium Cohort Study, International *Journal of Social Research Methodology*, *10*, 325-334.
- Plewis, I. (2007). *The Millennium Cohort Study: Technical Report on Sampling*. London: UCL Centre for Longitudinal Studies.
- Plexousakis, S S., Kourkoutas, E., Giovazolias, T., Chatira, K., & Nikolopoulos, D. (2019). School Bullying and Post-traumatic Stress Disorder Symptoms: The Role of Parental Bonding. *Frontiers in Public Health* 7, 10.3389.
- Ponnet, K. (2014). Financial Stress, Parent Functioning and Adolescent Problem Behavior:
 An Actor–Partner Interdependence Approach The Family Stress Processes in Low-,
 Middle-, and High-Income Families. *Journal of Youth and Adolescence, 43*, 1752-1769.

- Poonawalla, I. B., Kendzor, D. E., Owen, M. T., & Caughy, M. O. (2014). Family income trajectory during childhood is associated with adolescent cigarette smoking and alcohol use. *Addictive Behaviors*, 391, 383–1388.
- Poulain, T., Vogel, M., Sobek, C Hilbert, A., Körner, A., & Kiess, W. (2019). Associations Between Socio-Economic Status and Child Health: Findings of a Large German Cohort Study. *International Journal of Environmental Research and Public Health*, 16, 677-689.
- Poirier, M. J. P., Grépin, K. A., & Grignon, M. (2019). Approaches and Alternatives to the Wealth Index to Measure Socioeconomic Status Using Survey Data: A Critical Interpretive Synthesis. *Social Indicators Research*, 148, 1–46.
- Power, C., Manor, O., & Fox, A. J. (1991). *Health and class: the early years*. London: Chapman Hall.
- Public Health England. (2020). *Official Statistics: Child development outcomes at 2 21/2* years (Experimental Statistics). Public Health England.
- Public Health Wales. (2021). *Children and young people's mental well-being during the COVID-19 pandemic*. Public Health Wales.
- Präg, P. (2020). Subjective socio-economic status predicts self-rated health irrespective of objective family socio-economic background. *Scandinavian Journal of Public Health*, 48, 707-714.

- Prevoo, M J. L., Malda, M., Mesman, J.Emmen, R. A. G., Yeniad, N., Van IJzendoorn, M. H., & Linting, M. (2014). Predicting ethnic minority children's vocabulary from socioeconomic status, maternal language and home reading input: different pathways for host and ethnic language. *Journal of Child Language 41*, 963-984.
- Prithivirajh, Y., & Edwards, S. D. (2011). An evaluation of a stress management intervention for parents of children with Attention-Deficit/Hyperactivity Disorder. *Journal of Humanities and Social Sciences, 3*, 32-37.
- Proper, C., & Rigg, J. (2007). Socio-economic status and child behaviour: evidence from a contemporary UK cohort. CASE papers (125). Centre for Analysis of Social Exclusion, London School of Economics and Political Science. London: UK.
- Pryor, L., Strandberg,-Larsen, K., Andersen, A. M. N., Rod, N. H., & Melchior, M. (2022). Trajectories of family poverty and children's mental health: Results from the Danish National Birth Cohort. *Social Science & Medicine*, 220, 371-378.
- Quigley, M. A., Hockley, C., Carson, C., Kelly, Y., Renfrew, M. J., Sacker, A. (2012). Breastfeeding is associated with improved child cognitive development: a populationbased cohort study. *Journal of Pediatrics*, *160*, 25–32.
- Quon, E. C., & McGrath, J. J. (2014). Subjective socioeconomic status and adolescent health: A meta-analysis. *Health Psychology*, *33*, 433-447.

- Raffington, L., Prindle, J. J., & Shing, Y. L. (2018). Income gains predict cognitive functioning longitudinally throughout later childhood in poor children. Developmental Psychology, 54, 1232–1243.
- Ram, N., & Grimm, K. J. (2009). Methods and Measures: Growth mixture modeling: A method for identifying differences in longitudinal change among unobserved groups. *International Journal of Behavioral Development*, 33, 565–576.
- Ramírez-Luzuriaga, M. J., Hoddinott, J., Martorell, R., Patel, S. A., Ramírez-Zea, M.,
 Waford, R., & Stein, A. D. (2020). Linear Growth Trajectories in Early Childhood and Adult Cognitive and Socioemotional Functioning in a Guatemalan Cohort. *The Journal of Nutrition*, *151*, 206–213.
- Ray, C., Kalland, M., Lehto, Reetta, & Roos, E. (2013). Does Parental Warmth and Responsiveness Moderate the Associations Between Parenting Practices and Children's Health-related Behaviors?. *Journal of Nutrition Education and Behavior*, 45, 602-610.
- Reardon, S. F. (2011). The widening academic achievement gap between the rich and the poor: New evidence and possible explanations. In D. G. Duncan, & R. J. Murnane (Eds.). *Whither opportunity? Rising inequality, schools, and children's life chances (pp. 91–116).* New York: Russell Sage Foundation.

- Reardon, S. F., Valentino, R. A., Kalogrides, D., Shores, K. A., & Greenberg, E. H.
 (2013). *Patterns and trends in racial academic achievement gaps among states, 1999-*2011. Center for Education Policy Analysis.
- Reef, J., Diamantopoulou, S., van Meurs, I., Verhulst, F. C., & van der Ende, J. (2010).
 Developmental trajectories of child to adolescent externalizing behavior and adult
 DSM-IV disorder: results of a 24-year longitudinal study. *Social Psychiatry and Psychiatric Epidemiology*, 46, 1233–1241.
- Reinecke, J., & Seddig, D. (2011). Growth mixture models in longitudinal research. *AStA Advances in Statistical Analysis*, 95, 415–434.
- Reiss, F. (2013). Socioeconomic inequalities and mental health problems in children and adolescents: A systematic review. *Social Science & Medicine*, 90, 24-31.
- Ren, L., & Edwards, C. P. (2015) Pathways of influence: Chinese parents' expectations, parenting styles, and child social competence. *Early Child Development and Care, 185*, 614-630.
- Rescorla, L., Achenbach, T., Ivanova, M. Y., Dumenci, L., Almqvist, F., Bilenberg, N., Bird, H., Wei Chen, Dobrean, A., Döpfner, M., Erol, N., Fombonne, E., Fonseca, A., Frigerio, A., Grietens, H., Hannesdottir, H., Kanbayashi, Y., Lambert, M., Larsson, B., & Leung, P. (2007). Behavioral and Emotional Problems Reported by Parents of Children Ages 6 to 16 in 31 Societies. *Journal of Emotional and Behavioral Disorders, 15*, 130–142.

- Reynolds, L. M., Yetnikoff, L., Pokinko, M., Wodzinski, M., Epelbaum, J. G., Lambert, L.
 C., Cossette, M.-P., Arvanitogiannis, A., & Flores, C. (2018). Early Adolescence is a
 Critical Period for the Maturation of Inhibitory Behavior. *Cerebral Cortex*, 29, 3676–3686.
- Ribeiro, L. A., Zachrisson, H. D., Gustavson, K., & Schjølberg, S. (2015). Maternal distress during pregnancy and language development in preschool age: A population-based cohort study. *European Journal of Developmental Psychology*, 13, 20–39.
- Rijlaarsdam, J., Stevens, G. W. J. M., van der Ende, J., Hofman, A., Jaddoe, V. W. V., Mackenbach, J. P., Verhulst, F. C., & Tiemeier, H. (2013). Economic disadvantage and youn children's emotional and behavioural problems: Mechanisms of risk. *Journal of Abnormal Child Psychology*, 41, 125-137.
- Riva, V., Caruso, A., Apicella, F., Valeri, G., Vicari, S., Molteni, M., & Scattoni, M. L. (2021). Early developmental trajectories of expressive vocabulary and gesture production in a longitudinal cohort of Italian infants at high-risk for Autism Spectrum Disorder. *Autism Research*, 14, 1421-1433.
- Rivenbark, J. G., Copeland, W. E., Davisson, E. K., Gassman-Pines, A., Hoyle, R. H., Piontak, J. R., Russell, M. A., Skinner, A. T., & Odgers, C. L. (2019). Perceived social status and mental health among young adolescents: Evidence from census data to cellphones. *Developmental Psychology*, 55, 574–585.

- Rodriguez, E. T., & Tamis-LeMonda, C. S. (2011). Trajectories of the home learning environment across the first 5 years: associations with children's vocabulary and literacy skills at prekindergarten. *Child Development*, 82, 1058–1075.
- Rose, D. & Pevalin, D. (2003). A Researcher's Guide to the National Statistics Socioeconomic Classification. London: Sage Publications.
- Rowe, M. L. (2008). Child-directed speech: Relation to socioeconomic status, knowledge of child development and child vocabulary skill. *Journal of Child Language*, 35, 185– 205.
- Rowe, M. L. (2012). A longitudinal investigation of the role of quantity and quality of childdirected speech in vocabulary development. *Child Development*, *83*, 1762–1774.
- Rubio-Codina, M., Attanasio, O., & Grantham-McGregor, S. (2016). Mediating pathways in the socio-economic gradient of child development: Evidence from children 6–42 months in Bogota. *International Journal of Behavioral Development*, 40, 483–491.
- Russell, G., Ford, T., Rosenberg, R., & Kelly, S. (2014). The association of attention deficit hyperactivity disorder with socioeconomic disadvantage: alternative explanations and evidence. *Journal of Child Psychology and Psychiatry* 55, 436–445.
- Ryan, R. M., Claessens, A., & Markowitz, A. J. (2015). Associations between family structure change and child behaviour problems: The moderating effect of family income. *Child Development*, 86, 112-127.

- Sagatun Å, Heyerdahl S, Wentzel-Larsen T, Lien L. (2014). Mental health problems in the 10th grade and non-completion of upper secondary school: the mediating role of grades in a population-based longitudinal study. *BMC Public Health, 14*, 24406098.
- Sanvictores, T., & Mendez, M. D. (2022). *Types of Parenting Styles and Effects On Children*. Florida: StatPearls Publishing.
- Sarsour, K., Sheridan, M., Jutte, D., Nuru-Jeter, A., Hinshaw, S., & Boyce, T. W. (2011). Family socioeconomic status and child executive functions: The roles of language, home environment, and single parenthood. *Journal of the International Neuropsychological Society*, 17, 120-132.
- Schmerse, D., Anders, Y., Flöter, M., Wieduwilt, N., Roßbach, H., & Tietze, W. (2018).
 Differential effects of home and preschool learning environments on early language development. *British Educational Research Journal*, 44, 338–357.
- Shonkoff, J. P., Slopen, N., & Williams, D. R. (2021). Early Childhood Adversity, Toxic Stress, and the Impacts of Racism on the Foundations of Health. *Annual Review of Public Health*, 42, 115-134.
- Schoon, I. (2001). Teenage job aspirations and career attainment in adulthood: A 17-year follow-up study of teenagers who aspired to become scientists, health professionals, or engineers. *International Journal of Behavioral Development*, 25, 124–132.

- Schoon, I. (2010). Childhood cognitive ability and adult academic attainment: Evidence from three British cohort studies. *Longitudinal and Life Course Studies*, *1*, 241-258.
- Schoon, I., Bynner, J., Joshi, H., Parsons, S., Wiggins, R. D., Sacker, A. (2002). The Influence of Context, Timing, and Duration of Risk Experiences for the Passage from Childhood to Midadulthood. *Child Development*, 73, 1486-1504.
- Schoon, I., Cheng, H. & Jones, E. (2010). Resilience in children's development. In Children of the 21st Century: The First Five Years (eds K. Hansen, H. Joshi & S. Dex), ch. 14.
 Bristol: Policy Press.
- Schoon, I., Cheng, H., Jones, E., & Maughan, B. (2013). *Wellbeing of children: Early influences*. Institute of Education: London.
- Schoon, I., Jones, E., Cheng, H. and Maughan, B. (2012) Family hardship, family instability and children's cognitive development. *Journal of Epidemiology and Community Health*, 66, 716–722.
- Schoon, I., Nasim, B., & Cook, R. (2021). Social inequalities in early childhood competences, and the relative role of social and emotional versus cognitive skills in predicting adult outcomes. *British Educational Research Journal*, 47, 1259-1280.
- Schoon, I., & Parsons, S. (2002). Teenage aspirations for future careers and occupational outcomes. *Journal of Vocational Behavior*, 60, 262–288.

- Schoon, I., Parsons, S., & Sacker, A. (2004). Socioeconomic Adversity, Educational Resilience, and Subsequent Levels of Adult Adaptation. *Journal of Adolescent Research*, 19, 383-404.
- Schoon, I., Parsons, S., Rush, R., & Law, J. (2010). Children's Language Ability and Psychosocial Development: A 29-Year Follow-up Study. *Pediatrics*, 126, 73–80.
- Schuurmans, I. K., Luik, A. I., de Maat, D. A., Hillegers, M. H. J., Ikram, M. A., & Cecil, C.
 A. M. (2022). The association of early life stress with IQ-achievement discrepancy in children: A population-based study. *Child Development*, *93*, 1837-1847.

Schwarz G. (1978) Estimating the dimension of a model. Ann Stat, 6, 461–464.

Scottish Government. (2017). *The Child Poverty (Scotland) Act 2017*. Scottish Government. Scottish Government. (2018a). *Forecasting Child Poverty in Scotland*. Scottish Government. Scottish Government. (2018c). *Local Level Household Income Estimates 2018 Summary of Data Zone Level Results*. Scottish Government.

Scottish Government. (2018b). *Every child, every chance: The tackling child poverty delivery plan.* Scottish Government.

Scottish Government (2019). Scottish household survey. Scottish Government.

Scottish Government. (2020). No One Left Behind: Delivery Plan. Scottish Government.

- Scottish Government (2022). Achieving Excellence and Equity: 2022 National Improvement Framework and Improvement Plan. Scottish Government
- Scottish Government. (2022). *National improvement framework for Scottish education: Interactive evidence report.* Scottish Government.
- Scottish Government (2022b). *Best Start, Bright Futures: tackling child poverty delivery plan* 2022 to 2026. Scottish Government.
- Scottish Government. (2023). *Scotland's devolved employability services: statistical summary*. Scottish Government.
- Séguin, L., Nikiéma, B., Gauvin, L., Lambert, M., Tu, M. T., Kakinami, L., & Paradis, G.
 (2012). Tracking Exposure to Child Poverty During the First 10 Years of Life in a Quebec Birth Cohort. *Canadian Journal of Public Health*, 103, 270–276.
- Sharkins, K. A., Leger, S. E., & Ernest, J. M. (2016). Examining Effects of Poverty, Maternal Depression, and Children's Self-Regulation Abilities on the Development of Language and Cognition in Early Childhood: An Early Head Start Perspective. *Early Childhood Education Journal*, 45, 493–498.

- Shaw, D. S., Shelleby, E. C. (2014). Early-starting conduct problem: Intersection of conduct problems and poverty. *Annual Review of Clinical Psychology*, *10*, 503-528.
- Shelleby, E. C., Votruba-Drzal, E., Shaw, D. S., Dishion, T. J., Wilson, M. N., & Gardner, F. (2014). Income and children's behavioural functioning: A sequential mediation analysis. *Journal of Family Psychology*, 28, 936-946.
- Silver, R. B., Measelle, J. R., Armstrong, J. M., & Essex, M. J. (2005). Trajectories of classroom externalizing behavior: Contributions of child characteristics, family characteristics, and the teacher–child relationship during the school transition. *Journal* of School Psychology, 43, 39–60.
- Simonoff, E., Kent, R., Stringer, D., Lord, C., Briskman, J., Lukito, S., Pickles, A., Charman, T., & Baird, G. (2020). *Journal of the American Academy of Child & Adolescent Psychiatry*, 59, 1342-1352.
- Simons, L. G., Wickrama, K. A. S., Lee, T. K., Landers-Potts, M., Cutrona, C., & Conger, R. (2016). testing family stress and family investment explanations for conduct problems among African american adolescents. *Journal of Marriage and Family*, 78, 498-515.
- Singh, N. N., Lancioni, G. E., Medvedev, O. N., Hwang, Y. S., & Myers, R. E. (2020). A
 Component Analysis of the Mindfulness-Based Positive Behavior Support (MBPBS)
 Program for Mindful Parenting by Mothers of Children with Autism Spectrum
 Disorder. *Mindfulness*, 12, 463–475.

- Sirin, S. (2005). Socioeconomic Status and Academic Achievement: A Meta-Analytic Review of Research. *Review of Educational Research*, 75, 417-453.
- Slopen, N., Fitzmaurice, G., Williams, D. R., & Gilman, S. E. (2010). Poverty, food insecurity, and the behaviour for childhood internalizing and externalizing disorders. *Journal of the American Academy of Child and Adolescent Psychiatry*, 49, 444-452.
- Smith, D. G., & Lynch, J. W. (2003). Socioeconomic differentials. In Kuh, D., & Ben-Shlomo, Y. (Eds.). A lifecourse approach to chronic disease epidemiology.
 Oxford: Oxford University Press.
- Smith, J. P., & Smith, G. C. (2010). Long-term economic costs of psychological problems during childhood. *Social Science & Medicine*, 71, 110-115.
- Smith, N. R., Kelly, Y. J., & Nazroo, J. Y. (2015). Ethnic differences in cognitive development in the first 7 years: does maternal generational status matter?. *Journal of Epidemiology and Community Health*, 70, 506–512.
- Smith, T. E., & Graham, P. B. (1995). Socioeconomic stratification in family research. *Journal of Marriage and the Family*, 57, 930-941.
- Smokowski, P. R., & Evans, C. B. R. (2019). Consequences of Bullying in Childhood,
 Adolescence, and Adulthood: An Ecological Perspective. In: *Bullying and Victimization Across the Lifespan* (pp 59-86). Springer.

- Singh-Manoux, A., Adler, N. E., Marmot, M. G. (2003). Subjective social status: its determinants and its association with measures of ill-health in the Whitehall II study. *Social Science & Medicine*, 56, 1321–1333.
- Social Mobility Commission. (2020). Annual review and business plan: A year of innovation and collaboration. Social Mobility Commission.
- Sohr-Preston, S., Scaramella, L. V., Martin, M. J., Neppl, T. K., Ontai, L., & Conger, R.
 Parental Socioeconomic Status, Communication, and Children's Vocabulary
 Development: A Third-Generation Test of the Family Investment Model. *Child Development*, 84, 1046-1062.
- Song, X., & Yu, X. (2020). Occupation-Based Socioeconomic Index with Percentile Ranks. University of Pennsylvania Population Center Working Paper (PSC/PARC), 2020-59.
- Sonnenschein, S., Stapleton, L. M., & Metzger, S. R. (2013). What Parents Know About How Well Their Children Are Doing in School. *The Journal of Educational Research*, 107, 152–162.

Sorokin, P. (1959). Social and Cultural Mobility. New York: The Free Press.

Sosu, E. M., & Schmidt, P. (2017). Economic Deprivation and Its Effects on Childhood Conduct Problems: The Mediating Role of Family Stress and Investment Factors. *Frontiers in Psychology*, 8, 1580.

- Sosu, E. M., & Schmidt, P. (2022). Changes in Cognitive Outcomes in Early Childhood: The Role of Family Income and Volatility. *Frontiers in Psychology*, 13, 758082.
- Spann, M. N., Bansal, R., Hao, X., Rosen, T. S., & Peterson, B. S. (2020). Prenatal Socioeconomic Status and Social Support are associated with Neonatal Brain Morphology, Toddler Language and Psychiatric Symptoms. *Child Neuropsychology: A Journal on Normal and Abnormal Development in Childhood and Adolescence, 26*, 170–188.
- Stålnacke, S. R., Tessma, M., Böhm, B., & Herlenius, E. (2019). Cognitive Development
 Trajectories in Preterm Children With Very Low Birth Weight Longitudinally
 Followed Until 11 Years of Age. *Frontiers in Physiology*, *10*, 307.
- StatWales. (2019). Academic achievement of pupils aged 4 to 14 in core subjects, 2019. Welsh Government.
- StatsWales. (2019). *National level results, by area of learning and outcome*. Welsh Government.
- Straatmann, V. S., Pearce, A., Hope, S., Barr, B., Whitehead, M., Law, C., & Taylor-Robinson, D. (2018). How well can poor child health and development be predicted by data collected in early childhood?. *Journal of Epidemiology and Community Health*, 72, 1132-1140.

- Staunton, E., Kehoe, C., & Sharkey, L. (2020). Families under pressure: stress and quality of life in parents of children with an intellectual disability. *Irish Journal of Psychological Medicine*, 1-8.
- Stuart, M. Getting ready for reading: Early phoneme awareness and phonics teaching improves reading and spelling in inner-city second language learners. *British Journal* of Educational Psychology, 69, 587–605.
- Student Awards Agency Scotland. (2021). *Higher Education Student Support in Scotland* 2020/2021: Summary of financial support provided by the Student Awards Agency Scotland. National Statistics.
- Sullivan, A., & Brown, M. (2013). Social inequalities in cognitive scores at age 16: The role of reading. London: Institute of Education.
- Sullivan, A., Moulton, V., & Fitzsimons. (2017). The intergenerational transmission of vocabulary. London: Centre for Analysis of Social Exclusion.
- Sullivan, A., Moulton, V., & Fitzsimons, E. (2021). The intergenerational transmission of language skill. *The British Journal of Sociology*, 72, 207–232.
- Sun, W., Li, D., Zhang, W., Bao, Z., & Wang, Y. (2015). Family material hardship and Chinese adolescents' problem behaviors: A moderated mediation analysis. *PLOS one*, 10, e0128024.

- Sutton, T. E., Simons, L.G., Simons, R. L., Cutrona, C. (2017). Psychological distress, couple interactions, and parenting: A dyadic analysis of African American couples. *Journal* of Marriage and Family, 79, 850-864.
- Sylva, K. (1997). Critical periods in childhood learning. *British Medical Bulletin*, *53*, 185–197.
- Tabachnick, B. G., & Fidell, L. S. (2007). *Using multivariate statistics* (5th ed.). Allyn & Bacon/Pearson Education.
- Takizawa, R., Maughan, B., & Arseneault, L. (2014). Adult Health Outcomes of Childhood Bullying Victimization: Evidence From a Five-Decade Longitudinal British Birth Cohort. American Journal of Psychiatry, 171, 777–784.
- Taylor, C. (2017). The reliability of free school meal eligibility as a measure of socioeconomic disadvantage: evidence from the millennium cohort study in wales. *British Journal of Educational Studies*, 66, 29-51.
- Taylor, J. J., Grant, K. E., Zulauf, C. A., Fowler, P. J., Meyerson, D. A., & Irsheid, S. (2016).
 Exposure to Community Violence and the Trajectory of Internalizing and
 Externalizing Symptoms in a Sample of Low-Income Urban Youth. *Journal of Clinical Child & Adolescent Psychology*, 47, 421–435.
- Tamura, K., Morrison, J., & Pikhart, H. (2020). Children's behavioural problems and its associations with socioeconomic position and early parenting environment: findings

from the UK Millennium Cohort Study. *Epidemiology and Psychiatric Sciences* 29, e155, 1–14.

Thaning, M. (2021). Resource specificity in intergenerational inequality: The case of education, occupation, and income. *Research in Social Stratification and Mobility*, 75, 100644.

The Childcare Act (2006). 2006.

The Children's Services Co-operation (Northern Ireland) Act (2015). (2015).

The Education (Scotland) Act (2016). (2016).

- The Healthy Start Scheme and Welfare Food (Miscellaneous Amendments) Regulations (2020). The stationery office limited.
- Thompson, R., Kim, J., Litrownik, A. J., Briggs, E. C., Hussey, J. M., English, D. J., &
 Dubowitz, H. (2011). Early Adolescent Risk Behavior Outcomes of Childhood
 Externalizing Behavioral Trajectories. *The Journal of Early Adolescence*, *31*, 234–257.
- Tichnor-Wagner, A., Garwood, J. D., Bratsch-Hines, M., & Vernon-Feagans, L. (2015).
 Home Literacy Environments and Foundational Literacy Skills for Struggling and Nonstruggling Readers in Rural Early Elementary Schools. *Learning Disabilities Research & Practice, 31*, 6–21.

Townsend, P. (1979). Poverty in the United Kingdom. Allen Lane and Penguin Books.

- Townsend, P., Phillimore, P., & Beattie, A. (1988). *Health and Deprivation Inequality and the North*. Routledge.
- Toseeb, U., & St Clair, M. C. (2020). Trajectories of prosociality from early to middle childhood in children at risk of Developmental Language Disorder. *Journal of Communication Disorders*, 85, 105984.
- Troost, A. A., van Ham, M., & Manley, D. J. (2023). Neighbourhood effects on educational attainment. What matters more: Exposure to poverty or exposure to affluence? *PLoS ONE*, 18, e0281928.
- Tsal, Y., Shalev, L., & Mevorach, C. (2005). The Diversity of Attention Deficits in ADHD: The Prevalence of Four Cognitive Factors in ADHD Versus Controls. *Journal of Learning Disabilities*, 38, 142-157.
- Twenge, J. M., Campbell, W. K., & Foster, C. A. (2003). Parenthood and Marital Satisfaction: A Meta-Analytic Review. *Journal of Marriage and Family*, 65, 574-583.
- Ukoumunne, O. C., Wake, M., Carlin, J., Bavin, E. L., Lum, J., Skeat, J., Williams, J., Conway, L., Cini, E., Reilly, S. (2012). Profiles of language development in preschool children: a longitudinal latent class analysis of data from the Early Language in Victoria Study. *Child Care Health and Development, 38*, 341-349.

United Nations. (2023). Global Sustainable Development Report 2023. United Nations.

- University of London. UCL Social Research Institute. Centre for Longitudinal Studies,
 Millennium Cohort Study: First Survey, 2001-2003 [computer file]. 6th Edition.
 Colchester, Essex: UK Data Archive.
- Vahedi, A., Krug, I., Fuller-Tyszkiewicz, M., & Westrupp, E. M. (2018). Longitudinal associations between work-family conflict and enrichment, inter-parental conflict, and child internalizing and externalizing problems. *Social Science & Medicine*, 211, 251– 260.
- van Eldik, W. M., de Haan, A. D., Parry, L. Q., Davies, P. T., Luijk, M. P. C. M., Arends, L. R., & Prinzie, P. (2020). The interparental relationship: Meta-analytic associations with children's maladjustment and responses to interparental conflict. *Psychological Bulletin, 146*, 553–594.
- van Roode, T., Sharples, K., Dickson, N., & Paul, C. (2017). Life-Course Relationship between Socioeconomic Circumstances and Timing of First Birth in a Birth Cohort. *PLOS ONE*, *12*, e0170170.
- Vannette, D. L. (2017). Best Practices for Survey Research. *The Palgrave Handbook of Survey Research*, 331–343.

- Vasilyeva, M., Dearing, E., Ivanova, A., Shen, C., & Kardanova, E. (2018). Testing the family investment model in Russia: Estimating indirect effects of SES and parental beliefs on the literacy skills of first-graders. *Early Childhood Research Quarterly*, 42, 11–20.
- Vassiliu, C., Mouzaki, A., Antoniou, F., Ralli, A. M., Diamanti, V., Papaioannou, S., & Katsos, N. (2023). Development of Structural and Pragmatic Language Skills in Children With Attention-Deficit/Hyperactivity Disorder. *Communication Disorders Quarterly*, 44, 207–218.
- Violato, M., Petrou, S., Gray, R., & Redshaw, M. (2011). Family income and child cognitive and behavioural development in the united kingdom: does money matter?. *Health Economics*, 20, 1201-1225.
- von Hippel, P. T. (2018). How Many Imputations Do You Need? A Two-stage Calculation Using a Quadratic Rule. *Sociological Methods & Research*, 27, 1595-1605.
- Votruba-Drzal, E., Miller, P., Betancur, L., Spielvogel, B., Kruzik, C., & Coley, R. L. (2021).
 Family and community resource and stress processes related to income disparities in school-aged children's development. *Journal of Educational Psychology*, *113*, 1405–1420.
- Vukojević, M., Zovko, A., Talić, I., Tanović, M., Rešić, B., Vrdoljak, I., & Splavski, B.
 (2017). Parental socioeconomic status as a predictor of physical and mental health outcomes in children literature review. *Acta Clin Croat* 56, 742-748.

- Vygotsky, L. S., & Cole, M. (1978). *Mind in Society: Development of Higher Psychological Processes*. Harvard University Press.
- Wade M, Jenkins JM, Venkadasalam VP, Binnoon-Erez N, Ganea PA (2018) The role of maternal responsiveness and linguistic input in pre-academic skill development: a longitudinal analysis of pathways. *Cognitive Development*, 45, 125–140.
- Wadsworth, M., Santiago, C. D., Einhorn, L., Etter, E. M., Rienks, S., & Markman, H. (2011)
 Preliminary Efficacy of an Intervention to Reduce Psychosocial Stress and Improve
 Coping in Low-Income Families. *American Journal of Community Psychology*, 48, 257–271.
- Waldfogel, J. (2013). Socio-economic inequality in childhood and beyond: an overview of challenges and findings from comparative analyses of cohort studies. *Longitudinal and Life Course Studies*, 4, 268–275.

Walker, R. (2014). The shame of poverty. Oxford: Oxford University Press.

Washbrook, E., & Waldfogel, J. (2010). *Low income and early cognitive development in the UK*. London: Sutton Trust.

Weber, M. (1968). Economy and society. Totowa, NJ: Bedminster Press.

Weeks, M., Cairney, J., Wild, T. C., Ploubidis, G. B., Naicker, K., & Colman, I. (2014). Early-life predictors of internalizing symptom trajectories in Canadian children. *Depression and Anxiety*, 31, 608–616. Weisleder A, Fernald A. (2013). Talking to children matters: early language experience strengthens processing and builds vocabulary. *Psychology Science*, *24*, 2143–2152.

Weiss, B. A., & Dardick, W. (2016). An Entropy-Based Measure for Assessing Fuzziness in Logistic Regression. *Educational and Psychological Measurement*, 76, 986–1004.

Welfare Reform and Work Act. (2012). C.5.

Welfare Reform and Work Act. (2016). C.7.

Welsh Government. (2015). Curriculum for Wales. Welsh Government.

Welsh Government (2015). Child Poverty Strategy for Wales. Welsh Government.

Welsh Government (2022). *Childcare for 3 and 4 year olds: policy guidance for providers*. Welsh Government.

Welsh Government. (2022). Child poverty strategy: 2022 progress report. Welsh Government.

Welsh Government (2023). Consultation on the draft Child Poverty Strategy for Wales 2023. Welsh Government.

Well-being of Future Generations (Wales) Act (2015). Anaw.2.

- Wheeler, A. (2015, July 3). *What We Can Learn from Small Units of Analysis*. The Social Science Research Network.
- White, K. (1982). The relation between socioeconomic status and academic achievement. Psychological *Bulletin*, *91*, 461–481.
- White, R. M. B., Liu, Y., Gonzales, N. A., Knight, G. P., & Tein, J. (2016). Neighborhood qualification of the association between parenting and problem behavior trajectories among Mexican-origin father-adolescent dyads. *Journal of Research on Adolescence*, 26, 927–946.
- Whitfield, K., Betancur, L., Miller, P., & Votruba-Drzal, E. (2021). Longitudinal Links between Income Dynamics and Young Adult Socioeconomic and Behavioral Health Outcomes. *Youth & Society*, 53, 1181–1210.
- Williams, M. (2016). An old model of social class? Job characteristics and the NS-SEC schema. Work, Employment and Society, 31, 153–165.
- Willms, J. D., & Tramonte, L. (2019). The SAGE Handbook of Comparative Studies in Education. SAGE Knowledge; SAGE Publications Ltd.

Wilson, P., Bradshaw, P., Tipping, S., Henderson, M., Der, G., & Minnis, H. (2012). What predicts persistent early condict problems? Evidence from the growing up in Scotland cohort. *Journal of Epidemiology and Community Health*, 67, 76-80.

Wilson, S., & Durbin, C. E. (2013). Mother-child and father-child dyadic interaction:
Parental and child bids and responsiveness to each other during early childhood. *Merrill-Palmer Quarterly*, 59, 249–279.

- Xiaofeng, M., Wenhui, D., & Aibao, Z. (2018). The Link Between Parental Absence and Poor Reading Comprehension: Evidence From the Left-Behind Children in Rural China. *Frontiers in Education*, *3*, 71.
- Xie, S., Hubbard, R. A., Himes, B. E. (2020). Neighborhood-level measures of socioeconomic status are more correlated with individual-level measures in urban areas compared with less urban areas. *Annuals of Epidemiology*, 43, 37-43.
- Yaari, M., Mankuta, D., Harel-Gadassi, A., Friedlander, E., Bar-Oz, B., Eventov-Friedman,
 S., Maniv, N., Zucker, D., & Yirmiya, N. (2018). Early developmental trajectories of
 preterm infants. *Research in Developmental Disabilities*, 81, 12-23.
- Yap, M. B. H., Morgan, A. J., Cairns, K., Jorm, A. F., Hetrick, S. E., & Merry S. (2016).
 Parents in prevention: A meta-analysis of randomized controlled trials of parenting interventions to prevent internalizing problems in children from birth to age 18. *Clinical Psychology Review*, *50*, 138–158.
- Yeo, L. S., Ong, W. W., & Ng, C. M. (2014). The Home Literacy Environment and Preschool Children's Reading Skills and Interest. *Early Education and Development*, 25, 791–814.

- Youth Wellbeing NI. (2020). The mental health of children and parents in Northern Ireland: Results of the youth wellbeing prevalence survey. Youth Wellbeing NI.
- Wu, C. Y., & Lee, T. S. H. (2020). Impact of parent–child relationship and sex on trajectories of children internalizing symptoms. *Journal of Affective Disorders*, 260, 167–173.
- Zachrisson, H. D., & Dearing, E. (2015). Family Income Dynamics, Early Childhood Education and Care, and Early Child Behavior Problems in Norway. *Child Development*, 86, 425–440.
- Zahn-Waxler, C., Shirtcliff, E. A., & Marceau, K. (2008). Disorders of Childhood and Adolescence: Gender and Psychopathology. *Annual Review of Clinical Psychology*, 4, 275–303.
- Zahodne, L. B., Kraal, A. Z., Zaheed, A., & Sol, K. (2018). Subjective Social Status Predicts Late-Life Memory Trajectories through Both Mental and Physical Health Pathways. *Gerontology*, 64, 466-474.
- Zauche, L. H., Thul, T. A., Mahoney, A. E. D., & Stapel-Wax, J. L. (2016). Influence of language nutrition on children's language and cognitive development: An integrated review. *Early Childhood Research Quarterly*, 36, 318–333.
- Zdebik, M. A., Boivin, M., Battaglia, M., Tremblay, R. E., Falissard, B., & Cote, S. M. (2019). Childhood multi-trajectories of shyness, anxiety and depression: Associations with adolescent internalizing problems. *Psychology*, 64, 101050.

- Zeanah, C. H., Gunnar, M. R., McCall, R. B., Kreppner, J. M., & Fox, N. A. (2011). VI. Sensitive Periods. *Monographs of the Society for Research in Child Development*, 76, 147–162.
- Zhang, L., Cui, Z., Sasser, J., Carvalho, C., & Oshri, A. (2022)Family stress during the pandemic worsens the effect of adverse parenting on adolescent sleep quality. *Child Abuse & Neglect, 123*, 105390.
- Zhang, X., Chen, H.-C., Zhang, Y.-N., & Sun, B.-H. (2009). Family income and social competence in early childhood: Examining mediation and moderation effects. *Acta Psychologica Sinica*, 41, 613–623.
- Zhu, Z., Chang, S., Cheng, Y., Qi, Q., Li, S., Elhoumed, M., Yan, H., Dibley, M. J., Fawzi,
 W. W., Zeng, L., & Sudfeld, C. R. (2019). Early life cognitive development
 trajectories and intelligence quotient in middle childhood and early adolescence in
 rural western China. *Scientific Reports*, *9*, 18315.
- Zhu, Y., Chen, X., Zhao, H., Chen, M., Tian, Y., Liu, C., Han, Z. R., Lin, X., Qiu, J., Xue, G., Shu, H., & Qin, S. (2019). Socioeconomic status disparities affect children's anxiety and stress sensitive cortisol awakening response through parental anxiety. *Psychoneuroendocrinology*, *103*, 96–103.

Zuccotti, C. V. (2019). Ethnicity and neighbourhood attainment in England and Wales: A study of second generations' spatial integration. *Population, Space and Place, 25*, e2252.

Appendix

Ethical approval for this study

From: Eugenie Samier Sent: 10 December 2018 14:23 To: Edward Sosu; Simon Hunter; Scott Orr Subject: Ethics Approval

Type 1 Ethics Application Approval

Dear All

Testing the mechanisms between income dynamics and children's developmental outcomes

CI

Dr. Edward Sosu

Chief Investigator

Other Investigators:

Simon Hunter Scott Orr

I can now confirm full ethical and sponsorship approval for the above study.

Regards Eugenie A. Samier Chair, SoE Ethics Committee University of Strathclyde Level 5 Lord Hope Building 141 St James Road Glasgow G4 0LT